A Qualitative Study into the Impact of Outcomes Based Education on Engineering Educators and Engineering Education in the Technical Higher Education Sector in Ireland

Richard Gallery, BE, MEngSc, CEng, MEI

Department of Adult and Community Education

Maynooth University

Doctoral Thesis submitted in fulfilment of the requirements for the degree of Doctorate in Higher and Adult Education

May 2021

Supervisor: Dr Fergal Finnegan

Acknowledgements

I would like to acknowledge the patience and support of my family, Susan, Patrick and Christopher, as I took the journey through the research I present in this thesis, and for tolerating the sometimes quite philosophical Sunday dinner time discussion that resulted. I also wish to thank my employer, formerly Institute of Technology Blanchardstown, and now Technological University Dublin, for providing me with the space, and other support, I required to complete this work. I particularly want to thank my research participants for those wonderful conversations we had about their views of outcomes based engineering education. Every time I re-read the interview transcripts their lived experience of this comes to life for me. Further thanks go to Martin Meaghar and Pat O'Connor for lending me the benefit of their own, similar, experience when I needed it. I also wish to acknowledge the help, guidance and encouragement of my supervisor, Fergal Finnegan, and in particular a thoughtful phone call the day after a thesis meeting when the path forward was not as clear as it later became.

Finally, I wish to dedicate this thesis to the memories of my brother Damien, my father Christy, and my aunty Patsy who helped me to value education in the way that I do.

Answers

I kept my answers small and kept them near; Big questions bruised my mind but still I let Small answers be a bulwark to my fear.

The huge abstractions I kept from the light; Small things I handled and caressed and loved. I let the stars assume the whole of night.

But the big answers clamoured to be moved Into my life. Their great audacity Shouted to be acknowledged and believed.

Even when all small answers build up to Protection of my spirit, still I hear Big answers striving for their overthrow.

And all the great conclusions coming near.

Dame Edith Sitwell

Abstract

This thesis considers the effect the transition to outcomes based education (OBE) has had on engineering educators and engineering education in the Institutes of Technology (IoTs) in Ireland. Whereas engineering education research into learning outcomes largely focuses on how teaching may better align with their use, the focus of this research, the effect OBE has on both engineering educators and engineering education, receives little attention in the literature. I conduct this research as an engineering educator seeking to understand how we have been shaped by OBE, and how this is affecting the education of future generations of engineers. My research employs a qualitative methodology, in which I consider this impact as perceived and experienced by a sample of IoT engineering academics.

My research highlights the influence of the market in shaping engineering education, which can be regarded through Bernstein's (2000) concept of engineering as a region, facing inward to academia, but outward to the market, mediated by the professional bodies. This leads me to draw selectively on social realism, alongside my experience as an engineering educator, as a conceptual framework. This emphasises the need to gain comprehensive understanding of the historical context in which my research is situated, for which my literature review encompasses a number of inter-related socio-historical accounts: of the early development of Irish engineering education; of the establishment of the Regional Technical Colleges (which became the IoTs) to implement government policy that saw education, particularly technician education, as key to improving the life of the citizenry through economic advancement; the later development of the National Framework of Qualifications (NFQ) as part of a skills-focussed reorientation of higher education; and the reasons for, and consequences of, the adoption of OBE for engineering accreditation, internationally and in Ireland.

The fieldwork, comprising interviews and a focus group with engineering academics, reveals the perceptions of my research participants of the effect that OBE is having on their academic identity, including their approach to curriculum and pedagogy, and provides insight into the structure of engineering education, and the identity formation of students. I will show that my interviewees regard OBE as effective, in terms of: facilitating communication; improving access to education; impacting positively on pedagogy; and as a framework for curriculum design. However, my research critiques the assessment focussed pedagogy that they appear to have adopted as a consequence, questions the appropriateness of the 'language of levels' related to NFQ terminology that has emerged in our pedagogic discourse, and raises concerns about the impact on curriculum and the structure of engineering education.

i

Overview Table of Contents

Abstract	i
Overview Tal	ble of Contentsii
Detailed Tab	le of Contentsiii
Acronyms	xi
Figures	xiii
Tables	xiv
Chapter 1	Introduction1
Chapter 2	Engineering Education and Ireland's Technological Education Sector19
Chapter 3	Outcomes Based Engineering Education50
Chapter 4	A Conceptual Framework from which to consider IoT Engineering Education88
Chapter 5	Research Methodology106
Chapter 6	Influences and Prescriptions on Engineering Academics
Chapter 7	The Language of Levels
Chapter 8	Progression in IoT Engineering Education170
Chapter 9	The Impact on the Pedagogy of Engineering Academics
Chapter 10	The Impact on Knowledge, Skills and the Curriculum194
Chapter 11	Discussion
Chapter 12	Conclusions
Appendices	
Bibliography	

Detailed Table of Contents

Abstract	i
Overview	v Table of Contentsii
Detailed	Table of Contentsiii
Acronym	sxi
Figures	xiii
Tables	xiv
Chapter	1 Introduction1
1.1	The Research Question1
1.2	Outcomes Based Engineering Education2
1.3	Being an Engineering Educator6
1.4	The Fieldwork for the Thesis8
1.5	Drawing on Social Realism9
1.6	Why This Research is Important11
1.7	Key Findings and Contributions to the Literature12
1.8	Structure of the Thesis16
Chapter	2 Engineering Education and Ireland's Technological Education Sector19
2.1	Introduction19
2.2	Engineering and Engineering Education20
	2.2.1 Technician Education vs. Professional Engineering Education24
2.3	The Field of Engineering Education Research26
	2.3.1 Engineering Education Research in Ireland27
	2.3.2 Engineering Education Research and Learning Outcomes

		2.3.	3	Engineering Education Research and Engineering Educators	29
		2.3.	4	The Positioning of this Research in the field of Engineering E	ducation
				Research	30
	2.4	4	Th	e Establishment of Engineering Technician Education in Ireland	31
		2.4.	1	Engineering and Technical Education in Ireland Prior to 1960	31
		2.4.	2	Training of Technicians in Ireland (1964)	35
		2.4.	3	The Report on Training and Education of Technicians (1967) from the	e ICEI.37
		2.4.	4	The Investment in Education Report (1965)	
		2.4.	5	The Report of the Commission on Higher Education (1967)	
		2.4.	6	The Mulcahy Report (1967) and the Creation of the RTCs	40
	2.	5	En	gineering Education in the IoT Sector	41
	2.	6	Th	e Role of Engineering Academics in the IoTs	45
	2.	7	Со	onclusion	49
Chap	ter	3	Οι	utcomes Based Engineering Education	50
	3.	1	Int	troduction	50
	3.	2	Th	e Development of the Learning Outcomes Concept	50
	3.	3	Th	e Rise of National Qualification Frameworks	51
		3.3.	1	The Attractiveness of OBE as a Policy Instrument	53
		3.3.	2	Quality Assurance and National Qualification Frameworks	56
	3.4	4	Th	e Adoption of OBE based Accreditation for Engineering	57
	3.	5	Th	e Development of the NFQ	59
		3.5.	1	Teastas, the NQAI, HETAC and FETAC, and the Path to the NFQ	60
		3.5.	2	Policy Developments	62
		3.5.	3	European and International Influences	63

	3.5.	4 The Consultation Process	65
	3.6	The NFQ Revealed	58
	3.6.	1 Values	58
		3.6.1.1 Equality and Accessibility	69
		3.6.1.2 Comprehensiveness and Coherence	69
		3.6.1.3 Transparency and Simplicity	70
		3.6.1.4 Quality	72
		3.6.1.5 Relevance	73
	3.6.	2 The Engineering Standards	73
	3.7	The NFQ as a Policy Tool to Emphasis the Teaching of Skills	74
	3.8	El Accreditation: The adoption of LOs, and the Influence of Bologna	77
	3.8.	1 The Bologna Process and the Impact on Irish Engineering Education	79
	3.9	Implications for the Design of Engineering Curriculum	31
	3.10	Engineering Pedagogy and OBE	32
	3.11	Conclusion	36
Chap	ter 4	A Conceptual Framework from which to consider IoT Engineering Education	38
	4.1	Introduction	38
	4.2	Drawing on Social Realism	39
	4.3	From The New Sociology of Education to Social Realism) 0
	4.4	Knowledge) 2
	4.5	The Engineering Professional Bodies, Cognitive Interests, and Knowledge	
		Structures	9 4
	4.6	Curriculum	9 6
	4.7	Recontextualisation of Knowledge and the Pedagogic Device) 7
	4.8	Pedagogy and Identity Formation10	00

2	4.9	IoT Engineering Education Viewed as a Pedagogic Device	
2	4.10	Conclusion	
Chapte	er 5	Research Methodology	106
<u> </u>	5.1	Introduction	
[5.2	The Research Question	
[5.3	Selection of Research Participants	
[5.4	The Role of Document Analysis in my Research Process	
	5.4.	.1 Keyword Analysis for Knowledge and Skills	
	5.4.	.2 Readability Analysis	
[5.5	Selection of Data Gathering Methods	115
	5.5.	.1 Interviews	116
	5.5.	.2 Focus Groups	117
5	5.6	Conducting the Interviews	
5	5.7	Coding and Analysis	
	5.7.	.1 On the Inside Looking In and What I Didn't See	
	5.7.	.2 Presenting the Analysis	
5	5.8	The Focus Group	
5	5.9	Ethical Considerations	
	5.9.	.1 Contacting Potential Interviewees	129
	5.9.	.2 Data Storage	
5	5.10	Validity, Reliability and Limitations of My Research	131
[5.11	Conclusion	132
Chapte	er 6	Influences and Prescriptions on Engineering Academics	

	6.1	Introduction
	6.2	Implicit and Explicit Control of the Pedagogic Discourse: Hearts and Minds? 135
	6.3	Academics' Perception of the Effectiveness of the NFQ136
	6.4	Awareness of Impact of the Legislative and Policy Context of the NFQ137
	6.5	Academics Perception of Societal Impact of the NFQ137
	6.6	The NFQ and IoT Engineering Academic Management139
	6.7	Influence on Pedagogy141
	6.8	Course Design
	6.9	The Role of Engineers Ireland145
	6.9	.1 Views Critical of El147
	6.10	Conclusion149
Chap	ter 7	The Language of Levels150
	7.1	Introduction
	7.2	Gordon's Interview and the Language of Levels152
	7.3	The Use of the Language of Levels to Engage with the Regulative Discourse .154
	7.3	.1 Describing Awards and Programmes154
	7.3	.2 Referring to Graduates and Employment156
	7.3	.3 Transparency157
	7.4	The use of the Language of Levels to Engage with Both the ORF and the PRF158
	7.4	.1 Progression
	7.4	.2 Educational Positioning of the IoT Sector159
	7.5	The Use of the Language of Levels in the Instructional Discourse161
	7.5	.1 Referring to Students

7.5	.2 Approaches to Teaching and Student Projects	161
7.5	.3 Knowledge	163
7.5	.4 Modules	164
7.5	.5 Learning Outcomes	165
7.6	Criticisms of the Effectiveness of the Language of Levels	166
7.7	The Focus Group and the Language of Levels	166
7.8	Conclusion	169
Chapter 8	Progression in IoT Engineering Education	170
8.1	Introduction	170
8.2	Bill's Interview and Progression	170
8.3	Progression to Level 8 as the Aspirational Target for Students	173
8.4	The Effect on Technician Education	175
8.5	The Effect on the Higher Certificate	180
8.6	Progression and Equality of Access	181
8.7	Conclusion	182
Chapter 9	The Impact on the Pedagogy of Engineering Academics	
9.1	Introduction	184
9.2	An Emphasis on the Practical	185
9.3	Teaching to Assessment of LOs	186
9.4	The Influence of LOs on John's Pedagogy	188
9.5	Ayah's Managerial Perspective	191
9.6	The Focus Group Views on Pedagogy	192
9.7	Conclusion	193

Chapter 10	The Impact on Knowledge, Skills and the Curriculum	
10.1	Introduction	194
10.2	Engineering Knowledge	194
10.3	Knowledge Acquisition	
10.4	The Balance of Knowledge and Skills	
10.5	Recontextualisation of Knowledge in the Curriculum	201
10.	5.1 What has changed in the Design of Curriculum?	201
10.	5.2 Bringing an Academic's Personal Knowledge to the Curriculum	204
10.6	Conclusion	206
Chapter 11	Discussion	207
11.1	Introduction	207
11.2	Engineering Educators and Engineering Education in the IoTs	208
11.	2.1 Pedagogy	209
11.	2.2 Curriculum	214
11.	2.3 Structure of IoT Engineering Education	218
11.	2.4 Knowledge	221
11.	2.5 Contributing to the Debate Regarding Engineering Curriculum	222
11.3	The Effectiveness of the NFQ	223
11.4	OBE and the Relationship between IoT Engineering Education and	Society 226
11.5	Reflection on the Use of the Pedagogic Device	228
11.6	Conclusion	232
Chapter 12	Conclusions	233
12.1	Introduction	233

Detailed Table of Contents

	12.2	Contributions of my Research to the Literature	234
	12.3	Future Directions	238
	12.4	Final Reflection	241
Арре	endices		243
	Apper	dix A Level Indicators for the NFQ for Levels 6 to 10	243
	Apper	dix B Sample Focus Group Exercise	244
	Apper	dix C Readability Analysis of Programme Outcomes	245
	ITB	Mechatronics Higher Certificate Readability Analysis, level 6 NFQ	246
		Programme Outcomes	246
		Readability Analysis.	246
	CIT	Electrical Higher Certificate, level 6 NFQ	247
		Programme Outcomes	247
		Readability Analysis	247
	Air	Traffic Control Engineer, Level 6 EQF	248
		Programme learning Outcomes	248
		Readability Analysis	248
	Apper	dix D, HEA Data on Engineering Enrolments	249
	Apper	dix E The Bologna Process	250
Bibli	ography	/	252

Acronyms

ABET	Accreditation Board for Engineering and Technology
ACM	Association for Computing Machinery
ASEE	American Society for Engineering Education
BFUG	Bologna Follow-Up Group
BTech	Bachelor of Technology
CAO	Central Applications Office
CDVEC	City of Dublin Vocational Educational Committee
CEDEFOP	Centre européen pour le développement de la formation professionnelle (French: European Centre for the Development of Vocational Training)
CEVA	Council for Educational and Vocational Awards
CSO	Central Statistics Office
DES	Department of Education and Skills
DIT	Dublin Institute of Technology
EER	Engineering Education Research
EHEA	European Higher Education Area
EI	Engineers Ireland
ENAEE	European Network for Accreditation of Engineering Education
EQF	European Qualification Framework for Lifelong Learning
FDI	Foreign Direct Investment
FET	Further Education and Training
FETAC	Further Education Training and Awards Council
HEA	Higher Educational Authority
HEI	Higher Education Institute
HETAC	Higher Education Training and Awards Council
HoD	Head of Department
IEA	International Engineering Alliance
IBEC	Irish Business and Employers Confederation
ICEI	Institution of Civil Engineers of Ireland
ICT	Information and Communication Technology
IDA	Industrial Development Agency
IEEE	Institute of Electrical and Electronics Engineers
IEI	Institute of Engineers Ireland

IGIP	International Society for Engineering Pedagogy
IoT	Institute of Technology
ITB	Institute of Technology Blanchardstown
LO	Learning Outcome
NAE	National Academic of Engineering
NCEA	National Council for Educational Awards
NCVA	National Council for Vocational Awards
NFQ	National Framework of Qualifications
NQAI	National Qualifications Authority of Ireland
NQF	National Qualifications Framework
NVQ	National Vocational Qualification
NSF	The USA National Science Foundation
NZQF	New Zealand Qualifications Framework
OBE	Outcomes based education
OECD	Organization for Economic Cooperation and Development
ORF	Official Recontextualising Field
PRF	Pedagogic Recontextualising Field
QQI	Quality and Qualifications Ireland
RCSI	Royal College of Science of Ireland
RTC	Regional Technology College
SAQA	South African Qualification Authority
SCQF	Scottish Credit and Qualifications Framework
UCG	University College Galway, now National University of Ireland Galway (NUIG)
VEC	Vocational Educational Committee
VR	Virtual Reality

Figures

Figure 1-1 NFQ Level Diagram
Figure 2-1 A Timeline of the Development of Engineering Education in Ireland Prior to 1950
Figure 2-2 A Timeline leading to the Establishment of the RTC/IoT sector in the 1960s34
Figure 2-3 Number of Full-Time New Entrants to Higher Education in Engineering43
Figure 2-4 Full-Time Engineering Enrolments in the IoTs44
Figure 2-5 Part-Time Engineering Enrolments in the IoTs44
Figure 3-1 Values of the Framework69
Figure 4-1 Arena Created by the Pedagogic Device98
Figure 4-2 The Pedagogic Device of IoT Engineering Education, as Shaped by the NFQ and EI,
alongside other Policy Drivers103
Figure 5-1 Three C's of Data Analysis, Codes, Categories and Concepts122
Figure 11-1 The Pedagogic Device of Engineering Education in the IoT sector as Revealed
through the Analysis231

Tables

Table 3-1 Engineering Standards	.74
Table 5-1 Differences between Non-Probability and Probability Sampling 1	110
Table 6-1 Influence of the NFQ on the Interviewees 1	135
Table 6-2 Prescription and Influence from the NFQ and EI over Course Design 1	147
Table 7-1 The Use of the Language of Levels 1	151
Table 8-1 Influences of Progression on Engineering Education in the IoT Sector	183
Table 10-1 Recontextualisation of Knowledge 2	201
Table A-1 Level Indicators for NQF Levels 6-9	243
Table D-1 Enrolments in Full-Time Engineering Programmes in the IoT Sector2	249
Table D-2 Enrolments in Part-Time Engineering Programmes in the IoT Sector2	249
Table E-1 Comparison of the Bologna Framework, the EQF and the NFQ2	251

Chapter 1 Introduction

Engineering faculty, of course, will be on the front line of any change. National Academy of Engineering (NAE, 2001, p. 23).

1.1 The Research Question

This thesis considers the effect the introduction of outcomes based education (OBE) has had, and is having, on engineering educators and engineering education in the Institutes of Technology (IoT) sector in Ireland. I conducted this research from the perspective of an IoT engineering educator seeking to understand how we have been shaped by the introduction of OBE in engineering education, and how this is affecting the education of future generations of engineers. My research employed a qualitative methodology, in which I considered the impact of OBE as perceived and experienced by a sample of engineering academics drawn from the IoT sector.

Engineering is concerned with the creation and maintenance of artefacts for practical usage, through the use of tools, technology, methodology, skills and knowledge, where, however, the engineer is concerned not with "better and deeper knowledge, but better ends" (Poser, 1998, p. 85). Engineering education, with which my thesis is concerned, and which broadly comprises professional and technician engineering education, provides graduates to meet the market requirement for engineers. In Ireland, the IoTs and Technological University Dublin provide both technician and professional engineering education, whereas the traditional universities focus on the latter.

In order to position my research in relation to the field I consider the expansion of activity in engineering education research (EER) since the mid-1990s, initially in the USA but later on an international basis, and in which I note the development of an OBE approach to engineering accreditation (Riley, 2012) was highly influential. EER into the use of learning outcomes (LOs) largely focuses on how teaching may better align with their use. However, the focus of this study, the effect of OBE on both engineering educators and engineering education, receives little attention in the literature.

The literature review considers the formation of related educational policy, the policies themselves, and their implementation, beginning with developments leading to the creation of the technician education system in the Regional Technical Colleges (RTCs) in the 1960s. The

literature review highlights the importance of considering the role of market forces alongside the academic aspects of engineering education. This can be conceptualised through Bernstein's (2000) characterisation of engineering as a 'region', facing inward to academia and outward to the market, mediated by the professional engineering bodies. This led me to choose to draw selectively on Bernstein and other social realist researchers, alongside my experience as an engineering educator, in the research design and analysis. This allowed me to conduct the research from a standpoint that emphasises the importance of considering the role of power relationships in the adoption of OBE, and the influence this might have on the structure of engineering education.

My role as a head of department of engineering (HoD) in the IoT sector provided me with the opportunity to gain privileged access (Brinkman & Kvale, 2011) to my colleagues' experience of OBE, through fieldwork consisting of individual interviews and a focus group. This highlighted the effect OBE was having on their academic identity, including their approach to knowledge, curriculum, and pedagogy. It also provided insights into their views of the structure of our engineering education system and the identity formation of our students.

1.2 Outcomes Based Engineering Education

In 2004 IoT engineering transitioned to OBE (Riley, 2012, Walsh, 2018), aligned with the levels of Ireland's National Framework of Qualifications (NFQ) (QQI, 2014b), and Engineers Ireland's (El's) LO based accreditation processes (EI, 2014). LOs describe education in terms of competencies, what a graduate will know and be able to do when they complete a programme of study, or what a student will know and be able to do on passing a module.

The NFQ (QQI, 2014b) has 10 levels (see Figure 1-1), with qualifications placed at the appropriate level through being described in terms of LOs that must be achieved in order to attain the award. In the IoTs, awards are made at level 6 (higher certificate), level 7 (ordinary degree), level 8 (honours degree), level 9 (master's) and level 10 (doctorate). Engineers Ireland (EI) accreditation, which is now outcomes based, is the other key driving factor. Accreditation by professional bodies of qualifications from higher education institutes is important for engineering education, and is regulated internationally through a series of accords to which EI is a signatory (IEA, 2016). Although EI accreditation does not quite fall into

the category of a compulsory accreditation process (Friedman et al, 2017), in practice all of engineering higher education in Ireland engages with it (EI, 2018).

As an engineering academic and academic manager I helped implement the transition to OBE for engineering education. I encouraged, led, and cajoled on occasion, my department staff into the use of LOs, in replacement for our previous, content, based curriculum approach. I initially accepted OBE in the positive, progressive, terms in which it was presented, although I also associated it with the ongoing pervasive expansion of quality assurance into academia (Kenny, 2010).



Figure 1-1 NFQ Level Diagram (QQI, 2016a)

We engaged with LOs on the basis that they were government policy (NQAI, 2002b), that they were enshrined in statute (Irish Statute Book, 1999b) and that they had become the cornerstone of the professional recognition of our programmes (EI, 2014). The use of LOs was going to make it clearer to prospective students, employers, parents and professional bodies what it is that graduates of our programmes would be able to do (NQAI, 2003a). LOs would assist lecturing staff in selecting what they should teach, how they should assess, and make programme design more straightforward (Kennedy et al, 2006). Their encapsulation within the NFQ would make for clearer progression pathways for students (DES, 1998) and provide a basis for the up-skilling of the population as required for the' knowledge economy' (HESG,

Introduction

2011). The NFQ would be part of "the development of a lifelong-learning society" (NQAI, 2003a, p. 10), and would place the learner at the centre of education.

As I read deeper into the topic, and as I carried out the fieldwork for the thesis, discussing with my peers their experience of OBE, I came to the view that OBE is associated with fundamental change in IoT engineering education. Part of the reason OBE has been widely accepted by engineering faculty may be as simple as 'engineers like frameworks' (i.e. the idea that engineers have an affinity for clear and defined structure), but, regardless, we appear to me to have, at this stage, wholeheartedly adopted a LOs approach to education. My research suggests that, 12 years after the Collins et al (2009) report, which found that academics had not then fully bought into the LOs approach, the situation has changed quite considerably.

With hindsight, I believe that neither I nor my colleagues took too much trouble to consider whether OBE would really have all the positive influences that were put forward for its adoption, or indeed, what other consequences there might be. What was the effect on the education we were trying to offer and on the type of graduate we were hoping to produce? How was this affecting our pedagogy as engineering academics, and what influence might it be having on our approach to our epistemology in our own particular fields of engineering?

A further issue important to this research concerns the stratification of engineering education into technician and professional engineering education (Dempsey, 2017, 2018; McLaughlin, 1999; NAE, 2017), the latter of which is part of the pathway to Chartered Engineer. Prior to the introduction of the NFQ, the IoTs offered ladder programmes of engineering education (McLaughlin, 1999). This allowed those who technician graduates to continue their studies to gain honours engineering degrees, placing them on the path to Chartered Engineer. This is in contrast to the traditional universities, where all undergraduate engineering education enrolments are on honours degrees. The introduction of the NFQ, which promoted improved progression opportunities for students, would have been expected to reinforce this approach by the IoTs to engineering education. However, in parallel, and as a counter pressure to the NFQ concept of progression, EI, in adopting OBE for accreditation, increased the educational requirements for Chartered Engineering to master's degree (EI, 2012). My research provided an opportunity, through working with my peers in IoT engineering education, to explore their impressions of the effect of this, and related, issues. The research takes place in a context where the use of OBE is only part of the pressure on Irish engineering education, only one of the tools used to promote the skills agenda of government policy towards higher education (DES, 2016; HESG, 2011), and there is also pressure from industry (IBEC, 2015). This thesis will show that there is a longstanding policy trail¹ emphasising the linking of education in the IoTs to industry requirements, beginning with the very establishment of the sector (the Regional Technical Colleges). My research, through considering the effects of OBE on IoT engineering education, serves partly to examine the overall effect of government policy, as influenced by the market, on educational practice.

The use of OBE can be regarded as a manifestation of neoliberal influences on Irish higher education (Lynch, 2012), and in general on engineering education (Matos et al, 2017). Neoliberalism is a phrase which carries ideological connotations of the power of global capitalism in influencing and directing education policy, with a corresponding influence on public policy and education in Ireland (Lynch, 2012). A neoliberal perspective on education is that it should align with labour market requirements, emphasising accountability and the skills development of the individual for employment, rather than the content of the curriculum being led by disciplinary experts in universities or other educational institutes (Allais, 2014). OBE, with its output emphasis, is attractive from this perspective where it is seen to "ensure that curricula are more responsive to the needs of employers and learners" (Allais, 2014, p. 51). However, as will be further discussed later, OBE is also attractive to educational reformers through offering a 'student-centred' approach, and improved access to education (Allais, 2014). In Ireland this has meant that the NFQ also received support, indeed welcome, from this perspective, although, over time, the dominance of the neoliberal drivers, the reality of the industry/jobs/output focus, and the controlling boundaries the NFQ places around education, became apparent to some (Finnegan, 2016; Fitzsimons, 2017a).

This research speaks to my lived experience as an engineering academic. As an insider (Leckie, 2001), in researching the effect of OBE on engineering academia I was researching the effect it had had on me. Given my role as an academic manager, I was also researching the effect that I, in promoting the use of LOs, had had on others. This implied that, as I conducted what became at times a quite critical study of the effect of OBE, I would have to acknowledge that

¹ The *policy trail* concept takes cognizance of the increased importance of market influences and transnational organisations in determining government policy (Cort, 2014).

I had helped to implement something that I began to question some aspects of quite strongly.

1.3 Being an Engineering Educator

Qualitative research requires that the researcher situate themselves in relationship to the study and those being studied (Denzin & Lincoln, 2011; Leckie, 2001). As noted in O'Reilly and Kiyinba (2015), a qualitative researcher's axiology influences their view of the world they are researching, and their approach to the research process. For example, McNutt (2010) relates an insight in his methodology development, the need to include a "key component in this work – myself - my own beliefs, values and assumptions" (p. 78). In my research I had to ensure that I understood my values and beliefs with regard to education and engineering, and how they had been influenced by the LOs approach. These values and beliefs, which form my personal axiology, have been developed through my time as an engineering student, an engineering practitioner in industry, and through my time in academia.

My engineering education was oriented towards engineering science and mathematics, with some elements of engineering practice, inculcating strong positivist and rationalist facets to my epistemology, alongside some constructivist influence. Knowledge, aligned with the engineering worldview, was that which was seen to be useful (de Vries, 2006; Gallery, 1989; Korte, 2015; Montfort et al, 2014).

My subsequent industrial experience as a professional engineer and research scientist with Philips Research Labs (Walling, 2005) initially emphasised the development of my engineering problem solving abilities (Fink et al, 2005). However, I grew beyond that later in research into virtual reality (VR) (Gallery & Gibson, 1993), where we considered epistemological and ontological aspects of immersive VR systems (Brey, 2014; Gallery, 1999; Morie, 2008; Niiniluoto, 2011). In our research we were conceptualising new realities (Gallery, 1996), investigating the challenges within them, and the experiences people could have (Crogan, 2010), rather than regarding VR as an engineering problem to solve (Gallery & Gibson, 1993). This inculcated within me the importance of relating engineering knowledge to engineering practice, which is something I have brought with me to academia. However, my work in VR also made me more aware of who I was doing the engineering for, which had been somewhat lacking in my earlier engineering education (Johnston et al, 1996).

My principal experience as an academic has been acquired since my time in industry, joining

the nascent Institute of Technology Blanchardstown (ITB) (Irish Statute Book, 1999a) in 1999. Initially I focussed on "becoming a good academic [...] caring for the students, being research active and finding joy and contentment in this work" (Fitzmaurice, 2013, p. 621). In common with other academics (O'Byrne, 2009), I emphasised the relevance of my industrial experience in teaching, developing a pastoral care element to my professional identity, being interested in research, and actively participating in course design.

Becoming head of department (HoD) of engineering brought new responsibilities. I am expected to have considerable leadership skills (O'Sullivan, 2014), above and beyond that of academic leadership, with staff expecting the HoD to represent the department to advance its cause, provide strategic direction, to act as a role model and to protect staff autonomy, amongst others. However, the day to day role is dominated by administration and fire-fighting, where, it is suggested, that HoD's are "increasingly involved in administration to the detriment of leadership" (O'Sullivan, 2014, p. 182), which administrative duties often come under the guise of quality assurance (Kenny, 2010; Kenny et al, 2015). I feel that a HoD is a difficult position: where workload leads to a situation where we do not always provide the leadership we should to our staff, and restricts the time we can spend on research or teaching, i.e. where it is "difficult to balance each aspect" (Kenny et al, 2015, p. 92).

Much of my time is spent on matters concerning programme design and delivery. As an IoT engineering department we have focussed on programmes which offer students technology oriented careers, responsive to industry requirements. Within the programmes I have helped design and deliver we emphasise the use of theory and practice together. Whilst I can identify significant positivist and rationalist aspects in our programmes, there is also a pragmatist influence (Korte, 2015). This includes the use of problem based learning (McCabe et al, 2009), famously described and theorised by Dewey (1916). I view our graduates as differently positioned than I was on graduation, in particular more knowledgeable of engineering practice. Edström (2017) considers engineering education in terms of the tension between academic and professional values, "the analytical and practice orientation" (p. 40). I believe my educational values in this regard have largely achieved balance, reflected in a pragmatist facet in my epistemology, which brings more weight to the linking of theory with practice.

I thus approached this research on a personal level as an engineering educator immersed in the field I am researching, and, on a theoretical level, as an engineering educator conducting practitioner research using ideas from social realism to assist me in trying to understand that which I am researching. My epistemology combines elements of rationalism, positivism, constructivism (pragmatism) and managerialism, which in themselves can have tensions, for example between positivism and constructivism. However, they also lead to different expectations within an OBE engineering education.

From a managerial perspective, OBE provides an industry focus to our educational efforts, valuing the job-readiness of our graduates, which, at one level I welcome. However, as I will demonstrate later, it leads to contradictory policy goals that act against the promised efficacy.

From a rationalist perspective I regard it as important that engineering education has an emphasis on theory, in particular mathematics, whereas from a positivist perspective I look for an emphasis on experimentation and verification of theoretical postulates. Conducting this research has caused me to reflect on the underpinnings of various aspects of my work as an engineering educator in a new way, and, as I will argue later, the LOs concept is overly skills focussed, at the expense of theoretical knowledge. My constructivist aspect sees some benefit from competencies defined through LOs, given skills are important not only to technicians, but to all engineers. The LOs concept also takes into consideration that there is more than one way for an individual to construct their personal engineering worldview, which mirrors my experience as an educator.

1.4 The Fieldwork for the Thesis

The primary fieldwork consisted of a series of semi-structured interviews with fifteen engineering academics drawn from six of the fourteen IoTs. The majority of my research participants were engineering lecturers, from whom I was seeking to gain insight into their perception of the effect of OBE on themselves and the education they offer. I also invited some engineering academic managers² to take part, as they would provide a complementary viewpoint of their experience of helping to lead the implementation of OBE.

I chose a one-to-one interview approach for data-gathering to allow me to explore details of the effect that each of the interviewees perceived from the use of OBE. Although I framed the interviews within a consideration of their experience of OBE, the interviews also allowed me

² In the IoT sector the Engineering academic manager role, graded as Senior Lecturer 2 or 3, comprises both academic and management duties.

Introduction

to discuss their impression of more general effects on engineering education. I followed an open coding approach (Lichtman, 2014; Saldaña, 2009) in the initial analysis of the interviews.

I used a focus group formed from the interviewees to review my draft findings (Kamberlis & Dimitriadis, 2011). This enabled me to provide an opportunity for feedback to those who had been so helpful in my fieldwork regarding what had emerged from the initial analysis. In the focus group we explored the validity of my draft findings, and their group voice on these matters emerged. It provided a different scenario to the individual interviews, allowing points to be raised and emphasised within the focus group that might not have emerged otherwise. It also provided a political and pedagogising opportunity related to my findings. As I had begun to draw conclusions with regard to my research, I found that I actively sought out conversations with my colleagues with regard to my draft findings, on the basis that I felt that these were things that ought to be known and discussed, and consequently, I welcomed the opportunity to present these to the group. A draft finding from the analysis of the interviews was that a *language of levels* had emerged in the pedagogic discourse, and the focus group, in addition to providing insight into the drivers for its use, also allowed me to regard this language in action as my focus group participants made use of it in our discussion.

1.5 Drawing on Social Realism

A key driver for OBE in IoT engineering education that my research identified is the government higher education policy emphasis, driven through economic concerns, on skills. Allais (2014), in considering the policy drivers for OBE, writes that the "emphasis on the intertwining of education and economy explains the focus of policy makers on qualification reform and outcomes-based qualifications frameworks" (p. xvii). It also became apparent that market influences had been similarly influential in the adoption of OBE for engineering accreditation. As I have indicated, this can be conceptualised through Bernstein's (2000) characterisation of engineering as a region, which recognises the influence of both external and cognitive interests on engineering education.

Alongside my personal epistemology, my positioning as an engineering educator interested in policy formation and the impact on engineering education, I drew selectively on social realism to consider the policies that led to the adoption of an OBE approach to engineering education in the IoTs and the subsequent implications. Young (2008), in describing the basis

Introduction

for a social realist theory of knowledge, emphasised the importance of "detailed historical and ethnographic studies that can make explicit the contested character of intellectual fields" (p. 29). This suggested that, in order to answer my research question, it would be crucial to understand the socio-historical context (Young & Muller, 2010) in which IoT engineering education is situated. This concurs with Edström (2018), who, writing from an EER perspective, considers it "makes sense to develop a historical awareness, particularly to illuminate current efforts of engineering education development" (p. 38), identifying the need to consider organisations, vested interests and competing initiatives that contribute to this.

A social realist position acknowledges the social nature of knowledge, whilst at the same time accepts that there is "context-independent knowledge" (Young, 2008, p. xii). This leads to concern not just with what constitutes knowledge but also with the social-political issues and structures within which it is framed (Barrett & Rata, 2014a). In emphasising the importance of power relationships in determining access to knowledge through the curriculum (Bernstein, 2000), social realism provides a perspective from which to consider the formation and implementation of educational policy, and the stratification of knowledge in educational systems such as engineering.

Bernstein's (2000) pedagogic device models the power relationships and communication structures of an education system as a pedagogic discourse. The means of communication, and who controls them, are considered as important as the message. The pedagogic discourse that arises is regarded as constituted through the official knowledge³ of the official recontextualising field⁴ (ORF), alongside the instructional discourse of the pedagogic recontextualising field (PRF). Engineering education such as that in the IoTs can be represented as a pedagogic device (Klassen, 2019; Moodley, 2014), where official knowledge of the state, EI, and the market, influences the cognitive interests of engineering academics and the identity formation of engineering staff and students.

Initially I struggled with how to approach my research from a theoretical perspective. Drawing selectively on social realism, and in particular the pedagogic device, led to a resolution,

³ Bernstein used the phrase 'official knowledge' to refer, in an educational context, to government policy and associated organisational and legislative structures.

⁴ Recontextualisation was used by Bernstein (2000) to refer to the manner in which curriculum is formed from disciplinary knowledge and additionally the instructional approach that is used to teach it.

providing an analytical framework within which my fieldwork could be considered (Bernstein, 2000) and from which the impact my interviewees perceive from OBE could be regarded. It not only allowed me to draw on my experience and knowledge as an engineering educator in the research, but, in emphasising the importance of the socio-historical context, and the power-relationships and vested interests within the field, made it essential that I do so.

1.6 Why This Research is Important

This research is primarily for myself, as I seek to become a better engineering educator through developing a fuller understanding of the challenges, tensions, and opportunities facing the sector within which I work, and the power relationships which act to determine and influence those. In providing a perspective of the impact that OBE has had on engineering education I also intend it to be of value to my colleagues. It suggests areas where engineering academics could consider how their identities may have been shaped by the use of OBE, and may provide insight into their own professional development. I provide engineering academic management with a perspective which they may find of value in considering the profile of their staff, the structure and aims of the engineering education they offer, and how they might approach programme design and renewal in such a system. It will also inform their approach to staff recruitment, and their provision of professional development opportunities for staff. Given that engineering education is a significant part of the IoT sector, this research is also of relevance to wider academic management in the IoTs.

My research is a contribution to policy analysis in relation to engineering education. In researching the development of the NFQ, and how, alongside El's LOs based accreditation, it has affected engineering academics and engineering education, it provides a case-study of the different manners in which engineering educational policy can impact on practice.

The research also informs my colleagues in their role as members of EI, the engineering professional body, with regard to the influence that my research suggests LOs based accreditation is having on the structure and aims of engineering education.

Engineering education researchers will also find my research of relevance. The critical perspective into the impact on academics from the use of OBE shines a light on this underresearched facet of the field of engineering education research (EER).

Introduction

This research, which grew from my own interests as an engineering educator practitionerresearcher, focusses on the impact of OBE on our engineering education system. However, it will also be of broader interest to educational researchers considering the impact of OBE on academic identity. Building on earlier research on changes in academic identity in the IoT sector (Hazelkorn & Moynihan, 2010; Kelly, K, 2009; Kelly, M, 2005; Kenny et al, 2015; Moynihan, 2015; O'Byrne, 2009, 2011), my research investigates the impact on academics' perceptions of knowledge, their pedagogy and the balance of knowledge and skills in the curriculum they design.

This research is also of interest to social realist scholars. It shows how the development of a socio-historical perspective (Young, 2008; Young & Muller, 2010) of technical engineering education in Ireland was key in allowing me to contextualise my research into the impact of OBE on that education system. My research can be regarded as a case-study which draws selectively on aspects of social realism as a lens through which to consider the experience of a sample of engineering academics of OBE. In using Bernstein's pedagogic device as an analytical framework in qualitative research into IoT engineering education, my research extends earlier research (Klassen, 2018; Moodley, 2014) which proposed its applicability as a theoretical model for engineering education systems.

1.7 Key Findings and Contributions to the Literature

The engineering education literature contains some critique of LOs (Heywood, 2016; Wolff, 2015; Woolston, 2008), but generally reflects a positive view of their effectiveness (ABET, 2006; El, 2017a; Froyd et al, 2012; Owens, 2016a). Through taking a critical perspective, this thesis contributes to the field of EER in providing insight into the impact of OBE on the identity formation of engineering educators and on the nature and structure of engineering educators. My research suggests that the adoption of OBE has had profound effects on IoT engineering educators and educator, although the fieldwork provides evidence that my interviewees regard the NFQ and EI as only some of the drivers for change that they experience.

A finding from the fieldwork is that my research participants have given a positive, if at times qualified, acceptance to the NFQ and LOs, regarding this as an effective aspect of the overall structure of IoT engineering education. This effectiveness is perceived in: heightened

Introduction

consideration of pedagogy; promoting academic rigour in curriculum design; clarity for students on what is expected from them; and improved equity of access to education and progression. However, my research also sets out a critical view of the impact that can, at least partially, be attributed to OBE, including: regarding the type of pedagogy that has resulted; the appropriateness of the NFQ derived language of levels that has come into usage; and the impact on the technician identity and the structure of engineering education. A contribution to the literature is to demonstrate the manner in which OBE has been largely positively accepted by my research participants as part of the fabric of IoT engineering education

The professional bodies have a significant supporting role as communities of trust in the adoption of OBE (Tuck, 2007; Young, 2008), particularly in engineering education (Brent & Felder, 2003; Heywood, 2005, 2016). El promote the adoption of LOs through their accreditation processes, which forms part of the official knowledge of IoT engineering education (Klassen, 2018; Moodley, 2014). The fieldwork provides evidence that this has been a significant factor in my interviewees' acceptance of OBE, particularly for course design. A contribution to the literature is to provide evidence, through the views and practice of my interviewees, to support the contention of the key role EI, as an engineering professional body (ABET, 2006; Brent & Felder, 2003; Heywood, 2005, 2016; McLaughlin, 2001; Owens, 2016a), play in the adoption of OBE by engineering academics.

Changes in the pedagogic identity of faculty as the IoTs expanded their role (Hazelkorn & Moynihan, 2010; Moynihan, 2015; O'Byrne, 2009, 2011, 2014) can be regarded through the concept of pedagogic identity changing under the influence of the pedagogic discourse (Bernstein, 2000). The workload associated with the use of LOs is often characterised as quality assurance (QA) (Kelly, 2009; Kenny, 2009). However, the use of LOs is much more than a QA and workload matter. As part of the official knowledge that acts to assert control over the pedagogic discourse, it would be expected to have a significant effect on the manner in which engineering academics approach their role. The fieldwork provided evidence of my interviewees actively changing their pedagogic identity under the influence of OBE. Findings relate to the usage of the language of levels in their pedagogic discourse; the impact on their pedagogic practice itself; their view of the balance of knowledge and skills; and the impact on their regard for knowledge and the curriculum.

Kenny (2006) reported the embedding of LOs in the academic language of Irish higher education, as was revealed as an intended outcome of the NFQ by Collins et al (2009), and as is proposed will, in general, result from the use of qualification frameworks (Allais, 2010b; Raffe, 2011). A finding from the fieldwork is that my research participants have adopted the use of a language of (NFQ) levels in their pedagogic discourse. They use it as a language of engagement with the regulative, or policy, discourse, and also in the instructional discourse of their academic practice. A contribution to the literature is to provide evidence, through the fieldwork, consistent with Biesta's (2005) learnification, that suggests the emergence of a language of levels (Raffe, 2011) in consequence to the introduction of the NFQ.

A finding is to show that pedagogy received little attention during the development of the NFQ (NQAI, 2001a, 2001b, 2002c, 2003a), although a few voices raised concern (APEL, 2001; Connolly, 2001; HETAC, 2001; ITT, 2001). Considering IoT engineering education through the lens of the pedagogic device leads to the expectation that, constituted as part of official knowledge, OBE would have a significant impact on pedagogy. The fieldwork reveals that the majority of my interviewees acknowledge the implicit influence of the NFQ over their pedagogy, regarding this in a positive and constructive manner. For my research participants the use of LOs appears to promote a teaching to assessment oriented pedagogical approach. Where previously we assessed what we taught, now we teach towards what and how we will assess. Relating my findings to the literature (Deacon & Parker, 1999; Havnes & Prøitz, 2016; Muller, 1998; Rajaee et al, 2013; Torrance, 2007), suggests that, consequent to OBE, my interviewees have adopted performance oriented pedagogies (Bernstein, 2000). A contribution to the literature is to provide qualitative evidence, supported by related studies (Muller, 2004a; Rami, 2012) to support the contention (Muller, 1998) that a potential consequence of OBE is to influence educators to adopt performance pedagogical approaches.

A set of related findings emerged with regard to my interviewees' views on the role of OBE in curriculum. For the majority, the NFQ has not had a significant impact on the acquisition of new knowledge. However, they now perceive LOs (influenced by the NFQ and EI) as the key driver for the recontextualisation of knowledge in the curriculum, with, for some, knowledge secondary to LOs. Although the use of LOs was seen as a prescriptive process, for some academics, LOs themselves, if not written to be overly precise, can be enabling, facilitating them in bringing their individual expertise to teaching. A contribution to the literature is to

provide evidence, based on the views of my research participants, of the strong influence they experience from OBE over curriculum design, and the different modalities this takes.

Progression, in providing all students the opportunity to progress to level 8 degrees, acts to reconfigure the stratification inherent in engineering education. However, through the literature review I show that El's interpretation of the Bologna declaration (European Ministers of Education, 1999) was to raise the educational requirement for Chartered Engineer to master's degree (IEI, 2001d). This precipitated a structural effect on IoT engineering education, reinforcing the stratification of engineering education into technician and professional engineering. IoT engineering students pursuing five years honours degree were now required to spend at least another year in education in order to continue on the track to Chartered Engineering. The impracticality of this resulted in many such degrees being discontinued, as predicted by McLaughlin (2001), with the four year honours degrees that replaced them not suitable for accreditation towards chartered status. A related contribution to the literature is to demonstrate the dramatic change, since the NFQ's introduction, in the orientation of IoT engineering students are now enrolled on honours degrees (HEA, 2018).

A finding from the fieldwork is that, for my interviewees, progression is perceived as causing technician programmes to orient towards serving as pathways to higher qualifications. This is a contribution to the literature in providing qualitative evidence to support research that suggests that level 7 engineering programmes were moving to become precursors to level 8 programmes rather than educational ends in themselves (Llorens et al, 2014).

My research confirms that the NFQ is one of the principal tools to enable the government skills agenda for higher education, and a contribution to the literature is to confirm, in applying Corbell's (2014) methodology to NQAI (2003a), the skills focus of the NFQ. The international experience is that NQFs are often associated with policy conflicts (Allais, 2010a; Fernie & Pilcher, 2009; Havnes & Prøitz, 2016), and I show that in the case of the NFQ conflicting government policies are evident, as reflected in the experience of my research participants. The national requirement for a continuing supply of level 6 and 7 technician graduates (DES, 2016) appears at odds with the evidence from my interviewees that the NFQ concept of progression (NQAI, 2003a) has led to a more theoretical, less skills focussed, technician education. However, the analysis revealed conflicting opinions amongst my

interviewees with regard to the balance of knowledge and skills in curriculum: some considered there to be an insufficient emphasis on skills, whereas others felt the NFQ promoted an appropriate balance. I suggest the policy conflict between maintaining the supply of technicians (DES, 2016) and progression (NQAI, 2003a) is reflected in these differing attitudes. A contribution to the literature is to suggest this represents a manifestation of the pedagogic schizoid position (Bernstein, 2000) where market pressures conflict with the discourse of the discipline: contradictory educational policies simultaneously promote more skills, whilst orienting programmes to become more theoretical.

I will make the case that Bernstein's pedagogic device provides a framework within which the impact of the NFQ, and EI's accreditation requirements, on a sample of IoT engineering academics, can be regarded and critiqued. The views and practice of my research participants lead to a further claim, in that they provide evidence that in adopting performance pedagogical approaches under the influence of assessment focused OBE, the field of reproduction of the pedagogic device has become oriented towards a field of evaluation.

A further contribution to the literature is from a policy perspective, where I provide a critique of the effectiveness of OBE from the perspective of the technical engineering education sector in Ireland, which stands in comparison with QQI's largely positive recent review of the impact of the NFQ (QQI, 2017), and EI's similarly positive view of OBE (EI, 2014).

1.8 Structure of the Thesis

Chapter 1 introduces the research question, and how I am situated with regard to it as a practitioner researcher. I explain why this research is important, and who it may benefit. I provide an overview of my research methodology, findings and contributions to the literature.

In *Chapter 2* I provide an overview of engineering education, and I establish how my research is positioned with regard to the field of EER. I provide a brief history of the genesis of Irish engineering education, and describe the development in the 1960's of the RTC technician education system. I discuss the contemporary IoT engineering education system that has evolved from this, with particular attention to changes post-NFQ, and the pressures we experience as IoT engineering academics.

Chapter 3 describes the global rise of OBE, including for engineering accreditation. I discuss the development of the NFQ as part of a skills focussed re-orientation of Irish higher

education, and EI's adoption of OBE for accreditation of engineering programmes.

In *Chapter 4* I develop a conceptual framework based upon my experience and epistemology as an engineering educator interested in research into engineering education, drawing selectively on social realism as a theoretical framework.

Chapter 5 details my research methodology, a qualitative approach using interviews and a focus-group. I discuss the choice of data gathering and analysis techniques, selection of research participants, ethical considerations, and the implementation of the methodology.

This is followed by five chapters in which voice is given to the research participants' views of the impact of OBE, on themselves, their colleagues, and on our engineering education.

In *Chapter 6* I illustrate the manner in which the influence and prescription of the NFQ over IoT engineering education is perceived by my research participants. EI is shown to be a significant factor in IoT engineering academics largely positive acceptance of OBE.

Chapter 7 reveals the use by my research participants of a language of levels derived from the terminology of the NFQ, as they engage with the pedagogic discourse.

Chapter 8 examines the impact of progression, revealing my interviewees' perception that level 8 honours degree awards are the aspirational target for all undergraduates, with higher certificates and ordinary degrees now stepping stones towards this. The analysis considers the effect on how and what students are taught.

Chapter 9 reveals the change in the pedagogy of my research participants subsequent to the change to an OBE approach. The analysis shows that they have consequently adopted pedagogical approaches focused around teaching towards assessment.

Chapter 10 considers how OBE has affected my interviewees' relationship to knowledge, skills, and the curriculum. I examine their approach to knowledge acquisition, and explore their views of the appropriate balance of knowledge and skills in engineering education.

Chapter 11 discusses the research findings through the themes of: the effect of OBE on IoT engineering education; the overall effectiveness of the NFQ; and the implications for the relationship between engineering and society. I include a reflection on the use of the pedagogic device as a model for IoT engineering education.

Chapter 12 discusses my research journey, and describes limitations of the study. I present the contributions to the literature and discuss their significance. I provide suggestions for future directions of related research. I conclude with a reflection on what I have learned personally through carrying out the research.

Chapter 2 Engineering Education and Ireland's Technological Education Sector

Deprived of the past, the moment - the present -has little meaning, if any. Philip K. Dick (1966. p. 17).

2.1 Introduction

In this chapter I consider the development and evolution of the engineering education system in the Institute of Technology (IoT) sector in Ireland. In order to contextualise the research, I first discuss what is meant by engineering, which is stratified into technician and professional engineering. I argue that the epistemologies of engineers are multi-faceted, the influence of which is apparent in engineering education. This education is also strongly shaped through engineering's *raison d'etre* of productivity improvement (Trevelyan, 2019), encompassing the design, production and maintenance of artefacts for practical usage, and thus by its ties to the market. I explain how my research is situated in the field of engineering education research (EER).

Engineering education in Ireland prior to the establishment of the Regional Technical Colleges (RTCs) consisted of university education, principally accessible to the middle and upper classes, and a small, Dublin-centric, technical college sector. The formation of the latter was associated with the 19th century Mechanics' Institutes, which provided technical instruction to artisans, and which were themselves an arena of class-struggle and ideological conflict.

I discuss the socio-historical events, beginning in the 1960s, which led to the formation of an engineering technician education system in the RTCs, which were later to become the Institutes of Technology (IoTs). A guiding theme that emerges from my research was the national requirement for a technician education system to support an expanding economy, coupled with a desire to increase access to education for under-represented socio-economic groups. This mirrored the reasons for the development of binary higher education systems across Europe during the 1960s and 1970s (Christensen & Newberry, 2015; Triventi, 2013).

I review the development of engineering education in the IoTs, from conception through to the current day, and I present the significant changes to enrolment patterns that have occurred in recent years. I outline the role of engineering academics in the IoTs, drawing on the literature and my personal experience, and I discuss changes in our academic identity as the RTC/IoT sector expanded its role.
2.2 Engineering and Engineering Education

Engineering can be described as the "planning, designing, building and operation of efficient and economic structures, machines, processes and systems, based on the application of mathematics and natural sciences combined with knowledge of technologies and exercised with judgement and creativity" (Christensen et al, 2007a, p. 411).

Engineering can be considered to consist of four main engineering categories, chemical, civil, electrical and mechanical, each with their own sub-fields (WhatIsEngineering, 2016), although as technology and its application in society evolves a multidisciplinary approach can be regarded as more relevant rather than "focussed practice within traditional disciplines" (Duderstadt, 2008, p. 3). The different categories reflect industry requirements for engineering specialists, and, generally, are associated with dedicated courses in higher education (El, 2019c). From an educational and career perspective, engineering is stratified into professional (or chartered) engineering, and engineering technician/technologists (Dempsey, 2017; McLaughlin, 1999; NAE, 2017). Professional engineering management, costing, maintenance, or production. Typically, professional engineers are more theoretically oriented, and technicians/technologists more practice oriented (Land, 2012).

Through their education, and later professional practice, engineers experience "an identity formation process involving an acquisition of engineering discourse and engineering ways of thinking and doing" (Haase, 2014, p. 84). An engineer can be characterised as having a distinctive mind-set, or engineering habits of mind, which reflect these influences, where "thinking like an engineer" (RAE, 2014; p. 1) encompasses:

- systems thinking: being able to recognise interconnections between parts of a whole, and how these parts may interact;
- 2. adapting, or being able to test, analyse, reflect and reconceptualise;
- 3. problem-finding: being able to clarify requirements, verify existing solutions, and understand the context in which a problem is situated;
- 4. creative problem-solving: applying design techniques to solve engineering problems;
- 5. visualising: being able to relate abstract concepts to practical solutions and the understanding of the design processes associated with this;
- 6. improving: the process of applying engineering techniques to improve existing solutions and prototype new ones:

In order to understand this more fully I examine what these engineering ways of thinking and doing might be through considering engineering from a philosophical perspective. Considered somewhat of "a neglected sphere of interest in the thinking of engineering educators" (Heywood, 2011, section *Introduction*), in more recent times the role of philosophy in engineering has gained some attention (Christensen et al, 2007b; Guy, 2010; Heywood, 2011, 2016; RAE, 2008, 2010a, 2010b).

Comparison is often made between the philosophical basis of science and engineering (Coyle et al, 2007; Heywood, 2011; Lipton, 2010; Poser, 1998), where although both "are interested in the truth, they may not be interested in the same truths" (Lipton, 2010, p. 13). Physicists seek ways in which the world can be described and explained, through "theory - a set of propositions, a set of equations, a set of assertions" (Lipton, 2010, p. 8). For engineers "the ultimate output is an artefact" (p. 8), with engineering science primarily concerned as to how something may produce a useful effect, as opposed to why it produces that effect (Coyle et al, 2007).

Engineering can be conceptualised through a number of philosophical positions, including rationalism, positivism, empiricism, constructivism and in particular pragmatism (Korte, 2015), existentialism and transcendental idealism (Figueiredo, 2008; Grimson, 2007). A model where the engineer is, simultaneously, a sociologist, a scientist, a designer and a doer is suggested by Figueiredo (2008), which sees the engineer as working with epistemological stances appropriate to each facet. Grimson (2007) argues that "engineering uses knowledge in all its various forms and no special allegiance can be given to any one epistemological theory" (p. 99). He argues that, whilst engineering draws from science, art, nature, mathematics and architecture, it has its own "distinguishing features" (p. 90) from each of these, highlighting engineering's use of rules of thumb and approximations alongside theory, where:

the very essence that is engineering – to proceed at all, some assumptions or approximations have to be made if 'things' are to be designed and built. And there is great art in being able to use gainfully those theories that are known to be imperfect and to judge the extent to which rules of thumb may be safely deployed (p. 90).

Grimson's (2007) contention that engineering is not reducible to being considered from a single philosophical position is commensurate with my own experience (Gallery, 2015) that being an engineer encompasses being able to work with consideration of various philosophies

and epistemologies. However, the strong positivist and rationalist bases must be acknowledged, alongside constructivist influences.

Wolff (2015) advances the premise that "different forms of engineering disciplinary knowledge require different ways of thinking" (p. 197), and makes a distinction between the more practically oriented technician and the more theoretically biased professional engineer. Common to these differing but related, epistemological biases, which reflect the multifaceted manner in which engineering can be viewed from a philosophical perspective, is the engineering view of knowledge as that which is useful (de Vries, 2006; Montfort et al, 2014), with engineering inherently grounded in knowledge of technology and its application. Indeed, in reflecting on my own epistemology as a recently graduated engineer and postgraduate researcher, this view of knowledge was notably present in the title of my master's thesis (Gallery, 1989), which emphasised my contribution to engineering knowledge as "Development of a *practical* model of the cochlea using digital filters".

These philosophical considerations which provide a "conceptual basis for understanding engineering" (McGrann, 2008, p. S4H-32) have implications for engineering education. They influence how engineering academics approach curriculum development, raising questions about the control of, and responsibility for technology. This also gives rise to questions about who or what groups control the curriculum content (Edström, 2018), and the pedagogy used to deliver it, which are important for this research, A philosophical perspective gives rise to the fundamental question of 'what is the purpose of engineering' (McGrann, 2008), and situates engineering education in the societal area within which the engineering will be applied (Heywood. 2016). An overarching influence is that of educating the next generation of engineers (Haase, 2014), and philosophy helps define the research agenda in this regard (McGrann, 2008). This includes the questions as to how students "perceive technological artifacts" (McGrann, 2008, p. S4H-32), how they develop their ethical awareness, and how they understand the different natures of technological knowledge.

Engineering curricula provides the means to introduce students to engineering knowledge (Heywood, 2016), and are "the formal mechanism through which educational objectives are achieved" (Heywood, 2005, p. 3). They are principally comprised of: mathematics; engineering science; information systems; various aspects of technology; and design and engineering practice. Engineering graduates will be able to apply science, technology and

techniques to the solution of real-world engineering problems, and in support of engineered processes as deployed in industry, business, military and civil activity. Heywood (2008, 2016) proposes *screening* (Furst, 1958) as an approach to engineering curriculum design. In this approach design aims and objectives of a proposed curriculum are first screened against institutional educational and social policy, and then against the educational philosophy of the academic staff. The first step acts to reduce a potentially large number of aims and objectives to a more manageable size, and the second recasts them within the educational philosophy and pedagogy of the academic staff who will deliver the programme.

Engineering curricula have strong discipline specific elements, with associated discipline specific literature on curriculum development, e.g. software engineering (Fox & Patterson, 2013), mechatronics (Gallery, 2013), sustainability in engineering (Heeney & Foster, 2010). Ruprecht (2000) argues that curricula should not be driven solely by technical, industry or political concerns and supports including generalists and other disciplines in the decision making process of what is included in an engineering curriculum.

However, the factors that influence engineering education and curriculum go beyond philosophical, scientific and technological matters. Engineering, as identified by Bernstein (2000) in his conceptualisation of it as a region, is closely tied to the market, and the professional engineering bodies have an important influence. This market influence has profound implications, as will be discussed later, in relation to the adoption of outcomes based education (OBE) for engineering, as has become prevalent internationally, as influenced by industry (ABET, 1998a; Prados, 1992), national policies, international and European agreements, and the professional engineering bodies (Heywood, 2005).

Bernstein's concept of engineering as a region brings other aspects of the social realism of which he was a leading proponent into consideration for engineering education. Social realism emphasises the importance of knowledge and the curriculum in education. It recognises the social context of the creation of knowledge, whilst acknowledging that some knowledge has an importance and truth outside of the context of its creation. Social realism foregrounds the importance of power relationships in determining the structure and impact of that education. Other key considerations include the role of cognitive interests and communities of practice; pedagogy and identity formation of academics; and the concept of recontextualisation of knowledge in the curriculum. From an engineering perspective social

realism provides a framework within which the differing epistemological orientations and approaches of technicians (more practically oriented) and professional engineers (more theoretically oriented) can be conceptualised (Shay, 2012). Furthermore, it is argued it resolves the tensions between the positivist and constructivist facets of engineering epistemology (Wolff, 2015). This is important for IoT engineering education, with its tradition of providing pathways for technicians to continue their education to become engineers, and which, as will be shown, has been impacted by the introduction of OBE.

2.2.1 Technician Education vs. Professional Engineering Education

Professional degree curriculum have an initial theoretical basis, which is later built upon in theoretically informed practice. A graduate (such as myself in 1987), on the path to Chartered Engineering status, will be expected to have significant theoretical knowledge, but with a lesser emphasis on engineering practice (Shay, 2012). I certainly felt this lack of emphasis on engineering practice as a postgraduate researcher and as an early career academic in UCG in the late 1980s, where although I knew a lot of theory, I recognised I had very little idea what it was for. Indeed Johnston et al (1996) noted the dominant influence of engineering science in professional engineering education, to the neglect of engineering practice, and considered that "engineering teaching and scholarship [...] to a large extent they have remained isolated from the pragmatics of engineering as a professional practice" (p. 128).

In contrast, the pedagogical basis for technician curriculum can be described as based on practical knowledge taught alongside theoretically informed practice (Shay, 2012). In comparison to a professional engineering programme, a technician programme emphasises engineering practice over theoretical considerations. This encompasses a skills focussed education, complemented by theory to contextualise these skills, giving the students sufficient theoretical grounding should they choose to continue study to more advanced technician (technologist) or engineering programmes (McLaughlin, 1999).

The International Engineering Alliance (IEA , 2016) accords set out standards for engineering professionals and technicians/technologists, where the difference between the capabilities of the different engineering strata can be characterised (Wolff, 2015) on the basis of their problem solving capability "Engineers - complex problems; Technologists - broadly-defined problem; Technicians - well-defined problems" (p. 2). Wolff explains the difference between the three tiers in terms of: "engineers generally seen as being responsible for conception and

design, technologists and technicians for implementation". Technologists can be classed as technicians who have undertaken further education to degree standard, with a continued focus on the application of technology, as per EI accreditation regulations (EI, 2020a, 2020b). In the IoT sector we regard both higher certificate and ordinary degree holders as technicians, and I will generally use the term technician to refer to both technicians and technologists together, similarly to Wolff (2015), unless clarification is necessary.

The division of engineering education into professional and technical education can be regarded as a stratified system (McLaughlin, 1999; NAE, 2017), with technician education considered a lower tiered qualification. However, is this stratification a necessary approach to engineering education and the successful application of engineering? The distinction between the roles engineers and technicians may fulfil is not always fully defined (Lennox & O'Brien, 2013), with professional engineers and technicians in some cases, performing similar roles (Land, 2012; NAE, 2017). On the other hand McLaughlin (1999) places, in a typical engineering organisation, professional engineers in positions of authority over technicians.

Although USA and Irish engineering education are not fully aligned, their broad similarity allows for comparison, as evidenced by their respective professional bodies being signatories to the IEA (2016) accords. Dempsey (2018) believes the increased theoretical emphasis that developed in USA engineering education from the 1950s onwards was a consequence of engineering academics seeking to increase their status through ensuring engineering education contained "academic credentials deemed important by the scientific community" (p. 8). In 1955 the American Society for Engineering Education (ASEE) adopted the Grinter report, which proposed increasing the theoretical component of engineering degrees. An earlier draft also proposed a parallel, more practically oriented, technology focussed, degree, with equal weighting to that of a more theoretically focussed one (Dempsey, 2017; Seeley, 1999). Although this was identified as meeting the greater part of industry's requirements (Dempsey, 2017), the recommendation was discarded in the final draft (Seeley, 1999) of the report. The more theoretical engineering education that was adopted was designed, Seeley (1999) argues, "not to serve industry, rather to attract federal research funds" (p. 291).

However, Dempsey (2017) questions the continuing validity, in modern USA engineering education, of the current stratified structure, which can support the preservation of established interests and inequality of access. For example, although engineering technology

degrees offer a viable career pathway for African Americans, who are an underrepresented group in engineering, Dempsey considers that they do not always receive parity of esteem from their peers with bachelor of engineering degrees. In identifying both "regulatory and reputational" (p. 24) causes, and noting the dual track approach suggested in a pre-publication draft of the Grinter report, Dempsey argues for equal treatment of pathways to professional and technician engineering education.

There has been debate on this In Ireland, where, in 2005, the IoT Heads of School of Engineering, identifying that there was an issue with professional recognition of engineering technicians in Ireland, proposed to the IEI the creation of the professional title of Chartered Engineering Technologist. The proposal recognised the contribution engineering technicians made to industry and the high level of the work they carried out. This would have placed engineering technologists and professional engineers with commensurate levels of education and experience on an equal footing from an accreditation perspective (IEI, 2005a). The IEI however decided that chartered status should remain the preserve of the professional engineer, of which they state "a Chartered Engineer has status across the globe" (EI, 2019b).

2.3 The Field of Engineering Education Research

I needed to understand how my research into the effect of OBE on engineering educators and engineering education was situated in relation to the field of engineering education research (EER). This was essential to allow me to build upon the existing literature, contextualise my research in the context of the research themes identified in the field, and to identify the theoretical and empirical research gap that my research question would address.

The field of EER has undergone significant expansion since the 1990s, as evidenced by: the establishment of professorships, departments, research centres and doctoral programmes in engineering education; the creation of special interest groups within professional bodies; the increased number of conferences specialising in EER or incorporating this as part of their theme; and an increased number of submissions to engineering education journals; (Bernhard, 2018; Borrego & Bernhard, 2011; de Graaff, 2017; Johri & Olds, 2014).

Froyd and Lohmann (2014) associate this expansion with the development of ABET's (the USA engineering accreditation body) learning outcomes (LOs) based accreditation process, EC2000 (ABET, 1998a; Prados, 1992). As will be discussed in more detail later, in the 1990s

USA engineering education attracted strong criticism of its preparation of students for the workplace (Prados, 1992), and in response ABET advocated a move to a LOs approach for accreditation. To support this the USA National Science Foundation (NSF) provided significant funding for research into the use of LOs in engineering education, which served to fuel the expansion of EER in the USA (Daniels et al, 2011; Edström, 2016).

The American Society for Engineering Education (ASEE, 2019) *Journal of Engineering Education*, a key scholarly journal for EER (Edström, 2016; Lohmann, 2005), took a lead in promoting its expansion, publishing *The Research Agenda for the New Discipline of Engineering Education* (JEE, 2006). This identified themes for the discipline, comprising: epistemologies, or engineering thinking and knowing; learning mechanisms; pathways to diversity and inclusiveness; engineering education and institutional practices; and assessment and research methods (Johri & Olds, 2014, *Introduction*). However, Borrego and Bernhard (2011) characterise EER in the USA as being primarily concerned with how engineering is taught. They recognise the important role of engineering educators, but categorise related research as being concerned with how to assist them "improve their teaching" (p. 23).

The rise in EER in Europe is associated (de Graaff, 2017) with the European Society for Engineering Education (SEFI, 2019), a key objective of which is to connect those teaching engineering with relevant research (Edström, 2016). SEFI publishes the European Journal of Engineering Education, which, it has been argued (Edström, 2016), takes a less scholarly, more inclusive approach than the ASEE journal, with readers who are "engineering educators looking for inspiration, rather than researchers looking for references" (p. 976). Borrego and Bernhard (2011) characterise European EER as being concerned not just with how engineering is learned, but also with "what is taught and why" (p. 33), how engineering knowledge is recontextualised by the social context and pedagogy, how knowledge is selected, and how technology evolves and in turn helps change society.

2.3.1 Engineering Education Research in Ireland

Sorby et al (2014) review EER in Ireland through the lens of Fensham's (2004) framework for defining a research discipline, which identifies three sets of criteria. Structural criteria take such forms as academic recognition, journals, research centres, etc. Research criteria are associated with the intellectual coherence of the field, e.g. where groups of researchers reach consensus on research questions, their relative importance, how they might be solved, and,

indeed agree what might constitute a solution (Borrego & Streveler, 2014). Outcome criteria consider the implications of research for the practice of engineering education (Froyd & Lohmann, 2014).

In considering EER in Ireland, Sorby et al note the presence of dedicated engineering education researchers in higher education e.g. Seery (TERG, 2019), or others such as Brabazon (DCU, 2019), Dempsey (NUIG, 2019), Curran (UCD, 2019) who publish on EER in addition to their technical interests. In addition, the significant contribution of Heywood (TCD, 2017b), must be acknowledged. In further alignment with Fensham's framework, Ireland is also graduating PhDs focussing on EER, albeit in small numbers (Duffy, 2017; Goold, 2012; Kelly, 2010). However, in the absence of the funding levels available in USA, the number of academics engaged in EER is relatively low (Sorby et al, 2014).

Perhaps the largest activity EER activity is associated with TU Dublin's CREATE (2019) group, which aims to bring together staff and students "engaged in education research in engineering and technical discipline areas". The research interests associated with CREATE are complementary to my own study, including the epistemology and ontology of engineering (Duffy & Bowe, 2014), the relationship between technician and professional engineering programmes (Llorens et al, 2014), academics' perception of the importance of professional skills (Beagon, 2018), engineering ethics (Conlon, 2008, 2015), and the effects of change in higher education in Ireland on faculty (Kelly, 2010). As an indication of the level of activity, in 2019 CREATE organised a workshop on EER in conjunction with the UK and Ireland Engineering Education Network which included the research topics of combining accreditation and programmatic review, engineering skills requirements, and pedagogy (UK & IEE EER Network, 2019).

2.3.2 Engineering Education Research and Learning Outcomes

The view of outcomes based engineering education in the EER literature is generally positive (ABET, 2006; EI, 2014; Froyd et al, 2012; Owens, 2016a), with the application of LOs in engineering education considered part of the "fabric" (Froyd et al, 2012, p. 1349) of engineering education.

The tone for USA engineering education research in this regard is set out in the first chapter of *The Cambridge Handbook of Engineering Education Research* (Johri & Olds, 2014), where

OBE, contextualised within professional body accreditation, is described as having "many aspects of a scholarly approach to educational innovation" (Froyd & Lohmann, 2014, p. 6). However, some critical voices (Riley, 2012; Wolff, 2015; Woolston, 2008) have spoken against the accepted wisdom of OBE, and Heywood (2016) considers it is often not applied effectively. Klassen (2018) reports on attempts (allegedly influenced by ABET) to supress a presentation by Riley at ASEE 2016 critical of aspects of EC2000.

A European perspective on the use of LOs in engineering education is provided by the papers published for SEFI (2018), approximately 30% of which discuss the use of LOs. However, this is generally with regard to how aspects of education, such as innovations in teaching and learning, or in assessment, can be utilised to better meet LOs, or how to improve the design of curriculum using LOs. Indeed, only one paper in SEFI (2018) can be said to raise a critique of LOs, in that Virkii-Hatakka (2018) questions whether the LO approach is fully suitable for the teaching of more advanced engineering topics. The evidence from analysis of SEFI (2018) is that use of LOs appears indeed to be generally accepted as "part of the fabric of the engineering education community" (Froyd et al, 2012, p. 1349).

2.3.3 Engineering Education Research and Engineering Educators

Jesiek et al (2009) categorised the field of EER through considering a sample of papers from journals and conferences. Research into methods of assessment was one of the most represented research topics (15%), alongside teaching (8%) and learning (12%). Curriculum constituted the principal topic of 8% of the papers, with research into engineering faculty representing only 3% of the papers.

In order to evaluate more recent trends regarding research into engineering educators, consider SEFI (2018), which conference included *Educational and Organisational development* as a theme. However, analysis of the conference papers on the basis of their titles suggest less than 5% being concerned with matters directly related to engineering educators' professional development or the impact of educational policy, consistent with Jesiek et al (2009). Analysis of the 2017 SEFI annual conference (SEFI, 2017) reveals a similarly low level of research interest into engineering educators themselves, although papers such as Nyamapfene (2017) and Niemi et al (2017) reveal some awareness of the need for research into the pressures on engineering educators in today's engineering education environment.

2.3.4 The Positioning of this Research in the field of Engineering Education Research

As an engineering academic, I wish to understand the effect that the change to OBE has had on my colleagues and on myself. Although there is considerable EER research which considers the use of LOs, this is generally on the basis of how to improve that education to further align it with an OBE approach to engineering education. Research into engineering educators, a focus of this thesis, has a relatively small footprint within the overall field. Such research as there is into the impact of OBE on engineering educators tends to be on how better to engage faculty with the use of OBE, particularly through appropriate training (Brent & Felder, 2003; Cross et al, 2017; Felder et al, 2014; Laguador & Dotong, 2014; McKenna & Light, 2009). In contrast, this research, undertaken from my perspective as an engineering educator interested in policy formation and its impact on engineering education, provides a critical view of the use of OBE for engineering and its consequences for educators' approach to pedagogy, curriculum and knowledge. In considering the positioning of this research in relation to EER on Irish engineering education, it is complementary to Kelly's (2009, 2010) research into the impact of change in higher education on engineering academics.

Furthermore, in my academic leadership role as a HoD, I am concerned with how the relationship between technician and professional engineering has changed under the influence of Ireland's National Framework of Qualifications (NFQ) and Engineers Ireland's (EI's) OBE approach to accreditation. Building upon analysis of the structure of Irish engineering education (Bucciarelli et al, 2009; McLaughlin, 1999, 2001), my thesis adds to research on the relationship between technician and professional engineering education in Ireland (Llorens et al, 2014).

As an engineering educator, I am interested in what we teach to students on their journey to become engineers, and why we teach it (Borrego & Bernhard, 2011). My research investigates the impact of OBE on this through the views and experience of my research participants, engineering education practitioners.

I wish to acknowledge the significant contribution of Heywood to the field of EER, and to my own research, which draws on his publications into the philosophy of engineering (Heywood, 2011), assessment (Heywood, 2000, 2016), curriculum (Heywood, 2005, 2008), LOs (Heywood, 2008, 2016), and into professional bodies and accreditation (Heywood, 2005, 2016). My research is also complementary to the on-going research by Kyne (2019) into the

possibility of amalgamating programmatic review and EI accreditation, both of which are LOs focussed processes.

2.4 The Establishment of Engineering Technician Education in Ireland

I now consider the engineering technician education system in Ireland within which my research is situated. In order to understand the socio-economic history that led to the development of this effective engineering technician education system, I consider the genesis of engineering education in Ireland, and describe the pre-1960s Irish engineering education system (see Figure 2-1, p. 34). Then, through consideration of a series of reports and related government policies, I chart the path to the establishment of the RTC (IoT) sector (Figure 2-2, p. 34).

2.4.1 Engineering and Technical Education in Ireland Prior to 1960

The early development of engineering education in Ireland followed a stratified approach, with university engineering education for the upper and middle classes, and on the other hand the Mechanics' Institutes, and, later, technical colleges, offering technical education for the artisan working classes to allow them to engage with the requirements of industry.

Dooge (2006) links the development of university engineering education in Ireland to that in the UK, where, as part of an expansion from a military engineering focus to encompass a civil one, in the 19th century engineering professorships were established in the UK universities, e.g. University College London (1841), Kings College London (1838), Glasgow (1840). The earlier professors were themselves trained under the previous apprentice-like system (Bucciarelli et al, 2009), leading to an emphasis on practical training.

In Ireland similar developments led to the initiation of engineering courses in 1850 in Queen's colleges, Cork, Galway and Belfast, with Trinity having established a school of civil engineering and architecture in 1841 (Dooge, 2006). Although the Catholic university, which was to become UCD in 1908 (UCD, 2011), was not successful in its plans for engineering programmes, the government run School of Science included a faculty of engineering on its foundation (Kelham, 1967). This later became the Royal College of Science of Ireland (RCSI), and was incorporated into UCD (2011) under the 1926 University act (Irish Statute Book, 1926).

The RCSI programme was illustrative of university engineering education of the times, with an initial theoretical basis leading to a more practical focus in later years, comparable to contemporary professional engineering education. The programme comprised a year of general study emphasising maths, followed by two years of more specialised study of a practical nature, leading to a Diploma of Associateship (Kelham, 1967). Similarly Trinity's initial two year engineering course emphasised earlier study of theoretical matters followed by a year of study of engineering practice (Dooge, 2006), with a similar structure as the programme duration extended to a Diploma in Engineering in 1845 and a Bachelor of Civil Engineering in 1872 (TCD, 2017a). The professional identity being formed through these courses, a theoretical foundation later contextualised in practical studies, was different to the earlier apprentice-like training for engineers (Bucciarelli et al, 2009; Dooge, 2006), and attracted debate as to its suitability (Adelman, 2006).

The establishment of engineering as a profession began with the "cultivation of a professional image by organised groups of engineers" (Adelman, 2006, p. *xxx*), leading to the formation of engineering professional bodies. The Institute of Civil Engineers of Ireland (ICEI), established in 1835 (Cox & Callanan, 2006), received its Royal Charter in 1877 (Cox & Dwyer, 2014), with engineering firmly regarded as a profession by the end of the 19th century (Adelman, 2006).

Whereas in Britain industrialisation created a demand and interest in scientific education, Ireland was less developed, with the principal exception of the Belfast region, where engineering industry was supported by a workforce with strong engineering skills (Malley, 1981). Engineering graduates were in demand throughout the UK and its empire for infrastructure development, with positions available in the civil service (Adelman, 2006). Nevertheless the overall level of industrialisation in Ireland led to engineering programmes experiencing difficulty attracting students. However, from 1899 onwards demand increased, although the level of industrialisation entailed that, for the case of the RCSI in particular, the majority of graduates took up work abroad (Kelham, 1967). Nevertheless, by the beginning, of the 20th century a university engineering education system for the training of professional engineers was established in Ireland, accessible, generally, to the middle and upper classes.

Dooge (2006) describes the parallel 19th century UK Mechanics' Institutes movement for the technical education of craftsmen. This was mirrored in Ireland with the establishment of such institutes in Dublin, Belfast, Cork, Limerick, Galway and others (Cooke, 1999). Patrons of the institutes were drawn from "the leadership of the nobility and landed classes" (Thomas, 1979,

p. 70) who lent them "their respectability, influence and values", although some "supporters of the institutes were interested in the general cause of education" (Thomas, 1979, p. 69).

However, the Mechanics' Institutes were more than just a means to provide industry with skilled workers, it is argued they were also an instrument of social control (Royle, 1971, Shapin & Barnes, 1977; Turner, 1980), making accessible to workers knowledge aligned with manufacturers' needs (Cooke, 1999; Thomas 1979). Engels (1845) identified the ruling capitalist classes influence in the Mechanics' Institutes, describing the class struggle to assert control. This ideological conflict was apparent in the Dublin Mechanics' Institute, resulting in a withdrawal of patronage, which, amongst other factors, led to it ceasing to function in the 1830s, although it was later revived and became more established (Cooke, 1999).

The success of the 1885 Irish Artisans' Exhibition led to the founding of Kevin Street College of Technology in 1887, aided by a grant from Dublin Corporation, to "provide education for the working classes of the city" (Duff et al, 2000), and effectively replacing the Dublin Mechanics' Institute (Cooke, 1999). Initial programmes encompassed technical subjects such as mechanics, mathematics, machining, chemistry, and electricity.

The Agriculture and Technical Instruction Act 1889 enabled local authorities to fund technical education. This led to the expansion of the Kevin Street college, growing to offer courses in science, technology, art and commerce, and the establishment of further technical schools in Dublin and other parts of the country (Duff et al, 2000). However, the majority of these training institutes were in Dublin and not readily accessible to those from the provinces, although Gleeson (1956) contends that the "training bearing on the possibilities of employment" (p. 3) they offered that was quite sought after.

There were to be further changes in engineering related education in the state, e.g., the 1926 Commission in Technical training led to the Vocational Training act (1930) (McCarthy, 1977) and the establishment of the City of Dublin Vocational Educational Committee (CDVEC). This assumed responsibility for the Dublin technical schools which were to later form the DIT (Duff et al, 2000), with the only other significant technical college situated in Cork (Walsh, 2018). These technical schools prepared students for either the examinations of relevant professional institutes, or alternatively the state technical examinations (Gleeson, 1956).

Engineering Education and Ireland's Technological Education Sector

Institute of Civil Engineers of Ireland (ICEI) founded (1835)	Trinity School of Architecture and Engineering (1841) established	Royal College of Science of Ireland (1867) ICEI receives Royal Charter (1877) Irish Artisans' Exhibition (1885)		Vocational Training Act (1930) City of Dublin Vocational Education Committee, CDVEC (1930)
1825	1850	1875	1900	1925
•	Establishment of Queens Colleges and university engineering education (1850)	Kevin St College of Technology established (1887) Agriculture and Technical Instruction Act (1889)	•	Apprenticeship Act (1959)

Figure 2-1 A Timeline of the Development of Engineering Education in Ireland Prior to 1950



Figure 2-2 A Timeline leading to the Establishment of the RTC/IoT sector in the 1960s

The 1926 commission also led to the Apprenticeship act (1931), which "improved the situation in relation to apprenticeship" (McCarthy, 1977, p. 6). A further Apprenticeship Act (1959) established a National Apprenticeship Board, comprised of representatives from education, employers and employees. The apprenticeship board put in place a formal system of apprentice training, including the stipulation of the minimum educational requirements required to pursue an apprenticeship and a formal examination system, which had, hitherto, not been in place in all trades, and advanced proposals for specialist technical schools for the teaching of trades (OECD, 1964).

2.4.2 Training of Technicians in Ireland (1964)

In the late 1950s and early 1960s Irish government economic policy oriented "away from trade protection and towards export markets" (Barry, 2014, p. 214). This was partly motivated by the need to address the economic woes of the country. It was also influenced by the establishment of the European Free Trade Area (1960), the 1961 Washington conference on *Economic Growth and Investment in Education* (Hyland, 2014; O'Connor, 2014), and the UK's application for European Economic Community membership (1961), which threatened Ireland's preferential access to the UK market. This policy re-orientation (GoI, 1958, 1964; Hyland, 2014), supported by the Industrial Development Agency (IDA), to attract Foreign Direct Investment (FDI) led to a focus on the establishment of new industries.

It was apparent that these new industries required an educated, technically skilled, workforce (Barry, 2014), although what being a technician entailed was not clearly defined or understood in the Irish context (Latchford, 1962; Warren, 1961; White, 2001). Such technology oriented education as there was consisted of university engineering education, primarily accessible to the middle classes (Osborne, 1996)⁵, and a parallel, rather small, disparate, Dublin-centric technician training system (GoI, 1965; OECD, 1964) which had not been the subject of coherent national planning (OECD, 1964). To illustrate the small numbers involved, during 1960-63 only 219 students passed the state technician examinations, with relatively small numbers also taking City and Guild technical examinations. In the comparable year 1964, engineering enrolments in the universities totalled 1005 (McGoldrick, 1992).

⁵ The participation rate was relatively small however, less than 4% of the over-16 population in 1966 (CSO, 1970).

In order to understand the issues regarding technician education in Ireland, and answering the challenge that "one wonders if our higher educational authorities have the courage to admit that they simply don't know" (Warren, 1961, p. 22), the government, spearheaded by the interest and concern of the Minister for Education, Dr Hillery, commissioned an OECD survey of technician training in Ireland. A major theme of the resulting *Training of Technicians in Ireland* (OECD, 1964) report was that economic advance required a technically competent workforce, and the educational effort this required should be led by the state (White, 2001).

The report proposed that technician education should be "based on mathematics and the physical sciences" (OECD, 1964, p. 88), positioning this education towards engineering. In highlighting the paucity of information available regarding projected technician demand, the report encouraged the *Investment in Education* team (see below) to investigate this aspect.

In keeping with the economic drivers influencing government policy the report "strongly recommended that educational measures should be treated as a fundamental and essential part of economic development" (OECD, 1964, p. 101). A further recommendation, which likely influenced the later decision on the establishment of the RTCs, was that "steps should be taken to give people in western and southern Ireland access to technical education" (p. 106) as was already available in Dublin, where the bulk of existing technician oriented training was concentrated.

The report was received positively, with Dr Hillery remarking that it is "government's policy to make a substantial investment in education [...] We intend to make changes, we shall welcome changes, and we are not only willing but eager to have advice on the form some of these changes might take" (OECD, 1964, p. 102). In further support of the report's findings, an influential speech from Dr Hillery announced the creation of the RTCs (Hillery, 1963), "in the belief that they would align technical education provision with manpower needs" (McManus, 2016, p. 278). Although the speech emphasised a post-primary second level role, the RTCs were to form part of a new third level sector, as anticipated in their creation (Dr Hillery, interviewed in Rigney (2009, p. 43)), where suitable candidates could undergo "further training [...] for a post as a technician" (Hillery, 1963). The speech was notable for its emphasis on "equality of educational opportunity", which was a "duty of the state", and its view that courses (in business studies) should be "suitable for young women as well as young men". This speech influenced the Irish Government/OECD team preparing the follow-on

Investment in Education report to address the questions raised in the Training of Technicians report.

2.4.3 The Report on Training and Education of Technicians (1967) from the ICEI

The ICEI⁶ was also taking an interest in technician education, and in 1967 their council approved the *Training and Education of Technicians* report (Dooge, 2006), which was issued to government to add to the national policy debate (Rafferty, 1968), and informed the later Mulcahy report (GOI, 1967).

Unfortunately it was not possible to locate a copy of the ICEI report itself⁷. However they clearly considered it significant, with a commentary on it constituting one of the cornerstones of the ICEI President's 1968 annual address (Rafferty, 1968), and from which the salient points are reconstructed. Referring to the report, the President echoed the economic considerations influencing national policy, stating that "the shortage of technicians has had the effect of slowing down economic growth" (p. v). This shortage entailed that engineers were performing technician work (OECD, 1964), described as "economically indefensible" (Rafferty, 1968, p. v), and intellectually draining on the professional engineer.

The ICEI, informed by the *Training of Technicians in Ireland* report, recognised the moves to establish a national certification authority for technicians, but in the meantime proposed to endorse technician awards from approved colleges of technology once ICEI academic standards were being met (Rafferty, 1968). This led to the creation of a new membership category for engineering technicians who met appropriate standards (Dooge, 2006). The ICEI prerogative as the professional body for engineering technicians in addition to engineers was enshrined in the act that provided the ICEI successor, the Institute of Engineers of Ireland (IEI), with its charter (Irish Statute Book, 1969). Thus the ICEI, quite likely influenced by earlier developments in the USA (NAE, 2017), the UK (Heywood, 2016) and elsewhere, adopted, in their capacity as the professional engineering body in Ireland, a stratified approach to engineering education, and this approach was enshrined in the legislation that governed them. Although the new technician education system in the RTCs was not yet established, the ICEI successfully positioned itself as an arbiter of standards for Irish engineering technician

⁶ The ICEI amalgamated with Cumann na nInnealtóirí in 1969 to form the Institute of Engineers of Ireland (Cox & Callanan, 2006), and was to adopt the use of the name Engineers Ireland from 2005 (Cox, 2019).

⁷ EI no longer have a copy, the DIT library, another potential source did not have one, and R. Cox, recommended by EI's registrar due to his personal archive of the history of engineering in Ireland, was unable to source a copy.

education. This positioning was later to play a pivotal role in the introduction of OBE in engineering education in Ireland.

2.4.4 The Investment in Education Report (1965)

In response to the lack of data revealed in the *Training of Technicians in Ireland* report (OECD, 1964), the government commissioned the *Investment in Education* report (GoI, 1965) as part of a pilot study for an OECD international *Education Investment and Planning Programme* (Walsh et al, 2014). In commissioning the report Dr Hillery explicitly linked economic development and education, noted the increased emphasis this was attracting internationally, and indicated the importance this would have in the European context.

This report has been regarded as marking a turning point (Hyland, 2014; O'Connor 2014) towards recognising Ireland's interdependence on the wider world, which "was being transformed through the continual advancement of knowledge" (O'Connor, 2014, p. 193). Reorientation of the Irish economy was planned through a policy of rapid industrialisation using foreign direct investment, export driven, where education would be "a key facilitator in this transformation process" (Clancy, 2008, p. 123). The report considered the Irish education system through the dual lens of its capacity to meet future needs of industry, but also the societal need to improve access to education. The report was to supply "an adequate basis of relevant information" (GoI, 1965, p. *xxxiii*) to allow for "effective, decisions and policies".

Stark inequalities in Irish education were revealed. Students from semi- and un-skilled backgrounds constituted 25% of the population, but only 2% of university entrants (O'Connor, 2014). Furthermore, a deficit of qualified workers, particularly technicians, was projected for the planned industry expansion, leading to the conclusion of "the need for wider participation [...] and for longer retention of pupils in the educational system" (GOI, 1965, p. 390).

Two possible strategies were suggested to achieve this, the first being to increase participation in education in those population groups that were already inclined to participate through "an expansion of the existing structure of education" (GOI, 1965, p. 391). An alternative approach was to re-shape the educational system to enable more equitable access. This would be costlier than the first approach of "expanding upon traditional lines", but would have the advantage of addressing the inequality in the educational system.

The report envisaged a doubling of the technician requirement by 1971 with "a total demand of 5,000 for the decade 1961-71" (GoI, 1965, p. 209). It was suggested this could be tackled based upon a model used in some parts of Europe where there "exist two grades of professional engineers, one with a university degree, the other with a professional qualification below university level". This signalled the intention to adopt a stratified engineering education system to meet industry requirements.

An analysis of programme completion rates, graduate emigration, and other factors, led to an annual entry cohort of 1000 students being proposed to meet projected technician requirements. However, based upon the technical education structures currently available, the report expected a deficit of as much as 75%. The requirement for the expansion of technician courses and places to meet this demand was expected to be met by the already announced RTCs (Hillery, 1963). However, the role of technical education in the state, and the structures and roles of the future RTCs, had not yet been fully developed. This was to be further examined and defined in the Mulcahy report (GoI, 1967).

2.4.5 The Report of the Commission on Higher Education (1967)

In parallel with these developments, the Commission on Higher Education had been deliberating since 1960 to review and make recommendations into "university, professional, technological" (White, 2001, p. 42) and other aspects of higher education. In contrast to the *Investment in Education* report, which has been described as "the foundation document of Irish Modern Education" (Clancy, 2008, p. 124), the commission's report, published in 1967 amid criticism for the delay (Dáil Éireann Debate, 1966; White, 2001), had little impact (Coolahan, 1981). Its focus on societal and personal developmental aspects of education, and the view that higher education should "be looked upon as a good in itself" (White, 2001, p. 44), was out of step with the economic drivers that were on the ascendancy (Clancy, 2008).

The commission considered universities unsuitable for technological education (Osborne, 1996), concerned the required expansion would affect their "essential nature and functions" (CoHE, 1967a, p. 27). This can be interpreted as the commission rejecting the instrumentalist view of education (Young, 2008) that had arisen in government, instead wishing to continue with a "neo-conservative traditionalism" (p. 19) in which the identity formation of students would be determined by a traditional approach to curriculum (White, 2001).

Whilst conceding that university engineering education was linked to technology, it considered the requirement that technician education be firmly linked to industry was not something that universities "should be asked to assume [...] and [...] would not wish to assume" (CoHE, 1967b, p. 184). It was not that the commission did not recognise the growing importance of technical education and the importance of industry requirements, it was just that they considered it not for the universities (Osborne, 1996).

The government interpreted the report as positioning the universities as elitist organisations with little role in economic change (White, 2001, p. 50), and uninterested in providing technological education (Dr Hillery as interviewed in Rigney (2009)). Its proposals for technical education, the establishment of a Technological Authority and a new type of higher level institute, the New Colleges (Osborne, 1996) to focus on technical education (CoHE, 1967b; White, 2001), were rejected; instead the path to the already planned RTCs continued.

One important outcome from the commission was that it identified the lack of coordination and coherency in planning for 3rd level education in Ireland, and suggested in future there should be a unified approach to its development (Osborne, 1996). This led to its recommendation for the establishment of the Higher Education Authority (now part of the governance of the contemporary Irish higher education system).

2.4.6 The Mulcahy Report (1967) and the Creation of the RTCs

The steering group on technical education was established "to advise the Minister generally on technical education, and to provide the Department of Education Building Consortium with a brief for the technical colleges" (GoI, 1967, p. 5). In addressing this brief they were asked to "harmonise with any future thinking on third level technical education".

The report envisaged that the "long term function of the colleges will be to educate for trade and industry over a broad spectrum of occupations ranging from craft to professional, notable in engineering and science, but also in commercial, linguistic and other specialities" (GoI, 1967, p. 11). However, in the short term they would focus on "courses aimed at filling gaps … in the technician area". Technicians were regarded as an "intermediate position between the craftsman and professional" (p. 16), and some demand was envisaged from technicians and other graduates who may wish to further their education to gain professional qualifications. The Mulcahy report took issue with the recommendations of the Commission on Higher Education regarding lower entry standards into, and restricting the classes of awards from, the new 3rd level institutions. The report stated that "we do not agree with the Commission's view" (Gol, 1967, p. 27), being concerned that it might create "an undesirable dichotomy".

The report's recommendations included: the establishment of the RTCs; the creation of the regional vocational educational committees; the establishment of the National Council for Educational Awards (NCEA) for the setting of standards, the approval of programmes, and the awarding of certificates and diplomas. These recommendations, which built upon prior ministerial announcements, the *Training of Technicians in Ireland* report and the *Investment in Education Report*, were "largely adopted" (White, 2001, p. 58) as government policy.

The government moved quickly to implement this policy, opening RTCs in 1969 in Athlone, Carlow, Dundalk and Waterford, albeit with low student numbers to begin with. By 1977 nine RTCs were in place, with the remainder following in the 1980s (Osborne, 1996) (with the exception of ITB, created in 1999). In parallel the Dublin Institute of Technology (DIT) was established in 1978 on an ad-hoc basis (formalised under legislation in 1993) from the merger of the CDVEC higher education colleges (Duff et al, 2000).

2.5 Engineering Education in the IoT Sector.

Moving forward past the national policy discussions, analysis and planning in the 1960s that led to the establishment of a binary higher education system and the development of the RTCs, what sort of engineering education resulted, and how has it developed?

The universities were left to one side in the expansion of higher education in the 1970s through the creation of the RTCs (White, 2001). However they were, from the late 1970s, to commit, through an alignment with "government's priorities in higher education", (Coolahan, 1990, p. 11), to technological education, with government recognition of this leading (White, 2001) to full-time engineering undergraduates in the universities rising from 2891 in 1981 to 4373 in 1991. I myself enrolled in 1983 on an electronic engineering course in UCG created in 1979 as part of that expansion (IEI Western Region, 2000). The universities offered bachelor degrees in engineering, as well as master's degrees and PhDs. The DIT spanned all levels of the engineering education spectrum, including technician, professional degree, and

postgraduate engineering education. They were also the principal provider of apprentice training in higher education (Duff et al, 2000), a role shared with the RTCs.

The RTCs, through the Mulcahy report (GoI, 1967) were given the brief that, "to ensure their most effective contribution to the needs of society and the economy they must be capable of continuing adaption to social, economic and technological changes" (p. 11). It was acknowledged that the courses on offer would change over time to facilitate this adaptation, with the author's foreseeing "no final fixed pattern of courses in the Colleges" (p. 11). This brief was taken on board in the implementation of the RTC system, which has evolved from the earlier provision of pre- and post-leaving certificate courses and a largely vocational level of education (as proposed in GoI (1967)) to now offering apprenticeships, higher certificates, ordinary and honours degrees, master's, and PhD's (Kenny et al, 2015).

RTC engineering education focussed on technician programmes (McLaughlin, 1999) to meet industry requirements. Engineering students in the RTCs/IoTs during the NCEA era mainly pursued a ladder-based system of qualifications, consisting of two year national certificates, followed by a national diploma (a further year). Although through the 80s and 90s the RTCs encountered resistance from the department of education and others to overly expanding into degree provision (Thorn, 2018), for engineering diploma graduates two-year honours degrees became available (five years in total for an honours engineering degree) (McLaughlin, 1999). Provision of sub-honours degree offerings was particularly cost-effective, whilst strongly contributing to the needs of technology led industry as it expanded through FDI (Barry, 2005, 2007). Technicians were educated and trained for well-defined roles, being considered so important that a further higher education tier, the National Institutes of Higher Education, was established, partly with the purpose that technologist and technician roles should be "'status carrying in their own right" Clancy (1993) as cited in (Barry, 2005, p. 13).

Some of the RTCs adopted particular engineering specialisations, focussed around their remit of supporting their local region, e.g. polymer technology in Athlone, precision engineering in Sligo (McGoldrick, 1992), and aerospace engineering in Carlow. In the present day the main branches of engineering offered by the IoTs are in areas related to mechanical (eleven), electronics/computer engineering (eleven) and construction studies (ten)⁸. Additional

⁸ Source the web-sites of the Institutes of Technology

specialisations include agricultural engineering (IT Tralee), chemical engineering (Cork Institute of Technology) and manufacturing (Cork Institute of Technology and IT Tralee).

State investment led to the number of engineering graduates increasing by 40% between 1978 and 1983 (Barry, 2007). Engineering students comprised 30% of new entrants to the RTC sector in 1980, and 25% in 1998, showing the engineering focus of the institutes. The 1980s and 1990s engineering enrolments showed that the RTC/IoT sector met government objectives of meeting industry demand for technicians (Clancy, 2008). As measured in 2016-17, the IoTs continue to contribute strongly to engineering education, where the 10217 full-time undergraduate enrolments represented 60% of national enrolments in engineering programmes (HEA, 2016). Furthermore, the IoT sector holds a dominant position in part time engineering provision, with 85% of all enrolments (HEA, 2016).



Figure 2-3 Number of Full-Time New Entrants to Higher Education in Engineering

An analysis of new entrants to engineering programmes over the decade 2008-2017 (*Figure 2-1*) based on HEA (2021) data, points towards somewhat of a failing in more recent government policy (DES, 2016) to increase the number of engineering graduates. There has been a decline in the overall number of new entrants to engineering programmes (to 87%) in higher education from 2008-2017, with new entrants to the IoTs declining to 78% of the 2008 figure.

This drop is accompanied by a significant change in the percentage of IoT students pursuing full-time level 6/7 (technician level engineering s) vs. level 8 (honours engineering) programmes. *Figure 2-2* shows enrolments in level 6/7 programmes represented 70% of the total in 2008-09, but only 45% in 2017-18, with honours degree enrolments increasing from

30% to 55% in that period. This shows a dramatic structural shift in full-time IoT engineering education, with level 8 programmes now having the most enrolments.



Figure 2-4 Full-Time Engineering Enrolments in the IoTs

A complementary perspective is provided by considering part-time enrolments (*Figure 2-3*), where level 7 programmes remain the most popular offering nationally, mirroring my personal experience as an engineering educator involved in part-time programme provision.



Figure 2-5 Part-Time Engineering Enrolments in the IoTs

This can be explained in that the majority of part-time engineering students are seeking to further their career, with industry funding and encouraging education for employees that

furthers its business interests (Davies, 2008). Part-time level 8 honours degrees are important (some 413 enrolments in the IoTs in 2017-18) but to a lesser extent.

Barry (2007) suggests that Ireland's increasing affluence has promoted a preference for university programmes, which implies a corresponding increased interest in IoT honours degrees. It can also be argued that this is driven by the knowledge economy (OECD, 1996), where students see the benefits of engaging in longer programmes of study (Skilbeck, 2003). As will be shown later, it may also be partly an artefact of the NFQ concept of progression.

Notwithstanding the type of qualification that an IoT engineering graduate attains, as part of their education they will have received considerable practical training, in addition to having knowledge of engineering theory, and its application in practice. An IoT engineering graduate is expected to hit the ground running, whether in a technician role or in a professional engineering role. Indeed, Ireland's well-educated engineering workforce has helped attract inward investment, particularly in the electronics and software industries (Bruce, 1997), for which the IoTs can take considerable credit (McLaughlin, 2001). A comparable university graduate would have a more theoretical base, with less emphasis on engineering practice.

As I will explore later, do the structural changes discussed above reflect a shift from technician education to professional engineering education in the IoT sector, or is something more complex occurring? What are the implications for the stratification of engineering education and on the identity of the graduates being produced, who at level 8, would be expected to be more theoretically informed than in the past? How is this affecting the academic identity of the engineering faculty engaged with the development and delivery of these programmes?

2.6 The Role of Engineering Academics in the IoTs

A dichotomy of academic identity and profile amongst IoT lecturers was observed by O'Byrne (2009), focussing around the changes in the role of the IoTs towards adopting a research profile and beginning to teach higher level degrees, as formalised in the 1992 RCT act (Irish Statute Book, 1992). Those recruited pre-92 tended to have more industrial experience, be more concerned with teaching and imparting their experience to the students, have a pastoral care element to their professional identity, and had a relatively low level of interest in research. She found that they "feel that teaching must focus on the 'real' world in which the students will ultimately be expected to operate and be industry-led" (O'Byrne, 2009, p. 92.7).

Those recruited post-92 tended to have post-graduate qualifications, be more interested in research, and to lead and actively participate in course design. O'Byrne (2009) comments that:

the relatively simple, predominantly teaching-based identity which was developed by lecturers who worked in the Regional Technical Colleges prior to the 1992 RTC Act seems to have been replaced by a more complex and multi-layered professional identity built around a combination of roles in teaching, research and administration (p. 92.4).

Edström (2017) reported a similar dichotomy of engineering academics to that observed by O'Byrne (2009), but also views that we need some academics who "can simultaneously defend both the academic and the professional values" (p. 84) of engineering.

The IoT lecturer role consists of ten duties (TUI, 2016a, 2016b), where, however, the "majority of existing academic staff within the IoT sector have been employed to teach", (Hazelkorn & Moynihan, 2010, p. 194). Indeed, the Senior Lecturer position is explicitly named 'SL (Teaching)'. Despite this emphasis, teaching qualifications are not required for IoT academics, and the majority do not hold one (Donnelly, 2008), although the desirability of holding postgraduate qualifications is recognised and linked to promotion (DES, 2005). I will return to this important point regarding training in pedagogy in the fieldwork and the discussion.

Although one of the duties of a lecturer is to engage in research and consultancy, not all lecturers are research active. Establishing research programmes in the IoT sector (O'Byrne, 2011) is not a straightforward process, with restrictions including funding, teaching hours, and "the lack of a reward structure for researchers", (p. 21) making becoming research active a difficult experience. Indeed, as a new IoT engineering academic I engaged in application focussed research (Carroll & Gallery, 2006; Gallery & Shakya, 2004; Gallery et al, 2009), graduating three research masters students. However, as research funding dried up in the post-Celtic tiger collapse, and my HoD duties became more demanding, I found it difficult to find time to be research active, prior to more recently commencing my doctoral studies.

Nevertheless, the institutes are increasing their research profile, and delivery of postgraduate programmes, with Moynihan (2015) identifying that engineering academics in particular felt increased pressure to carry out research. This increased research emphasis brings with it pressure to recruit new academic staff with PhD qualifications, and is gradually leading to a diminution of the level of industrial expertise of the academic staff of the IoTs. This may have longer term implications for the industrial relevance of IoT programmes (Hazelkorn &

Moynihan, 2010), which has always been a strength of the sector, with Beagon (2018) reporting that there is evidence to suggest that engineering academics with industrial experience place a higher value on professional skills, including technical skills.

On becoming an IoT engineering academic in 1999 I embraced this emphasis, seeking to relate the theory and practice I was teaching to my industry experience, to the benefit of my students. I noted previously that, on graduation, and as an early career academic in UCG in 1988, I had felt somewhat isolated from engineering practice. This prompted me to move from academia to Philips Research laboratories (Walling, 2005), where I gained the industry experience which firmly situated me as an engineering practitioner (Gallery, 1999a, 1999b; Gallery et al, 1992, 1999; Gallery & Ballesty, 1999; Gallery & Bliss, 2000; Gallery & De Bruyn, 2002; Gallery & Kuijpers, 2003; Gallery & Trew, 1992, 1994, 1997). In commencing lecturing as an IoT academic to our first year students, I distinctly recollect an epiphany as I related what I was teaching to how I had approached a problem in industry. In doing this I suddenly recalled my doubts regarding my ability to lecture on what I felt was real engineering in UCG post-graduation. I was now teaching my students with confidence about how, as a practicing engineer in industry, I had approached using the technology I was lecturing them about.

However, the conditions laid out for an IoT consortium to become a technological university (Irish Statute Book, 2018) include a requirement for a minimum percentage of postgraduate research students, further shifting the staffing emphasis from industrial experience to research. Other pressures on engineering academics (Kelly, 2009, 2010) include economic pressures, the changing needs of society (the drive towards more knowledge based business/industry), increased competition from universities, accreditation requirements, and increased QA requirements (including the use of LOs). Results of a survey (Kenny et al, 2015) indicated that academics felt pressure from being asked to perform more administrative work, which they considered non-productive in relation to their academic responsibilities.

A particular concern raised by Kelly (2005) was that lowering of academic standards may follow from increased access to education. Kelly proposed that this may lead to lecturers needing to change their pedagogic approach, particularly for mature students, who bring a "challenging dimension to the teaching experience" (p. 217), including "differences in learning styles, interests, attitudes and approaches to learning", all of which must be catered for. A key contractual role for engineering lecturers is course development (McLaughlin, 1999). The IoTs pride themselves that in their programme development they "develop innovative responses to the needs of society in general and industry in particular" (p. 102). On the establishment of my own institute in 1999 (Irish Statute Book, 1999a) I participated in the development of technician and professional engineering programmes in computer engineering and mechatronics in response to industry and student demand. Later we expanded, under my leadership, into the provision of on-line mechatronics engineering programmes, specifically aimed at those working in industry. More recently the growth in pharmaceutical industry in our hinterland has led to my involvement in the establishment of related engineering programmes, and the demand for skills related to data centre operations has brought about an expansion into programme provision in that area (ITB, 2018).

This course design is now undertaken within the scope provided by the NFQ, and El's LO based accreditation requirements, which I have earlier reflected on my role in promoting. The requirement that all higher education programmes be validated within the framework has been categorised as an aspect of quality assurance (Kelly, 2009; Kenny 2009), and a "diversion of academic time away from academic issues" (Kenny 2009, p. 38). Although I question the use of qualification frameworks, I argue strongly that such use is principally an academic rather than a QA issue, the implications of which go way beyond an added administrative overhead. Nevertheless, there is no disputing that the use of LOs brings additional workload:

course leaders must adapt to this language of outputs and standards, must learn to work with the requirements of identifying learning outcomes and specifying course objectives, and in the process must produce seemingly endless new reams of documentation about their courses. (Barnett & Coate, 2005, p. 29).

However, we should expect more than just extra work-load from the use of OBE, which, as it shapes the pedagogic discourse (Bernstein, 2000), will have a consequent effect on academic identity. It is not just that there is more to do, staff will move to do things in a different way as their values and identity adapt to new pressures and requirements, as has been suggested happened during earlier changes in the role of the IoTs (O'Byrne, 2009, 2014). Kenny (2010) links the effect on curriculum and pedagogy, asking will "utilising a common language of 'learning outcomes' have any effect on pedagogical practice" (p. 45). This will be considered in more detail later in the literature review, and in the fieldwork.

2.7 Conclusion

I have described what is meant by engineering and differentiated professional and technical engineering and the associated education. In considering engineering from a philosophical perspective, I identified the multi-faceted nature of the epistemological influences. I advanced Bernstein's characterisation of engineering as a region, inward facing to the knowledge and skills provided through academia, but outward facing to market demand, as a means through which engineering education can be conceptualised and considered.

I reviewed the field of EER, considering it from international, and national perspectives, and illustrated how my research into the impact of OBE on engineering educators and engineering education is positioned to contribute to this important aspect of the field.

I then outlined the creation of the engineering technician education system in the RTCs in the late 1960s, which was associated with the policy goal of increasing economic capacity through the development of a well-educated populace (Clancy, 2015; Garvin, 2004; Osborne, 1996). A further influence was a recognition of inequalities in the education system, and the need to improve access for those from working class backgrounds (Osborne, 1996).

The predominant focus of IoT engineering education since its creation was on the provision of two and three year technician programmes. In more recent years there has been a dramatic change, to where the majority of full-time enrolments are now on honours degree programmes. This structural change raises questions regarding the stratification of engineering education and the identity of the graduates being produced. It further raises a question as to the effect on the academic identity of IoT engineering educators, who experience pressures related to increased emphasis on research, the moves towards the creation of technological universities, increased administration and QA requirements, and in particular the introduction of OBE for engineering education in the IoTs.

In the next chapter I will consider the conception and aims of OBE, and how it has attained such a prominence internationally, in particular in engineering education. I will then discuss the aims, development and implementation of the NFQ, alongside the parallel adoption of an OBE approach for engineering accreditation by EI, in order to understand the contemporary context within which IoT engineering academics deliver an education to their students.

Chapter 3 Outcomes Based Engineering Education

In the world of engineering education [...] it is difficult to raise questions about the phenomenon of outcomes-based assessment. Donald Woolston (2008, p. S4G1).

3.1 Introduction

The use of learning outcomes (LOs) is now firmly established in engineering education in Ireland. There are two pillars to this: our National Framework of Qualifications (NFQ) which mandates an outcomes based education approach (OBE) for higher education in Ireland; and the parallel adoption by Engineers Ireland (EI) of an OBE approach for accreditation.

In order to consider how this has arisen, I describe the origin and global rise in the use of OBE, and I discuss why National Qualification Frameworks (NQFs) are attractive to those developing and implementing government educational policy from an instrumental, industry focussed, economic viewpoint, but also from one in keeping with the concept of a learner centred education. I then describe how engineering education internationally has adopted OBE for accreditation, beginning with, and influenced by, the decision by ABET, the USA accreditation body, to fundamentally re-cast its accreditation approach as a LOs based one.

I then return to the national context, reviewing the policy basis (including European influences), development, structure, and implementation of the NFQ from an engineering education perspective, including a critique of its pedagogical basis. I discuss El's parallel adoption of an OBE approach to accreditation, influenced by European developments and international accords. As a further manifestation of the influence of OBE on IoT engineering education, I show how the Bologna process to develop a European higher education area was interpreted by EI as requiring a change to the educational requirements for Chartered Engineer, resulting in structural changes in Irish engineering education.

3.2 The Development of the Learning Outcomes Concept

The conception of LOs can be traced through Taylor's time and motion studies (Allais, 2014) and the development of the scientific curriculum (Bobbit, 1918). The scientific curriculum was to be discovered, rather than developed, by the "curriculum-discoverer" (Bobbit, 1918, p. 11) through observing practitioners in the field, distilling the things that they do, the knowledge that they employ, the judgements that they make, the skills that they have etc. This allowed

the establishment of educational objectives, or LOs, for carrying out these tasks. Education was to focus on "actual-life situations, a task distinctly different from the cloistral activities of the past" (Bobbit, 1918, p. 10). Tyler (1949) similarly proposed designing curriculum and instruction around assessment of educational objectives, emphasising studying real-life situations in curriculum design. Tyler proposed evaluation of educational objectives as a means of: curriculum planning and review; influencing what is to be learned; monitoring student performance and making consequential adjustments to what is taught; and evaluating the school.

Tyler assisted in the development of Bloom's (1956) taxonomy, where Bloom links curriculum development firmly to the ideas of Tyler, focussing on student achievement of learning objectives. Bloom's taxonomy orders cognitive skills in a hierarchy of 6 levels, from low level cognitive acts such as recall of knowledge, through increasingly more complex levels of mental ability, the highest being evaluation:

- 1. Knowledge
- 2. Comprehension
- 3. Application
- 4. Analysis
- 5. Synthesis
- 6. Evaluation (Bloom, 1956, p. 18)

Blooms taxonomy became very influential since its development in the 1950's, and is now considered part of "mainstream educational thinking" (Allais, 2014, P. 32). The widespread adoption of Bloom's taxonomy does not seem to have occurred due to deep engagement with his ideas, but rather, can be traced to the provision of substantial increased USA federal funding in the 1960s for childhood education. This led educational institutions in the USA to adopt the use of Bloom's taxonomy, in tandem with Tyler's concept of educational objectives, to design curriculum that could be used to meet government reporting requirements (Marzano & Kendall, 2006; Munzenmaier & Rubin, 2013).

3.3 The Rise of National Qualification Frameworks

Qualification frameworks incorporate the LOs approach in a hierarchical framework within which qualifications can be mapped, and compared. The UK NVQ framework, the first of the outcomes based NQFs (Young, 2008), incorporated a competencies and levels approach that

was hugely influential as NQFs became a global phenomenon (Allais, 2014; Heywood, 2016), including for the European Qualification Framework (EQF) (Clarke & Winch, 2015).

Although developed for vocational education, Jessup (1991), one of the chief architects (Cort, 2010) of the UK NVQ framework, argued for the applicability of its outcomes based approach to all forms of learning (Heywood, 2016; Young, 2008). The NVQ incorporated the concept of a hierarchical arrangement of LOs across five levels (Jessup, 1991). Competencies to be achieved by the learner become more demanding as the level increases, with increased "breadth and range of competence; complexity and difficulty of competence; requirement for special skills; ability to undertake specialized activity;" (p. 20). The NVQ's primary purpose was "to facilitate transfer and progression" (p. 20), allowing awards to be grouped together at levels reflecting similar competency. Differentiating it from earlier LO models, including Bloom's, Jessup stated that: "a crucial aspect of the new model is that assessment is regarded as the collection of evidence from any relevant source" (p. 145). This concept that learning outside the classroom can be evaluated through a qualification framework, facilitating access (Young, 2003) was influential in the development of the NFQ (NQAI, 2003a).

Bloom's taxonomy is used to link the concept of educational objectives with that of the hierarchical structure of qualification frameworks (Allais, 2014), where it is commonly used as part of the design of LOs. In this approach 'verbs' from the taxonomy are used to provide the basis for LOs. Modules mapped to lower levels of a framework use verbs drawn from, for example, the *knowledge* and *comprehension* levels of Bloom's taxonomy, whereas verbs from *synthesis* and *evaluation* might provide the basis for LOs of final years of honours or master's degrees (Kennedy et al, 2006). However, different disciplines have their own logics and terminology (Young, 2008), and the taxonomy does not supply such context (Allais, 2014). As an engineering educator I find the application of the generic hierarchy of cognitive abilities from the taxonomy to engineering problematic, a concern shared by Heywood (2016), who considers the taxonomy may not be suitable for all aspects of engineering education. Furthermore, QQI (2013) sound a note of caution, stating that "taxonomies [...] can help to express intended LOs. However, when such tools are inappropriately used or slavishly applied they can do more harm than good." (p. 16).

The development of various types of NQFs commenced in the 1990s, leading to the establishment of the Scottish Credit and Qualifications Framework in 2001, the South African

Qualification Framework (SAQA) in 1998, the New Zealand Qualifications Framework (NZQF) in 1992, and the Irish NFQ in 2003 (CEDEFOP, 2015b). The NZQF was "the first attempt to introduce a unified comprehensive national qualifications framework of 8 levels" (Allais, 2010a, p. 32). The proliferation of qualification frameworks since then is dramatic, as of 2015 over 150 countries implemented NQFs (CEDEFOP, 2015a).

The Irish NFQ is thus part of a transnational inter-related set of developments in which qualification frameworks of various sorts are being developed, established and implemented to influence, change and reconceptualise the delivery of education. In Europe a regional framework, the EQF (EC, 2008a, 2008b), allows national signatories to design and implement their own NQFs, aligned with the "EQF level descriptors (knowledge, skills and competence)" (CEDEFOP, 2015a, p. 10). Europe also has the parallel Bologna framework (ECA, 2020) (see Appendix E), which also promotes OBE. Although, as will be shown later, the Irish NFQ was developed in advance of, and contributed to the discussions that led both to the European Qualification Framework (EQF) and the Bologna framework, the development of the NFQ was in turn influenced by the New Zealand NZQF (Duff, 2011; NQAI, 2002a), which itself was guided by the structure of the NVQ (Priestly & Higham, 1999).

3.3.1 The Attractiveness of OBE as a Policy Instrument

The proliferation of NQFs is considered a consequence of neoliberal influenced educational policies, and the expansion of higher and further education (Young, 2003). An NQF can be attractive from a policy perspective (Young, 2008) for the following reasons: providing flexible qualification types in keeping with the needs of a modern economy; to facilitate learners to engage with life-long learning; and improving equity of access to education and ameliorating the effects of the stratification of knowledge in education systems by making it feasible for anyone to progress to the highest level of qualification. Further perceived advantages can be that the use of an NFQ may not disrupt "existing privileged routes to university" (p. 230) and provides employers an increased say in what was to be taught.

In support of these aims, OBE related policies garner support from what appear to be diametrically opposed viewpoints (Allais, 2014). Commercial interests seek skills-oriented, job ready, graduates (workers), painting subject based curriculum as "out of touch with the needs of industry" (p. 27) and "contributing to industrial decline". LOs are seen as facilitating a

curriculum which, through mapping objectives to real-world tasks, ensures the job-readiness of graduates. However LOs are also associated with learner centred education, where a student's education proceeds at their own pace, choosing what they want to learn and when to learn it, building upon their knowledge of their everyday world (O'Neill & McMahon, 2005).

A learner centred education emphasises learning rather than teaching, with the teacher as a facilitator (Collins & O'Brien, 2011). Although this can be traced to various influences (O'Neill & McMahon, 2005, Allais, 2014), it is often associated with Dewey (1907), who proposed that the learner become "the centre around which" (section *51*) education is organised. This positioned a learner-centred approach as a progressive move to replace the medieval concept of learning as a passive act (Allais, 2014).

Learner centred education encompasses a continuum of teaching and learning possibilities, incorporating more student choice, involvement, and control over their own education (O'Neill & McMahon, 2005). Some advocates regard pedagogy as the learner-centred aspect, with curriculum defined through the knowledge associated with the discipline (Allais, 2014). However, others advocate a constructivist view (Attard et al, 2010; O'Neill & McMahon, 2005), where both the curriculum *and* associated knowledge are learner centred (Allais, 2014). Attard et al (2010) regard there to be two parameters to learner centred education: the use of innovative teaching approaches which focus on how best the student may learn; and the use of LOs, where the focus of education moves from the teacher to what the student can achieve (O'Neill & McMahon, 2005). This encompasses a shift in the power relationship from the teacher to the student (Hodge, 2010; O'Neill & McMahon, 2005).

It is noteworthy that the learner centred concept often appears alongside that of LOs in government policy (Allais, 2014), as in the development of the NFQ (GoI, 1992, 1995; NQAI, 2003a) where a skills agenda was driven alongside "the holistic development of the individual" (GoI, 1995, p. 5). As I discuss later, the presence of these disparate views of the purpose of education in a LOs system may create tensions, with consequent pedagogical implications (Muller, 1998).

Raffe (2009) proposed there are three types of NQF. A communications framework seeks to support an existing system of qualifications, providing "tools for change but does not try to drive change directly" (Raffe, 2013, p. 148), e.g. the Scottish SCQF (SQA, 2017). A

transformative framework can be used where a new system of qualifications is envisaged, e.g. the South African National Qualification Framework (SAQA, 2017), conceived in the postapartheid state (Allais, 2014, p. xvi). A third type, a reforming framework, combines aspects of the communication and transformative frameworks. Similar to a communications framework, it starts with the existing education system, but will have "specific reform objectives" (Raffe, 2009, p. 5) (e.g. support for progression, quality assurance), and, consequently, will be supported by legislation and related statutory agencies e.g. the Irish NFQ.

Related policy objectives (Allais, 2010b; Bjornavold & Coles, 2010) include:

- comparison of different qualifications, nationally and internationally;
- consistency of qualifications across providers;
- transparency through programme LOs, as to what constitutes a qualification, for graduates and their employers;
- facilitating stakeholder engagement;
- formal certification of prior experiential learning through the mapping of a person's work experience, training and general accumulation of knowledge onto the framework;
- promoting a student-centred learning approach;
- aligning a programme with labour market demands through LOs related to the role it is envisaged graduates will fulfil;
- facilitating transfer of students between educational institutions;
- serving as a basis for quality assurance (QA) of programmes;
- providing clear indication of progression opportunities for students;
- as a mechanism to implement policy initiatives and reforms.

Contradictions arising from NQF associated policies have been reported: Havnes and Prøitz (2016), in discussing OBE, regard "policy and pedagogy as potentially conflicting frames of reference" (p. 220); Fernie and Pilcher (2009) question whether "intended and unintended consequences" (p. 227) have been fully considered. A key contradictory aspect arises between the policy emphasis that educational programmes should meet the needs of the knowledge economy, vs. the emphasis on skills, and the consequent downgrading of knowledge that is associated with NQFs, leading to "narrow qualifications without theoretical components" (Allais, 2010a, p. 12). In the case of the NFQ, as will be seen, many of the policy drivers discussed above are evident, sometimes with associated contradictory positions.

Young (2008), although quite critical of OBE, does not fully reject qualification frameworks
from a policy perspective, but considers that they can only achieve a small part of the policy objectives with which they are associated, unless the larger problems they claim to be able to solve are addressed comprehensively. For example, the concept of allowing students to progress from one qualification level to the next only is meaningful if the higher value job or professional opportunities these higher level qualifications claim to lead to are put in place. As will become clearer later, this has particular ramifications for IoT engineering education, which I will explore through the fieldwork and in the discussion chapter.

3.3.2 Quality Assurance and National Qualification Frameworks

I believe that both HEI administrators and academics often lose track of what is meant by QA for higher education, which I consider should fundamentally be about supporting and enhancing academic programmes. However, this view is not universally shared (Fitzsimons, 2017a), for example from the perspective of someone concerned with managing a QA system, quality can be categorised as "the distinguishing characteristic guiding students and higher education institutions when receiving and providing higher education" (ESIB, 2002, p. 12). On the other hand, from an academic perspective, QA may be regarded as an administrative overhead (Kenny, 2010), whereas in fact it may be deeply effecting how and what we teach our students. A critical point is that one's view of what QA is for can lead to the fundamental question of "the purpose of higher education" (ESIB, 2002, p. 10).

In Europe QA is associated with the Bologna declaration (European Ministers of Education, 1999)⁹ and the moves to create a qualification framework based European higher education area (Fitzsimons, 2017a). QA of higher education in Europe is based on European standards (ESG, 2015), which were explicitly created to support OBE. In considering QA's role in supporting NQFs, its origins in the marketplace (ESIB, 2002; Fitzsimons, 2017a) raises the likelihood of this taking an instrumentalist turn, rather than supporting the student centred approach promised by some advocates of LOs (Fitzsimons, 2017a). Indeed it is considered that a principal aspect of QA for qualification frameworks is "ensuring that intended LOs have been assessed and met when qualifications are being awarded" (Murray, 2013, p. 7). Thus QA in European higher education, measuring and reviewing compliance with the NQF approach, is a key aspect of supporting the economic focus of OBE (Fitzsimons, 2017a).

⁹ See Appendix E

3.4 The Adoption of OBE based Accreditation for Engineering

In addition to the pressure experienced from the proliferation of NQFs, a further critical driver for the use of OBE in engineering education (Heywood, 2016, Matos et al, 2017) is the global move towards LOs based accreditation. This first arose in the USA, where their national engineering accreditation body, ABET (2018a), accredit both engineering technician (ABET, 2018b), and additionally professional engineering (ABET, 2018c) programmes.

Many USA employers in the 1990s considered engineering graduates lacking in design capability and manufacturing knowledge, and weak in innovation and teamwork (Prados et al, 2005). In response ABET's president, highlighting concerns of leading engineering educators (Prados, 1992), called for "fundamental change" (p. 1) in engineering education, to be supported by a new, radical, accreditation approach, realised through LOs.

ABET's subsequent review of their accreditation process, centred round a series of stakeholder workshops in 1994-95 (Prados et al, 2005), led to the recasting of ABET accreditation criteria around "a limited set of education objectives for any engineering program" (ABET, 1995, p. 6), devolving responsibility for details of these objectives, and curriculum content, to the educational institutions. Following pilot evaluations, EC2000 (ABET, 1998a), an OBE accreditation system, became mandatory (Prados et al, 2005). OBE accreditation criteria were also developed for USA technician programmes (ABET, 1998b), a sector where ABET are extremely influential, whilst not totally dominant (Frace et al, 2016).

ABET view OBE accreditation as a success, with a review in 2006 stating that "the implementation of the EC2000 accreditation criteria has had a positive, and sometimes substantial, impact on engineering programs, student experiences, and student learning" (ABET, 2006, p. 12). McCullough (2007), however, is critical of ABET's review for relying upon a (self-reporting) methodology which ABET themselves rejected as unsuitable for accreditation. The ABET report highlights graduates "better understanding of societal and global issues, their ability to apply engineering skills, group skills, and understanding of ethics and professional issues." (p. 13). An increase in interest in development of teaching and learning capability was identified in academics, which was thought to be partly driven by faculty seeking to better understand how to conduct assessment. Indeed, the engineering education literature contains a generally positive view of LOS (ABET, 2006; EI, 2014; Froyd et

al, 2012; Owens, 2016a). Given ABET's role as a key professional engineering body for USA engineering, it would be expected to play a significant role in bringing about the acceptance by engineering educators of the use of LOs (Young, 2008). El have a similar influence on Irish engineering educators, which I will explore in depth in the fieldwork.

ABET consider they provide "world leadership" (Milligan, 2014, p. 6) in engineering education, which alternatively has been described as a 'colonial' project spreading neoliberal views (Matos et al, 2017). ABET's influence is pronounced in international engineering accords (IEA, 2015a, 2015b), agreements of national engineering professional bodies. Signatories mutually recognise each other's accredited graduates, and agree to adopt a LOs approach which "categorise what graduates should know, the skills they should demonstrate and the attitudes they should possess" (IEA, 2015a, p. 14). The encapsulation of ABET's LO accreditation approach in these accords is considered a guiding influence in the adoption, internationally, by engineering professional bodies of LOs for accreditation (Augusti et al, 2011; Heywood, 2016; IEA, 2015b; Memon et al, 2009; Prados et al, 2005) including in Ireland (IEI, 2003).

EC2000, and more generally the use of OBE in engineering education, is not without criticism. Matos et al (2017) see neoliberal influences to the fore in engineering education. Neoliberal influences are also specifically recognised by Akera (2017) for EC2000, but also in terms of how he considers European engineering education is adapting to market pressures. The resultant use of LOs in accreditation can be characterised as reflecting a power struggle over the control of knowledge in engineering. This influences academics perception of knowledge, assessment and their academic identity (Matos et al, 2017), which matters, as my literature review progressed, became key considerations for my research.

Engineering student identity formation (Haase, 2014) can be considered as a series of epistemological transitions (Winberg et al, 2012), and I have a particular regard as an engineering educator for the importance of considering the journey alongside the destination in becoming an engineer. In OBE, the focus of curriculum and teaching is, inevitably, on assessment of whether specified outcomes have been achieved, and what I perceive as an over-focus on assessment, and over-emphasis on skills, concerns me. Of course student identity formation still takes place in such a process, and outcomes cannot be achieved without the journey towards them. However, Riley (2012) speaks of ABET's approach valuing "product over process in traditional engineering education" (p. 6) and Cheville (2016)

critiques the OBE approach "which concerns itself primarily with efficiency of means but does not address or question the aims of education" (section *Conclusions*). In my own experience, and others (Adams & Forin, 2013; Socha et al, 2003), we learn much about engineering practice from reflecting on mistakes made as a student, engineering practitioner, and educator. As I will explore later in relation to my research, I fear that in an assessment focussed system this reflective aspect to education might be somewhat lost, although research such as Bowe and Duffy (2010) suggests this can be addressed through explicitly building reflection, alongside LOs, into the educational model.

Similarly, Woolston (2008) rejects the manufacturing input/output model to education resulting from LOs, and decries the increased administrative efforts resulting from EC2000. Indeed, Prados (2004) himself warned of the dangers of an increase in bureaucracy consequent to the new accreditation process, which resonates with earlier discussion of the increased administrative load pursuant from LOs experienced by IoT engineering academics.

A questioning voice confirms the influence the engineering professional bodies are having on engineering education: "the accrediting agencies are having a profound impact on what individual faculty and departments do. That is much time is spent, some would say an inordinate amount of time, by faculty showing how the programme outcomes dictated by the agencies are being met." (Heywood, 2016, p. 286).

3.5 The Development of the NFQ

Having reviewed the international context I now move forward to discuss the development of Ireland's NFQ, one of the two pillars of Ireland's OBE based engineering education.

Ireland saw increased state monitoring and intervention in higher education since the 1980s, as government policy sought to reorient it to a more economic focus, under neoliberal influences (Mercille & Murphy, 2015). For example, the *National Development Plan 1994-1999*, which emanated from the National Economic and Social Forum, was tasked with developing "economic and social policy initiatives" (NESF, 1993, p. 29). The 1992 Green Paper on education (Gol, 1992) put forward the view that education was to fulfil the "need to develop students for life as well as work, in a social and economic environment that is rapidly changing" (p. 3), and that it should "prepare young people adequately for work".

The linking of the social and economic good continued in the 1995 White Paper on education

(Gol, 1995), which took the philosophical position that education policy should comprise:

the articulation, nationally, of a statement of broad educational aims, which focus on nurturing the holistic development of the individual and promoting the social and economic welfare of society, including the provision and renewal of the skills and competencies necessary for the development of our economy and society (p. 5).

This led directly to the development of an NQF to assist with the implementation of a skills focussed educational system.

3.5.1 Teastas, the NQAI, HETAC and FETAC, and the Path to the NFQ

The path to the NFQ was associated in legislative and policy development terms with the moves, reflected in the discussion and announcements in the 1992 Green Paper and the 1995 White Paper, towards replacing the National Council for Educational Awards (NCEA) with Teastas, a new national certification body.

The Green Paper (GoI, 1992) proposed significant changes in higher education, including the establishment of a Council for Educational and Vocational Awards (CEVA), incorporating the NCEA and the National Council for Vocational Awards (NCVA). The NCEA had been created on an ad-hoc basis in 1972 (OECD, 2006) in response to the Mulcahy report (GoI, 1967), to oversee qualifications, being later established on a statutory footing (Irish Statute Book, 1979). The NCEA developed a formal QA system of awards, including processes for examination boards (NCEA, 2001), and periodic programmatic reviews of academic schools and departments (White, 2001). The QA processes associated with the NCEA have been characterised as "important in giving public assurance as to the quality of the work in these new institutions and helped to build their successful public profile" (OECD, 2006, p. 243). The NCVA had been created more recently in 1991 with responsibility for awards in the further education sector (Trant, 2002) and to provide certification of vocational training.

The Green Paper proposed that higher education adopt modularisation and a credit transfer system, which it was expected would facilitate mature students entering third level education, and play a significant role in workplace up-skilling (GoI, 1992). The Green Paper used language strongly suggestive of qualification frameworks. Awards were to be "modular, *graded by levels*¹⁰, and standards based" (p. 116), with "courses at lower levels [...] being

¹⁰ My emphasis.

reorganised into potential stepping stones to higher levels" (p. 114) consistent with an NQF approach incorporating levels and progression. The Green Paper also made reference to "accreditation of levels of knowledge, skills and competences" (p. 113), which language was to be retained in the development and implementation of the NFQ.

The concept of an overarching national body with responsibility for awards in higher and further education was further developed in the 1995 White Paper (Gol, 1995), where the Report on the National Education Convention (Coolahan, 1994) was referenced as supporting a "unified National Awards framework" (Gol, 1995, p. 88) of certification. One of the proposed functions of Teastas (previously CEVA) was to be "responsible for the establishment, direction, supervision and regulation of a national qualifications framework" (Gol, 1995, p. 89). The creation of an NQF was revealed as official government policy.

The White Paper also continued the theme of encouraging a modular approach to higher education, which it assumed would "enable mature and part-time students to study for qualifications while remaining in full-time employment" (GoI, 1995, p. 101). Although modularisation is not intrinsic to the use of LOs, it is often linked with them (Adam, 2004), and in Ireland, modularisation, qualification frameworks and LOs all came to the fore in the same time-frame and in the context of the same policy discussions.

Following on from this, Teastas was established first as ad-hoc body (1995) and later on a statutory basis by the Qualifications Act (Irish Statute Book, 1999b). This act created the National Qualifications Authority of Ireland (NQAI) (Teastas renamed again), the Higher Education and Training Awards Council, HETAC (which subsumed the NCEA) and the Further Education and Training Awards Council, FETAC (which subsumed the NCVA) (OECD, 2006). A consultation process regarding the establishment of the NFQ was initiated, with the Minister for Education remarking upon the active role of "the technological sector of higher education" (Dáil Éireann Debate, 26 March 1998). HETAC, FETAC and the NQAI were all later to be subsumed into Quality and Qualifications Ireland (QQI) in 2012, which took on their collective responsibilities (Minister for Education and Skills, 2012).

HETAC now took on responsibility for conferring awards at third level, replacing the NCEA. It had a more overseeing role, with the act allowing the devolution of awarding powers to

specified higher education institutes¹¹. The replacement of the NCEA by HETAC can be considered through the concept of "community of trust" (Young, 2008, p. 134). It takes time for qualifications to be trusted by society, including by those who seek to attain a qualification, and those who employ or otherwise interact with those who hold them. The replacement of an awarding body by another can damage trust in the old awards and create a difficult starting point for the building of trust in the new awards, a concern raised with regard to HETAC (Granville, 2003; White, 2001). However, the NCEA itself did not have the fulsome support of the RTC sector (Coolahan, 1994; Cullen, 1996; Walsh, 1999). Although the IoTs/RTCs were, by this point, principally concerned with third level programme provision, apprentice programmes they also delivered would fall within the remit of FETAC.

The act created the statutory basis for the NFQ, mandating the NQAI to "establish and maintain a framework [...] for the development, recognition and award of qualifications [...] based on standards of knowledge, skill or competence to be acquired by learners" (Irish Statute Book, 1999b, section 7). The NQAI was required to promote standards of higher and further education, and "promote and facilitate access, transfer and progression".

3.5.2 Policy Developments

The decision to develop the NFQ had an immediate influence on government educational policies. Indeed from the vantage of 2005, it appeared to the OECD that "in Ireland the introduction of a national framework of qualifications is the central concept in a comprehensive reform of the qualifications system" (OECD, 2005, p. 35).

A Green Paper on adult education and lifelong learning (DES, 1998) explored the role an NQF would have. Deficits were identified in the recognition of prior learning, and learning in the workplace. The NCVA was praised for its pioneering work in OBE, with modular structures, and transparency through module descriptors, being particularly highlighted. The proposed NFQ legislation was discussed, and described as "a major development for adult education" (p. 103). Amongst advantages it proposed would follow were the facilitation of transparency and the use of standards of educational achievement. A further advantage would be that it would support the use of LOS, which would in turn facilitate the recognition of work-based and experiential learning. A further perceived advantage was that the NFQ would allow

¹¹ I had a lead role in delegation of awarding authority to Institute of Technology Blanchardstown's (ITB, 2006).

learners from different educational backgrounds to join programmes at levels appropriate to their prior achievement and would allow for multiple modes of assessment.

The follow-on White Paper on adult education, entitled *Learning for Life* (DES, 2000), further discussed the role an NQF would have in regard to adult education. Adult education was seen as a key component of lifelong learning, where citizens of the state were expected to engage in "a continuum of education from the cradle to the grave" (DES, 2000, p. 32). The NFQ was to be "a fundamental ingredient in a comprehensive system of lifelong learning" (p. 155). In particular, the NFQ would facilitate the recognition of informal, experiential and work based learning through the "development of mechanisms" (p. 62) for accreditation of same. The White Paper viewed that "participation by disadvantaged groups in third-level education" (p. 142) would be facilitated by the promotion of access, transfer and progression by the NFQ.

The continuing interest in Irish government policy considerations in lifelong learning, and the role the NFQ could play, is reflected in a contemporary OECD study of the role of national qualification systems in Europe in promoting lifelong learning (OECD, 2005). Themes included the use of NQFs to reform and manage education systems, QA, and recognition of experiential learning. As part of this activity an OECD (2003) report praised the impact they anticipated the NFQ would have on lifelong learning in Ireland, particularly in introducing coherency and flexibility to Irish qualifications, with clear lines of progression. The report noted that the open nature of Ireland's economy created a requirement for Ireland's qualification system to reflect international developments, and for there to be "comparability and compatibility" (p. 65) with European qualifications.

3.5.3 European and International Influences.

The influence of European matters on Irish education policy was quite pronounced. In the 1992 Green Paper there was a European dimension to the proposed adoption of modularisation and a credit transfer system, which would facilitate "mutual recognition by different educational institutions of each other's academic courses standards" (GoI, 1992, p. 187), enhancing student mobility. Duff (2011) links the Green Paper's discussion on the European context directly to the provisions of the Maastricht Treaty (European Communities, 1992) "encouraging mobility through the recognition of qualifications" (Duff, 2011, p. 4). Duff further regards the treaty as placing obligations on Ireland of "utilitarianism in education,

certification and *qualifications framework arrangements*^{12;} [...] and quality assurance" (p. 5).

European influence was also apparent in the 1995 White Paper (GoI, 1995), which, although it quoted from the Maastricht Treaty that cooperation on European initiatives would fully respect member states control over their education systems¹³ also committed the state to participate fully in all European education initiatives⁷.

Dáil records reveal this European influence continued in the development of the NFQ: in a 1998 debate (Dáil Éireann Debate, 12 Feb 1998), the Minister for Education, Deputy Martin, made clear that on-going significant contacts with the European Commission, reflected a "mutually supportive position" regarding the establishment of the NFQ.

Although it is clear that the EC were aware of and supportive of the development of the NFQ, Ireland's efforts were in advance of overall European developments. The overarching EQF was not proposed till 2005 (Tuck, 2007), by which time the NFQ had already been established and implemented. Ireland was the first country to reference its NQF to the EQF (NQAI, 2009). Indeed, QQI view that "Irish experience of the NFQ has been a strong influence on the evolution of the EQF and Irish experts have been active contributors to a great many policy discussions and peer learning events in Europe" (Coles, 2016, p. 22): see Bergan (2003). Ireland was also a key contributor to the parallel Bologna process to develop a European higher education area (see *Appendix E*), which Duff (2011) considered committed Ireland to adoption of a system of comparable degrees implemented using the NQF approach.

As part of the development process, and in line with OECD recommendations, the NQAI considered international developments. A working paper (NQAI, 2002a) discussed the significance of the Bologna declaration, which, with its concept of first and second cycle degree's (bachelor's and master's) constituted "a partial framework for Higher Education in Europe" (p. 5). It also reviewed NQFs in the UK, Australia, South Africa and New Zealand, the latter being particularly interesting to Ireland due to the comparable size of the countries (Duff, 2011). Of further interest, however, was that the NZQF acted as a "tightly structured

¹² My emphasis

¹³ Vocational training, which OBE was developed to support (Jessup, 1991, Allais, 2014; Heywood, 2016), was an area in which the Maastricht treaty's provisions respecting member states control over their education systems was less strictly adhered to, e.g. a 1995 European Commission White Paper (CEC, 1995) called for "generalising vocational training and strengthening cohesion in the European Union". (p. 21).

system of regulation [...] where a single central authority has statutory responsibility for all qualifications, except those of universities" (NQAI, 2002a, p. 2), paralleling the later Irish experience (NQAI, 2003a). This signalled that state control over qualifications was an important factor for the NFQ, as it was previously for the NCEA (Seanad Éireann Debate, 1979).

3.5.4 The Consultation Process

The consultation process leading to the establishment of the NFQ now intensified (Granville, 2003), with the NQAI publishing *Towards a National Framework of Qualifications: A Discussion Document* (NQAI, 2001b), to guide the discussions (NQAI, 2003a). This described how the Qualifications Act (Irish Statute Book, 1999b) would support lifelong learning, the statutory bodies created under the act (NQAI, HETAC and FETAC), and their responsibilities. The principles that would underpin the NFQ were set out, "transparency, simplicity, equality, relevance, comprehensiveness and flexibility" (NQAI, 2001b, p. 9). The document further discussed the processes, policies and criteria that were under consideration for the NFQ.

The document is notable for a lack of specific discussion with regard to the philosophy of education that was being proposed through the NFQ, although the focus on LOs implied both instrumentalist and student-centred approaches. The NFQ would need to "cater for all types of learning and for all learners in a comprehensive way." (NQAI, 2001b, p. 23). However, there was almost no discussion on how this learning might be achieved (as opposed to how it was to be assessed): with the document making over 400 references to 'learning' or 'learners', but only one to 'teaching', where it was anticipated it would be necessary to "adapt curricula and teaching methodologies" (p. 52). The manner in which educational institutions might need to adapt in order to implement OBE was also not addressed in a comprehensive manner, although there was an acknowledgement that the way they provided learning might change.

The importance of professional bodies was considered, particularly their role in setting standards. It was proposed that, where an award certified a recipient to carry out a particular occupation, it should be possible to clearly identify its correspondence with the appropriate professional standards. However, it was proposed this would be done through liaison with the providers of such awards rather than in consultation with the relevant professional body.

Signalling a significant issue that was to arise during the implementation of the NFQ, the document pointed out disquiet regarding the meaning of awards such as certificate and

diploma, and highlighted the need to reconsider their naming. The role of employers was discussed, who were expected to have "a large part to play in developing and maintaining the framework" (p. 59), and it was suggested that the framework, through "setting standards for knowledge skills and competence" would have a significant economic influence.

This discussion document, and submissions relating to it made to the NQAI by interested parties (88 published submissions (NQAI, 2001a)), were used as the basis for a public forum in February 2002, hosted by the NQAI, affording the 300 attendees the opportunity to address the issues raised in the discussion document (NQAI, 2001b).

A presentation for the forum summarised the issues raised in the submissions (NQAI, 2002c). This focussed on the principles of the framework (e.g. whether accessibility and progression should be additional principles), and guidelines for its use. However, broader and somewhat contentious issues also came to the fore, e.g. conflicting views on the balance between "the economic and instrumental objectives of education and training" and "broader humanistic ideals of human learning" (NQAI, 2002c, p. 53). This was reflected in views that "employability represents only one aspect of learning" (p. 34) and those that considered "employability vital for framework relevance". There was also a concern that past awards should not be devalued by the introduction of the framework, which was certainly influenced by the submissions from the Institute of Engineers of Ireland (IEI) and the Heads of School of Engineering of the IoTs.

The IEI (2001a) submission, although generally supportive, expressed concern that the award titles of National Certificate and National Diploma might be replaced, potentially devaluing the contribution holders of such awards made to the economy and multinational companies in particular. They also drew attention to the knock on effect on international agreements that IEI had with regard to such qualifications. The IEI accepted the principles of OBE, explaining that this mapped on to the manner in which it already described the competences of Chartered Engineers, and accepted that these competences could be achieved by a variety of routes. However, their submission raised concern regarding the National Certificate, which it was concerned might get re-classified into further education. The IEI stated that adequate staffing and facilities were essential if academic standards were to be maintained, i.e. outcomes based programmes still required capable educational institutions (Allais, 2014).

The submission of the Heads of Engineering of the IoTs (CoHE IoT, 2001) stressed the

importance of maintaining linkages to professional bodies and their requirements. They also advocated the retention of the awards of National Certificate and National Diploma. They further advocated for "a clear description of each standard in terms of knowledge, skill and competence for each level of award so that course designers have a defined learning outcome to target when designing a course" (CoHE IoT, 2001). This is significant in that this approach was later adopted at both generic (QQI, 2014a) and discipline specific level (QQI, 2014c).

An analysis of the submissions (NQAI, 2001a) to the forum reveals almost a complete lack of comment on the pedagogical implications of the framework, or for staff training that might be required. Although only a few submissions commented on pedagogy (4 out of 88), those that did were quite insightful and questioning. These addressed the fundamental issue that the framework was not just a mechanism for implementing awards, but had the potential for significant impact on teaching and learning, and the educational philosophy and the associated pedagogy needed to be thought through. For example, "the learner-centred focus is welcome. It is envisaged that this will have implications for the adult education pedagogical approach" (Connolly, 2001), and "the pedagogic principle set to underpin the overall framework ought to be articulated from the outset" (ITT, 2001). Perhaps the most articulate was a recommendation that "the epistemological and pedagogical positions underpinning the Discussion Document be fully articulated so that there is convergence of understanding before mechanisms for implementation are further developed" (APEL, 2001). These submissions showed an appreciation of the influence the NFQ would have on the pedagogic discourse (Bernstein, 2000), and that this did not appear to have been taken into account. HETAC (2001), in differentiating its role from FETAC's, suggested that differences in pedagogy between vocational and academic programmes should be considered (Young, 2008).

However, the vast majority of the submissions were silent regarding pedagogy, focussing on the structures of the NFQ, the implications as they saw them for themselves, and for lifelong learning, without considering how the framework might influence the means of education (other than to recognise a focus on assessment). Reports from the forum discussion groups (NQAI, 2002d), and the submission summary (NQAI, 2002c), were similarly silent regarding pedagogy. Collins et al (2009) report that Irish academics felt that the timeframe within which the NFQ was developed limited consideration of pedagogical matters, and that subsequent attempts to debate the pedagogical basis of the NFQ were perceived as resistance rather than constructive dialogue. However, my research has shown that there was indeed adequate time, it was just not regarded as an issue requiring discussion by most stakeholders.

This consultation process was followed by the publication of the proposed policies and criteria that would govern the NFQ (NQAI, 2002b). This also set out the underpinning values and principles, and defined the concepts of standards of knowledge, skills and competences as this related to the framework. An outline of the framework was given, based upon levels (without defining how many), with awards-types mapped into the different levels. The discussion regarding award names was re-visited, making it clear that change was coming, without reaching a conclusion. Despite identifying the confusion that existed regarding exactly what was meant by a certificate award, the document did not address the issue that further confusion might be created by introducing new award names to replace old ones.

3.6 The NFQ Revealed

The consultation process culminated in the publication of the *Outline National Framework of Qualifications* (NQAI, 2003b), which revealed the NFQ as a 10 level framework, and where there had been some renaming of award types, as had been anticipated. The NFQ was formally established by NQAI in 2003 (NQAI, 2003a), under The Qualifications (Education and Training) Act (Irish Statute Book, 1999b), to "control access to the market for education and training qualifications" (QQI, 2017, p. 8).

Awards are mapped to the 10 levels of the NFQ (see *Figure 1-1*) through level indicators (NQAI, 2009), which are generic statements, for each level of the NFQ, of the standards of knowledge, skills and competence represented by generic awards mapped to that. Level indicators are available from QQI in grid-form (NQAI, 2011), and are included in *Appendix A* for levels 6-10. As can be seen, knowledge, skills and competence have been expanded into:

- Knowledge Breadth and Knowledge Kind;
- Know-How & Skill Range and Know-How and Skills Selectivity;
- Competence Context, Role, Learning to Learn and Insight.

Level descriptors provide sets of generic LOs which must be met for an award to be mapped to a level of the framework, consistent with the approach taken in most NQFs (Coles, 2016).

3.6.1 Values

An exploration of the founding values of the NFQ (NQAI, 2003a) provides context for the

vision that was in place at its establishment, and how this has matured (Coles, 2016; QQI, 2013, 2016a). In doing so I highlight those aspects particularly germane to my research.

3.6.1.1 Equality and Accessibility

Equality of access (NQAI, 2003a) refers to allowing disadvantaged groups access to all levels of education, but also to the recognition of prior learning, whether achieved in an educational establishment or through life/work experience. This latter aspect assumes that it is possible to fully equate knowledge gained outside of education with knowledge gained in education. This goes straight to the heart of a LOs approach, in that it focuses on whether LOs have been met, as opposed to what and how someone has learned (Jessup, 1991). On the other hand critics of LOs deny the equivalency of experiential learning and all academic knowledge, and argue for the essential role of academic institutions in providing an academic education (Allais, 2014; Young, 2008; Young & Muller, 2014).



Figure 3-1 Values of the Framework (NQAI, 2003a, p. 11)

In my own experience, experiential learning can be used effectively for advanced entry, where the new entrant must subsequently undertake formal academic learning and assessment to gain an award. However, assessment of prior learning can be difficult (FIN, 2011), and although I have helped develop ad-hoc approaches for particular situations that have been quite effective, I consider it of limited applicability, albeit important for those who benefit.

3.6.1.2 Comprehensiveness and Coherence

The framework is designed to support awards reflecting "personal and cultural development" (p. 12). However, the NFQ is particularly tasked with facilitating awards that represent new types of employment, and that reflect the effects of the rapid change in technology and roles

in existing employment areas. The relationship between different awards in the framework should be apparent, which is designed to facilitate progression opportunities for learners (NQAI, 2003a), as enshrined in the enabling act (Irish Statute Book, 1999b).

There are some issues with the comprehensiveness of the NFQ. Confusion arises with the difference between an Advanced Craft Certificate (trade) as compared to a Higher Certificate, both placed at level 6 (Collins et al, 2009). For example a level 6 Higher Certificate award is generally sufficient for progression into an add-on level 7 programme, whereas a level 6 Advanced Craft Certificate is not. Further confusion arises in that whilst level 5 and 6 awards from the further education and training (FET) sector are accepted for entry to first year of some higher education programmes, entry to second year (QQI, 2016c) may be allowed for holders of some FET level 6 awards. SOLAS (2014) acknowledge this confusion, and the need to build upon the NFQ to "deliver more seamless progression pathways" (p. 170) to higher education, in order to meet progression targets from FET.

Although an NQF composed of levels can be regarded as representing a vertical stratification (Triventi, 2010) of educational qualifications (Adelman, 2009; DHET, 2014; Teustsch, 2010), progression pathways seek to counter the elitism that can be associated with this segmentation (Singh, 2017; Young, 2008). Pre-NFQ, the IoTs had all taken steps to introduce degree programmes, often allowing progression for national diploma graduates to honours degree. The NFQ formalised the situation, aiming "to facilitate all learners to receive recognition and facilitate progression" (NQAI, 2003a, p. 10). A qualification at any particular NFQ level thus represents a graduate's current educational achievement, as opposed to a limits on what they have been able, or allowed, to achieve. Given the marked shift in engineering enrolments in the IoTs from level 6 and 7 technician programmes towards level 8 engineering degrees noted earlier in *Chapter 2*, this signposted the need to consider, in the fieldwork, the effects the concept of progression might be having.

3.6.1.3 Transparency and Simplicity

Transparency means it should be clear to all stakeholders, staff, students, employers, etc., what an award on the framework is, what someone should be able to do on completing an award, and what they are required to have done to commence studying on a programme (NQAI, 2003a). This is to be achieved through the use of module and programme outcomes.

I consider the concept of transparency problematic. It may be a useful attribute if regarded in a limited fashion, e.g. to allow academics to compare the standard of one programme against another. Although transparency can be regarded as contributing to the student centred facet of an NQF (Attard et al, 2010), that potential students can understand programmes LOs specified by domain experts is disputed: "what you should know as someone who is qualified cannot be explained to any unqualified person by giving them a list of LOs" (Young, 2011, p. 97). Cort (2010), in analysing the efficacy of the EQF in achieving its policy goals of transparency, comparability and portability of qualifications, suggests that qualifications specified in terms of LOs are "an opaque language of qualifications" (p. 309) for both educators and students.

Heywood (2016) questions, in the case of engineering education, whether LOs are readily understandable by all faculty, and considers they may not be interpreted to have the same meaning by those who develop them as by those teaching or learning them. He reports on a study (Squires & Cloutier, 2011) which identified differing perceptions by instructors and students of competencies taught and assessed in engineering. Their findings questions whether students and academic staff share mutual understanding of LOs, which they speculate may be due to "differences in perceptions" (p. 237.41.13) regarding knowledge. Indeed the assumption that engineering students could fully understand LOs written by experts is not consistent with their education being a process of graduate identity formation (Haase, 2014) involving a series of epistemological transitions (Winberg et al, 2012).

However, despite these criticisms, the value of transparency has become more important in the NFQ over time, e.g. Coles (2016) where the "aim of the NFQ structure is to enhance transparency and trust in qualifications." (p. 7)¹⁴. These are said to be important "so that qualifications can better support economic and social policies such as skills supply to the labour market, lifelong learning and social inclusion", (Coles, 2016, p. 7). In line with other qualification frameworks, trust and transparency are achieved in the NFQ by three factors:

- the classification of awards by type and level, thus allowing the qualification system to be communicated to all stakeholders;
- 2. the specification of the QA procedures to be used to validate awards;
- 3. through acting as a point of contact between those involved with national

¹⁴ The Qualifications Act (Irish Statute Book, 1999b) calls for a framework focussed around knowledge, skills and competence. Recasting this as being aimed at transparency and trust seems to obfuscate this.

qualifications and those who have international qualifications.

Simplicity refers to restricting the names allowed for awards in the NFQ. However, the replacing of national diplomas by ordinary degrees provides scope for confusion between what exactly a degree is (i.e. honours or ordinary). The IEI moved from their earlier position arguing for the retention of the National Diploma title, advocating for its replacement with Bachelor of Engineering Technology awards (IEI, 2004), and they continued to object to the decision that the IoTs should use title of Bachelor of Engineering for a level 7 qualification (IEI, 2005b). However, their representations to HETAC and the NQAI did not lead to change. The DIT could choose their own award titles, and instead selected the title of Bachelor of Engineering Technology. I myself consider this more appropriate, and I was involved in an unsuccessful attempt to initiate a national discussion on this issue in 2012, along with HETAC, IoT Ireland (the representative body for the IoTs), and other colleagues from the sector.

3.6.1.4 Quality

The importance of QA was highlighted during the policy development leading to the NFQ (Fitzsimons, 2017b). QA processes are a "central issue" (NQAI, 2003a, p. 13) underpinning the NFQ, and are intended to "permeate all aspects of learning and awards", including in institutional review, programme design and validation, module design, programmatic review, external examining, exam paper production and monitoring of student progress (QQI, 2016b).

Under the NFQ QA has become a pervasive aspect of academic life in the IoTs. QQI recognise the significant impact of LOs on QA processes (Coles, 2016) and that the use of LOs is not without criticism. However, Coles goes on to argue that there are clear benefits (in terms of objectivity and transparency) from QA, and, as an open question, asked "should the NFQ do more to encourage the use of LOs in curriculum specification, qualification specification and quality assurance processes?" (p. 11).

Kenny, however, (2009) argues that the quality previously achieved through the excellence of specialist academic institutions has been displaced from its central role in higher education through the introduction of OBE, echoing arguments made by Allais (2014). Fitzsimons (2017b), in recognising the tension between the learner centred aspect of the NFQ, and its employability focus, considers that, although "QA has the capacity to support divergent philosophies" (p. 28), what she terms the "employability discourse" is dominant. I have

participated in QA as an academic, a head of department, external examiner, and a registrar. In these roles I have being involved with exams, programmatic reviews, EI accreditation, and institute reviews for QQI. On reflection, I can certainly identify times when student centeredness came to the fore, in exam appeals procedures, in student representation at course boards, and in ensuring the student voice is heard in reviews. However, in terms of QA of how and what we teach, by far the dominant factor has been a procedural approach in which the use of LOs is a principal consideration.

3.6.1.5 Relevance

The NFQ is intended to be relevant to all stakeholders: students; teachers/lecturers; employers; and to provide support for awards that encourage an active participation in society (NQAI, 2003a). It is also intended to prepare students for employment, and where appropriate, to certify their competency to engage in professional practice. Such awards should demonstrate a clear mapping to "occupational or professional standards" (p. 14). Moving forward from the introduction of the NFQ, QQI (2013) regards the primary objective of a qualification system as "to prioritise the support of human development" (p. 22). However, they give significant weight to the "employability objective" (p. 23), emphasising learning to learn skills to cope with rapidly changing workplace requirements.

3.6.2 The Engineering Standards

A set of supplementary standards were developed, including the award standards for engineering (QQI, 2014c), which effectively create an engineering-specific sub-framework (Coles, 2016). They further subdivide each aspect of the generic standards level descriptors of Knowledge and Know-How and Skills-Range, on the basis of Mathematics, Science, Information Technology, Design and Development, Business Context and Engineering Practice. The standards can be regarded as a language of learning (Biesta, 2005) associated with the NFQ (Kenny, 2006), emphasising process (stipulating outcomes) (Cheville, 2016; Riley; 2012; Woolston, 2008) rather than knowledge (Allais, 2014).

As an embedded systems engineer, I find these engineering specific LO based standards in turn quite generic, lacking disciplinary context (Wolff, 2015). For example, in *Table 3-1* the engineering standards for Knowledge-Breadth for level 7 and 8 are shown. The *design and development* descriptors state what is obvious to an engineering educator, that a graduate of

a level 8 programme should be knowledgeable and be able to apply appropriate design techniques as they might require them in the solving of engineering problems. The standards could thus be considered to be of little value to curriculum design other than ensuring it follows a LOs approach, aligned with the levels of the NFQ. However, they are more than just a QA mechanism to map a lecturer's expertise into a curriculum. The control over knowledge these standards represent, the skills focus they enforce, and the emphasis on process (Biesta, 2005) act to influence us as academics in relationship to the curriculum and instruction. This is an aspect of OBE I was to consider in the fieldwork.

Table 3-1 Engineering Standards	(QQI, 2014c)
---------------------------------	--------------

	Level 7	Level 8
Knowledge- Breadth	Specialised knowledge across a variety of areas	An understanding of the theory, concepts and methods pertaining to a field (or fields) of learning
Mathematics	Has knowledge of the underlying theory essential to the mathematics used for common engineering problems in the particular sub-field of engineering	Has knowledge and understanding of a range of mathematical methods and the underlying theory relevant in the particular sub-field of engineering.
Science	Has knowledge of the fundamental principles of science appropriate to the particular sub-field of engineering.	Has comprehensive understanding of the fundamental principles of appropriate scientific knowledge and the extent of their applicability to engineering problems.
Information Technology	Has knowledge and understanding of the role of ICT for engineering. Has sufficient knowledge and understanding of ICT to adopt it to solve common engineering problems.	Has knowledge and understanding of the role of ICT and its application to the particular sub-field of engineering. Has sufficient knowledge and understanding of ICT to adopt it to solve complex engineering problems.
Design and Development	Has knowledge of the essential elements of the design process and design methodologies relevant to common engineering problems in the particular subfield of engineering.	Has knowledge and understanding of the essential elements of the design process and methodologies relevant to complex engineering problems in their particular subfield of engineering.
Business- Context	Has basic knowledge and understanding of management and business in the context of common industrial practice.	Has knowledge of management and business in the context of complex industrial practices and in the context of technological innovations and change. Understands the different operational and managerial structures in companies and the context of employment legislation, trade unions and public and private bodies.
Engineering Practice	Has knowledge of specific codes of practice in common engineering problems, including the role of design factors. Has knowledge of the codes of practice relating to hazards and operational safety. Understands the need for operational safety by design and good working practices.	Has knowledge of current engineering practice at project and management levels.

3.7 The NFQ as a Policy Tool to Emphasis the Teaching of Skills

The use of LOs can lead to radical forms of social constructivism (Young, 2008), reducing curriculum to a question of who is in control (employers or educators/students) of particular parts of it. For example, in post-Apartheid South Africa where, in the interests of promoting access to education, curriculum was replaced entirely by LOs promoting generic skills, the development of the citizenry, the integration of school and everyday learning and knowledge

selected based upon the views of the educator and the local culture of the students¹⁵ (Hoadley, 2015). On the other hand, if industry is allowed too much influence an instrumental viewpoint may prevail (Young, 2008). Rather than education being based upon cognitive interests, curriculum is side-lined and replaced by skill-oriented LOs constructed to reflect employers' views on what constitutes appropriate knowledge for the workforce.

Although the NFQ was developed with the dual aims of supporting student centred learning, and promoting a skills focus in education to support the economy, it appears the latter is the key motivation (Allais, 2017). Lynch (2006) argues that the massification of higher education in Western countries, which has resulted in significantly higher standards of living, requires major, and ongoing, state investment to support it. This results in the marketization of higher education, in which "the values of the commercial sector can be encoded in the hearts of the university systems and processes almost without reflection" (Lynch, 2006, p. 6). The NFQ can be considered an aspect of this, inculcating a pronounced skills focus in education, with knowledge relegated to a secondary role (Allais, 2014; Corbell, 2014; Young, 2008; Young & Muller, 2010). Kenny (2010), in considering the Irish context in relation to international educational policy developments, states that "NQFs provide for a human capital accumulation currency mechanism. Human capital can be quantified into units of knowledge, skills and competency, mediated by 'learning outcomes' and placed on the hierarchical currency table of national frameworks" (p. 72)

This human capital focus leads to a clear emphasis on skills at a national policy level, even our Minister for Education is, in fact, a Minister for Education *and Skills*. National strategy for higher education considers that "a high proportion of the skills that we need now in the workforce are high-order knowledge-based skills" (HESG, 2011, p. 4), equating knowledge and skills. An Irish government key advisory body on industry requirements is the Expert Group on Future *Skills* Needs (Skills Ireland, 2017). Ireland's national skills strategy closely links skills with economic success, the personal development of individuals and society (DES, 2016).

Ireland's national strategy for higher education (HESG, 2011) emphasises its role in providing the skilled graduates needed for the knowledge economy. The concept of lifelong learning is

¹⁵ Young, who was involved in this as a consultant, relates that "the teachers were underqualified. Expected to create knowledge themselves, they didn't know what to do. It was a disaster", as interviewed in Wilby (2018).

an inherent aspect of the knowledge economy¹⁶, which is "characterised by the need for continuous learning" (OECD, 1996, p. 13). Indeed, this learning is seen to be of the highest importance "in determining the fate of individuals, firms and national economies." (p. 23). Shannon (2019) contends that Irish policy makers embraced the lifelong learning concept "to develop common strategies for employment and education policy" (p. 100), and I have shown it was a key concept during the development of the NFQ. However, it is a particular type of learning that is considered to be of most value (Shannon, 2019), that of "learning new skills and applying them" (OECD, 1996, p. 23)¹⁷. It is contended that this skills emphasis is motivated by the neoliberal belief that employers' requirements should drive the curriculum and it is the role of the state to support learners in meeting these (Allais, 2014).

The focus on skills in programmes implemented using LOs establishes that someone can competently carry out specific tasks, which is normally equated with vocational education. However, this is problematic for other types of education, particularly more advanced studies, where specifying curriculum using LOs downgrades the knowledge that may be required to achieve the specified competencies, and curtails the role of those teaching (Young, 2008).

I applied the keyword analysis approach of Corbell¹⁸ to the policies and criteria related to the establishment of the NFQ (NQAI, 2003a), which constitutes a key node in the policy discourse (Corbell, 2014). I examined the document's use of the word 'knowledge', finding that of 133 occurrences, 95 are associated with the word 'skills' e.g. 'knowledge and skills'. Thus knowledge is quite bound with skills in the NFQ, consistent with the knowledge economy perspective outlined above. Although NFQ LOs encompass knowledge, skills and competence, in fact knowledge is somewhat side-lined through its equation with skills. Indeed, Corbell suggests that the phrase 'learning outcomes' is a proxy term for 'knowledge and skills', part of the skills discourse of Irish government educational policy for lifelong learning (Shannon, 2019). The NFQ is thus a key enabler for the knowledge economy, through its promotion of knowledge and skills *together*, as part of life-long learning. A finding is that the NFQ is one of

¹⁶ In a knowledge economy "knowledge is now recognised as the driver of productivity and economic growth, leading to a new focus on the role of information, technology and learning in economic performance" (OECD, 1996, p. 3)

¹⁷ Regarded by some as a key player in the promotion of neoliberalism (McPhail, 2008; Mahon, 2010), the OECD's prominent role in the development of Irish technician education, and support for the NFQ, has already been discussed. McPhail (2008) describes the OECD transition to a neoliberal organisation in the 1970's, regarding it as asserting that "society's well-being rests upon economic imperatives" (p. 43), promoting state intervention to support this. Savage (2017) regards the OECD as having a leading role in reforming education "for the sake of increasingly economic productivity" (p. 152).

¹⁸ See section 5.4.1.

the principal tools to promote the government skills emphasis in higher education.

The emphasis on skills is concerning from an engineering perspective, where a balance of disciplinary knowledge and skills is required (Winberg et al, 2012; Wolff, 2015). This provides the student with the capability to engage with professional practice, rather than providing them with "short-term, utilitarian and instrumental ideas of education, geared only to the minimal preparation of practitioners" (Winberg et al, 2012, p. 20).

These concerns have particular relevance to the IoT sector, with its traditional emphasis on engineering practice supported by theory. Young (2008) contends that attempting to describe both work-based learning and theoretical learning using the same OBE approach masks the critical epistemological differences between the two types of learning. There is a distinction between academic and vocational education (Shay, 2012), which is formed from both codified disciplinary knowledge and "implicit and sometimes tacit knowledge acquired in workplace" (Young, 2008, p. 144). The skills focus leads to "a steady weakening of boundaries, a de-differentiation of knowledge and institutions" (p. 18) as the boundaries between vocational and academic qualifications are weakened. This leads to a concern (Allais, 2014) that vocational programmes can become pathways to education rather than employment, and provides a perspective from which to consider the NFQ concept of progression between higher certificate, ordinary degree and honours degree programmes.

3.8 El Accreditation: The adoption of LOs, and the Influence of Bologna

I now move on to discuss the second pillar of Ireland's OBE engineering education system, El's accreditation requirements, which is now LOs based. This is an important factor to consider in understanding the impact of OBE on engineering education in the IoTs. In addition, El's interpretation of Bologna, and resultant further change in accreditation requirements, was significant for the structure of Irish engineering education.

EI is the sole body in the state licenced to award the title of Chartered Engineer (Irish Statute Book, 1969), and may also admit engineering technicians as associate members, in line with the earlier positioning of the ICEI (Rafferty, 1968). EI consider its remit is to "establish and maintain standards of professional engineering and engineering education" and "act as the authoritative voice of the engineering profession in Ireland" (EI, 2017b). EI (then the IEI) decided, in the early 1980s, to accredit programmes itself, where previously this had been conducted by UK engineering institutions, and commenced to accredit programmes in the universities and the RTCs (Cox & Callanan, 2006, Cox, 2019). This activity was regarded as having "greatly enhanced the image and status of the organisation" (Cox & Callanan, 2006, p. 75) and facilitated it having an "influential voice in subsequent negotiations about the equivalence of qualifications in Europe and elsewhere". The IEI also began to move beyond the traditional links to the UK, with increased international contacts, particularly with the USA. This was to lead to the IEI being one of the initial signatories in 1989 of what was to become known as the Washington Accord, along with the USA, Australia and New Zealand, with the UK and Canada signing at a later date (Cox, 2019).

El accredit programmes as suitable for the titles of Engineering Technician, Associate Engineer, and Chartered Engineer. El stipulate input standards for each title, being particularly concerned that, generally, those admitted to programmes suitable for the path to Chartered Engineer should have an honour in Leaving Certificate maths¹⁹. Becoming a Chartered Engineer typically requires gaining appropriate accredited qualifications, acting as a practicing engineering for a number of years, followed by an interview to assess suitability.

EI developed revised LOs based accreditation criteria in 2003, in which they acknowledged the influence of other engineering bodies, and the Washington Accord (IEI, 2003). EI is also a signatory to the Dublin and Sydney Accords, agreements on the equivalency of accredited programmes for engineering technician and technologist respectively (IEA, 2016). Influenced by ABET's accreditation requirements, these accords all developed requirements for signatories to undertake accreditation visits (IEA, 2016) to evaluate programmes for accreditation, and that criteria for accreditation should be programme outcome based.

These international agreements required EI to move from its earlier position of considering NCEA approval of certificate and diplomas as sufficient for accreditation (IEI, 2001b). Instead, from 2006, accreditation would be based upon evidence of meeting programme outcomes as evaluated by a visiting panel of experts (IEI, 2004). It was, however, presented to the IoT Heads of School of Engineering as consequential to a lowering of standards in their

¹⁹Given the basis of the learning *outcomes* approach, it is difficult, in my opinion, to justify the honours maths rule at *input* to a programme, although I agree that proficiency at maths is essential for engineers.

programmes following the introduction of new progression opportunities for students (McNutt, ITB, personal communication, October 2003). In any case it marks EI moving to establish firm control over standards in technician education through their accreditation process (Matos et al, 2017), and was to reinforce the use of LOs in IoT engineering education.

El also participates, through the ENAEE (overarching European engineering professional body) (ENAEE , 2015) in Eur-Ace (Augusti, 2009), a European project to establish a badge of quality for engineering programmes. ENAEE also stipulates the use of LOs to determine the capability of engineering graduates and the accreditation of programmes, as influenced by ABET (Augusti et al, 2011; ENAEE, 2017), and within the context of compatibility with the Bologna framework. As a member of the ENAEE, EI, in their accreditation requirements, has "based its approach firmly on the basis of programme outcomes" (EI, 2014, p. 3).

3.8.1 The Bologna Process and the Impact on Irish Engineering Education

The Bologna Process led to the establishment of the European Higher Education Area (EHEA), which, alongside the EQF, reinforces the OBE approach in Europe. The development of the Bologna framework is described in *Appendix E*, where the important Irish contribution culminated in the Dublin descriptors that enshrined the use of LOs (Feeny & Horan, 2016).

Implementation of the Bologna '3+2' structure (three year bachelor degree followed by two year master's degree) has impacted in different ways on European engineering education. In Germany it is associated with concerns including the loss of well-regarded degree titles, graduate work-readiness, and continued retention issues (Schuster & Hees, 2010). It has been argued that Bologna has had little impact in France (Orivel, 2005), where under the Bologna process engineering schools have continued to offer their old programmes, with the old five year diplomas renamed as master's degrees (Campus France, 2016): "the majority of university departments have simply changed the names of their diplomas, but not their content" (Orivel, 2005, p. 13).

El's interpretation of Bologna, in addition to supporting the adoption of a LOs approach to accreditation, had an impact on the structure of Irish engineering education. A 2001 submission to government (IEI, 2001d) set out options for awards suitable for accreditation for Chartered Engineer that would be compatible with Bologna. These included retaining the status quo, and changing the accreditation requirement to master's degree. El expressed a

particular concern that Ireland's bachelor degree route to accreditation should align with developments in Europe to change the accreditation benchmark to that of master's degree. This attracted some criticism, with Kelly (2001) arguing "the existing 4 year university engineering degree is fully (100%) in compliance with Bologna and there's no reason to change it"²⁰ (p. 25). Professor Kelly later made it clear he considered this move by El unrelated to Bologna, but, rather, an attempt to restructure Irish engineering education (SEFI, 2002).

However, ultimately EI interpreted the Bologna framework as requiring that a professional engineering degree, suitable for accreditation purposes, should be at master's level (EI, 2012), and recommended this be implemented using the Bologna 3+2 structure. Consequently universities (and some of the IoTs) established taught master's programmes for many of their engineering bachelor degree graduates in order to satisfy EI's qualification requirements for Chartered Engineer (EI, 2017a). However, despite the prediction by EI's Registrar that the 3+2 structure would "become the norm", (Bucciarelli et al, 2009, p. 6), with the exception of UCD those universities and IoTs that offer accredited engineering degrees have retained their four year Bachelor degrees and added a one year add-on master's (EI, 2017a).

The accreditation changes have had a significant effect in the IoT sector. Previously IoTs could seek accreditation for five year honours degree programmes, graduates of which would be eligible for chartered status. Such degrees were structured as three years of technician education and training, followed by a more theoretical final two years of education to professional engineering standard. The additional requirement of at least a year to gain a master's degree made the route to chartered status through such five-year programmes seem over-long. Consequently, many such programmes in the IoTs were discontinued, as predicted by McLaughlin (2001), and supplanted by four year honours degree engineering programmes. Such programmes do not generally form part of the accreditation route for receiving the title of Chartered Engineer, instead being suitable for the EI title of Associate Engineer. Thus El's interpretation of Bologna acted to curtail the teaching of programmes leading to chartered status in the IoTs, reinforcing the stratification of engineering education.

²⁰ I participated in this event as a rather novice IoT academic, which is how I located the reference. I distinctly recall Professor Kelly's intervention, which in my recollection, was the only real challenge to the proposal.

3.9 Implications for the Design of Engineering Curriculum

IoT engineering educators have to consider, when designing curricula, two separate, but related, LOs based, standards, the Engineering Standards (QQI, 2014c) backed by the legislative basis of the NFQ and the professional standards influence brought to bear by EI (2014). El's accreditation procedure evaluates a programme's suitability for accreditation, and in itself, is not strictly speaking a curriculum design guide. However, in practice anyone designing a new engineering curriculum, or updating an old one, must consider the implications of the El accreditation criteria. For example, see Forero et al (2011), where El programme outcomes are discussed seamlessly in the context of the design of module LOs.

The accreditation guidelines, sitting alongside the HETAC standards for engineering, define programme outcomes to be addressed in accreditation, including: science and mathematics; discipline-specific technology; software and information systems; design and development; engineering technology practice; and social and business context. Each programme area will have a subset of modules associated with it, with the modules LOs forming conglomerate programme outcomes which are assessed to demonstrate "the engineering technology graduate's achievement of the stated Programme Outcomes" (EI, 2014, p. 13).

As an example of EI generic programme outcomes, consider Associate Engineer (EI, 2014, pp. 11-14). These, which map to the NFQ level 7 outcomes, include knowledge of relevant maths, engineering science and technology alongside capability to: identify and solve engineering problems; contribute to the design of engineering solutions; work as an individual or a team in multi-disciplinary settings and undertake lifelong learning; and be able to communicate regarding engineering activities to engineers and others. Additionally, associate engineers should have awareness of related ethical issues.

The two, inter-related, LOs based standards are reinforced through separate review process. The first, programmatic review, is a periodic five yearly review required of programmes stipulated by QQI, an intensive effort by academics to revitalise their programmes for the next cycle e.g. see ITB (2017). The second is EI accreditation, which, also, has a five yearly cycle.

This discussion on curriculum leads into consideration of the pedagogical approaches that IoT engineering academics use to deliver such curriculum, with Barnett and Coate (2005) regarding the two as "intertwined", (p. 79), asking "where does the development of the

curriculum end and pedagogical strategies begin?".

3.10 Engineering Pedagogy and OBE

In considering the effect that OBE might have on our pedagogy as engineering educators, in the Irish engineering education context in which my research is situated I have shown that little attention was given to pedagogy in the development of the NFQ. However, it has received some attention in more recent years at European, and Irish, policy level. CEDEFOP (2009) places LOs above and in control of both pedagogy and curriculum, with equal gravitas to the mission of the host institute. EC (2010) expects that educators should "adapt pedagogy and training and assessment methods, to align them more clearly to LOs." (p. 23). In Ireland, higher education strategy (HESG, 2011) expects that "in the design of courses and programmes, HEIs should ensure alignment and balance between learning outcomes, pedagogy and assessment" (p. 62).

The proposal that pedagogy should be aligned with LOs is suggestive of the concept of 'constructive alignment' (Biggs & Tang, 1999), an OBE approach which seeks to provide a cohesive approach to teaching, learning and curriculum. In constructive alignment, which follows from the ideas of Tyler (Biggs, 2014), LOs, and in particular the 'verb' used to describe them, are used to drive the selection of pedagogical approaches that "address the verb" (Biggs & Tang, 1999, p. 100) on the basis that this will align pedagogy with LO. Assessment tasks should also make use of the 'verb', focussing the student, and assessors, attention to the criteria implicit in the LO. Teaching activities, and what students do, are "aimed at achieving the outcomes by meeting the assessment criteria" (UCD, 2021).

Constructive alignment has been applied as an approach for engineering pedagogy in OBE (Biggs, 2014; Turunen & Byers, 2012; Marusik, et al., 2019), although not without criticism as to its effectiveness (Nightingale et al., 2007). Within Irish engineering and technology education, there is some direct evidence in the literature of the application of constructive alignment in particular engineering education contexts. For example: Murphy et al. (2019) consider the presence of constructive alignment in an engineering programme can be used as a measure of whether it is meeting its educational goals; Keenahan and McCrum (2021) describe its effectiveness as the basis for the design of a first year module taught to

engineering and architecture students; and Buckley et al. (2020) propose it as a suitable approach for the teaching of design in second-level technology courses.

However, Bowe and Duffy (2010), fellow IoT engineering educators, present constructive alignment as an implicit, mainstream, aspect of the practice of engineering education. In considering the validity of this perspective, it must be acknowledged that IoT engineering academics have had some exposure to the concept, given that constructive alignment features as an approach to linking pedagogy and curriculum in guides to the use of OBE developed for higher education (Lahiff, 2006; DIT, 2009; LIT, 2018). Of particular significance during the introduction of OBE in Ireland was Kennedy (2007), highlighted in Collins et al (2009) as a reference point in Irish higher education for the use of LOs and, in particular, for introducing the concept of constructive alignment.

Kennedy (2007), which was quite influential in my own IoT engineering department, describes a three-step process of constructive alignment in which you approach curriculum, pedagogy and assessment through:

- 1. Clearly defining the learning outcomes.
- 2. Selecting teaching and learning methods that are likely to ensure that the learning outcomes are achieved.
- 3. Assessing the student learning outcomes and checking to see how well they match with what was intended. (Kennedy et al, 2006, p. 68).

Rather than the deepening of our approach to consideration of curriculum, pedagogy and student achievement that was claimed would result through this approach, my own experience as a HoD leading the transition to OBE was that, given the imperative to redesign curriculum, our emphasis was, instead, on an instrumentalist approach that focussed on adapting to the use of verbs in writing LOs (Kennedy, 2007). The contemporary 2006 ITB engineering department programmatic review (ITB, 2006) panel report confirms the pressure we experienced at the time to re-write syllabi using LOs and the emphasis this was engendering in us in how to assess these LOs. Furthermore, in retrospect it is notable that it did not report on a reflection from my department on how OBE was going to affect our approach to teaching engineering.

Moving forward to where OBE is now firmly established in IoT engineering education, a sample of programmatic review documents from engineering departments, schools and

faculties from across the sector (ITB, 2015; LIT, 2017; IT Sligo, 2019; GMIT, 2019; CIT, 2021) reveals that, although there is some discussion of pedagogy (often in the context of assessment of LOs, e.g. the number and timing of assessment events linked to LOs (ITB, 2015)), there is no mention of constructive alignment. It is perhaps telling that, now that OBE is quite firmly established in IoT engineering education, a programmatic review panel report for Limerick IT Electrical and Electronic Department recommended that "the verbs used in the Learning Outcomes in a number of the level 8 modules need attention" (LIT, 2017, p. 9).

The concept of constructive alignment has further influence in Irish engineering education through the Conceive-Design-Implement-Operate (CDIO, 2021) initiative. CDIO is a model for engineering education (Andrews et al, 2011; CDIO, 2021; Heywood, 2016) developed contemporaneously with ABET's EC2000 in which constructive alignment is implicit (Edström & Malmqvist, 2010). CDIO makes use of the OBE concept but looks beyond this in incorporating "the full lifecycle of product, process and system development and deployment" (Edström, 2018, p. 52). As examples of the use of CDIO in Irish engineering education: Coyle & Rebow (2009) describe CDIO's application in the development of a Masters programme concerned with sustainability and energy systems, and were influential in a contemporary decision by the Dublin Institute of Technology to join the CDIO initiative; Hyland et al. (2018) apply CDIO in engineering describe it as feature of their approach to mechanical engineering education (IT Sligo, 2019), but do not propose it to have wider applicability within their engineering faculty.

UK and Ireland CDIO activity is sufficient to sustain an annual conference to share ideas and experiences, where, in 2020, workshops considered engineering ethics and the technician role (CDIO, 2020). However, only two Republic of Ireland Universities, with 11% of Irish engineering students in 2019-20 (HEA, 2021) and no IOTs, are currently members of the CDIO initiative (CDIO, 2021). Given this excludes TU Dublin, Munster Technological University and UCD, who between them had 45% of Irish engineering education enrolments in 2019-20 (HEA, 2021), it appears that CDIO is not an overarching influence on Irish engineering education. In comparison, as discussed earlier, all Irish Universities and IoTs engage with Engineers Ireland accreditation.

More generally, and as discussed earlier²¹, engineering education research is generally accepting of the OBE approach to curriculum and pedagogy. A wide range of pedagogic techniques are described in the literature for use with OBE engineering education, including: problem based learning (Mitchell et al, 2019), action research (Bhat et al, 2019), agile methods (Myers, 2016), didactic approaches (Sedelmaier & Landes, 2014), service learning (Ropers-Huilman et al, 2007), and approaches designed to improve students' reflective capacity (Bowe & Duffy, 2010). IGIP (2021), the International Society for Engineering Pedagogy, offers certification as an engineering pedagogue. IGIP advance a perspective of engineering pedagogy as fundamentally oriented around a goal-oriented approach based on taxonomies such as Bloom's (Rüütmann, 2017), where the "teacher's main didactic task is to find an optimal teaching method" (p. 3) to "achieve a given instructional objective".

In considering this I ask myself as an engineering educator whether I feel that what and how I teach my students should be defined by the alignment of teaching with objectives. Surely it is the body of knowledge and practice associated with a subject, my own experience and knowledge of it, and the reflective capacity in this regard that I have developed as an engineering practitioner and educator that should inform my teaching approach. Similarly, Wolff (2015), who is critical of OBE, argues that engineering curriculum structures should instead reflect the organising structures of the disciplines from which the region is formed, with clearly defined pedagogic practice required to develop the problem-solving characteristics of each engineering qualification. Indeed IGIP concede the role of the subject matter under consideration in the selection of a teaching methodology (Rüütmann, 2017). Rather than this discipline-oriented approach being evident in the IoTs consequent to the NFQ, Kenny (2009), a fellow engineering educator, argues that "teaching is reconstructed to meet the defined LOS" (p. 24).

In suggesting alignment of pedagogy with LOs, HESG (2011) proposes that "academic staff should make full use of the range of pedagogical methodologies [...] and be qualified as teachers as well as in their chosen discipline." (p. 11). However, if alignment of pedagogy with LOs is a requirement then the range of teaching approaches available may be somewhat restricted. As Finnegan (2016) writes "neoliberalism and the linked phenomenon of outcomes

²¹ P. 28

based assessment have narrowed the educational imagination" (p. 55). For example, consider the DIT *Postgraduate Diploma in Third Level Learning and Teaching* (DIT, 2017), designed for new academic staff. The indicative syllabus covers a range of models for curriculum design and assessment. Given that module design and delivery in the DIT was firmly LOs based, the LOs oriented aspects appear most likely to influence academics taking the programme.

I am concerned that the pedagogic basis of the NFQ, so firmly linked to assessment, like other NQFs (Deacon & Parker, 1999) is quite limiting. Pedagogy is not dictated by EI's accreditation criteria, which is "left to the HEIs" (EI, 2014, p. 6). However, given the LOs focus, it would be expected to reinforce the influence of the NFQ. Whereas Kenny (2010) questions whether LOs might "have any effect on pedagogical practice" (p. 45) I considered, given the lack of attention to pedagogy in the drivers for OBE in Irish engineering education, that, as suggested in the literature (Finnegan, 2016), there may be significant implications. I was to explore this with my research participants in order to understand their views and experience of how our pedagogy as engineering educators has been affected by the NFQ.

3.11 Conclusion

The NFQ and EI are revealed as dual pressures on IoT engineering academics to engage with OBE. A skills agenda, reflecting international and European developments, dominates national policy for higher education, for which the NFQ is a key enabler. EI, similarly influenced by international developments, adopted an OBE approach for accreditation of engineering programmes, and, through their interpretation of Bologna, introduced new accreditation requirements leading to structural change in IoT engineering education.

It is important not just to acknowledge, but to consider the implications of these dual pressures on IoT engineering education. How are the regulatory nature of the NFQ, and the influence that EI have as an engineering professional body, perceived by engineering educators, and what impact do they have? The NFQ concept of progression, alongside the skills focus of OBE, raises concerns regarding the weakening of boundaries between the level 6, 7, and 8 programmes, which traditionally represent a ladder from more vocationally focussed programmes to more academic focussed ones. What impact have the structural changes resulting from El's interpretation of the Bologna framework had? And what of the student journey of identify formation as they progress from one level to the next?

The literature review also highlighted the need to consider the effect the NFQ, alongside El's accreditation requirements, may be having on curriculum design, and whether engineering educators' view of knowledge has been affected. Neither the NFQ nor El's make stipulations regarding pedagogy, but the literature suggests the assessment focus has implications. I show the NFQ value of transparency to be a problematic concept, given that LOs may not be interpreted with the same meaning by either educators or students. Above all else, these considerations suggest that the NFQ and OBE based accreditation requirements do more than introduce additional workload through increased QA, they would be expected to have tangible effects on our academic identity, and on the education we offer.

In the next chapter I will discuss the conceptual framework that my literature review, and my consideration of my own role in the implementation of OBE, led me to develop to assist in contextualising these complex, inter-related, aspects of our engineering education system, in order to explore them in the fieldwork.

Chapter 4 A Conceptual Framework from which to consider IoT Engineering Education

"I'm an Engineer," he said simply, as if it would explain everything. April Adams (2011, Kindle Edition, "Two 8", para. 30).

4.1 Introduction

My review of the development of Institute of Technology (IoT) engineering education, and the dual pillars of its outcomes based education (OBE) approach, highlighted the international and national policy influences and other interests that act within this arena. In order to develop a fuller understanding of this complex field I was researching I sought to develop a conceptual framework which reflected my own experience as an engineering educator, and allowed me to analyse and learn from it. Although I initially struggled to find an adequate theoretical grounding for my research, I was later drawn to Bernstein's (2000) characterisation of engineering as a region, as described earlier, inwards facing towards academic, and outward facing to the market, mediated by the engineering professional bodies, to assist in conceptualising what I was reading and observing. This social realist perspective foregrounds the importance of the consideration of the power relationships between both internal and external forces, and the historical conditions within which they arise, in developing understanding of education systems (Muller, 2004b; Young, 2008), as has been proposed for engineering education (Akera, 2017, Edström, 2018).

In my conceptual framework I draw selectively on social realism's contribution to the sociology of knowledge, particularly the concepts and research of Basil Bernstein (1971, 1999a, 1999b, 2000, 2001) and Michael Young (1971, 2002, 2003, 2005, 2008, 2010, 2011, 2014; Young & Muller, 2010, 2014). This influence emphasises the importance of knowledge and the curriculum in education. In recognising the social construction of knowledge, it foregrounds consideration of the social and political issues and structures within which our knowledge as engineering educators is constituted and framed (Barrett & Rata, 2014a), issues which are very much to the fore in the use of OBE (Allais, 2014, Matos et al, 2017; Young, 2011). Alongside social realism's acknowledgement of the manner in which pedagogic discourse acts to shape academic and student identity (Bernstein, 2000), this provides a framework, building upon my own experience as an engineering educator, from which the impact of OBE on engineering educators and education can be regarded and critiqued.

4.2 Drawing on Social Realism

Matos et al (2017) speak of the importance of considering power relationships in understanding the structure of engineering education, and identify that "accreditation maintains a continuing power struggle over the control for regulatory power of knowledge" (section *Governance and Accreditation*). They regard LOs based accreditation as promoting a "neoliberal turn" (section *Relationship of ABET to Engineering and Engineering Education*) in international engineering education, raising questions regarding the control of access to knowledge, the identity formation of graduates, and academic identity, including approaches to assessment. In considering the role of qualification frameworks, Young (2008) regards the introduction of OBE as attempts by government to exert increased control over education, and to break with past practices, which have been identified as some of the drivers for the introduction of the Irish National Framework of Qualifications (NFQ) (Allais, 2017).

My conceptual framework allows engineering education to be critiqued from this perspective, drawing on the educational sociology lens provided by social realism, which has been applied in research into a number of aspects of engineering education (Case, 2015; Nudelman, 2018; Shay, 2012; Smit, 2016; Wolff, 2015). My principal social realist influences are Michael Young (1971, 2002, 2003, 2005, 2008, 2010, 2011, 2014; Young & Muller, 2010, 2014) and Basil Bernstein (1971, 1999a, 1999b, 2000, 2001), two key proponents. In order to contextualise their influence I first discuss social realism from an historical perspective charting the changing viewpoints over time of Michael Young (1971, 2002, 2008).

I then describe important theoretical aspects, with a particular emphasis on how I make selective use of the ideas of Bernstein, who made significant contributions to the educational sociology of curriculum and pedagogy through: his understanding of how communication is affected by power relationships: how this influences pedagogy and the curriculum; and how this acts to shape the identity of those teaching and those being taught. Young looked to Bernstein as a principal theorist, identifying his key contribution that "you cannot have a curriculum theory without a theory of knowledge" (Young, 2014, section *Introduction*). Young (2014) proposes that research into curriculum theory should investigate the processes associated with Bernstein's concept of recontextualisation²², where competing pedagogic

²² See p. 10 and p. 93

and official recontextualising discourses highlight the tension between educators and government and other official agencies in shaping the curriculum.

4.3 From The New Sociology of Education to Social Realism

In order to understand contemporary social realism, it is necessary to consider the intellectual journey that led to it, commencing with the new sociology of education, which emerged from Young's earlier research. This stressed the importance of considering the curriculum and pedagogy in research into the sociology of education, and of understanding the competing power relationships between different groups involved in the specification of the curriculum.

The publication of *Knowledge and Control* (Young, 1971), which included contributions by Young (1971) from an anti-positivist perspective, and Bernstein (1971) and Bourdieu (1971) from structuralist perspectives (Hoadley, 2015; Young, 2002), marked the founding of the "new sociology of education" (Young, 2008, p. 3). This asserted that curriculum and pedagogy should be the focus of a re-orientation of the sociology of education (Young, 2002). Young advocated a social constructivist perspective in which power relations within what he terms the "dominant institutional order" (Young, 1971, p. 34), consisting of "economic, political, bureaucratic, cultural and educational" groups, were conceived as central to the selection of the curriculum content. The stratification of knowledge which follows from this leads to a view of the curriculum as being constructed to reflect the knowledge of the powerful, and thus a source of unequal access to education. Indeed, Young (1971) questioned whether the very basis of academic curriculum, "bookish learning" (p. 38), was anything other than a manifestation of "the values and beliefs of dominant groups" (p. 38).

The new sociology of knowledge sustained considerable criticism from conservative advocates of traditional educational approaches, where in proposing that the curriculum was inherently socially biased it was seen as "challenging something almost sacred" (Young, 2002, p. 40). The new sociology of education was also criticised for its characterisation of teachers as the principal agents of change, divorced from policy makers and policy implementation (Young, 2002).

Young (2008) was to later acknowledge criticisms of the theoretical limitations of the new sociology of education (Moore & Muller, 1999) as a form of voice discourse which eschews the "epistemological grounding of knowledge" (Young, 2008, p. 3) in favour of the

"standpoints of [...] dominant social groups". Legitimation of knowledge claims moves from "what is known (and how) to who knows it" (Moore & Muller, 1999, p. 200). Thus a principal theoretical problem was that the new sociology of education ignored the fact that although the knowledge in the curriculum may reflect the vested interests of controlling groups, it may also be a source of "real understandings" (Young, 2002, p. 44): 'powerful knowledge' as opposed to 'knowledge of the powerful' (Young & Muller, 2010).

Having later engaged in a considerable body of work on policy related matters (Hoadley, 2015), Young was also to accept criticism (Moore & Muller, 1999) that the theoretical weakness of the new sociology of education was compounded by the insulation of those proposing it (university academics) from "both policy and practice" (Young, 2002, p. 47). For example, the new sociology of education provided "no criteria for developing and assessing curricula alternatives" (Young, 2002, p. 45). Indeed, taken to an extreme it could be used to argue for a curriculum totally based on experience, or indeed none at all, i.e. "deschooling", (p. 44). In reflecting on his earlier questioning of the traditional classroom based curriculum, Young also began to see a value beyond the social stratification that may have led to it: recognising the role that formal schooling can provide in inculcating knowledge that students can make use of in the wider-world (Wilby, 2018; Young, 2002).

However, despite its relatively short life (Young, 2002), and theoretical limitations (Moore & Muller, 1999; Young, 2002, 2008), the new sociology of education was important (Hoadley, 2015) in contributing to a debate as to what counted as educational knowledge, and seeking to move the focus of the sociology of education to curriculum and pedagogy. It also made a distinctive contribution (Young, 2002, 2008) in identifying the considerable influence power relations had on knowledge in the curriculum. Bernstein (2001) contends that though the approach of the new sociology of education may have been flawed, the problems it identified continue to require attention, and I consider them important for the research in this thesis.

Building upon his earlier writings, the criticism it attracted, his subsequent experience in educational policy (Hoadley, 2015), and the research of Bernstein (1999a, 2000), Young (2008) set out a vision of social realism as a sociology of education with knowledge at its heart, whilst fully acknowledging and considering the social circumstances that leads to its creation. The subsequent perspective on knowledge can be viewed in terms of the epistemological dilemma, where knowledge is regarded from either a positivist viewpoint, where it is
absolute, and "located outside of society" (Young, 2008, p. 25) or a relativist standpoint, which, in "privileging the exclusivity of particular experiences" denies the "possibility of knowledge that goes beyond [...] experience". Social realism transcends this dilemma through acknowledging that, although all knowledge is created in a social context, it is not reducible "to the activities and interests of those who produce or transmit it", (Young, 2008, p. 94), and some knowledge will have a truth and value outside of the context of its creation.

4.4 Knowledge

My personal ontological and epistemological positioning, which is shaped through my education and experience by the engineering view of the utility of knowledge (de Vries, 2006; Montfort et al, 2014), is a key aspect of my conceptual framework. Drawing usefully upon social realism, which emphasises the importance of knowledge and the curriculum in education (a key consideration in my research) entails my engineering worldview of what constitutes knowledge should be commensurate with the social realist perspective.

As a professional engineer and engineering educator (Winberg et al, 2012), I regard knowledge, to a significant extent, as that engineering knowledge, of technology, engineering science, practice etc., which we will teach our students on their path to becoming engineers. However, an engineer also requires the ability to operate successfully within and outside organisational structures, using the tacit knowledge that Heywood (2016) characterises as *know-how*, but which also involves *know-who*. Thus as an engineering educator knowledge is also my understanding of how to get things done, and who with, which has inculcated within me a consciousness of the organizational structures and power relations within our sector, and the ability to navigate within these to achieve ends.

Social realism borrows from the critical realism (Archer et al, 1998) position on knowledge, developed through consideration of ontological realism, epistemological relativism and judgemental rationality (Maton & Moore, 2009; Moore, 2014). The key idea of the new sociology of education, the importance of power relations in the selection of knowledge for the curriculum, is maintained, but knowledge no longer reduces to "statements about knowers" (Moore & Muller, 1999, p. 190).

Ontological realism, the principle that there is a "reality beyond the symbolic realm" (Maton & Moore, 2009, p. 4), is a recognition that, although socially constructed, our knowledge of

the world reflects, although is not identical to, our experiences and beliefs regarding a reality that exists (Smit, 2016). Engineering is concerned with the production of artefacts (Lipton, 2010) through which we interact with that reality (Lawlor, 2016), relationships within which the assumption of ontological realism can be considered implicit (Hector et al, 2018; Zelic & Stahl, 2005).

Epistemological relativism views knowledge as "not necessarily universal, invariant, essential Truth – we can 'know' the world only in terms of socially produced knowledges which change over time and across socio-cultural contexts" (Maton & Moore, 2009, p. 4). As science, technology, and their application move forward new engineering knowledge displaces the old in professional practice through its dissemination by engineering practitioners, professional engineering bodies, engineering educators and others, an epistemological relativism.

Judgemental rationality completes the framework within which social realism regards knowledge, asserting that decisions as to what constitutes knowledge are made on a rational basis. Archer et al (2016) describe this in terms of "we are able to, and required to, adjudicate between rival or competing accounts, and there are often relatively objective reasons for affirming one model over another". Sayer (1992) suggests practical adequacy as providing a rational basis for such decisions. i.e. "to be practically adequate, knowledge must generate expectations about the world and about the results of our actions which are actually realized" (p. 69). Arguably this is the implicit model used by many engineering practitioners, who will select one technique, technology or piece of science over another on the basis of practical adequacy (Tilley, 2016), through their own experience or that of their fellow-professionals, and, may regard some things as not useful at all, and thence not engineering knowledge.

Judgemental rationality has profound implications for education, in that it follows firstly that there is a body of "established powerful knowledge" (Moore, 2014, p. 37) to be taught. This is the means through which engineering education introduces students to the body of engineering knowledge (Heywood, 2016) they require in their identity formation (Haase, 2014). Secondly, the principles that led to the production of this knowledge need themselves to be taught through their embedding in pedagogy in the classroom (Moore, 2014). Frezza and Nordquest (2015) speak eloquently of what this can mean for engineering education:

an engineer's education and practical experience will make him or her familiar with an abundance of patterns of data and of relations of causality and dependence – and these

patterns can be brought to bear both in achieving insights and in establishing the conditions that must be met for something really to be so (section *Patterns of Knowing*).

4.5 The Engineering Professional Bodies, Cognitive Interests, and Knowledge Structures

Social realism highlights the role of the engineering professional bodies as communities of trust (Young, 2008) for engineering as a region (Bernstein, 2000), and this is important for the analysis offered in this thesis. I explore this through social realism's contention that there are two types of social interest that regulate the production and dissemination of knowledge. The first are 'external' interests associated with wider societal developments, for example economic and political pressures. The second are 'internal' cognitive interests associated with the creation and acquisition of particular knowledge, for example academic disciplines and professional bodies, and which form a contested intellectual field. Social realism see's specific value in understanding the historical events that influence, help form, and contribute to change in these social interests. This leads to complementary aims, on the one hand to understand how external interests may exert power to influence the curriculum, and on the other hand to understand the influence that the social organisations that correspond to the cognitive interests will in turn have in shaping wider society (Young, 2008l; EI, 2017c). An implication for my research into the consequence of OBE for engineering educators and education is that I must be cognisant of the shaping of engineering by society but also that of society by engineering (Heywood, 2017, Pawley, 2019).

The social organisations that form cognitive interests can be considered through the ideas of Bernstein (1999a), who distinguishes between two types of knowledge discourse, the first of which are horizontal discourses, which reflect the common-sense, experiential knowledge that people will acquire as they go about their every-day social activities. The second are vertical discourses, which contain the coherent knowledge base of an academic discipline, organised through "specialised symbolic structures of explicit knowledge" (Bernstein, 2000, p. 160). These knowledge structures are typically acquired in academic settings in accordance with pedagogy and recontextualisation of knowledge associated with the academic discipline (Young, 2008). Critically, this means that the coherency of an engineering curriculum relates not only to the knowledge and skills selected for the curriculum, but also to the "associated pedagogic practice" (Wolff, 2015, p. 5).

Bernstein's describes academic disciplines as singulars, intellectual fields with their own "texts, practices, rules-of entry, examinations, licences to practice [...] etc." (Bernstein, 2000, p. 52). For some singulars, e.g. the humanities, knowledge is horizontally organised through "a series of specialised languages with specialised modes of interrogation and specialised criteria" (p. 161). In others, e.g. the sciences, knowledge is hierarchically organised, with increasing abstractions at higher levels incorporating concepts from lower ones (Wolff, 2013).

Concern has been raised regarding the impact of OBE on hierarchically organised knowledge (Allais, 2014, Wolff, 2015). Whilst acknowledging that the EQF explicitly includes knowledge as one of the headings under which LOs are specified, Allais (2014) proposes that it is information rather than knowledge that is being assessed through such LOs. Knowledge "is viewed as information or facts—something that can be broken into little bits which can be selected and combined at will." (Allais, 2014, p. xx), not taking into account the hierarchical nature of knowledge in many disciplines.

Regions comprise groups of singulars, and "face inward" (Bernstein, 2000, p. 55) to academic disciplines, and "outward" to the field of practice (e.g. engineering). Engineering knowledge is comprised of the disciplinary knowledge of its constituent singulars, but also has its own knowledge, particularly in engineering practice. The concept of design (of artefacts) epitomises engineering practice, which integrates knowledge from across the singulars that constitute the engineering region. However, design also builds upon the experience of engineers, and the application domain of the artefact, in doing so making substantive use of social skills (Wolmarans, 2017). Consequently, although some engineering knowledge is hierarchically organised, e.g. design of systems, other knowledge, although part of a vertical discourse, will be horizontally organised (Wolff, 2013; Wolmarans, 2017), and some engineering knowledge will even be part of a horizontal discourse (Wolmarans, 2017).

Regions may include 'communities of trust' with associated powers of accreditation and standard setting, of which the engineering professional bodies are a particular example. These provide an objectivity to the knowledge associated with a region and a measure of the standard of those admitted to the discipline as they move from being students to graduates (Young, 2008). Furthermore, they have a role in the establishment of professional identity (Wolmarans, 2017). Singulars and regions, and their associated knowledge, may well have elitist origins, but the cognitive interests intrinsic to them have a real value in the production

of knowledge and its recontextualisation in the curriculum (Young, 2008). This highlights the influence of the engineering professional bodies on the curriculum through their adoption of OBE for accreditation. My conceptual framework reflects that I consider the curriculum not just a source of education and knowledge about engineering (Heywood, 2016), but, as do others (Matos et al, 2017, Pawley, 2019), a site of "ideology and power" (Edström, 2017, p. 15). This recognises that engineering cognitive interests have an associated emancipatory potential, a critical consideration given the manner in which society is shaped by engineering (Homan, 2020).

4.6 Curriculum

Introducing students to engineering knowledge is a key curriculum objective of engineering education (Heywood, 2016; Wolff, 2015). This position of the importance of knowledge in the curriculum is shared with social realism (Young, 2008), with Allais (2014) considering that "knowledge should be the starting point of curriculum design" (p. *xv*). However, whether for technician or professional engineering education, the application of that knowledge in engineering practice through skills is critical (Edström, 2017; McLaughlin, 1999; NAE, 2001, 2017). It is argued that at times engineering education struggles to find the appropriate balance between knowledge and skills (Dempsey, 2017; Dooge, 2006; Edström, 2018; Heywood, 2016; Johnston et al, 1996; Prados, 1992; Seeley, 1999). Debate on this matter has led directly to the adoption of OBE for accreditation of contemporary engineering education, with the pendulum between the analytic and practice orientations which Edström (2018) identifies currently ascended towards the skills maxima.

The needs of the market have been shown to be, historically, a strong influence on engineering education (Heywood, 2016; Akera, 2018). However, Young (2008), in reference to Bernstein (2000), raises a concern regarding regions: the "readiness on the part of professionals and [...] scholars to respond to whatever the exigencies of the markets require in the shaping of their intellectual fields" (p. 155), which influences include OBE. One could argue that the fact that market oriented neoliberal influences lead to OBE and its skills emphasis (Barnett & Coate, 2005; Allais, 2014) is just another manifestation of this influence from a curriculum perspective. Certainly OBE, despite concerns as to its reinforcement of the neoliberal agenda (Allais, 2014, Matos et al, 2017; Pawley, 2019), is largely regarded, as I have shown, as a positive influence on engineering curriculum (ABET, 2006; EI, 2014; Froyd et al,

2012; Froyd & Lohmann, 2014; Owens, 2016a).

However, engineering, in providing us with the artefacts required for, and enabling the infrastructure of, our modern society (Lawlor, 2016), is a human endeavour (ASEE, 2020) in which we must strive for "morality in our actions" (Einstein, as quoted in Dukas & Hoffmann (2013, p. 95). It has been argued that the neoliberal influence "abdicates engineering's moral choices to the market" (Pawley, 2019, p. 452), preventing engineers from engaging fully with important debate on "the purpose of engineering education, and who should decide".

Pawley (2019) sought to provoke discussion regarding the effect of the neoliberal influence on engineering education, rather than offering solutions, and addressing this fully within this thesis is not feasible. Through my research I can offer some insight to contribute to the debate, in seeking to understand, for a sample of IoT engineering academics, whether they consider the skills emphasis of OBE leads to an appropriate, balanced, curriculum approach, suitable for the education of engineers in our modern Irish society. A particular concern is whether OBE has led to the de-emphasis of knowledge identified by the social realists (Allais, 2014; Young, 2008). Furthermore, is there a diminution in the discipline specific cognitive and pedagogic elements of the production and teaching of engineering knowledge (Wolff, 2015)?

4.7 Recontextualisation of Knowledge and the Pedagogic Device

Bernstein (2000) regarded pedagogy as the transformation of knowledge into a discourse associated with teaching that knowledge, and also the teaching, or pedagogic practice, itself. This transmission of knowledge is influenced by power relations between vested interests associated with the pedagogic discourse, and encompasses: who the knowledge is to be transmitted to and where; alongside what is to be transmitted and how. Bernstein (2000) considered pedagogic discourse embeds two associated discourses, an instructional discourse related to disciplinary knowledge and skills, and a regulative discourse related to social order.

The concepts of 'classification' and 'framing' model power and control relationships within the pedagogic discourse (Bernstein, 2000). Classification is used to describe the degree of insularity between academic disciplines. A more traditional academically oriented discipline in which strong boundaries exist between subjects, and where the academic discipline is clearly defined, is said to be strongly classified. The degree of classification, strong or weak, is associated with power relationships within and external to the discipline. Framing refers to the degree of control within the discipline regarding pedagogy. Where there is strong control by dominant actors within the field over what is taught, how it is taught, and when it is taught there can be said to be strong framing.

Bernstein (2000) conceptualised the pedagogic device to address whether there are "general principles underlying the translation of knowledge into pedagogic communication" (p. 25), and in which the concepts of classification and framing are situated. It models the knowledge practices associated with pedagogy. The field of production is the site where new knowledge is produced, for example universities. The field of recontextualising is where disciplinary knowledge is selected and sequenced, based upon intrinsic and extrinsic interests, to become pedagogic discourse, and to form the curriculum. However, recontextualisation refers not just to what to teach, but also to the selection of the theory of instruction. The field of reproduction is the site where pedagogic practice occurs, i.e. where the curriculum, in tandem with pedagogy, is used by educators to teach students.

Field of Production

Field of recontextualising

Sites where knowledge from the field of

production are selected, re-arranged and

become

pedagogic

to

Field of reproduction

and

Sites for creation of new knowledge

Distributive rules

Recontextualising rules

Evaluative rules

Sites of teaching

learning

Figure 4-1 Arena Created by the Pedagogic Device(Maton, 2014, p. 48)

transformed

discourses.

The pedagogic device acts as an arena (Bernstein, 2000, Maton, 2014) within which the classification and framing of knowledge are mediated through a hierarchical set of rules: distributive rules define the power relationship between different social groups in terms of access to knowledge, practice, and consciousness, and regulate "who may transmit what to whom and under what circumstance" (Bernstein, 2000, p. 31); recontextualisation rules regulate the transformation of disciplinary knowledge into pedagogic discourse: evaluation rules constitute the pedagogic practice itself, and act to influence the consciousness of those who acquire the knowledge (e.g. the identity formation of engineering students).

The recontextualisation rules lead to a pedagogic discourse encapsulating an instructional discourse (what knowledge is available to be taught) and a regulative discourse (what *should*

be taught). The recontextualising field is formed from two sub-fields, the first being the official recontextualising field (ORF), corresponding to the educational apparatus and influence of the state, and which, for engineering education, has been extended to include the professional bodies (Klassen, 2018; Moodley, 2014). Bernstein uses the phrase "Official Knowledge" (p. 65) in relation to this field, as that which is expected to construct moral and motivational dispositions in academics and students, leading to "particular performances and practices". The second field is the pedagogic recontextualising field (PRF), corresponding to those engaged and concerned with pedagogy. Bernstein views the ORF as typically dominant, as the state and its agents attempt to exert control over "the pedagogic discourse and its social contexts" (p. 33), curtailing debate over "pedagogic discourse and its practices". This shapes both academic and student identities, and influences academics' pedagogic practice.

However, the PRF, and its associated cognitive interests, also contribute to the pedagogic discourse. For example, Bell and Stevenson (2006) acknowledge the central role of the state in defining educational policy, but that though academics may be influenced by policy, they also, through "personal characteristics, expertise [...] and opportunity" (p. 21) influence its formation and implementation. Similarly, in considering the effect of policy on educators, Coffield et al (2007) write that "policy makers are not writing upon a blank slate, but on a page already taken up with 'ecologies of practice', past and present initiatives and specific local factors" (p. 728). Thus the pedagogic discourse (Bernstein, 2000) that arises through OBE is being shaped not just by official knowledge of the ORF, but also by the academic staff.

I argue for the applicability of this model to engineering education through considering Heywood (2016) who emphasises the importance of knowledge in the curriculum. He speaks of "the curriculum process" (Heywood, 2005, p. 3), to which both "learning and instruction" are central, corresponding to the field of reproduction. Heywood (2005) acknowledges: the social nature of what he terms the curriculum process; the interdependence of engineering academics; the potential for conflict between competing internal, cognitive, interests (the instructional discourse which leads to the PRF); and the importance of regulatory influences such as government funding priorities and the engineering accreditation bodies (the ORF).

The emphasis on what is to be taught to new generations of students is decided through the distributive rules of the field of production, which is largely at the direction of the state with a view to meeting the needs of industry. The ORF is constituted through both the influence

and direction of the state, but also is influenced, for engineering, by our professional bodies (Klassen, 2018; Moodley, 2014). Together these act to influence: what knowledge is selected; why that knowledge is selected, and how that knowledge is recontextualised in pedagogy; and to influence the identity formation of both academics and students.

Tyler (2004) proposes Bernstein's ideas relating to pedagogic discourse provide a means to analyse market-driven educational policies, their implementation and impact, and Klassen (2019) has shown how it can be extended to consider higher education and the professions. Thus the pedagogic device provides a framework through which OBE, and the competing interests involved in influencing and controlling IoT engineering education, can be considered. Sitting alongside the insight provided through Young's deep interest in policy and its implications, this provides me with a means to analyse the role of the various influencers and regulators concerned with OBE in Ireland's IoT engineering education system.

4.8 Pedagogy and Identity Formation

Bernstein (2000) defines pedagogic identity as "the result of embedding a career in a collective base" (p. 66), and as being shaped by educational reform. Pedagogic identity emerges from the pedagogic practices resultant from the interaction of dominant social interests in both the ORF and PRF.

Bernstein (2000) contrasts different types of pedagogic practice that may emerge within the pedagogic device, influencing the pedagogic identity of the teacher and student. The first of these is termed competence or therapeutic, associated with learners having control over the pace, selection and sequencing of their learning. In a competency model highly trained teacher/facilitators assist students in moving towards their personal goals. It is assumed that all may become competent, that there is a "democracy of acquisition" (p. 43), and that learners actively participate in the construction of their own individually-oriented knowledge. However, in a performance model, strongly classified, emphasising assessment and grading, students have little control over what, when, where, or the pace at which, they learn. Performance models reflect pedagogic practices in knowledge fields with strong classification where grading, and relative performance of learners, is a key aspect. Bernstein regards performance models as associated, primarily, with singulars and regions.

Bernstein (2000) cautions of the tendency for regions to be dominated by market pressures,

which can cause regionalisation to become "a crucial recontextualising procedure" (p. 60), and Heywood (2016) identifies and challenges the utilitarian model of engineering education that has dominated post-ww2. Bernstein (2000) concludes that regionalisation, and its susceptibility to market pressures, leads to an erosion of the "inner commitments and dedications" (p. 62) of introjected identities associated with the disciplinary discourse to be replaced with "state-promoted instrumentality" (p. 61).

I earlier discussed the changes in academic identity as the Regional Technical Colleges (RTCs) expanded from their initial role (Hazelkorn & Moynihan, 2010; Moynihan, 2015; O'Byrne, 2009, 2011, 2014). A finding is that these changes are consistent with Bernstein's concept of pedagogic identity changing under the influence of a changing pedagogic discourse. The introduction of OBE, constituted as part of the ORF of IoT engineering education, would be expected to have a significant effect, above and beyond being a workload and quality assurance (QA) matter, on the pedagogic discourse, and pedagogic identity.

4.9 IoT Engineering Education Viewed as a Pedagogic Device

Knowledge is "socially produced and acquired in particular historical contexts, and in a world characterized by competing interests and power struggles" (Young, 2008, p. 88). Through my research I have shown that an understanding of the socio-historical context of IoT engineering education's adoption of OBE encompasses: the development of the engineering technician education system in the RTCs and its subsequent evolution into contemporary IoT engineering education; a review and critique of the development globally of OBE; a discussion of the skills focus that arose in Irish government education policy and that led to the development of Ireland's NFQ; and the concurrent adoption by EI of an OBE accreditation approach influenced by international developments in engineering accreditation, and also by the Bologna process.

The policy debate and formation which led to this can be regarded as the struggle over, and the establishment of, the distributive rules of the pedagogic device of IoT engineering education. Control of this is associated with the official knowledge of government policy and associated organisation and legislative structures. This includes Quality and Qualifications Ireland (QQI), the NFQ, the Higher Education Authority (HEA), the Department of Education and Skills (and the skills emphasis in government educational policy), the various RTC and IoT acts and other legislation, and the governance structures of the IoTs. Official knowledge is also formed from the influence of EI, through their accreditation requirements. The IoT engineering education system, can be represented, as per Klassen (2019), as a pedagogic device, *Figure 4-2*.

The distributive rules governing the structure of this education (Kwok, 2017; Lim, 2014) include the NFQ, which is used as a key policy tool in the government's skills agenda for higher education, i.e. who is taught what. EI, through their accreditation requirements, and the affect that their interpretation of Bologna has had on the structure of engineering education, also form part of the distributive rules, i.e. where things are taught.

The discourse of the ORF, where factors related to the use of OBE compete to exert control over the pedagogic discourse in the field of recontextualisation, is formed, *inter alia*, from the influence and prescription of the NFQ and EI, particularly over course design and course structures. Key aspects are set out below.

- The government skills agenda for higher education, which is enabled through OBE, and which is linked to the concept of lifelong learning.
- The NFQ value of *transparency*, through which the use of LOs is mandated, is a key factor in course design and programme review (Kenny, 2006).
- The nomenclature of the NFQ, and the QQI Engineering Standards form a language of learning (Biesta, 2005), influencing academics' engagement with LO's. However, I will show that the voices of the interviewees reveals this goes deeper, in their use of a language of levels (Raffe, 2011) to engage with policy *and also* academic practice.
- The structure of the NFQ, and the foregrounding of progression.
- El's accreditation requirements reinforce the use of OBE (Forero et al, 2011; Kenny, 2006), and additionally influence the structure of engineering education.
- El also have a further, complementary, influencing role as the professional body of the community of practice of engineering.

In examining this struggle over the pedagogic discourse (Bernstein, 2000) I regard the NFQ and EI, alongside government policies and economic imperatives, as having both an implicit influencing aspect and an explicit prescriptive aspect (Ingram, 2016; Slough-Kuss, 2015) on engineering education and educators. This dual aspect of control over Irish education is recognised by Kenny (2006), who views policy implementers as seeking to "*influence*²³ and *direct*²³ the higher education environment" (p. 311) and the work of academics.

²³ My italics

Field of Production

Sites	for	Creation	of	New
Knowledge				

Distributive rules: The selection of skills and competencies as what is to be taught to new generations of students can be seen as a consequence of the dominant influence over the distributive rules by neo-liberal oriented economic interests.

The NFQ promotes the teaching of skills in higher education, in line with government policy.

El's accreditation requirements increase vertical stratification in engineering education.

Field of Recontextualisation

The **ORF** is formed from the influence and the control of the

NFQ, QQI, HEA, and also EI's accreditation requirement, i.e. by

Government Policy, in particular the skills driven knowledge

Transparency, a key under-pinning of the NFQ, manifests as

use of learning outcomes, leading to new course design

The nomenclature of the NFQ, helping to communicate policy and shape implementation, including the level descriptors and

Engineers Ireland require OBE approach for accreditation

The Discourse of the Official Pedagogic Field

economy and related economic imperatives

Official Knowledge

requirements

the engineering standards

Progression is foregrounded

Skills being emphasised

The Bologna declaration

Aspects of this discourse include:

IoT Engineering Education as a site where knowledge and skills from the field of production are selected, re-arranged and transformed to become pedagogic discourses. This includes programme and module design.

Recontextualising rules

The Discourse of the Pedagogic Recontextualising Field

The **PRF** incorporates the cognitive interests of academics

The NFQ would be expected to influence the pedagogic discourse

Pedagogic identities of academics are being re-shaped by the **ORF**

Field of reproduction

The IoT Engineering sector as

sites of teaching and learning

Evaluative rules

OBE approach to teaching and learning.

Identity of new graduates is being formed.

How is the technician identity being affected?

Figure 4-2 The Pedagogic Device of IoT Engineering Education, as Shaped by the NFQ and EI, alongside other Policy Drivers, adapted from Bernstein (2000)

In Collins et al (2009), the framework is described as "embedded in academic policies" (p. 10), reflecting an explicit prescriptive aspect. In constituting part of the dominant, regulative, discourse of the ORF, the NFQ exerts explicit control over the pedagogic discourse through defining new course structures and validation requirements which *must* be followed. Similarly, El's use of LOs within their accreditation processes is quite prescriptive. Slough-Kuss (2015) characterises this as the ORF describing what education "should be" (p. 19).

On the other hand the influencing nature of the framework is suggested where understanding of the framework is said to depend upon "the level of engagement of individuals" (Collins et al, 2009, p. 16), and the "the *insinuation*²³ of the Framework into the daily processes of teaching and learning" (p. 49). The more inclusive engagement referred to by Collins et al seeks to connect educational policy with academics as "social actors" (Slough-Kuss, 2015, p. 23), *influencing* lecturers in how they approach course design, pedagogy, knowledge acquisition and recontextualisation of knowledge.

The resultant pedagogic discourse acts to re-shape the pedagogic identity of academics: in their attitudes to knowledge (Allais, 2014; Barrett & Rata, 2014b; Young, 2008), curriculum and pedagogy (Bernstein, 2000; Muller, 1998), and may impact on the stratification of engineering education itself. This will have consequences for the education we teach, and in turn the identity formation of our engineering students. The pedagogic device provides a model within which the fieldwork can be conceptualised, allowing consideration of the pedagogic discourse that has arisen (Bernstein, 2000).

4.10 Conclusion

I have shown how my conceptual framework draws selectively on social realism, alongside my experience as an engineering educator. This foregrounds the influence of power relationships on the structure of that education and the selection of knowledge in the curriculum. Engineering is regarded using Bernstein's concept of a region, which highlights the influences of the professional bodies and the market in engineering education.

Key to my conceptual framework is the use of the pedagogic device to model the discourse of IoT engineering education. In order to establish the "rules" (p. 27) which regulate the behaviour of the pedagogic device of contemporary IoT engineering education, I drew on the socio-historic accounts I developed earlier, and the understanding of the power-relationships in this contested field these accounts, and my own experience, engendered.

The distributive rules of this pedagogic device are determined by official knowledge of the state, EI, and market influences. Alongside the cognitive interests of engineering academics, these additionally contest the field of recontextualisation. The field of evaluation consists of the sites of teaching and learning for IoT engineering education, and is where the identity formation our engineering students takes place. This pedagogic device will be used to frame the analysis of the fieldwork, and to situate critique of OBE, as I examine its impact on my interviewees and the engineering education system they work within.

This provides a conceptual framework from which the political and socio-economic considerations that lead to the use of OBE, including El's role as the engineering professional body, can be regarded and critiqued. It acknowledges the stratification inherent in engineering education and the power relations at play, foregrounding related issues, such as progression and student identity formation. It allows consideration of our identity formation as engineering academics, in terms of the discourse we engage in as pedagogues, and with regard to how we view knowledge, skills, and their recontextualisation in the curriculum.

Chapter 5 Research Methodology

Beware of first- hand ideas!' exclaimed one of the most advanced of them. 'First-hand ideas do not really exist. They are but the physical impressions produced by love and fear, and on this gross foundation who could erect a philosophy? Let your ideas be second-hand, and if possible tenth-hand, for then they will be far removed from that disturbing element - direct observation. E. M. Forester (1909, part III, para. 3).

5.1 Introduction

I designed the research methodology to allow me to examine how engineering academics in the Institute of Technology (IoT) sector perceive the impact of outcomes based education (OBE), on themselves and their colleagues, on the education they teach, and on the identity formation of their students. Given my training as an engineering educator, with strong positivist and rationalist influences, following a quantitative approach might have seemed more natural (McNutt, 2010; Nudelman, 2018). However, as I was introduced to the concepts and possibilities of qualitative research, the rich understandings it enables, including for engineering research (Kelly & Bowe, 2011), offered a powerful approach to gaining insight from my engineering academic colleagues at the "front line" (NAE, 2001, p. 23) of OBE. Furthermore, I have, through my time as an engineering, and academic, manager, seen the value in gaining an "individual's point of view" (Denzin & Lincoln, 2011, p. 9) in order to develop understanding of shared issues, a strength of qualitative research. I concluded that it would be very difficult to understand the field I was researching without using a qualitative approach to gain access to the ideas, impressions and actions of my colleagues and peers in the sector: quantitative methods, such as surveys, would just not provide the rich data I needed for my research to progress.

As is not uncommon in qualitative research (Hewlett, 2013), my research had not been fully planned, and, as it progressed I struggled with a theoretical framing which would allow me to make sense of the complex field my literature review, and the fieldwork, began to reveal. My literature review grew from being an analysis of the National Framework of Qualifications (NFQ) as it related to IoT engineering education to encompassing a number of inter-related socio-historical accounts: of the genesis of engineering education in Ireland; of the development of the RTC/IoT sector in response to government policy that saw education as playing a key role in improving the life of the citizenry through economic advancement; the

later development of the NFQ as part of a skills-focussed reorientation of higher education; and the reasons for, and consequences of, the adoption of OBE for engineering accreditation, internationally and in Ireland. As I read further into the literature, and began to analyse data from the fieldwork and grappled with how to interpret it, Bernstein's (2000) concept of engineering as a region, alongside social realist critique of OBE, offered explanations which resonated with what I was reading and observing. This led me to develop the conceptual framework described in *Chapter 4*.

My conceptual framework reinforced the importance of following a qualitative methodology. This research approach allows us to seek understanding regarding what it is about the social structures in which effects are observed that have caused them to come about (Sayer, 1992), and offered an approach to accessing the knowledge of practitioners in the discourse (Sayer, 2000) of IoT engineering education. Van et al (2008), from an engineering perspective, regard a qualitative approach as ideal for research that explores the thought processes and dispositions of teachers and students, allowing for the generation of descriptions and understanding of their actions in context. Hewlett (2013), from a social realist viewpoint, highlights the relevance of a qualitative approach in research challenging the dominant viewpoint of curriculum. This has a particular resonance with my research, which examines issues related to the privileging of the dominant discourse (Tilley, 2019) of learning outcomes (LOs) in engineering education. These arguments gave me confidence that my research methodology would connect my theoretical paradigm to "strategies of inquiry" (Denzin & Lincoln, 2011, p. 14) and "methods for collecting empirical material", through working with my research participants to generate the understanding I sought.

Representing the engineering education system in the IoTs as a 'pedagogic device' provided a focus for my lines of inquiry. My fieldwork was particularly concerned with the role played by the NFQ (QQI, 2014b) and EI's accreditation requirements as 'official knowledge' (Klassen, 2018; Moodley, 2014), although other government policy, legislation and associated organisational structures also contributed. I used the framing provided by the pedagogic device to examine how these influences on the pedagogic discourse act to re-shape the pedagogic identity of academics, on their attitudes to knowledge, curriculum, pedagogy, the identity formation of students, and on the stratification of engineering education itself.

This chapter discusses the choices I made in designing the fieldwork and analysis phases of

the research methodology, describes the implementation of the associated research methods and concludes with a discussion of ethical issues.

5.2 The Research Question

The research question is "what effect has the use of OBE had, and is having, on engineering educators and engineering education in the IoT sector in Ireland". A serious of related subquestions emerged from the research process which I explored with my research participants, taking cognisance of the claims of proponents and critics of OBE.

To begin with I expanded the research question to consider aspects such as:

- how do engineering academics perceive the influence of the NFQ and EI?
- what other policy driven changes to engineering education are at play in the sector?
- what acceptance has there been by academic staff to the use of OBE, and how is this manifest?
- are any of the problems that were suggested in the literature evident?
- how are the NFQ principles of transparency and progression seen to have impacted on engineering education?

However, as the literature review progressed (*chapters 2-4*), and my conceptual framework began to solidify, deeper, more reflective, questions emerged:

- how are these factors affecting the pedagogic identity of the engineering faculty?
- what effect has OBE had on: pedagogy; curriculum design; and knowledge building?
- what changes are seen to have occurred to the structure of engineering education in the IoT sector since the implementation of the NFQ?
- do these changes reflect a shift from technician education to professional engineering education in the IoT sector, or is something more complex occurring?
- what are the implications for the stratification of engineering education and the identity of the graduates being produced?

The questions I posed in the interviews were also influenced by aspects of IoT engineering education highlighted by my research participants. In particular, the emphasis, in the first interview, by Bill on the concept of progression caused me to foreground my inquiries in that regard. A further notable interview topic that emerged from the analysis of the earlier interviews was to ascertain my research participants views of the 'language of levels' that was to become apparent in the analysis. This not only confirmed my interviewees were cognisant of its use, but, in one case (George) led to discussion as to whether it was appropriate for representing graduate identity. As the interviews progressed, the importance of Engineers

Ireland (EI) influence also became even more apparent than it had through the literature review, ensuring I gave my interviewees adequate space to share their view of this with me.

5.3 Selection of Research Participants

Having chosen avenues of inquiry, my selection of interviewees was guided with a view to allowing the areas of inquiry that I had identified to be investigated (Denzin & Lincoln, 2011; Sayer, 1992). To achieve this I used my knowledge of the field to address a number of crucial issues, including ensuring that those I recruited would be in a position to contribute effectively to my research (Denscombe, 2003), and that I would be able to recruit a sufficient number.

Engineering lecturers are experts in particular facets of engineering and engineering education, but, also, they share common roles, aims and means of enacting those. I decided that research participants should be my colleagues in the sector, practitioners in the field, from whom I hoped to gain access to the insight and understanding I sought. My research participants would have either or both of the following roles:

- engineering academics, from whom I would seek to understand their experience of the effect of OBE.
- academic management/leaders who would be able to provide a complementary viewpoint of their experience of helping to lead the implementation of OBE.

I considered whether probability or non-probability sampling was appropriate (Saunders, 2012) for the selection of research participants, and in addition what constituted the sampling frame I should use (Denscombe, 2003). Given the nature of the research, the resources available, and the research question, the rich understandings offered by non-probability sampling, see *Table 5-1*, seemed most suitable in the search for answers to the research question. This offered an approach in which I could gain a new perspective on the use of OBE in engineering education, revealing and illuminating key themes, and which would support theoretical generalisations (Saunders, 2012).

In non-probability sampling the choice of potential research participants is "based on the researcher's judgement regarding those of the population's characteristics that are important in relation to the data required to address the research aim" (Saunders, 2012, p. 39). All of my prospective research participants, i.e. lecturers, heads of school, heads of department, had lecturing duties, were involved with course design and delivery, and with knowledge

building associated with that, i.e. they helped form the 'pedagogic recontextualising field' (PRF). Management would bring an additional perspective on managing the change to OBE, where they act as an interface between the 'official recontextualising field' (ORF) and the PRF. However, the predominant viewpoint that I sought was that of lecturers' perception of the change in engineering education, and how they perceived this impacted them. Consequently, I chose a ³/₃ to ³/₃ split in sampling between lecturers and academic management.

Difference	Non-Probability	Probability	
Specification of Population	Not Necessary	Essential as is a sampling frame	
Basis of sample choice/selection	Researcher's judgement	Random	
Basis of generalising from sample	If undertaken theoretically findings may be transferrable	Statistical representation	
Nature of aim usually addressed	Exploratory, answered using rich understandings	Explanatory, answered using statistical inferences	
Sample size	Relatively small (other than for quota sampling)	Relatively large	

Table 5-1 Differences between Non-Probability and Probability Sampling (Saunders, 2012, p. 39)

The sampling frame consisted of a relatively homogenous group (Morgan D., 2008), for which a sample size of twelve has been suggested as sufficient (Ando et al, 2014; Boddy, 2016; Fugard & Potts, 2016; Saunders, 2012), particularly where the research is seeking to understand, and develop themes, related to their shared experience (Guest et al, 2006), as was the case with my own research. I decided to aim for fourteen research participants in total, falling into one or more of the inter related categories of lecturers, heads of department, and heads of school involved with engineering programmes in the IoTs. The decision to aim slightly above the minimum sample size allowed for a situation to arise, where, post-interview, a research participant might withdraw from the process.

There is a significant gender disparity in engineering in Ireland (Fingleton et al, 2014), with females representing 20% of undergraduate students, but, only 9% of those who attain Chartered Engineering status. In my selection of research participants I endeavoured to ensure I had a female voice, representative of the sector, with three of the fifteen interviewees (20%) being female, which included one in an academic management position.

I did not set out to select academics by discipline, as, in terms of the conditions under which we serve as IoT engineering academics, and the pressures we experience as educators from the NFQ, and EI, we appear to me a relatively homogeneous group. Take up of the invitations was by academics involved in the delivery of either electronic or mechanical related engineering programmes, or with one of these as their original engineering specialism, and civil/structural engineering academics were not represented in my fieldwork. This reflected the academic profile of my own department, and it is apparent, that, in leveraging my knowledge of the sector to identify potential interviewees, I favoured these disciplines.

This is significant in that, although EI accreditation is particularly relevant for employment as a civil/structural engineer, it is less so for electronic or mechanical engineering graduates. To illustrate this consider EI (2019a), a list of those companies that EI accredit for continuing professional development for their employees, which is dominated by civil engineering oriented employers. In my own experience as an electronic engineering graduate the issue of EI accreditation has never been raised in any job interview I have participated in, either as an employer, or potential employee. Although, as will become clear from the fieldwork, the pressures of EI accreditation on engineering education are very important for my interviewees, it would be expected that the relevance to employment would be raised. This leads to a limitation for my research findings, discussed, below.

The physical environment, the people with which an interaction takes place, and the objects with which someone interacts in the gaining of an experience, are intertwined. Situating the fieldwork in a physical location related to a research study helps to stimulate memories of the conditions of the experience in order to help the interviews flow. On this basis I decided the fieldwork should be situated in each participant's IoT.

I used an informal approach, using my contacts in the field, to identify potential research participants. However, conscious of power issues associated with my position as an academic manager in the IoT sector, I followed a two stage process to invite and discuss participation with potential research subjects, as discussed below in section 5.9 on ethical considerations.

5.4 The Role of Document Analysis in my Research Process

In considering curriculum, Young (2008) stresses "its fundamentally social and historical basis" (p. 19). Documentation proved an important source of historical data, providing insight into the genesis of Irish engineering education, the state of play of this education pre-1960s, and the subsequent development of the RTC/IoT sector. I also used documentation in my

consideration of how the ideas that led to OBE had evolved over time, and in examining the factors leading to the global proliferation of national qualification frameworks (NQFs). Historical records further illuminated: the policy development that led to the NFQ; international influences; the consultative process that was followed; and the structure and values of the NFQ as it became a statutory instrument. In considering the pedagogical basis of the NFQ, available evidence indicated this was afforded little discussion, even though my research suggested it was important. In investigating the implementation of OBE in IoT engineering curriculum I made use of programme schedules mapped to the NFQ, e.g. ITB (2016). I further used documentation to assist in developing an understanding of the global move towards LOs based accreditation in engineering education, and El's adoption of same for Irish engineering education. This historical data also illuminated the social pressures that influenced the policy formation and implementation that the records describe.

I also applied two document analysis techniques in my research. I used Corbell's (2014) keyword analysis methodology to examine the co-occurrence of the terms "knowledge" and "skills" in NQAI (2003a) to support and develop an argument regarding the NFQ as a policy tool in promoting a skills agenda in higher education. I made use of readability analysis (Reck & Reck, 2007), as applied to curriculum LOs, to support an argument that the NFQ concept of transparency (of LOs) was not a valid concept in consideration of the accessibility of their meaning to prospective students, or students in the early years of study on a programme.

5.4.1 Keyword Analysis for *Knowledge* and *Skills*

Corbell's (2014) keyword analysis methodology is concerned with "issues and problems" (p. 110) relating to knowledge, which are "significant for the development and social distribution of knowledge through the vocational curriculum" (p. 110).

Corbell's premise is that the co-occurrence of the words 'knowledge' and 'skills' in a text representative of a particular policy discourse is indicative that the individual meanings of the words may have been superseded, for that policy discourse, as the words occur *together* as a *binomial*²⁴, with a new joint meaning. In order to establish what meaning this binomial might have, Corbell (2014) notes the change in dictionary (Oxford University Press, 1992, 2012)

²⁴ A "commonly occurring pair of words linked by a conjunction which works within a sentence as [...] a single word" (Corbell, 2014, p. 115)

definitions of knowledge from 1992 to 2012, where in the latter version one of the associated definitions includes 'skill', and a further definition frames knowledge squarely in the context of the knowledge economy; on the other hand a more abstract definition of knowledge as representing that which is known has been dropped when compared with the 1992 edition. Corbell draws the conclusion that this is indicative that the phrase 'knowledge and skills', is in fact, a proxy for 'skills'.

To investigate the significance, keyword analysis searching for the co-occurrence of the words of interest is applied to documentation associated with the policy discourse. Corbell proposes this can be facilitated through the identification of what he terms a nodal text (a key policy document) associated with the policy discourse, to which keyword analysis is then applied to quantify the degree to which the conflated term is present, and to allow consideration of the contexts within which it is used. Corbell demonstrates the effectiveness of the technique through the analysis of a nodal text associated with the policy discourse on education and the knowledge economy in Australia, finding an overwhelming use of the words 'knowledge' and 'skills' together. This juxtaposition of knowledge and skills in policy documentation implies an associated dilution of the more abstract meaning that can be associated with knowledge (in line with the changes in dictionary definitions). Corbell (2014) concludes that "the voice of knowledge is being silenced in through its conflation with skills in the lexical item knowledge and skills" (p. 118), which he particularly associates with the use of NQFs.

In my research I had a concern, that, regardless of the lifelong learning and student-centred policy aspirations associated with the introduction of the NFQ, and although NFQ LOs should encompass knowledge, skills and competence (NQAI, 2002b), the primary driver was to increase the skills focus of Irish education. Corbell's keyword analysis as applied to the use of 'knowledge' and 'skills' together offered an approach to investigate this. In line with Corbell's methodology I identified NFQ (2003), a compendium of the policies and criteria related to the establishment of the NFQ, as a suitable nodal text, and applied keyword analysis, as described in *section 3.7*, to support the argument that a skills agenda was the primary focus of the NFQ.

Although I have made use of keyword analysis in a similar policy context to Corbell, I note that it has the potential for wider applicability. In order to utilise it in the analysis of other policy discourses the following steps are suggested: the co-occurrence of keywords of significance in the policy discourse under consideration should be identified; their meaning, separately and as a binomial, established; the significance of the meaning of the binomial in the policy discourse determined; and appropriate nodal documentation identified to allow the significance of the usage to be considered.

5.4.2 Readability Analysis

One of the claims associated with LOs is that they promote transparency (one of the values of the NFQ) with regard to qualifications (Cort, 2010), for all stakeholders, including students and prospective students. However, as discussed earlier (*section 3.6.1.3*), there are compelling arguments in the literature as to why transparency of LOs is not achievable for non-domain experts (Cort, 2010, Young, 2011, Heywood, 2016), for whom the language of levels and learning provided by NQFs through LOs would be largely opaque. In the fieldwork I was to discuss the validity, importance and application of transparency with my research participants.

Readability analysis (Reck & Reck, 2007) offered a further route to explore this through providing an analytical tool to measure the readability of programme LOs from curriculum documents. This allows their readability, and hence transparency, to be quantified, and thus the accessibility of LOs to non-experts to be ascertained. Alongside investigating transparency in the fieldwork and literature, I was to make use of this technique to examine particular claims for transparency: that prospective students should be able to understand LOs sufficiently well to make an informed decision regarding entering a programme of study (Kennedy, 2007; Lindholm, 2009) and additionally that LOs should be easily comprehensible by a student in the earlier years of study on a programme (Kennedy, 2007; QQI, 2018).

In considering prospective students, Lindholm (2009) links LOs directly to the marketing of a University's programmes: "learning outcomes communicate to prospective students, their parents, and the public what is valuable about (an) academic program." (p. 9). Armstrong & Lumsden (1999), in investigating the impact of a university's marketing material on prospective students, suggest that "most of all, the materials need to speak the (prospective) students' language" (p. 90). Similarly, QQI (2018) propose that LOs should be comprehensible to students, although I counter argue that, for students in earlier years of study, their epistemological journey has not yet progressed sufficiently for them to be fully familiar with the terminology of their academic discipline as will be used in the specification of the LOs.

In order to demonstrate the complexities of programme LOs for qualifications mapped onto NQFs a readability analysis of the programme LOs of a number of such programmes is presented in *Appendix C*. Such programme level LOs are readily available for IoT programmes mapped to the NFQ, e.g. ITB (2016), MTU (2021), DKIT (2021), from which a critical point related to the value of readability analysis for my research emerges. On the one hand, in keeping with the NFQ value of transparency the IoTs make their LOs based programme curricula readily available through open-access web interfaces. On the other hand, this only has value for non-expert readers if the programme LOs are, indeed, easily comprehensible, or transparent, to them.

To carry out the readability analysis I used an on-line utility (Readability Calculator, 2017), which takes as input the text of the programme LOs I was considering. Through the use of a number of readability algorithms, Coleman-Liau, Flesch Kincaid Grade Level, Simple Measure of Goggledygook (SMOG), Gunning Fog Index, and the Automated Readability Index, the utility provides indices representing the readability of the input text in terms of the number of years of formal education needed to easily understand the text (Reck & Reck, 2007). A note of caution is sounded in the literature, in that "domain specific attributes such as complex terminology or language can direct readability scores towards higher values than the actual complexity of the text warrants" (Reck & Reck, 2007, section *Introduction*). However, I argue that in my research into the transparency of LOs, which may well contain such complex terminology, a useful measure of readability is precisely one in which the presence of such domain specific attributes is taken into account.

Detail of readability analysis of a selection of programme LOs is provided in *Appendix C*. For the sample I considered, readability analysis indicated that, rather than the meaning of the programme LOs being accessible to prospective students with a pre-university education, or indeed to students in the earlier years of study of a programme, a much higher level of education (at least to graduate level) would be required to easily understand them. I make use of this in the discussion in *section 11.3* in relation to claims by QQI (2018) for the transparency of LOs.

5.5 Selection of Data Gathering Methods

As I briefly began to discuss earlier, in selecting a research methodology a "social researcher

is faced with a variety of options and alternatives and has to make strategic decisions about which to choose" (Denscombe, 2003, p. 3). In developing my research methodology, I had to consider the research questions (Ryan, 2006), the type of data that would most suitably lead to answers to these questions, and approaches to gathering this data (LeCompte et al, 1993). In my research I needed to select methods that would allow academics' views to be gathered and analysed. Having considered possible approaches to generating the qualitative data required for the research, I chose a semi-structured interview approach, followed by the use of a focus group.

5.5.1 Interviews

Interviews, in which the researcher works with research participants to generate accounts related to the research issues, are the predominant data collection tool for qualitative research (Peräkylä & Ruusovouri, 2011), and have been usefully applied in the field of engineering education research (EER) (Kelly & Bowe, 2011; Lauritsen, 2012; Karwat et al 2014; Sheppard et al, 2006; Van et al, 2008). Denscombe (2003) points towards the attractiveness of interviews for data gathering, given that it uses the conversational skill of the researcher and the interviewee. However, he distinguishes a conversation from an interview, where the research subject has consented to take part and have their participation on the record, and the agenda is set by the researcher. Interviews can be used to gain in-depth information on a topic, often from a relatively small number of participants, and are suitable in situations where the researcher requires information on "emotions, experiences and feelings", "sensitive issues" (Denscombe, 2003, p. 165) or where the information would be "based on privileged information" held by key individuals. In my research key individuals, engineering lecturers and management, hold privileged information with regard to how they, and the engineering education system they work within, have been affected by OBE. This strongly suggested interviews as a data-gathering technique for my research.

In considering what type of interview technique to use, structured (Firmin, 2001b), semistructured (Denscombe, 2003, Adams, 2010), or un-structured (Firmin, 2001a). I reflected on my aims. Most importantly, I wanted to ensure that my research participants' experience was the "focus of the interview" (Adams, 2010, p.20). It was key that the interviews would go into sufficient depth to get to the detail of the various ways my research participants might have perceived engineering education has changed under the influence of OBE. Additionally, I anticipated that they would have views and experience with regard to OBE that I had not considered, and it was essential that I provide them with the opportunity to discuss these.

On this basis I chose to use semi-structured interviews, which consist of a defined list of questions to addresses a pre-determined list of topics (Denscombe, 2003), but the interviewer uses their discretion with regard to what is asked. Open questions encourage the interviewee to expound on topics raised, and allow the interview conversation to flow and develop. Semi-structured interviews should be used "in research questions where the concepts and relationships among them are relatively well understood" (Ayres, 2001, p. 811), which was to an extent true of the research proposed in this project. Ayres (2001) further states that "the development of rich, relevant data rests on the interviewer's ability to understand, interpret, and respond to the verbal and nonverbal information provided by the informant" (p. 811), where, given I was working in the field, I was well-positioned to do.

5.5.2 Focus Groups

In a focus group research participants are collectively interviewed and group conversations are enabled (Kamberlis & Dimitriadis, 2011). Focus groups also allow pedagogical interaction with the participants, and furthermore, can have a political dimension, where they can be used to raise group consciousness regarding a topic (Liamputtong, 2011) towards enacting change. Kamberlis and Dimitriadis (2011) propose that focus groups work best when all three facets are in play, where the synergy of the research, pedagogy and politics are exploited to produce data and effects that cannot be realised through individual interviews.

A focus group would provide the opportunity for a collective voice to emerge with regard to the impact of OBE. This could reveal more, or something different, than interviewing individuals, which could only be ascertained by utilising them in conjunction. Additionally, I considered that focus groups could be used in reviewing and expanding the initial findings of my research, after individual interviews had been undertaken and analysed (Morgan, 1998).

I also considered the political and consciousness raising facets of focus groups. As a former registrar in ITB and currently HoD in TU Dublin, I had to respect the power and political relationships between myself and other academics, and also other registrars and IoTs. A focus group concerned with the impact of OBE on engineering education and educators could be seen as a political statement within the sector, and as something I was trying to drive change

in. Similar issues existed for McNutt (2010), a Head of School of Informatics and Engineering, who used focus groups in his research into the habitus of educational technologists. McNutt regarded this political dimension positively for his participants, which could "contribute to local initiatives designed to consolidate the status of the individuals as practitioners within the field." (p. 86). Similarly, Fitzsimons (2015) reported that the focus groups used in her research had a strong political and pedagogising element for some participants, and provided them with a "feeling of empowerment and energy" (p. 356).

Having considered the political dimension to focus groups, and having reviewed the positive view discussed above, my concerns were mitigated. I realised that, in the same way that a focus group has a political facet to it, so do my studies, readings and research towards a doctorate (Ryan & Walshe, 2014). Indeed, as my fieldwork progressed, and I began to draw conclusions with regard to my research, I found that I actively sought out conversations with my colleagues related to these issues, on the basis that these were important topics for us to consider, and I welcomed the opportunity to run a focus group when it arose.

As discussed below in the section on ethics, the use of focus groups also has a power relation dimension to it, where "focus groups can (and often do) mitigate or inhibit the authority of the observer, allowing participants to 'take over' or 'own' the interview space" (Kamberlis & Dimitriadis, 2014, p. 324). This further supported their use, given my concerns about power relationships with the research participants.

I decided to use a focus group as a means for reviewing and expanding the initial findings of the research, inviting those I had already interviewed to take part.

5.6 Conducting the Interviews

I commenced each interview by seeking to establish with the research participants how they came to be an engineering academic in an IoT, what they currently do as an academic, and seeking to elicit from them what they thought had changed consequent to the introduction of OBE. Prompts I used to ensure important aspects were covered included:

- what did they see as their role, e.g. course design, assessment, delivery, and research?
 ("how did they become a lecturer" was an opening line in each interview);
- how did they relate to the use of theory and practice together? This concerned both the pedagogical use of theory and practice together to aid learning, and also with the use of theory and practice together in the teaching of engineering practice;

- what impact has OBE had, in terms of how they approach course design, delivery assessment, pedagogy, and their own approach to knowledge building;
- how do they see what they do now to be different to before the NFQ, and how do they relate to any perceived change;
- how they keep to the forefront of developments in their technical fields?
- understanding how technologies, techniques and theories can be combined together and applied to achieve practical effects, often in the development of engineering systems to achieve particular desired effects;
- their research interests;
- what influence have EI had?

In preparing for interviews, Plummer (2001) advises being concerned with the practicalities of the interview location, arriving on time, ensuring the functionality of the recording device, and planning in advance how to conduct the interview. He stresses being an empathic and good listener: considering aspects of the interview that you as an interviewee would struggle to answer; and reflecting on why these may provide that tension and how to move through it. Plummer suggests showing alertness and interest, and considering how to deal with resistance to addressing particular topics. Denscombe (2003) on the other hand believes "the good interviewer is able to tolerate silences during the talk" (p. 177). This was all useful advice which I considered in advance of the interviews taking place. In particular I felt the most important thing to do to avoid silences and tensions was to build a rapport with each research participant at the beginning of the interview, as discussed below, and, in general, to use our shared experiences to move through times in the interview where the conversation faltered. Although the interviews flowed quite freely for the most part, I did find that, on occasion, with no discernible pattern, a topic I raised did not elicit a strong response. In such cases I tried, often successfully, to bring the conversation to life through speaking with them on our shared experience of the topic, and where that was not successful I moved on to the next question. I was, however, conscious that I was there to get their perspective, rather than share my own, and I was careful to try and maintain my role as interviewer.

I conducted fifteen interviews in total between June and November 2016, of which four were academic management, one former academic management, and the other ten lecturing staff. I had an uptake of approximately 50% on interview invitations, which I issued in groups of 4-7 in order to pace the interview schedule. I noticed that a positive response to an interview invitation was more likely from those with which I was personally acquainted, although a number (six) were not known to me previously. Furthermore, in considering those who responded positively, I note the majority (fourteen) had been involved in engineering research of one form or another. The interviewees came from six of the IoTs, with participants from individual IoTs numbering five, four, two, two, one and one respectively. Of the fifteen research participants, nine had experience as IoT engineering academics prior to the transition to OBE, five had joined whilst that transition was ongoing (i.e. before a complete cycle of programme renewal through programmatic review and/or accreditation), with only one having joined the IoT sector when the use of OBE was more fully established. All interviewees signed the consent form as per the agreed ethical protocol.

Equipment failure resulted in the audio recording for one interview (Grace) being lost. Taking heed of Denscombe's (2003) advice I checked that the equipment was working prior to the interview commencing. However, the malfunction occurred and the data was lost. After that I used two recorders. The interview still yielded interesting data, in the form of a memo of the interview agreed as part of the process of informing the interviewee that the data had been lost, the salient point of this memo having been captured in my fieldwork journal. The net result was that I had fourteen interviews fully transcribed, and one additional interview which, although not transcribed, still contributed valuable data.

The interviews were transcribed using a professional service. Upon receipt of the transcripts they were anonymised, but otherwise unchanged, and forwarded to the interviewees for comment, as agreed in advance. This gave each participant the opportunity to review the interview transcript, and, ultimately, they could ask for this not to be used in the thesis. The transcripts describe aspects of a research participant's life, and their views related to their profession, and there was a possibility that an interviewee would later disagree with the use of their transcript. However, I received no requests to amend or withdraw the interviews, although one asked that I be cognisant of where they might have spoken on sensitive matters, and be judicious with regard to how I used their words, but left it to my discretion.

Research subjects were anonymised using the first names of famous engineers, Bill (Gates), Gordon (Moore), Ayah (Bdier), Vitruvius (Marcus Vitruvius Pollio), Rudolf (Diesel), Isambard (Brunel), John (Logie Baird), Kees (Bulthuis), Grace (Hopper), Henry (Ford), Thomas (Edison), Edwin (Armstrong), Edith (Clarke), George (Stephenson), Alan (Oppenheim). In suggesting that the issue of pseudonym selection should be of primary concern, Clarke (2006) considers that the choice of pseudonym can have connotation of "class" (p. 6). In choosing names for the interviewees that equated them with a class of famous engineers I showed that I was regarding the opinion of them all equally. Additionally, the choice of pseudonyms reflected that I wanted to "select names that would at least in some way resonate with them" (Saunders et al, 2015, p. 621), and that choosing *important* names for them would draw attention to the significance with which I regarded their contribution. I received feedback from two of the interviewees on this, who expressed sentiments of being proud to be associated with the engineers in question. I preserved gender in the anonymisation, on the basis that it had been observed that from the perspective of research participants a minimal desirable attribute for an anonymising reference appeared to be gender (Clarke, 2006).

I had the interviews transcribed verbatim, (Cavendish, 2011), which preserved the emotion and meaning implicit in the language of the interviewees. This did not affect the readability of the transcript texts themselves, rather the meaning that the interviewees were trying to impart was often easier to discern, for example as they developed a theme by repetition of points (Corden & Sainsbury, 2006). Indeed, in reference to the repetition of words by speakers, in itself a type of filler word, Clark and Wasow (1998) suggest that it solves the dual problems of "how to speak in a timely fashion and yet how to speak smoothly. Repeating a word deserves our respect as an efficient and effective way of dealing with these problems." (p. 239). This implies that the use by the interview subjects of repeated words as fillers helped them to develop the cogency of their responses to my questions.

The interviews largely followed the points laid out above, many of which were raised organically through the interviews, as the topics were quite inter related and did not always need to be specifically addressed. However, as anticipated some prompting was required to steer the interviews to ensure we covered all topics I had identified, and in each interview there were some points I had to specifically raise towards the end. In keeping with the framing of the interviews I commenced by discussing with each interviewee their motivation for becoming a lecturer. This made it clear to them that: the interview was about *their* personal views; it established a rapport with them, in that we often had shared/similar experiences; and it seamlessly brought us into discussion directly relevant to the research. Andrews et al (2019), remark on the difficult task of getting their colleagues in engineering education to engage in reflective discussion regarding their pedagogy, and the importance of framing the

discussion to encourage this. I believe my fieldwork achieved a similarly rich approach to data gathering to that which they reported.

As the interviews progressed, and I carried out the coding of the initial interviews, some topics emerged which I had not anticipated, but which seemed, on reflection, germane to the research. Consequently, later interviews covered some additional topics, in particular the use by lecturers of the 'language of levels' that has come about due to the NFQ.

5.7 Coding and Analysis

The aim of coding is to identify concepts that will help explain complex social phenomena (Denscombe, 2003, p. 20) under consideration in the research. Lichtman (2014) describes the application of generic, or open (Saldaña, 2009), coding as applied to data gleaned from interviews with research subjects. In this approach, an interview is read and analysed in order to identify categories and codes that represent data within those categories. The codes themselves are either words or phrases, arrived at through a careful reading of the first (and usually several more) interviews.





Denscombe (2003), in discussing the initial choice of codes and categories, explains that, to begin with, researchers may use theoretical concepts, or perhaps professional judgment/hunches to arrive at these. Indeed, "as one proceeds through the initial coding of the data, there is usually much potential for pursuing a variety of themes and issues" (Benaquisto, 2008, p. 87). Saldaña (2009) suggests that a provisional list of codes can be chosen in advance "to harmonize with your studies conceptual framework or paradigm and

to enable an analysis that directly answers your research questions" (p. 49). I started with a small provisional list of codes to assist in commencing the process.

The researcher's knowledge of the field is used to decide upon the saliency of the codes, although analysis tools such as the frequency of occurrence of codes in the interview may also be used as a guide. Some codes may overlap, and be seen to represent similar aspects, some may be found to be redundant. Initial codes are examined "removing redundancies, renaming synonyms, or clarifying terms" (Lichtman, 2014, p. 253). Codes are grouped together into categories, which can then be used to identify concepts to assist in developing an understanding of the phenomenon under investigation.

This approach to coding offered a suitable approach for my research, supporting the analysis of interviews in order to identify concepts, which could then be further analysed to develop themes related to the research question. I noted it had been utilised by McNutt (2010), which paralleled my own research in that it was also a research study conducted with IoT academics.

I envisaged that the categories and themes that emerged could lead to consideration of policy documents relevant to the inquiry. This stage is an opportunity to link the views of the participants to the policy, pressures and results that constitute the overall field within which the game is being played (Bourdieu, 1990). For example, as will be seen later, the issue of the balance of skills and knowledge being taught became quite important in the research, which led to national policy in this regard being interrogated and discussed.

I had planned that the process of coding would be carried out in phases, initially coding using two transcripts. After reviewing the initial coding I expanded the coding scheme to include many more aspects of consideration, forming initial categories. I returned to the field with the initial categories, allowing the later interviews to be used more effectively. I kept a log in which to keep track of insights gained while analysing the data, and which serves to show my developing line of thinking (Denscombe, 2003). My approach seemed to broadly follow the following description of coding:

rarely is the first cycle of coding data perfectly attempted. The second cycle (and possibly the third and fourth, and so on) of recoding further manages, filters, highlights, and focuses the salient features of the qualitative data record for generating categories, themes, and concepts (Saldaña, 2009, p. 8).

The various versions of coding I had included the following:

- phase 1 coding of two interviews;
- phase 2 re-coding of these two interviews;
- phase 3 coding of new interviews as transcripts were generated, and revision of earlier coding of interviews to support new codes that emerged;
- phase 4 coding where codes were re-arranged into themes as part of initial draft of the analysis chapters, and some further re-coding took place;
- phase 5 coding where interviews were re-coded to fully support all aspects reported in the analysis chapters, with ultimately 424 codes in total.

The categories that resulted, once all interviews had been undertaken, transcribed and coded, were then analysed in the context of the conceptual framework and the literature review, to arrive at themes, and ultimately conclusions, as to the effect that OBE is having.

5.7.1 On the Inside Looking In and What I Didn't See

As I carried out the interviews for the fieldwork, I was conscious that aspects of the initial interviews "could serve as starting points for analysis *as well as for further data collection*²⁵." (Charmaz, 2006, p. 3), and I actively looked for new sub-themes that might emerge, which could add to the topics I was discussing with the interviewees. I was conscious of my insider role (Leckie, 2001), but, I had neglected to fully think through the implications of this, in particular, that "individuals are often so enmeshed in their own experience that the adequate distance required to know their experience is not available" (Dwyer & Buckle, 2009, p. 59).

As coding progressed themes focussing on the major points of discussion emerged, with the analysis of each interview influencing how I approached the next. However, it was only in analysing the seventh interview that I noticed the preponderance of the use of the term "level", initially in that seventh interview, but, in retrospect, across all of the interviews, to describe aspects of the pedagogic discourse. On reflection I realised that since the introduction of the NFQ, I, my colleagues and peers in the IoTs, have, gradually, adopted what I will refer to as a 'language of levels' as part of our vocabulary as engineering academics.

In considering, from a methodological perspective, why I did not notice the significance of this earlier, I concluded that it is because as an insider to the research I am quite comfortable this usage. I believe that this realisation was a milestone in my development of a researcher identity, becoming an observer in the world as per Denzin and Lincoln's (2011) definition of

²⁵ My emphasis.

qualitative research. As part of my professional identity as an engineering educator I was quite comfortable with the language of levels, however, as a researcher, I wanted to understand more about why we used it and the implications this may have. My approach to the research needed to be dual-faceted, on the one hand, my professional identity allowed me to comprehend it "from the insider's perspective" (Fetterman, 2001a, p. 347), for which the interview approach is well suited (Fetterman, 2001b). On the other hand, from a researcher/observer perspective I needed to be able to "objectify patterns of behaviour" (Fetterman, 2001a, p. 347) to gain the insight I sought.

I also further highlight the importance of the language of levels from a methodological perspective. As will become apparent in the analysis, the language of levels pervades the pedagogic discourse of the fieldwork. Consequently, in order to usefully analyse the fieldwork, it was necessary to gain a comprehensive understanding of the phenomenon.

5.7.2 Presenting the Analysis

The analysis of the fieldwork is presented through themes identified in the coding, illustrated by the voice of my research participants as they related to those themes in the interviews. I place emphasis on the voices of some individuals, including a number of longer narratives, on occasions where they serve as particular exemplars of a theme (Gordon on the language of levels, and John regarding pedagogy), or offer particular salient insights (Ayah's managerial perspective). With one interviewee, Bill, I believe together we developed significant new knowledge related to progression during his interview, and I use themes that emerged from this to frame the analysis in this regard.

The focus group was at times supporting of the themes that emerged from the interviews. However, on other occasions their discussion generated a new perspective, and, where this was apparent, I highlight the focus group contribution.

As discussed earlier, Grace's interview recording was lost through equipment failure. Consequently, although her interview was not available in the analysis, I was able to make use of our agreed memo of the meeting, and I present Grace's contribution in the analysis where it is possible. For completeness, where I provide tabular overviews of the interviewees' views I include Grace, even though it was not possible to represent her views fully.

5.8 The Focus Group

Four of the interviewees accepted the invitation to the focus group, Thomas, Isambard, Vitruvius and Gordon, a sufficient number to proceed (Barbour, 2007). As the participants were all IoT engineering lecturers, this also met the criterion that the focus groups "should be homogenous in background" (Morgan (1998) as referenced in Barbour (2007, p. 59)). However, as I could attest from the analysis of the interviews with the same participants, not in "attitudes". Barbour contends that differences in attitude can lead to argument within a focus group, which, once it is kept within limits, can assist group members and the facilitator in teasing out the perspectives behind differing opinions, and facilitates understanding.

In preparing for the focus group I considered Fitzsimons (2015) description of how using focus groups contributed to an understanding of the concerns, views and attitudes of her participants in relation to the research. I also revisited the use of focus groups by McNutt (2010), where, given his research participants came from a similar pool to mine, his ethical concerns with regard to anonymity and informed consent partially mirrored my own.

In planning how to stimulate the focus group conversation I considered Barbour's (2007) suggestions of how to moderate focus groups, and the choice of stimulus material, and the use of reflective exercises by Fitzsimons (2015). However, as I struggled with how to structure the session, and what exercises to use to drive the focus group, I took heed of descriptions of stimulus materials that did not have the desired effect (Barbour, 2007; Fitzsimons, 2015). I concluded that the key to the selection of stimulus material was to be aware of who my audience was. Given I was an insider to the group I was interviewing I was confident that once the material was structured such that it would elicit my interest as an engineering academic, that it would similarly do so for the other participants. I consequently chose to use a set of 1-2 page hand-outs, structured around the themes of:

- the language of levels;
- the NFQ and EI;
- progression;
- pedagogy;
- knowledge.

These contained tables, upon which those presented later in the analysis chapters were later based, which related the engagement/agreement of my interviewees with the themes I had

identified, along with associated questions that I wished to consider. A sample hand-out is provided in *Appendix* B. I chose this approach on the basis that, as engineers, my research participants would be quite comfortable with the use of tables, in particular to analyse situations (hence the supplementary questions). I was confident this approach would facilitate the participants in engaging with the topics and facilitate me in addressing the questions I had for them, and stimulating the discussion I desired (Morgan, 2008). The approach proved quite fruitful.

Following Barbour's (2007) advice on planning a focus group, I considered using a note-taker or assistant moderator. However, I decided that it would not be necessary, and that I would lead and moderate the discussion myself, and take notes using a flip-chart. I also recorded the focus group discussion, and later had it transcribed for use in the analysis. The focus-group was held in the ITB boardroom, and refreshments were supplied to the participants.

Prior to the focus group commencing I re-iterated the confidential nature of the process (which had been included in my letter of invitation to the participants), and all the participants signed the informed consent form agreed in the ethical approval protocol.

The focus group resulted in an animated and interesting discussion, the relevance of which to the research is shown in the analysis chapters. As agreed earlier with the participants, the session lasted approximately 1 hour and 30 minutes. The participants had received copies of the hand-outs prior to the focus-group, to give them the opportunity to read them in advance, but, I had made it clear that this was not necessary. As discussed previously, focus groups have research, pedagogical and political dimensions to them, and the research dimension was quite ascendant in the interviews. One participant did remark that they themselves were much more aware of the issues we were discussing through being involved in the research, which confirmed to me that the pedagogical element was present to an extent.

5.9 Ethical Considerations

I adopted the three principles proposed by Denscombe (2003) in working with research subjects, similar to the guiding ethical principles used by O' Malley (2012):

- 1. protecting the interests of participants;
- 2. avoiding deception;
- 3. ensuring informed consent.
Clandinin (2006) highlights the added importance on anonymity and confidentiality in a research methodology where people are sharing complex issues related to their experience. Research participants were recruited from across the IoTs, with their identities anonymised using pseudonyms to preserve confidentiality. This makes it difficult to infer the identities of research participants through consideration of published material associated with my research, primarily due to the fact that there are more than a thousand IoT engineering academics. Nevertheless, although anonymity can be attempted, it cannot be guaranteed. A note to this effect was included in the consent form, and this was discussed with participants.

Lynch et al (2012) in discussing the identity formation of senior management in the education sector, writes that "many identified themselves as change agents and were positive about bringing a more strategic vision to their organisations" (p. 112). Regardless of how I positioned myself in an interview my role as an academic manager nevertheless influenced me, and no doubt influenced those who I interviewed. There was a risk that the power relationship might unduly influence the interview, where "overlooking the complex power dynamics of the social construction process may, however, seriously impair the validity of the knowledge constructed", (Kvale, 2006, p. 485). I realised that I may be perceived as a change agent with a strategic view, as opposed to a colleague interested in the participants' views.

I was encouraged by Kvale's (2006) view that interviews can produce useful knowledge despite the presence of a power relationship. Bringing academics together in a focus group, where their group voice could emerge, and their strength in numbers would act to mitigate any adverse consequences of the power relationship, provided a suitable check (Kamberlis & Dimitriadis, 2014). Furthermore, the concept of academic freedom for IoT academic staff is established in legislation (Irish Statute Book, 2006), and well understood within our sector. Thus academic staff, whilst still in a relationship with me where there is a power imbalance, had significant protection to allow them to speak their mind.

The use of focus groups returned me to the issue of confidentiality. As discussed in Hofmyer and Scott (2007), confidentiality of the individuals who constitute a focus group is compromised, as the researcher cannot control the actions of those who form the group. In adherence to the principle of informed consent (Denscombe, 2003), participants were made aware of this potential issue. However, the participants in the research were all academics used to dealing with confidential matters, who I believed could be trusted to maintain confidentiality, and who would trust other academics to do the same.

Another aspect of protecting the interests of the interviewees that I needed to consider was that the "dominant position of the interviewer may lead to an invasion of the subject's privacy" (Kvale, 2006, p. 497). It would be possible that an interviewee might disclose personal information to me that they, or I, might later regret. Although I hoped it to be unlikely, I took the precaution of including the Samaritan contact number in the consent form, in case either I or the interviewees later felt the need to talk through issues that might arise.

I was committed to not coercing anyone to take part in the research, and I believe the contact approach which I document below, ensured this.

I was quite cognisant of confidentiality issues as related to data protection, and I describe comprehensive data protection mechanisms below.

In conducting my research I analysed documentation for programmes from ITB and other IoTs, and also from HETAC, QQI, EI and others, the majority of which is publicly accessible, although in a few cases I held personal copies of otherwise unavailable documents. However due to my position as HoD in ITB (now TU Dublin) I was careful to consider the effect of any critical analysis I undertook, in order to differentiate my views as a researcher from views I hold through my role as HoD in TU Dublin.

Ethical approval was sought, and received, from the Maynooth University ethical approval committee, and as an ITB employee at the time, also from the ITB ethics committee.

5.9.1 Contacting Potential Interviewees

When I designed the ethical protocol I was Registrar at ITB, and not in a direct line of authority over any of the potential participants in the programme. However, in my approach to seeking research participants I acted on the basis that I could be perceived to be in a strong power relationship with relation to all the participants, and that anyone I approached should be empowered to decline the invitation.

In order to address this I followed a two stage selection process to invite and discuss participation with potential research subjects. Initial contact was through email, identifying myself, explaining the research project within which I was engaged, and asking the potential research participant that, if they were interested in discussing their possible participation, to reply to my email. At this point there were three possibilities:

- they would reply in the negative, in which case I would respond by email thanking them for their consideration, and make no further attempt to contact them on this matter; however, I received no responses in this category;
- 2) they would not reply, in which case I regarded this as a negative response, and I did not have any further correspondence on the matter with them;
- 3) I would receive a positive response. At this point I contacted the potential interviewee further by email to discuss the project in more detail, and to arrange an interview date. My experience of this was that once the initial positive contact was made an interview date would be agreed and an interview followed at some stage.

I had decided that if I received more than 18 positive responses it would not be possible to interview everyone (given time constraints), and I would have to make an informed choice with regard to selection. If I had received an oversubscription I would have biased selection towards those who have spent most time working as engineers, respecting their greater continuity of experience, consistent with Sargeant (2013). However, the matter did not arise, a total of fifteen positive responses were received, and all resulted in an interview.

The recruitment approach was effective in recruiting the desired number of research participants. In addition, I believe the ethical protocol achieved the desired effect of minimising the power relationship. One interview request, to a colleague I have known for some years, but always in a position of authority in relation to them, elicited no response. Although initially I speculated as to the reasons for this, I realised that, for whatever reason, they were not in a position to participate and I should accept that. However, at the same time I recognised that the contact protocol had empowered them to ignore my request, precisely the effect it was designed to achieve, turning this into a positive experience.

5.9.2 Data Storage

Storage and retention of data gathered during the fieldwork was carried out in with a view to protecting the identity of the participants, ensuring it is only used for the purposes of this thesis, and that it is deleted at an appropriate time (Maynooth University, 2015). Recorded interviews are stored in a secure IT system. Research participants were informed of storage arrangements, and of a delete date for the data. Participant data is stored using the

anonymising identifiers agreed with them. A key to translate between anonymising identifiers and actual names is held in a separate, secure location.

Data will be retained for ten years after the thesis has been completed, in accordance with Maynooth policy. The electronic folder within which the data is retained will have a delete-by date indicated on it as part of the title, for which I will set a reminder in my electronic calendar.

5.10 Validity, Reliability and Limitations of My Research

In considering the reliability of my findings, the sample size, as discussed above, is sufficient (Ando et al, 2014; Boddy, 2016; Fugard & Potts, 2016; Guest et al, 2006; Saunders, 2012) to allow for the identification of themes that are important to my research participants, a selection of IoT engineering academics biased towards those from the Electronics/Mechanical disciplines, as a group, as well as individuals.

In following a qualitative approach my research allows for a deep exploration of my research question, generating understanding of how my research participants have experienced the social phenomenon (Denzin & Lincoln, 2011) of OBE in IoT engineering educations. Although qualitative research does allow scope for theoretical generalisations (Saunders, 2012), in adopting a realist stance (Sayer, 1992) in my conceptual framework it follows that the validity of my findings must be considered within the context of the socio-cultural and historic structures (Maton & Moore, 2009) of IoT engineering education within which I have observed and developed them.

Although my findings have validity for the IoT engineering education system, I do not consider that they generalise across all of Irish engineering education. In particular, in *Chapter 8* I will show the influence of progression on my research participants as they teach across NFQ level 6, 7 and 8 undergraduate engineering programmes. This pressure with regard to progression would not be experienced by university academics for whom the undergraduate offering is level 8 only. My research will also show that teaching across these different levels has contributed to the phenomena of the language of levels (*Chapter 7*) evident in the discourse of my research participants, and, similarly, I would not expect it to be evident in a similar fashion in that of 'traditional' university academics.

A limitation of my research, given it is situated in IoT engineering education, arises from the exclusion of IoT civil engineering academics from the sampling frame. Although, as I will show

in *Chapter 6*, perceptions of EI's influence on IoT engineering education were pronounced in the fieldwork, a somewhat negative theme arose as to its relevance for graduate employment (*section 6.9.1*). As discussed earlier (*section 5.3*) EI accreditation is much more important for employment for graduates of civil engineering than for mechanical or electronic engineering. Given that none of my research participants were civil engineering academics, I do not consider my findings with regard to the relevance of accreditation for graduate employment to be applicable for this discipline.

My findings, as will become apparent in the analysis (*Chapters 6-10*) and the later discussion (*Chapter 11*), are quite related to the specific case or Irish technical engineering education, and the policy and other pressures we experience as educators. Other jurisdictions will experience similar pressures in their engineering education systems, and although my research will speak towards OBE related issues in other such engineering education contexts, care must be taken in any such comparisons.

5.11 Conclusion

In this chapter I expanded the research question, based upon avenues of interest identified in the literature review, and as considered through the conceptual framework. I explained how this led to my selection of a qualitative research methodology for the fieldwork, and to my selection of the research participants, IoT engineering academics. The details of the research methodology were presented: i.e. semi-structured interviews with engineering academics, followed by coding and analysis, with the use of a focus group to review the draft findings. Two document analysis techniques, which were used earlier in the literature review, and later in the discussion, were also described, along with my rationale for using them.

I then discussed the implementation of the research methodology, highlighting issues regarding conducting the interviews including: the iterative nature of the coding and analysis; the manner in which I present the analysis; being an insider to the research; ethical considerations, and the holding of the focus group.

The scene is now set for the analysis of the fieldwork that my research question, literature review, conceptual framework, and research methodology has led to.

Chapter 6 Influences and Prescriptions on Engineering Academics

Let us take as an illustration recent changes in medical and engineering curricula, which bring out the ways in which the characteristics and content of curricula are influenced by the changing values and interests of the controlling groups involved. Michael Young (1971, P. 34).

6.1 Introduction

Issues of power and control were quite central to my consideration of how my research participants, and the education they offer, have been impacted by the introduction of outcomes based education (OBE) in our IoT engineering education system. My literature review highlighted the importance of considering not just the role of the National Framework of Qualifications (NFQ), but also that of Engineers Ireland (EI), alongside other drivers, including the global nature of the developments that led to both the NFQ and EI's OBE accreditation requirements. Drawing from social realism in my research provided a perspective from which I could consider the related power relationships and their significance. Through concepts such as stratification of knowledge (Young, 2008), pedagogic identity, and recontextualisation (Bernstein, 2000), and in stressing the importance of understanding the socio-historical context (Young, 2008), this influenced the design and implementation of my research methodology to examine how these power relationships might manifest, and what effect they might have.

The importance of exploring issues of pedagogy and power relationships in engineering education (Edström, 2018) with my research participants, in order to develop a fuller understanding of their experience of OBE, was borne out through the fieldwork, where issues of power and control came up repeatedly in our discussion. For example, Ayah, in reflecting on the use of the National Framework of Qualifications (NFQ), remarks *"this is the framework you have to stick to"*, George considered that *"the state gives Engineers Ireland certain powers"* as he discussed the importance of accreditation, and Henry, in considering *"the learning outcomes or NQAI way of doing things"* felt that *"you're trying to direct people"* with regard to pedagogy. This influenced the development of my conceptual framework to further emphasise the importance of consideration of the power relationships at play in the field I was researching, and led to my adoption of the lens of Bernstein's (2000) pedagogic device as a model for the pedagogic discourse in Institute of Technology (IoT) engineering education.

As discussed earlier (*section 5.8*) I structured the focus group session around a number of themes that emerged in the initial phase of the analysis, namely: the NFQ and EI; language of levels; progression; pedagogy; knowledge. This served as a useful basis to facilitate the group voice that emerged in the focus group, and later allowed me to advance the analysis and relate it to the literature. Consequently, I used these themes, having expanded their scope somewhat to allow inclusion of all relevant findings, for presentation of the analysis in this thesis.

This first chapter of the analysis considers how the 'official knowledge' of the NFQ and Engineers Ireland (EI) supports and controls the dominant discourse of OBE. To begin with, the analysis examines *where* my research participants experience the influence and prescription (Slough-Kuss, 2015; Ingram, 2016) they associate with the NFQ and EI's accreditation requirements, and considers their impressions of the effectiveness of OBE. Later chapters consider their perceptions of *how* that influence and prescription has had an impact on their pedagogic identity and practice, and on the structure of engineering education.

In *Chapter 7* the analysis examines the phenomenon of the language of levels, the usage of which by my interviewees will be evident throughout this current chapter, and I will consider how and why they employ this in their pedagogic discourse.

In *Chapter 8* I develop the theme of progression that emerged from the analysis, which, in this current chapter (6) I show is regarded by my research participants as important in facilitating access and reducing elitism in engineering education. In *Chapter 8* I further analyse the fieldwork to consider the impact of the NFQ concept of progression on how and what we teach our students.

In *Chapter 9* I consider the pedagogy my research participants have adopted in our LOs based engineering education system. Whereas in this current chapter (6) I reveal that my research participants perceive a significant influence over their pedagogy from the use of OBE, in *Chapter 9* I consider how their approach to pedagogy has adapted in consequence.

The impact on engineering academics' approach to knowledge and skills, and their recontextualisation in the curriculum, which the analysis in this current chapter will show is firmly oriented towards a LOs approach, will be considered in *Chapter 10*.

6.2 Implicit and Explicit Control of the Pedagogic Discourse: Hearts and Minds?

Tuck (2007) sets out criteria for the successful implementation of a national qualification framework (NQF), including: ensuring an effective legislative framework is in place; that government policy is coherent; that account is taken of the realities of implementation; and that adequate funding is in place to support it. Additionally, Tuck (2007) proposes that:

For an NQF to be really effective, hearts and minds – genuine and active support – have to be won. Otherwise, there is a danger that education and training institutions and social partners will simply pay lip-service to the framework. It is increasingly accepted that successful implementation depends on communities of trust (p. 25).

In analysing the fieldwork, I conclude that my interviewees offered a positive view of the NFQ, often somewhat qualified, but largely 'genuine and active support' (Tuck , 2007).

To begin with I present findings which illustrate the manner in which my research participants view the influence and prescription of the NFQ, and what they perceive it exerting control over in the pedagogic discourse. I later discuss how this is reinforced by EI, who form part of the community of trust for engineering academics, through their accreditation requirements.

	Research Participants														
Aspects of NFQ's Influence	Isambard	Ayah	<i>кее</i> з	flobuR	Vitruvius	Bill	spmoqT	Gordon	Edwin	george	Edith	Клиен	uųor	Alan	Grace
NFQ seen as effective															
Societal impact and progression															
NFQ associated with semesterisation															
NFQ used as a Management tool															
Awareness of Legislative and Policy Context															
NFQ influences Pedagogy															
LOs influence course design															

Table 6-1 Influence of the NFQ on the Interviewees

The following reasons were offered by my interviewees for their positive impression of the NFQ (see *Table 6-1*): it is seen as effective; the use of LOs encourages more consideration of pedagogy; learning outcome's (LOs) influence course design and delivery; the NFQ offers management a tool for working with staff, including in advising lecturers regarding pedagogy, and in terms of allocation of resources; and it is seen by some to foster a societal role for the institutes, promoting equity of access to education and improved progression for students.

Although largely positive, for some their views were qualified by concerns that management had not fully bought in to the NFQ, and that the framework could be too prescriptive. Another concern related to the introduction of semesterisation, which although not linked directly to the NFQ, was seen as by many of the interviewees as consequential from it.

A sample of my research participants' impressions related to the NFQ and IoT engineering education are given below, sometimes qualified with criticism, and in one case a somewhat negative stance is presented.

6.3 Academics' Perception of the Effectiveness of the NFQ

My interviewees generally offer a view of the framework as a well-functioning, integral, aspect of engineering education, with interviewees using phrases such as "very structured" (Vitruvius), "it works very well" (Gordon), "a fantastic tool" (Rudolf), "everybody's using it" (Edith). Indeed, there was considerable agreement that the NFQ provides a structure within which they can work effectively (eleven of the interviewees).

This view was exemplified in John's interview, which showed his embracing of the NFQ across a range of matters, including: *"it is actually well structured, right and it makes some useful points"*; indicating *"how we should design or specify a curriculum"*; *"the know-how and skill and the competence strands* [...] *are really important to employers"*; and *"it comes down to pedagogy, you know and it comes down to human nature in one sense that we have people who lecture who haven't gone through formal teaching and learning programmes, we were never taught to teach"*.

A dissenting voice, Thomas, expresses a negative opinion of the NFQ, where, rather than seeing a deep impact, he perceived a more superficial interaction, where they don't "*impact* on our day to day business in here" except "when we're doing programmatic review and [...] we're making sure our LOs fit into the descriptors [...] and that's about it. Maybe they should play more of a role in quality assurance".

However, Thomas's interview demonstrated he was quite immersed in the use of the language of levels (see *Chapter 7*) e.g. stating, in discussing the types and focus of programmes that an IoT should deliver: "*at level 6 and level 7 where we should be, we're applying [...] technology*". Similarly, Thomas was quite comfortable with the use of LOs in programmatic review. He also had considered that the NFQ came about "*to link into a*"

European framework to allow transfer of degrees and for qualifications". Although not positive about the NFQ, it clearly had caused him to reflect on his role as an engineering academic, and is quite comfortable discussing engineering education with the terminology and structures provided by, and as a consequence of, the framework.

6.4 Awareness of Impact of the Legislative and Policy Context of the NFQ

Twelve of the interviewees had given consideration to the impact of the legislative and policy context of the NFQ, and the prescriptive effect this has had. For example, Rudolf states that the "framework probably puts a control around the environment in terms (of) which module and programme development happens". John is accepting of the regulatory nature of the NFQ as part of the society within which he lives and works, influencing how academics approach "learning and assessment". Edwin regards the NFQ as being something you have to use ("there's no choice"), and indeed that "It's embraced by everybody", which he regards as a "good thing". In terms of understanding why he thinks it must be used, he considers "It's like the smoking ban, isn't it", i.e. a legislative requirement.

Edith has adopted quite a pragmatic approach to the NFQ, neither embracing nor rejecting it, but clearly feels that "you do have to use it", regardless of its pros and cons, similarly recognising that it is part of the legislative framework within which the IoTs operate. Bill viewed the NFQ as a "new system" which "changed the nature of things", and Vitruvius feels that the NFQ is a "very formal framework" that was "imposed".

Kees brings a management perspective, describing how "a learning outcome framework" can be used by the "Department of Finance, or the Department of Education & Skills [...] to justify how tax payer's money is used", allowing the economic imperatives of value for money and the supply of appropriately trained graduates (at different NFQ levels) to be measured.

6.5 Academics Perception of Societal Impact of the NFQ

Bill perceived the advent of the NFQ as a discontinuity, with there being a before and after, where "we suddenly changed over". He feels that the NFQ, albeit at the cost of increased complexity, has introduced a more satisfactory system allowing comparison of programmes, making progression opportunities for students clearer, and supporting lifelong learning. Bill also feels that the NFQ has improved access to education, and made the system less elitist. Henry expressed similar sentiments: "the student gets high points and they go and do their

university degree and the student who doesn't get the points and they can come here and they get a really good education".

The related societal role of the IoTs was a topic discussed by seven of the interviewees. For example: Edwin, in discussing mature students entering education to pursue a "level 7 or level 8 degree" stated "they all come to have a second chance in life, and [...] they do fantastically"; or Alan "I try and get the students [...] (to) believe in themselves in terms of [...] 'If I work at this long enough I'll be able to figure it out'. If I achieve that irrespective of any learning outcomes or any of the rest of it I think it is actually a good job".

The theme of supporting access and lifelong learning also emerged in the focus group, with the participants feeling that, whereas previously in Irish higher education an elitist system held sway, the NFQ has levelled the playing field. Indeed, even the use of the terminology of level 6, 7, and 8 to represent qualifications was put forward as "*putting everything on an even keel*". The fieldwork thus provides some evidence that the policy aims of using the NFQ to support access and lifelong learning are apparent to engineering academics.

However, the increase in numbers in third level, which is associated with access, also drew criticism, e.g. "from a business perspective of these institutions it is in their interests to keep people in the system as long as possible" (Alan). Thomas, despite seeing a role for the NFQ in improved access to education, particularly for apprentices, felt that there are "more and more people going to third level [...] twelve, fifteen years ago only those who wanted to go went. Now the career guidance in the schools is fill out the form anyway".

Although equality of access is a fundamental value of the NFQ (NQAI, 2003a), the massification²⁶ of third level identified by Thomas and Alan is not fully attributable to the NFQ, which forms only part of state educational policy. This points towards the complexity of my interviewees' views of the NFQ, which are often intertwined with other facets of our education system. As further example, some ten of the interviewees conflated semesterisation and modularisation with the NFQ, even though they are not directly related. To illustrate this consider Kees, who, in response to a question about the effectiveness of the NFQ, stated *"I would have arguments about the efficacy of the semester. I would have arguments about the efficacy of the semester. I would have arguments with regard to modules"*, however afterwards stating that *"whatever issues that*

²⁶ Massification refers to widening of access to education to increase participation rates. (Hazelkorn et al, 2015)

might be there with regard to modularisation, semesterisation, they're not framework issues". These views provide evidence of the confusion that can arise in decentralised higher education systems, such as in Ireland, in the implementation of policy (Viennet & Pont, 2017).

6.6 The NFQ and IoT Engineering Academic Management

The manner in which academic management perceive the influence and purpose of the NFQ in their dual-faceted role as *academic managers* and *academic leaders* (O'Sullivan, 2014), is explored in this section of the analysis.

Kees played a leading role in the introduction of the NFQ in engineering education in his IoT, where he recounted organising seminars, guest speakers, but above all else dealing with the hard issue of how to "motivate faculty staff". Some took on re-writing their modules with enthusiasm, but, others had to be "dragged kicking and screaming into doing it". Kees regarded Heads of School and Department as the key people involved, "the front line" in moving academics to OBE. Kees describes the QQI Engineering Standards as a "very good job", which made module development more disciplined, and which engineering academics found quite effective. He regards modules developed under the NFQ as being clearer in their aims, and that it is easier to see if a module has delivered on these aims.

Ayah considers the NFQ assists students to understand what they are doing, and assists their lecturers to help them achieve success, i.e. promotes a learner centred approach. As an academic manager she feels responsibility for the student, and, also, recognises the benefit of lecturers having clear direction about what and how to teach. Although she considers the LO approach quite restrictive, she regards it as effective for level 6 and 7 programmes, but less suitable for levels 8 and 9. She may have to tell her staff that *"this is the framework you have to work within"*, but also feels the *"need to leave a little bit of room"* to allow academic freedom. On the one hand Ayah is conscious of her responsibilities as an academic manager to help implement the NFQ. On the other hand, her interview shows awareness of her role as an academic leader to support her staff and help them work and grow as academics.

Isambard, from a management perspective, regards the NFQ as assisting in providing clarity of purpose and implementation to a programme, and making the desired graduate attributes clear. He also considers it benefits programme design, and assists in the allocation of limited resources within a restricted timescale. Isambard considers the NFQ assists students on their educational journey, moving from more prescriptive lower levels where the student is "taught to follow instruction and a methodical process" to where, later "things become more complex". He considers that the NFQ "has made the depth a bit shallower but it has been rigorous in the sense that it has been able to allow for measurement of exactly what your output is, what they know, what they can do, and you can stand over that". He speaks of graduates in terms of "what product are you getting in the end", reflecting earlier concern raised that OBE leads to a focus on process (Woolston, 2008), and suggestive of the strong framing (Bernstein, 2000) associated with performance pedagogy (Muller, 1998).

These views from academic management, which concur with my own experience, suggest that academic management see themselves in a dual role with regard to the NFQ. As academic managers they promote the use of the NFQ as part of engineering education in the IoTs. However, as academic leaders, they see their role, to an extent, as guardians of the academic integrity of what is being taught, how it is being taught, who is teaching it and who it is being taught to, which can cause them to question aspects of the use of the NFQ.

In addition to academic management acting to promote the NFQ with their staff, the analysis revealed they may also use the NFQ to assist with staff training, and to advise their staff on matters related to what and how they teach. Rudolf considered that "*I would never associate the framework with a management tool*". However, in comments representative of sentiment also expressed by Vitruvius and Ayah he states: "*I see it with new staff that come in [...] as part of their induction that I give them, and say this is a level 7 module that you're teaching, here's where it sits, so you understand [...] the type of approach that you need to take"*.

Similarly Kees described trying to use the NFQ to influence a staff member who he felt needed to adjust their pedagogy: "I would argue that [...], at least if you (have) got a framework, you've got, potentially, the ability to start trying to shape" their teaching approach.

However, other interviewees were critical of the effectiveness of management's approach to the NFQ, e.g. "*I think management here don't understand the NFQ" (John), and "most of what it seems to me [...] middle management does is kind of fire fight"* (Henry). Alan's experience as a relatively new lecturer is that he *"learned by doing"*, not relating, as he described his introduction to his new role, any experience of having the NFQ and its use explained. Edith, in describing her first interaction with the NFQ, explained that she did not know *"anything* about the framework until really we did the first (EI) accreditation" Similarly Thomas sees little management buy-in to the NFQ: "the only time it ever comes out is when we're doing programmatic review" and programme and module outcomes need to be updated.

6.7 Influence on Pedagogy

Havnes and Prøitz (2016) suggest that the use of LOs, in addition to being a "central aspect of a policy initiative" (p. 206) is a "pedagogical undertaking". The interviewees acknowledge the influence of the NFQ on pedagogy in terms of how it had influenced them personally, but also in terms of the influence on their profession.

As an example, John, who has embraced the NFQ, reflected on the development of his personal pedagogical position within the context the NFQ provides:

I [...] like it because it gives me a context in which to enact what I want to do [...] having said that, I like it because I have a range of enactment skills now that I didn't have ten years ago in terms of looking at different learning theories, being more open to summative and formative assessment. I am now prepared to pull reflection in which is key in the NFQ.

Although he considered that the NFQ is quite specific in terms of the course structures that have to be used, he still felt considerable autonomy in that lecturers "can enact that specification any way we want in our classroom". In doing so he drew a distinction between "the specified curriculum, the enacted curriculum and the experienced curriculum".

Vitruvius, in referring to the direct and constructive effect the NFQ has had on his pedagogy, stated that "I find if you know what the student is meant to be doing afterwards [...] then that tells me a lot about how I should teach them", using LOs to guide his teaching.

Bill perceives: "when you start bringing stuff in like LOs [...] then people actually should or can or ought to be thinking about – what is the best overall educational plan here rather than just the necessary nuts and bolts of the syllabus". Bill approached this change enthusiastically, explaining that he always considered "how can I do my job better. And even at this stage I wouldn't have even remotely exhausted how you could be a better lecturer or a better educator", which he would be doing "as an engineer who has become an educator".

Isambard takes an overarching perspective, linking LOs, pedagogy, and assessment, in which a strong sense is evident that he sees this as prescribed through OBE: *"if you are developing an assessment process for a particular module, you actually have to build in the pedagogy,*

the requirement that will enable the LOs to be achieved".

Twelve of the interviewees recognised the influence of LOs on pedagogy, whether having a perception of influence on self, on the profession, or both. For some (ten interviewees), similarly to Vitruvius, they see how they themselves have been influenced e.g. "You've set the list of learning outcomes and you have to basically cover those in your class, in some shape or form, whether its lab, lecture [...] or exams" (Edith) and "I have come around to the idea of learning outcomes. I think it's good, it changes the way you think about teaching" (Henry).

However, some (five of the interviewees) see, similarly to Isambard, an influence on their profession, e.g. "when you have put learning outcomes into it, suddenly we had to think about education really" (Bill). Ayah, when speaking of level 6 and 7 programmes had a positive view in that the use of LOs "tells the lecturer what you need to teach and how you're going to do it and that's fine". However, she was critical of their use for level 8 and 9 programmes, where the LO focus means "**they** are quite tied, **their** administration workload is heavier, everything, **their** assessments, their exams, everything is tied into these six or seven learning outcomes".

A majority (twelve) of my interviewees recognise that change has occurred in pedagogy under the influence of OBE, using positive language such as "*it gives me a context*" (John) and "*I have come around to the idea of*" (Henry). For some there was a sense of enablement: giving John a framework in which to develop his pedagogy; and Bill feeling he grew to be a better educator. This leads to a finding that OBE acts as an implicit influence over the pedagogy of the majority of my interviewees, that they largely regard this as constructive, and it has encouraged them to give more consideration to how they approach teaching. The manner in which their pedagogy has been affected through this influence is analysed in *Chapter 9*.

6.8 Course Design

This section analyses my interviewees perceptions of the NFQ's influence over course design (curriculum) through their use of the LO based QQI Engineering Standards (QQI, 2014c), with a later section considering their perceptions of EI's complementary role. The manner in which the use of LOs has shaped the interviewees' approach to knowledge and skills in the curriculum, a key consideration for engineering education, is considered in *Chapter 10*.

All fifteen of the interviewees acknowledged the implicit influence of the NFQ on the use of LOs in course design. A representative sample of their views are presented below, illustrating the positive benefits they perceive.

Gordon regards the use of LOs in course design as very positive, allowing him to focus on teaching a smaller number of relevant curriculum topics in depth, as opposed to covering a larger curriculum in a shallower manner. Kees feels they introduce more academic rigour, regarding there to be *"more discipline associated with"* module and programme design using LOs, a sentiment shared by Rudolf. Vitruvius is similarly positively inclined towards the use of LOs in course design, particularly in terms of knowing what is expected of the students. George sees benefit from LOs in terms of understanding students' achievements, regarding LOs as *"a good summary of what they should achieve at the very end"*.

Edwin, is conscious that, in module design, "we have to be aware of these LOs". Henry acknowledges that in course design he has to "start with the LOs", and then later "think about the content". Even Thomas, who views the influence of the NFQ somewhat negatively, states that "we go from the top down, programme out, with the LOs and how can you achieve a LO".

The use of LOs has, also, an explicit, prescriptive control over course design. Analysis of the interviews showed a broad acceptance amongst the interviewees (nine) that the NFQ is playing a significant controlling role in terms of specifying structures within which programmes must be designed and delivered. A sample of their views are given below.

John feels that "the NFQ [...] structures how we should design or specify a curriculum". Rudolf also viewed the NFQ as being quite controlling of module and programme development, and stated that "the framework provides [...] boundaries" for course design. Course development is where academics decide what knowledge, skills etc. are represented in the LOs of modules and thence represent the competencies of a graduate. These boundaries are, thus, boundaries around, inter alia, knowledge.

Ayah, although not referring directly to boundaries, felt that, at level 8, more flexibility than that allowed by the NFQ was necessary in course design, and, indeed, was quite critical:

as you go higher up, I think it stifles the whole process and I think its stifling innovation and its stifling research and I think that problem is that they're going out into the real world where they don't actually have had the chance or the space to actually think outside of

these very prescriptive LOs.

Ayah's use, and re-use, in her interview, of the term "*stifling*" is quite pronounced, reinforcing her perception of prescription. The NFQ has been previously described as "being too prescriptive, reducing the creativity and autonomy of the programme developer. Uniformity and conformity become the dominant mantra" (Kenny, 2009, p. 24). This resonates with an excerpt from George's interview, where, referring to writing LOs he states that "*I would be conscious that it has to be of the correct level*".

Kees, viewed programme design under the NFQ as prescribing a focus on LOs. In discussing module design under the NFQ, he viewed that "for someone with a particular qualification, say at Level 7, that graduate should be able to demonstrate a series of competencies. [...] and that enabled then, programmes to be developed that were focused on that".

Isambard, whilst agreeing with Kees's view of graduate competencies, has a somewhat nuanced approach to the meaning of LOs in the curriculum. On the one hand, he feels that any two academics teaching a module may, in interpreting them "*draw examples from different spheres and you are free to do that*", echoing John's description earlier of the difference between the specified curriculum and the enacted curriculum. However, this is only insofar "*as the LOs* [...] allows you to achieve the same end". Ultimately the module LOs, rather than the lecturer's expertise, provide a high level specification for what is to be taught, where "the defining thing is the LOs" (Bill).

Alan also views the use of LOs as a key part of module design, and that the NFQ causes him to "*adhere*" to the use of Bloom's taxonomy (the "*verbs*") in the design of modules and later to "*adhere*" to what is written in the module descriptor.

These views suggest that the NFQ is a significant factor in the increased "emphasis on control and oversight" (Kenny et al, 2015, p. 102) of the working environment for IoT engineering academics. Havnes and Prøitz (2016) succinctly capture the dominant nature of an NQF in describing that LOs constitute "a system 'from above'" (p. 220).

An earlier finding was that the NFQ is one of the principal tools used to support a policy-driven re-orientation of Irish higher education towards having a more skills focus. Complementing this earlier finding, the analysis leads to a finding that, through its LOs focus, the NFQ is perceived by engineering academics as an explicit control that is placing boundaries around what is to be taught, supporting the skills focus of government policy for higher education.

6.9 The Role of Engineers Ireland

In exploring El's role in the pedagogic discourse a particular focus of the interviews was the duality of El accreditation and programmatic review, alongside El's influence on programme design. Their influence goes further, however, than accreditation: Kees related that the QQI Engineering standards (QQI, 2014c), were created via a process *"under the El umbrella, of linking the generic learning outcomes associated with the National Qualifications Framework with engineering learning outcomes"*. Isambard, whilst acknowledging this direct relationship, felt this is often not entirely clear to all engineering academics.

As an example of this duality, Henry in comparing a forthcoming EI accreditation visit to a recent programmatic review felt "we would be much more focused on that than we would be on the programmatic review". Henry also regarded EI accreditation as the main driver for updates of modules. He viewed EI's introduction of evidence driven accreditation as a positive change, whereas his view of programmatic review is that "we don't spend enough time reflecting on the course as a whole [...] I don't know if any of the IoTs do".

George, expressing similar sentiments to Edith, states that "I find that what we concentrate on a lot is the EI criteria, mainly because we try and get our courses validated by EI. And so, their criteria are equally as important as the framework, if not more important". An even more pronounced view of the importance of EI accreditation is given by Grace, who related that, in her institute, the term level 8 is generally associated with EI accredited programmes. However, she emphasised this approach did not generalise across the IoT sector.

Bill discussed trying to map programmes into El "LOs and their programme outcomes, to satisfy what they hope to see for accreditation purposes". Ayah refers to academics, in designing programmes, "being forced more into [...] the (El) programme outcomes". Alan discussed his experience of El accreditation as a relatively new IoT lecturer. He regarded this as a strong influence on module design, informing "the LOs", and "almost dictates in some respects what you should be doing". Isambard speaks of "what is required by El", and Gordon felt that "it challenged us I think in ways that the programmatic review didn't".

The analysis showed that EI accreditation requirements are regarded (by eleven of the interviewees) as more important in course design than the NFQ or institute level course

design and validation processes. Indeed, for Grace, for whom, unfortunately, the interview recording was lost, the principal topic noted in the agreed memo of the meeting was the impact of the influence of EI accreditation on course design and structure. Of the remaining interviewees, three still recognised the significant importance of EI for engineering academics, and one of the interviewees did not comment on the subject.

This focus group were informed of the analysis pointing towards the strong influence of EI, and were asked "What influence do you feel that EI have had on the adoption of the learning outcome approach by engineering academics"? They confirmed the view of the extensive influence of EI in this regard, with the discussion centring on the significance of this influence.

Part of the discussion concerned the mobility of engineering graduates, with a feeling that "*El accreditation certainly does give a greater mobility*" to students wishing to work abroad, causing students and graduates to place a value on El accreditation. A dissenting voice felt that *"the graduates I would be in touch with who are still in Ireland have never heard a mention of El*". However, another participant felt that part-time engineering students, who are generally in employment, were very conscious of accreditation.

In considering programme design, one focus group participant felt that EI programme area descriptors "*fit in flawlessly into programmes outcomes*". Another questioned the relevance of EI, who they recognised however as "*an external body who has a huge influence on what we're teaching our students*", in particular in terms of the maths content of programmes. They went on to question whether this significant maths content was of value to the students, when, instead, they could be taught more skills.

In discussing whether EI, through their adoption of LOs for accreditation, have influenced engineering academics acceptance of LOs, there was consensus that EI's influence meant that "we really have no choice". This is reinforced by practice across the sector: "it's a combination of EI and every other college [...] So maybe EI on its own wouldn't have that influence".

These views point towards the significant influence of EI's OBE accreditation on the design of IoT engineering programmes. There was also a strong element of compulsion felt by the interviewees, who, in their individual interviews, used language such as "dictates", "forced into", "drive", "required", and "challenged"; and when together in the focus group, "no choice", "we're a small fish in a big pond and we have to do what we're told.". It is not just

that EI have accreditation *requirements*, but that clearly the interviewees saw them as overarching requirements.

Table 6-2 provides an overview of where my interviewees' perceive: the NFQ is *prescriptive* with regard to course design; LOs *influence* them in their approach to this; and EI are prescriptive regarding course design. All experience an element of prescriptive control from one or both of the NFQ and EI, with the exception of Vitruvius, who does experience implicit control. These views are consistent with the NFQ and EI constituting 'official knowledge'.

						Res	earc	h Pa	ırtici	Research Participant								
	Isambard	Ayah	Kees	Rudolf	Vitruvius	Bill	Thomas	Gordon	Edwin	George	Edith	Henry	nhol	Alan	Grace			
NFQ LOs influence course design																		
NFQ prescriptive for course design																		
El prescriptive with regard to Course design																		

Table 6-2 Prescription and Influence from the NFQ and EI over Course Design

6.9.1 Views Critical of El

A number of interviewees (five) questioned the relevance of EI accreditation for graduate employment, whilst still recognising its influence on engineering education. It is significant that any engineering academics would question the relevance of accreditation from their professional body. However, the interviewees were predominantly from the fields of electronic and mechanical engineering, where, as discussed earlier, for electronic in particular, and also mechanical engineering to an extent, although accreditation is an important consideration for engineering education, it is less so for employment.

For example, Alan, speaking of El's role in his field of ICT, states, of his time in industry, that, regardless of the importance of El accreditation from an academic perspective, *"I have interviewed people in industry and […] the thought never even entered my head to turn or evaluate if this course was accredited by El"*.

Similarly, Edith, on a personal professional development level, with regard to pursuing Charted Engineering status, asked herself "to what benefit am I doing this [...] we don't need it for our working environment", considering EI "biased towards civil engineering". George considers that, whilst he supports the idea of EI accreditation, for graduates "it is hard to know how important it is because I think most people who go out and work in the industry, especially

in electronics, they don't need to be a member, for most of the industry"

Edwin states that "the benefits to me as an academic I know, it's a prestigious accreditation", but at the same time feels that, for some engineering degrees such as electronics "I don't think the employer asks [...] is that EI accredited or not". Thomas feels that accreditation will be of relevance to few graduates: "I'm not sure how many will actually avail of that Chartered Engineering status from a programme.", and that if you dropped it "you could redesign your courses [...] and you could eliminate some content that's only in there to satisfy the IEI".

In the focus group some negative views of EI's influences were expressed, with regard to the emphasis on maths mentioned earlier, but also concerning the quality assurance system they use to evaluate evidence for accreditation, with some of the participants questioning the value of their approach. One participant stated that "*I can't say I've ever given EI a moment's thought during the delivery of anything that I've ever taught*", but on the other hand conceded that EI's accreditation requirements had a huge influence on programme design, and thence on what he would deliver.

El is thus seen as a particularly strong influence, which, in parallel with the NFQ, has reinforced the adoption of a LOs approach to course design amongst my research participants, even though not all saw El's accreditation requirements as fully relevant for employment. Even in this case they still influence the structure of engineering education, and most IoT engineering academics participate in five year accreditation cycles (QQI, 2019).

One interviewee expressed a particularly critical view of the impact of El's interpretation of Bologna on IoT engineering education. In their view, one reason El raised their Chartered Engineering educational requirements to master's degree was to "*keep a separation between the types of qualifications that were coming from IoTs, and coming from universities*". I have heard, and shared, such views before in a private capacity, however, the accessible literature contains little of a critical nature in this regard, with the exception of Kelly (SEFI, 2002), who described this as an attempt to "change our engineering degree structure" (p. 23) that had "nothing to do with Bologna". These views support my earlier argument that El's interpretation of Bologna acted to curtail the provision in the IoTs of programmes accredited on the path to Chartered Engineer, reinforcing vertical stratification in engineering education.

6.10 Conclusion

The analysis presented above leads to a finding that a significant majority (twelve) of my research participants have given a positive, albeit qualified, acceptance to the role the NFQ plays as part of IoT engineering education, and that they regard its OBE approach as effective. This reflects my own view of the NFQ as I assisted in leading the transition to OBE in ITB engineering. Of the remaining interviewees, only Thomas, was recognisably negative, with the other two recognising pros and cons to the NFQ. Of the interviewees who are academic managers, all were generally positive regarding the framework, albeit somewhat critical on occasion. For those interviewees I regard as holding neutral views, in one case they had joined the IoTs post-NFQ, and what I viewed as a neutral stance could instead be a reflection that the NFQ is just part of the system as they know it. All of my interviewees, it was clear, whether they held positive, neutral, or negative views regarding its effect, considered the NFQ a significant factor in engineering education.

The fieldwork also provides evidence that EI's LOs based accreditation has been a significant additional factor in promoting the acceptance of OBE in engineering education, although engineering academics do not always regard this as important for graduate employment.

This chapter outlined the control experienced by my interviewees over their engineering education system, through official knowledge promoting OBE, which Kees referred to as *"that control issue we talked about earlier, right. Fascinating, fascinating area*". Further chapters of the analysis consider how this has impacted the pedagogic discourse and the pedagogic identity of my research participants.

Chapter 7 The Language of Levels

Because without our language, we have lost ourselves. Who are we without our words? Melina Marchetta (2009, p. 65).

7.1 Introduction

As I carried out preliminary analysis of each interview I was keen to see what themes might emerge, and use these insights to guide subsequent interviews. My initial analysis showed that they were yielding rich data regarding the perceptions of my research participants of the impact of outcomes based education (OBE). However, only in coding the 7th interview (John) did I notice that in our discussion on the pedagogic discourse of engineering education we both repeatedly referred to National Framework of Qualification (NFQ) levels in describing various aspects of Institute of Technology (IoT) engineering education. In subsequently reviewing the significance, it became apparent that this was a common feature of all the interviews I had undertaken. I was quite comfortable using this terminology as it was employed in our pedagogic discourse, which explained why I had not noticed it previously.

To reflect on the phenomenon the analysis had revealed I returned to the literature which suggested that the policies associated with NQFs bring with them an associated terminology (Allais, 2010b; Biesta, 2015; Raffe, 2011). This aims to support features of qualification frameworks such as access, transfer, progression, approaches to teaching, employment opportunities, student choices, etc. (Allais, 2010b; Raffe, 2011). In observing the international experience of national qualification frameworks (NQFs) Raffe (2011) considers their effectiveness depends on the "widespread understanding and fluent use of the 'language' of learning" (p. 70), and refers directly to "a language of 'levels'" (p. 75). Within Europe "LOs are perceived as a key approach to the establishment of a common language, an understanding of learning as quantifiable units" (Kenny, 2006, p. 197).

I had already noted it was intended that the nomenclature of the NFQ would become part of the language of Irish higher education (Collins et al, 2009), with Kenny (2006) referring to a "language of learning outcomes" (p. 187). In the fieldwork I observed the term 'level' being used by all fifteen interviewees, not just to describe NFQ levels themselves, but also other aspects of engineering education, *inter alia*, students, programmes, graduate attributes, jobs and even knowledge. This strongly suggests that IoT engineering academics, including myself, have, consequent to the NFQ, adopted a 'language of levels' as part of our academic vocabulary.

As would be expected given the policy driver for OBE, this language of levels is used by my research participants as they engage with policy discourse as modelled through the concept of the official recontextualising field (ORF) This includes in referring to awards, programmes, graduates, transparency, interactions with employers, progression and the structure of higher education.

The use of the language of levels, however, goes deeper than being part of the policy discourse, it has also become part of the language of everyday academic practice of my research participants. Considering this through the lens of the pedagogic device, it is part of their instructional discourse in the pedagogic recontextualising field (PRF) and in the field of reproduction, displacing more traditional terminology associated with engineering education. In this context it is used to refer to progression, the educational positioning of the IoTs, students, knowledge, projects, pedagogy, module design and delivery, and even the production of lecture notes and other teaching material. A summary of these different usages is provided in *Table 7-1*.

			Isambard	Ayah	Kees	Rudolf	Vitruvius	Bill	Thomas	Gordon	Edwin	George	Edith	Henry	nhol	Alan	Grace
Language of Policy Discourse		Awards															
		Programmes															
		Graduates															
		Transparency															
		Jobs															
	ctice	Progression															
		Positioning of IoTs															
	: Prae	Students															
	Language of Academic Practice	Student Projects															
		Knowledge															
		Pedagogy															
		Modules															
	Lang	Learning Outcomes															

Table 7-1 The Use of the Language of Levels : including where it is used to reflect discourse on policy, where it is used as part of academic practice, and where these two areas overlap

The focus group considered the language of levels part of the academic norm, and discussed aspects of its usage. However, they extended the discussion to encompass *why* it was used, which, they felt, is because everyone uses it, and you need to use it to be understood. Indeed it is *"confusing now when you try to talk (about) our current system and you don't use levels"*.

To begin with I illustrate the manner in which the language of levels has become an integral part of the totality of the pedagogic discourse through considering Gordon's interview, which I use as an exemplar of the usage of the terminology.

7.2 Gordon's Interview and the Language of Levels

Gordon's interview took place before the analysis revealed the language of levels. However, when I re-analysed his interview in the context of the phenomenon, I realised it had been an integral part of our dialogue as we discussed our education system. Our conversation is indicative of the manner in which my interviewees, and I, used this language in our pedagogic discourse in the interviews. I believe that for someone familiar with IoT engineering education the interview is quite cogent. However, I suspect it might be somewhat opaque to engineering educators from the Irish universities²⁷, who moved much later as a sector to embrace the principle of mapping of programmes to LOs (FIN, 2009). Additionally, they do not have the same concern regarding the relationship between undergraduate engineering programmes at level 6, 7 and 8. Furthermore, as the focus group suggested, the language of levels would be expected to be quite unclear to those from other jurisdictions.

The extracts below illustrate the manner in which Gordon used the language of levels in referring to: courses; student projects; students; modules; pedagogy; knowledge; graduates; curriculum and assessment.

In discussing the use of the NFQ to guide the design and delivery of programmes Gordon's usage of the language of levels goes beyond merely associating courses and projects with levels, it is also used to consider what the student is expected to achieve, what might be taught to them, and how it might be taught to them:

we very, very often talk about, 'Well is this a level 7 project or is this a level 8 project?' [...] we have guidelines that we share [...] that [...] would match the difference between level 7 and level 8, for instance the stages in the product development process where we have a

²⁷ With the exception of TU Dublin, formed from the amalgamation of three IoTs.

marketing department, the engineering department, the fabrication department I guess and they would be working together [...]

To be very over simplistic about it, I find at level 6 might match to the fabrication. Level 7 might be fabrication plus design. So, a level 7 student might need a statement of requirements to do what she does. Then coming up to level 8 I would feel that the input into the problems that they would solve would be more, the questions that they would be able to answer would be more open.

Gordon discussed the instructional discourse using the language of levels: to differentiate between student projects in different years of programmes; to differentiate the educational aims of such projects; and in discussing their technical aspects with his peers.

In responding to a question regarding how his approach to knowledge in the curriculum is influenced by the levels in the framework Gordon firmly associates knowledge with LOs. NFQ levels may be used to consider the depth of the competencies expected of the student:

I think the way I would put together the modules would be in terms of the questions that the students should be able to answer at the end of it. And what they should be able to do at the end of it, like given this thing, can they do that thing? [...] So, for instance a practical where you are given a thing, you're asked to predict how it works and then you're asked to measure it and then compare the two. I think that's as applicable at level 6 as at it is at level 7 and at level 8. Perhaps the depth might be different

Gordon differentiated LOs by level:

NFQ levels can map pretty well onto [...] some kind of taxonomy, like Blooms taxonomy where there are key words, at level 6 it could be, measure, analyse, graph. At level 7 it could be things like, compare, investigate [...] I find those words very, very useful to guide me in how assessment gets done and that then feeds back into how the stuff is delivered.

Gordon discussed his interactions with an EI accreditation panel using the language of levels, considering pedagogy, and what could be expected of a student:

one of the panel [...] said to me 'I deliver C++ or object orientated programming in second year through C++. I deliver Java and algorithms at the moment, well in Java of course, but he suggested at level 8, would it be a thing to say to a student, 'OK you know how to programme in C++, there's a really good book about Java, now go to it.' And you know that could be valid.

Towards the end of our discussion I asked Gordon if there were any particular aspects of the NFQ he wished to highlight:

the upside is, yes. Like I say, the definition of the difference between level 6/level 7/level 8 is very useful to me in designing modules and even in thinking about courses and thinking

about the kind of graduate that comes out of these, that's great, that's very clear. Downside, I find the modular structure of it [...] tends to de-emphasise in a student's mind that it's all interconnected.

Gordon concluded the interview by stating, with regard to the opportunity the interview gave to share his views of the NFQ, that "*I have had the chance to give a pretty full account of what I think about it*", an account throughout which he made use of the language of levels.

7.3 The Use of the Language of Levels to Engage with the Regulative Discourse

I now move on to illustrate the manner in which the language of levels is used by all of my interviewees in their engagement with the regulative discourse of the ORF, in reference to awards, programmes, graduates, the NFQ value of transparency, and employment.

7.3.1 Describing Awards and Programmes

Kees experienced the tension between the NQAI and EI regarding the naming of level 7 qualifications as ordinary degrees, explaining HETAC insistence's "that they were going to regularise and minimise the number of different types of qualifications". However, notwithstanding the official name changes, and the earlier disagreement by EI (IEI, 2001a), the analysis suggests that it is now quite common, in academia, for awards to be described using the terms level 6, level 7 etc. (twelve of the interviewees).

Henry's words were reflective of the views of Bill and Edwin: "the degree word now doesn't have the same meaning it used to have but level 6, 7, 8 does have currency" and we have "replaced using old words like certificate, diploma, degree with these new levels".

Edwin had doubts as to the effectiveness of such descriptions, feeling that the effort to attain a degree is diminished in describing it as a level. He expressed concern that *"nobody asks anymore questions. How long? Where? Who gave it to you?"*. He concludes the levels concept is not sufficient to fully comprehend the basis for an award. The terms PhD or doctorate have significant traction in academia (Park, 2007), bringing with them connotations of effort, experience, research, peer review and academic excellence, but Edwin considers *"a level 10 degree, and in three words, it's minimised somehow, the huge effort to get there"*.

Alan described that, prior to joining the IoT sector, "I was aware of the cert and diploma but I had no understanding, I had never heard of level 6, 7, 8, 9, 10 until I actually started here". This suggests that although the IoT 's may make heavy usage of levels terminology, it may not have been adopted by all stakeholders. On the other hand Edith discussed dealing with companies seeking graduates at particular NFQ levels (7 and 8).

Continuing with the theme of stakeholders understanding the use of levels to describe awards, Rudolf noted an issue with students returning to education. He found they did not always understand the new terminology "we often confuse them because we are speaking at level 6 or level 10 and they are talking about diplomas".

George's interview was noteworthy in that he seemed to avoid the use of the term level to describe awards. When queried about this he emphatically explained that he did not consider levels to be appropriate for describing qualifications and graduate capability, stating that "*I liked the idea of the cert, the diploma and the degree. I liked that concept because [...] I had a good idea what that actually meant*". However, George did use the terminology on occasion, for example in differentiating universities and IoTs and their students.

What emerges clearly, however, is that regardless of the concern with regard to the use of the term level to describe awards indicated by five of the twelve interviewees, even those with such disquiet accepted that this usage was now prevalent.

The use of levels to describe programmes (by thirteen of the interviewees) further illustrates the manner in which the NFQ has impacted on engineering education.

For example, Gordon regards the levels of the NFQ as allowing differences between programmes, and their future graduates, to be conceptualised at design stage. George, in viewing programmes as levels, regards it as easier to decide that they are fit for purpose, i.e. whether "that degree is of the correct level and that degree isn't".

Henry sees the term level as useful at CAO²⁸ application time, i.e. applicants have "*their level* 8 form, their level 6 and 7 form and I think that's a very good thing", as, he feels that, for prospective students "*we've made deciding on courses on what to do much more difficult*", as there are now so many to choose from. However, Rudolf feels that students are not overly familiar with using levels to describe programmes, and that "*they have no real idea what level* 6, level 7 or 8 mean", or how long it will take to complete them.

²⁸ The CAO, or Central Applications Office (CAO, 2017), is Irelands principal application route for potential students who wish to study in the university or IoT sector.

Ayah used the term level to refer to higher certificates and ordinary degrees, where "at level 6 and 7 we could probably try and be a little bit more global in the approach rather than being so prescriptive". This prescriptive view aligns with the NFQ level 6 indicator that graduates should be able to "formulate responses to well-defined abstract problems" (NQAI, 2011).

To summarise the significance, my research participants were quite comfortable in the use of the term level to represent various aspects of programmes, implying that the policy led discourse of representing programmes using NFQ levels has been accepted by academics.

7.3.2 Referring to Graduates and Employment

Government policy regarding supply of graduates is often couched in terms of NFQ levels, (HESG, 2011; Oireachtas Library & Research Service, 2014). Graduates were referred to as levels by eleven of the interviewees, principally in discussing graduate capabilities.

Alan compared graduates, and their consequent employment, using their programme level designations, in doing so highlighting situations where the role they take up is not commensurate, in his opinion, with the level of their qualification: *"I know guys that are coming out of here with level 7s who are doing the same job as a level 8, no doubt. You know? I can think of one example last year"*. This may have occurred due to an element of confusion amongst employers with regard to what degrees at different levels mean. Kees described in the early years of the NFQ *"we would have seen evidence (of that). I don't think it was widespread."* and I would share a similar experience of the early years of the NFQ.

Isambard, similarly to Edwin and Kees, used the term level to describe graduate capabilities:

a level 9 is [...] somebody who can follow through a structured research process, whereas in a level 10 which is a PhD, you expect that is somebody who can bring in novelty and they can bring in new knowledge and they can push the frontiers.

At the other end of the NFQ level spectrum of higher education, George also viewed higher certificate graduates in terms of their capability: "*people used to be fairly skilled at level 6*".

Kees and Grace described professional engineering graduates as level 8 graduates. Henry, on this subject, feels that "you can't just say a professional engineering degree, full stop the way you did at one time so you have to say [...] what level degree and where is it from".

Edith discussed graduates in the context of the reduction in government funding of third level, relative to student numbers, that has occurred since the hard recession in the aftermath of

the Celtic Tiger, and difficulties this has created (IBEC, 2015; Expert Group in Future Funding for Higher Education, 2016; HEA, 2017a):

the government are asking for all these levels of students coming out, level 6, level 7 [...] but they're not contributing to the colleges [...] our labs and I'm sure your labs are the same, they could do with a bit of a facelift; they could do with an upgrade [...] with maintenance.

Nearly half (seven) of my research participants use the language of levels to refer to the type of employment that someone completing a course might take up in industry. This usage is consistent with the emergence of the language of levels in the national policy discourse to refer to up-skilling and employment (IBEC, 2015; DES, 2016).

Bill succinctly linked levels and jobs, referring to the end result of an honours degree as "you're an engineer level", which Ayah similarly referred to as "professional level", also describing the capabilities of higher certificate graduates as "operator technician level". Thomas similarly used the language of levels in discussing how he views the technician role, stating that level 7 "doesn't give you technician level".

Edwin described that, when large companies visited his IoT seeking to employ graduates they look to fill posts at level 7 and 8. In this case, the language of levels has also become part of the discourse with key stakeholders, the large employers who not only provide jobs for many graduates, but also contribute to (and sometimes dictate) programme design. Edith also provided evidence of this, speaking of a recent recruitment drive from industry in her IoT, stated *"they were looking for technicians, so they specifically said 'we only want level 7'*".

Rudolf uses the language of levels to articulate a different position regarding industry engagement with the framework:

They struggle with the ladder, the progression pathway, 6, 7, 8. You know I think that is a challenge for them. So, certainly the bigger employers you know it's honours degrees, it's a language that they talk about it and they are comfortable in that but really they have very little knowledge of the framework and its purpose.

This usage of the language of levels to refer to employment provides further evidence of the manner in which the NFQ, as part of the regulative discourse of the ORF, is influencing educators, inculcating a jobs and skills focus.

7.3.3 Transparency

Eight of my research participants used the language of levels to discuss transparency.

Edith refers to the usefulness of the language of levels for prospective students and their parents "that are sending their kids to college, they can see what levels they can achieve". Edith described further evidence of levels and transparency, where "at graduation yesterday the President was talking about different levels" in a speech to students and their families.

Isambard felt that the NFQ "has actually enabled a very clear demarcation of the different levels of the awards", which, in terms of transparency, allows awards from different HEI's to be compared against one another. Edwin's interview also discussed transparency being applied to compare HEI's, their qualifications, and their students in terms of levels.

Rudolf was sceptical about how transparent programmes were, to prospective students, to current students, and also to employers:

some of the HR and the training managers certainly in the engineer space understand the level 6 space because they may be looking to develop programmes where they can up-skill craft-workers or operatives, but that middle ground and level 7 and the validity of it, I would have a sense that its zero or nothing.

Henry, on the other hand, felt that although "*employers have now become a little bit confused*" with regard to what a Bachelor of Engineering award means, "*the level does help clarify that to some extent*".

The language of levels is used by over half of the interviewees in reference to transparency. All but one felt the NFQ facilitated transparency, but even Rudolf, who did not fully agree with this, discussed his concerns using the language of levels. This provides further evidence of the use of the language of levels in academic engagement with the policy discourse.

7.4 The use of the Language of Levels to Engage with Both the ORF and the PRF

This section discusses how the language of levels is used by my research participants in referring to aspects of the pedagogic discourse that encompass both the ORF and the PRF.

7.4.1 Progression

Progression is part of the national policy discourse for further and higher education (GoI, 1992; Irish Statute Book, 1999b; NQAI, 2003a; OECD, 2003; SOLAS, 2014), and a key aspect of the NFQ value of *Comprehensiveness and Coherence*. Furthermore, progression raises fundamental issues for IoT engineering education, as will be considered in later chapters. The use of the language of levels in relation to progression is thus both a means for engagement

with national policy discourse, and also of relevance to academic practice. Over half (eight) of the interviewees referred to progression in terms of levels, predominantly in discussing the intended policy impact of supporting students in moving upwards through NFQ levels. However, as will be discussed later (Chapter 8) my interviewees also perceive progression as having a wider impact on the education they offer, and they made use of the language of levels in this context also.

Thomas felt that progression was positive for tradespeople, who, previously "stopped at their particular level with their apprenticeship". He considered the NFQ required IoTs to remove barriers to their progression, and create paths to allow them to further their education.

Bill firmly linked progression to the concept of levels, comparing it to the previous system where there "*wasn't an automatic progression route*", and also felt the NFQ acted to remove barriers. He used an engineering analogy to explain his view:

with the National Framework, the fact that you have levels; if you've levels in anything like a tank, it's always seen that you can take up any of these levels and you can move between them. So level 6 automatically means you should be going to level 7 and 8.

Several of the interviewees similarly discussed progression between NFQ levels 6, 7 and 8. Edith sees "everybody's into level 7 and then level 8 and it's a feeder route to getting your degree, whether its honours or ordinary". Ayah described her personal experience of progression in a similar manner "I would have done the ladder system so the level 6, 7, 8 and then 9".

Isambard, however, rather than focussing on aspects related to the policy discourse, reveals his perception of the impact of progression on teaching, where he states that:

if someone who is teaching has interpreted correctly then they would need to recognise that at some stage through all adjacent levels, for example between level 6 and level 7, there is bound to be some unifying [...] overlaps in what you teach or in the way the learning outcomes are assessed or even in the way the syllabus is interpreted.

This is an important point I return to in *Chapter 8,* when I discuss the wider impact my research participants perceive progression has had on our engineering education system.

7.4.2 Educational Positioning of the IoT Sector

This subsection shows the language of levels as used by five of the interviewees to describe the IoT sector, in terms of where it is situated from an educational perspective, where it should stay, and where it might evolve to. This is related to the analysis in *Chapter 6* of the societal impact of the NFQ, which discussed the concerns of the research participants with regard to their students, funding for the sector, and the impact of massification.

Ayah, herself a graduate of a ladder based system, uses the language of levels to differentiate the IoTs and the universities, describing that "*the level 6, 7, 8 and then 9*" of the ladder system made her "*very aware as a student of a LOs approach*", more so than if she had been a university student. Thomas uses the term level in a similar context, feeling that the IoTs should be focussing on teaching technology "*at level 6 and level 7 where we should be*".

Alan on the other hand came "from the university system" and "didn't even know what level 6 & 7 was". He was somewhat aware of, but not fully appreciative of, the differences between the IoT sector and the university sector before he joined an IoT:

I actually learned more in six months in industry than in terms of applied focus than I did in four years (in university). I will be honest, whereas when I started here I was shocked with the level of contact in lab environments.

To further illustrate, he described an experience as a student, where IoT graduates had joined his university class: a lecturer asked a question of the class with the accompanying instruction that *"I don't want anyone from an IoT to answer this"*, on the basis that their different knowledge base made the answer obvious to them.

Edith uses the language of levels in a similar context, speculating that it may be the case that in a future *"technological university they just do level 8 onwards"*, i.e. the creation of technological universities from the merger of IoTs might cause a shift in programme offerings. Henry conjectures that, *"the differentiation between us and the universities is we are now going to produce level 8 students and they'll produce level 9 students"*.

Those of the interviewees who used the language of levels in the context of the IoT educational positioning used NFQ levels in expressing views of the past, the present, and what may be the future of the IoT sector. This was done at both a policy level (e.g. the technological university, funding) and in regard to the type of education offered in the IoT sector, e.g. in terms of laboratory work, the teaching of technology, and the type of students.

7.5 The Use of the Language of Levels in the Instructional Discourse

The language of levels was used by the interviewees in discussing their academic practice, to refer to: students; student projects; approaches to teaching; modules; and knowledge.

7.5.1 Referring to Students

The interviewees (ten) used the term 'level' to refer to students, in terms of situating them at the stage they are at in their programme of study, and their educational needs.

Alan identifies students as levels, ascribing different educational needs, to the different levels. He differentiates how he would "present the content to a level 6 or 7 students versus how you direct the student at level 9 to actually go on and almost figure it out themselves".

John identifies his research master's students as *"level 9 research students"*. Edith referred to students as levels in discussing whether some programmes are too academic, explaining that she regarded the focus as *"wrong for the level 7 students, because I do think there should be a more trade element for a lot of those students"*.

Ayah considers that "level 6, its very much hands-on practical focused, very prescriptive, stepby-step approach, a lot of hand holding" (Ayah). With similar concern, Vitruvius "will supply more backup material to the level 6 students". In contrast, level 8 students "need to be able to go out into the world and solve problems and navigate through different systems that they've never seen before" (Ayah).

The key points that emerge from this analysis is that lecturers regard students, at least partially, as being at a "*level*", and seek to educate them appropriately on that basis.

7.5.2 Approaches to Teaching and Student Projects

Levels were used by seven of my research participants to describe different contexts within which the students are educated, and how different types of learning and teaching are associated with the levels.

Isambard, reflecting sentiments shared by Ayah advocated, that "as you move from the lower levels to higher levels of the NFQ, they need to move from prescription and give more or less open-ended problem solving type of challenges". Thomas similarly recognised differences in the teaching approach required at the different levels "at level 8 [...] its [...] about creativity and that has to come from the individual and level 7 its more guidance based". Although he

felt that "there's a lot to be said for guiding a student a bit more even at level 8 as well as opposed to just letting them off on their own and submit something".

Vitruvius similarly regards levels as a guide to teaching students, going so far as to use it to explain to students the amount of self-directed learning he expects from them. When the students question "the level of which they must go and find stuff out themselves [...] I can have that argument and ultimately I can pull out the NFQ framework documents and quote them at them". Alan also adapts his teaching "based on the level" he is "trying to teach it at".

A small number of interviewees (two) directly used the language of levels when describing lecture notes. For example, Vitruvius describes how he considers lecture notes:

When I read over my notes over the course of the summer, before I launch them again, I'm always trying to think "Now hold on. I keep writing this material that I deliver for level 7, level 6, but they're the same notes: something's up there.

He further explains he uses the levels the students are studying at to decide how in-depth and challenging his lecture notes should be.

The analysis reveals that some of my research participants actively consider the manner in which they are teaching students in the context of the level at which they are teaching. This is further illustrated in considering projects, which for engineering students, are critical pieces of work. They provide a student the opportunity to integrate the knowledge, practice, technology and application that they have learned across modules, and demonstrate their engineering ability in the production of an engineering artefact (Lipton, 2010). At honours degree it might typically be worth 20% of the marks for the year. As a guide to their importance, I can say, based upon my industry experience, that the most likely interview question a new graduate engineer will get is 'tell me about your project'. Where engineering student projects were raised by my interviewees (four), the language of levels was integral to the discussion.

Gordon differentiates projects across years using the language of levels: "*is this a level 7 project or is this a level 8 project*". He uses levels in describing the essential metacharacteristics of student projects, stating that "*a level 7 student might need a statement of requirements to do what she does*" whereas "*a level 8 student would be able to get [...] a* general statement of [...] design a bread-maker [...] that can wake a person up with fresh bread'."

For Bill, "a level 6 project is year two", which involves getting the student to build, test, and verify an engineering solution to a problem. He described teaching students "this is how you would do a project, you just need to follow this in doing, you build something, you get it working". He considers projects more self-directed at level 7, with this implicit in the level. Finally, at honours degree level, Bill states "you're very much on your own", i.e. projects are very much self-directed at this stage, with this implicit in using the term 'level 8 project'.

Bill described the relationship *between* these different types of project in terms of the language of levels. He views progression as having changed the nature of what it is being delivered, ensuring the project supervisor establishes linkages with the student in terms of what went from one year to the next. Projects at levels 6, 7 and 8, are all part of "*one system*". Thomas similarly describes projects at different stages of a programme in terms of what can be expected at different levels, and the guidance the supervisor will give.

Not all agreed that projects were clearly differentiated by the level concept, with Alan stating that for "*project work, I have to say, there is a significant blurring of lines between level 7 and level 8*". However, he still considers this in terms of the language of levels.

To conclude, although a relatively small number of lecturers used the language of levels to discuss engineering student projects, those who did were using it in considering the depth a project should have, and the relationship between projects at adjacent levels. This is a key point related to the instructional discourse, which I was to revisit in the focus group.

7.5.3 Knowledge

Lecturers' acquisition of new knowledge, their recontextualisation of that knowledge into curriculum, and their related pedagogy, are key aspects of academic practice. Eight of my interviewees discussed knowledge in terms of NFQ levels, with for some (two), levels used to represent or designate the knowledge.

Alan is comfortable using levels to represent types of knowledge, stating, in relation to graduates who were proficient in practical skills, that *"they would have struggled with [...] level 8 theoretical concepts"*. In acquiring knowledge to teach at a particular NFQ level, Alan
states "I am teaching this at level 6 so I need to learn it to that level". In the recontextualisation of that knowledge into teaching, Alan states that he would "learn as much as I can about it myself and then I adapt based on the level I am trying to teach it at".

Bill was very conscious of the level he would be teaching at when he acquired new knowledge, describing that new material "had to be learned enough to be able to teach it, maybe not to be an expert in it, but enough for the level that you were teaching".

For some interviewees knowledge is considered from the point of view of assessment, e.g. Isambard, describing post-NFQ, of syllabi and LOs being written to "allow for assessment at a particular level". Edwin considers that, as external examiner, he evaluates the assessment of knowledge in terms of "the level that it's pitched at".

Vitruvius uses NFQ levels in terms of revising his notes, and, similarly to Alan and Bill, regards the NFQ level as helping to define the knowledge required to teach. Also, mapping on to Isambard and George's views, he sees the role the NFQ levels plays in the recontextualisation of knowledge into syllabi. Ayah also sees the role of the NFQ in curriculum design, recounting how she found it restrictive having to design level 8 programmes using the NFQ.

Considering knowledge in terms of levels causes me to reflect on how we regard knowledge as engineering educators. On the one hand, I have a concern with the contingent nature of the use of NFQ levels to provide an epistemological perspective for engineering knowledge, rather than this being something with a disciplinary basis. More commonly, my research participants viewed their recontextualisation of knowledge in the curriculum in terms of the level it was to be taught at. Bernstein's (2000) concept of recontextualisation identifies the role of pedagogues in selecting and sequencing disciplinary knowledge for the curriculum, and the use of levels to consider knowledge, has become, for some of my research participants, part of this process. I will consider the implications of this further in *Chapter 10*, when I consider my research participants' views of knowledge, skills and the curriculum, and also in the discussion in *Chapter 11*.

7.5.4 Modules

Modules are designed, mapped, and validated at a particular level on the NQF. Referring to modules as levels, is thus to be expected, but does illustrate another of the multiple ways that 'level' is used in common parlance by academics. I observed this usage in five of the

interviews, which I illustrate below with examples.

Of his early years teaching Alan states "I taught computer networks, cloud computing architecture at level 9 and I taught computer networks at level 8 for BSc in software design". Alan also discussed lecturers "inheriting and then owning certain modules at different levels".

As noted earlier (p. 140), Rudolf, as an academic manager, discusses modules with new staff in terms of levels to guide them in their approach to teaching. Rudolf also refers to modules he teaches himself in terms of levels, talking, for example, of "*a level 8 module that I am teaching online*".

7.5.5 Learning Outcomes

Not surprisingly, some of the interviewees (eight) use the language of levels in relation to LOs.

Mirroring the earlier discussion on forms that projects take at the different NFQ levels, Ayah describes differences between the LOs of different years of a course in terms of levels:

the biggest jump is from level 7 to 8 where you're now going into more analytical type phase and often I think students struggle an awful lot more when they move from 7 to 8 [...] there probably could be a softening of lines between the LOs at 7 and 8 in order to equip students to be more self-directed learners at the level 8 stage or possibly that we could have a little bit more at the level 6 rather than very kind of prescriptive type modules.

Alan has a related stance comparing "what you're supposed to be able to achieve on a level 6 LO in one subject versus a level 8 or 9 in the same subject". Alan prefers "the higher level LOs because it is [...] different, it is closer to what I am trying to describe here in how to think". For level 6 LOs, which are often quite skills based, he feels that "sometimes the lower level LOs [...] they don't explain" the context fully, i.e. "what skill is this proposing".

George feels that LOs assist lecturers: "when you have taught for a while you do have a feeling about what level they should be at the end of the course [...] the LOs are quite useful because they can say what, how much in depth the actual content is". Isambard regards LOs, combined with levels, as making it clear to a lecturer what they can achieve within the time available to deliver a module. Kees is even clearer that the QQI Engineering Standards give direction to the lecturer "you could look at a cell in the matrix in terms of programme outcomes, level and read what you were supposed to do". This sentiment is mirrored to an extent in a comment by Vitruvius, in referring to programme validation documents: "you can go to those Word documents, show what the outcome must be, and then delve into the framework and it'll tell you the level at which you might challenge, the level of independent learning".

Kees concern was with how LOs and levels guide the lecturer, whereas Vitruvius's considers this should be in terms of how they can teach the student. Rudolf holds a more student centred view, regarding that "you help the students achieve LOs at that particular level".

7.6 Criticisms of the Effectiveness of the Language of Levels

A disquiet regarding the language of levels is visible in some of the interview transcripts. Rudolf's perceives part-time lecturing staff "*don't understand, level 6, level 7, level 8*", ascribing this to the NFQ not being part of the context of their non-academic lives. He expressed similar concerns regarding industry understanding of NFQ levels.

Henry, in discussing the use of the terms level 6, 7 and 8 to describe graduates, states that:

I think we have muddied the waters [...] I've had employers ask me [...] explain [...] to them what's going on because they are looking for a particularly person and they're saying what should I look for [...] they may be looking for a technician.

As discussed earlier, George did not feel describing a graduate as a level was an appropriate reflection on their achievements. A similar concern was expressed with the representation of a PhD using the phase 'level 10'.

Concern was also expressed with the NFQ principle that the mapping of awards to levels can be used to compare awards directly, without considering the awarding or educating body.

7.7 The Focus Group and the Language of Levels

The focus group discussion led to a deepening of my understanding of the language of levels, which I also highlight in considering the effectiveness of the research methodology. In this stage of the fieldwork I sought to enable a group conversation that would shed more light on the phenomenon I had observed. I presented a preliminary finding that the language of levels had become part of our pedagogic discourse, used to refer to both policy and academic practice. To support this I illustrated the manner in which my interviewees had used the terminology when I spoke with them individually. The focus group agreed that the use of the language of levels was pervasive in our education system. Rather than exploring in depth how this terminology was used, the principle discussion that emerged was of reasons that it had become so important, for which the following key points of consensus were established.

An overarching theme was that the focus group felt everyone is using the language of levels, and in order to be understood, individual academics must use it also.

The use of the terminology by students, who seemed to have adopted the level of the course they are doing as part of their student identity, was regarded as a key driver:

I use it now, because the students use it. Because they tell me I'm doing the level 7, or I'm doing the level 8. I would be old school, and I'd still call the level 7 the diploma and I have to correct myself, the ordinary degree [...] They're talking in numbers now.

Student usage was seen to be influenced from two directions outside of higher education: "that's coming from the career guidance teachers; you do a level 7, you do a level 8"; and also "It's feeding in as well from industry [...] similar grades in Company X, you know where, they have a system of levels. That's right yeah, for their engineers"

The group felt that the language of levels brought clarity to understanding the difference between an ordinary and an honours degree. Indeed, they felt that it is "confusing now when you try to talk about our current system and you don't use levels; it's a lot easier to say, 6, 7, 8". The focus group consensus was that it is less likely that master's and PhD's would be described as NFQ levels than undergraduate awards, speculating that this might just be because they did not "have many level 9s and level 10s" in their IoTs.

In discussing the international mobility of graduates, the focus group was concerned that the perception of an award in the "*international arena*" was important, but where the terms level 6, 7, 8 etc. to describe programmes have little meaning. Indeed, even Ireland's NFQ levels, referenced to the EQF, are differently numbered than the EQF levels. One group member highlighted the impact on the work they do for EI. They stated that "*I would use the levels secondary to the primary description of a degree. For example, BSc NFQ level 7, simply because in whatever work you do, you look at how it's going to be perceived in the international arena, rather than the national arena".*

Further disquiet was expressed by one participant, who, although accepting of the usage of the language of levels to describe awards, harked back to formal naming conventions: "although I speak in levels now, I still prefer the old system, and I think we have simplified education by giving it numbers. I would prefer to refer to as a bachelor in science or a bachelor of engineering, or an ordinary degree, as opposed to just going numbers".

There was a sense from the focus group discussion that they considered the NFQ improved access to education, inculcated a sense of inclusivity and facilitated recognition of IoT engineering qualifications: remarking *"it's putting everything on an even keel"*, as earlier noted in *Chapter 6*. The use of the terminology of levels, as part of the NFQ, was seen by the group to counter elitism: *"I think the terminology changed; the old system […] was an elitist system, it was for an elite portion of society"*.

Related to this concept of countering elitism, the focus group considered that the use of the language was a reinforcing influence regarding progression: *"if you just start talking about level 6, 7 and 8 qualifications [...] it's also emphasising the progression between them [...] It gives it that greater sense of you simply went as far as you could do"*.

Supportive of a finding from the interviews, the focus group considered the language of levels assisted in clarifying the relationship and distinction between student projects at different levels: "In the project presentations, we kind of say that's a bit level 7, where's the level 8'ness in that. [...] It's also just easier to use, it's just easier to say". A consensus was that projects for different years are defined by different levels, which aids and clarifies communication.

I attempted to explore the relationship of the usage of the language of levels to the regulative discourse of the ORF through discussion of its usefulness in engagement with quality assurance. The focus group, in recognising that the language of levels had a role in this, steered the conversation to the academic practice of course design. One focus group member commented that *"it makes it easier to design programmes, or to lead a team"*, and another felt that the language of levels was effective in mapping from programme LOs to module LOs in curriculum design.

The focus group discussion confirmed my preliminary finding that the language of levels had been adopted by my interviewees as part of their pedagogic discourse, and also, offered insights into why it might have occurred. It also added another facet to the fieldwork, in which I was both a participant and observer. My observation of the use of the language of levels in action in a group discussion regarding our engineering education system provided direct evidence of engineering academics, including myself, using it as an effective means of communication in our pedagogic discourse.

7.8 Conclusion

A policy goal of the NFQ is to create a culture where lecturers, students, and employers use the nomenclature of the framework as a language of engagement with higher education (Collins et al, 2009). A finding is that the analysis shows that my research participants have adopted a language of levels as a consequence of the NFQ, as part of a language of policy discourse, and as a language of academic practice. A summary of the use by academics of the language of levels is captured in *Table 7-1*. Whether academic management or lecturers, the use of the language of levels to engage with matters related to policy discourse is quite pronounced. This reflects Kenny et al's (2015) finding that Irish academics are keenly aware of the impact of government policy on their role, and their ability to perform it effectively.

Edwin considered that "everybody's talking with these levels", with its use pervasive in the interviews and the focus group. A finding, enabled through the focus group discussion, is that the language of levels is so pervasive and important that not only *can* it be used to discuss academic matters but rather it is *essential* that academics do so in order to be understood. The level concept has become so integral to our pedagogic discourse I believe it would have been quite difficult to usefully analyse the fieldwork without developing the understanding of the phenomenon of the language of levels I have presented.

A further finding is that the language of levels is seen in a positive manner, enabling effective communication between academics, and supportive of some of the aims of the NFQ. In particular, it is seen to support improved access to education, in doing so countering elitism in the system. It is also seen to be a significant factor in emphasising progression for students and academics, the impact of which is considered in the next chapter of the analysis.

Chapter 8 Progression in IoT Engineering Education

Nothing would be done at all if a man waited until one could do it so well that no one could find fault with it.

John Henry Newman (1851, part 7, para. 1).

8.1 Introduction

The importance my research participants ascribe to student progression in IoT engineering education was a topic they chose to emphasise, highlighting its relationship to other interview topics. This chapter builds upon an earlier finding that the level terminology of the NFQ foregrounds progression for my interviewees. Through this I identified a perception amongst a selection of my research participants that progression stresses for them the importance of ensuring earlier years of all undergraduate study primarily prepare a student for later years, whereas previously the delivery of engineering higher certificate, national diploma, and honours degrees programmes was more compartmentalised.

This chapter explores in more depth my research participants' perceptions of how progression has influenced their approach to teaching of level 6 higher certificate and level 7 ordinary degree students. This leads into analysis of the manner in which they consider that IoT engineering technician education has changed under the influence of outcomes based education (OBE). An earlier, related, finding was that the language of levels, in emphasising progression, acts to improve equity of access to education and counter elitism, and I further consider my research participants' views of the relationship this has to progression.

To begin with I analyse progression through Bill's eyes, who, in the first interview of the fieldwork, wove progression, and the effect he perceived it had on the engineering education that he had made a career, and on him as an educator, in a powerful narrative. Commencing the analysis in this chapter with Bill's interview is more than a matter of presentation: in gaining his perspective of the impact of progression I developed a new understanding of this aspect of the National Framework of Qualifications (NFQ), which I use in framing my further analysis of the fieldwork regarding progression.

8.2 Bill's Interview and Progression

Bill described being on a personal journey throughout his career as an educator, stating that "over the years then I think I've reinvented myself I don't know how many times". He explained

that he was always considering "how can I do my job better. And even at this stage I wouldn't have even remotely exhausted how you could be a better lecturer or a better educator". In relating his experience as an engineering educator in the Regional Technical Colleges (RTCs) in the 1980s and 1990s, Bill described how a national certificate could lead to a "decent enough sort of technician level job in some industry". A further year to gain a diploma could allow a graduate "to be a really good technician", who could then choose to enter a 2 year add-on engineering degree, at which point education started "to prepare someone to be an engineer".

Bill used the term 'sudden' in relation to the introduction of the NFQ, the consequent change in how students were educated, and the progression opportunities it introduced; highlighting the strength of the change he perceived. Bill described how the NFQ introduced a new structure to Institute of Technology (IoT) engineering education: programmes (and students and graduates) became defined in terms of level 6, 7 and 8, as opposed to previously being programmes for technician, advanced technician, and professional engineering students. Progression became an item of "*a bigger relevance and a newer relevance*" compared to pre-NFQ days. Students of higher certificates (level 6) began to be seen as destined to do ordinary and honours degrees, rather than these sub-degrees being ends in themselves. Bill associated the NFQ not just with increased progression opportunities for students, but as making such progression automatic.

Bill's pedagogy, and, in his opinion, those of his peers, changed as a consequence, and he "started doing things differently". Bill perceived that the progression that was now built in from years two to three to four (level 6 to 7 to 8) caused him to link modules taught in earlier years seamlessly to those running in later years of programmes. He considered he, and his peers, now "taught in terms of progression". He explained his view that "the minute you put in a progression system from FETAC, HETAC and levels, then you immediately start changing the way you're thinking. Well, I did anyway".

As articulated earlier, Bill used student projects as a particular example (See p. 163). Projects structures became clearly defined from one level to the next: the make/build/test/verify of level 6; the more self-organised but still directed level 7 project; and the largely self-directed projects of level 8; with successful completion of each the "*next stage of your progress*". For Bill, the level of the project came to define the pedagogy of the lecturers in relation to

projects. Bill, influenced by progression, changed to view projects across years as being part of a single system, as opposed to separate modules on different programmes. For example, a student taking a level 6 project in year two is in *"training for year three"*. Bill, an experienced lecturer, regarded this change in emphasis as *"interesting and new for me"*.

In discussing the manner in which level 8 projects prepare a student for employment, Bill feels "you're going to have to present stuff on time to your boss, and they're not going to be impressed if it's not done on time. So right now is your time to shine on that one". However, preparation for employment was only mentioned by Bill with regard to level 8 projects. It seems that Bill now expects students to remain in education till completion of an honours degree, with students on higher certificates and ordinary degrees being educated from the outset to be professional engineers.

In concluding the interview Bill re-stated his views on the change he had seen from the introduction of the NFQ: "we went to this fluid system that goes through all the level 6, 7, 8's and 9's, and then progression is almost a paramount thing". He explained that, as opposed to the previous rather elitist system, people who previously would not have been educated to the same extent are now "getting the chances that they wouldn't have (previously)", albeit it raised a concern with him on maintaining academic standards.

An overarching theme in Bill's interview linked progression and the NFQ to a change in his perception of what he was teaching his students to be and how he was teaching them. A number of related sub-themes emerged from my consideration of Bill's interview:

- how progression to level 8 has become the aspirational target for students;
- the effect of progression on technician education, including the effect on higher certificates;
- how progression has effected pedagogy;
- the manner in which progression has effected how student projects are regarded by engineering academics.
- progression and equality of access to education;

In the following sections I develop these sub-themes through examining how they were viewed by the other interviewees, and how the focus group related to them, with the exception of the effect on pedagogy which I consider in *Chapter 9*, and the effect on how engineering academics regard projects, which has already been considered in *Chapter 7*.

However, Bill's interview leads to more than just a set of sub-themes to guide my exploration of the fieldwork. It is noteworthy in itself in illustrating how an individual academic, in regarding their career as a personal journey as an educator, perceived the NFQ, and progression in particular, as such a defining, positive influence on how he approached being an educator. Bill's interview shows evidence of his pedagogic identity being influenced by the skills focus of the NFQ, expecting of the students that *"when you get to level 8, you're certainly expected to have very detailed writing skills for reports, organisation, timetabling, to make sure you get the thing done in time"*. His interview also shows evidence of his pedagogic identity being influenced by the student-centred, lifelong learning policy drivers, e.g. he is trying to find *"new ways of doing things that were more relevant to students"*. An earlier finding was that the change in pedagogic identity of academics as the RTCs expanded their role was consistent with Bernstein's concept of pedagogic identity changing under the influence of a changing pedagogic discourse. It is a further, related, finding that Bill's interview provides evidence of an engineering academic embracing the change in their pedagogic identity brought about through the influence of the NFQ, and progression in particular.

8.3 Progression to Level 8 as the Aspirational Target for Students

The analysis suggests that over half of the interviewees (eight) perceive that level 8 has become the aspirational target for all new entrants to IoT engineering programmes, whether initially registered on level 6, 7 or 8.

Rudolf considers the concept of progression as well-understood by staff and students. However, he regards this as not the case for external stakeholders, stating, for example, that *"parents struggle with it"*, where they want their children to go straight into honours degrees, rather than progressing through the ladder system. Rudolf similarly felt that employers²⁹ struggle with the concept.

Kees, in regarding progression within the ladder system, expects students to aspire towards the end-goal of an honours degree, "stacking" up the intermediate qualifications en route. In some cases, the interviewees felt that, from the students' perspective, it was an imperative, rather than an aspiration, to progress to level 8. Thomas regarded level 8 qualifications as the "end goal" of all students, and that "if you don't make it [...] your exit award is your ordinary

²⁹ See p. 146.

degree, you couldn't cut it". This sentiment is mirrored in Edwin's comment that "*a lot of students now, they feel like if they have only a level 7 degree, they've nothing*".

Edith recognised the opportunities that progression offers to students, but felt that progression had introduced some educational issues, proposing that not all students were placed to engage well with level 8, and that *"there should be a more trade element for a lot of those students [...] They're going in to be a technician, so they need the practical skills"*.

Ayah questioned the manner in which studying at one level of the NFQ prepares the student for the next, in the context of whether a student who commences a level 6 programme is, from the outset, being prepared for a professional engineering qualification. Ayah considers that achieving awards at different NFQ levels requires particular, and "completely different", mind-sets in the students, which mirrors the position on the differing epistemologies of technician and professional engineers articulated earlier in the literature review (Shay, 2012; Wolff, 2015). She expresses a concern that "as the student progresses up the ladder more, it can leave them quite vulnerable in that now they feel they're exposed and they're lost at level 8, 9, 10 if they're not applying the LOs approach". She feels that the education required for associate engineer, which is quite prescriptive and aimed at "operator technician level", needs to be adapted to prepare students properly for progression to level 8 where they can aspire to professional engineering status.

In the focus group I queried whether the level 6 and 7 qualifications (are) becoming merely precursors to professional engineering qualifications (level 8)? This sentiment was supported by the focus group discussion, with a consensus that "we now see a level 7 as a person on a journey to a level 8 as opposed to being an exit award³⁰. So, we're not training technicians anymore". One focus group member went so far as to say that a level 7 graduate appears to be someone "who couldn't make it to level 8 for whatever reason, they consciously make a decision to exit at level 7, saying 'Great, I'm not a qualified technician, that's my ambition".

The fieldwork provides evidence that a significant proportion of my research participants view level 8 as the aspirational qualification for all undergraduate students. Some of the

³⁰ Completing a 3 year engineering level programme in an IoT usually allows a student to receive an ordinary Bachelor of Engineering degree. However, completing the first 3 years of a level 8 programme may entitle them to receive an exit award at level 7 if they so wish, although it is unusual for them to take this exit.

interviewees expressed concerns with smoothly moving, from an educational perspective, between the different levels, feeling that the prescriptive nature of the earlier levels does not prepare a student well for level 8. Others queried the appropriateness of what is being taught at the lower levels as a consequence of progression.

The difference, indeed conflict, between level 7 and level 8 graduate attributes was also raised. The importance of this is considered further below.

8.4 The Effect on Technician Education

A sub-theme that emerged from Bill's interview was that, post-NFQ, "you felt that even at third year level, you were teaching somebody to be an engineer, rather than a technician". In exploring this, the analysis revealed the perception of two-thirds (nine) of the interviewees that progression has led to a change in higher certificate (level 6) and ordinary degree (level 7) engineering technician programmes towards being regarded as pathways towards honours degrees. This is consistent with a survey of first year level 7 engineering students in the DIT which revealed over 75% already intended, on completion, to transfer to an associated level 8 degree (Llorens et al, 2014). It is further consistent with the data on IoT engineering enrolments presented in *Figure 2-4*, which shows a shift in emphasis to honours degrees.

It has already been argued that this change might be attributed to the increased affluence of the country (Barry, 2007), and to the benefits students see of holding an honours degree to allow them to engage with careers in the knowledge economy (Skilbeck, 2003). However, it may also be at least partially an artefact of the NFQ concept of progression. This section explores the impact the interviewees perceived from this refocusing of IoT engineering education towards honours degrees.

Alan discussed students who, he felt might make very good technicians, but would have struggled with more theoretical level 8 topics. He felt that the technician identity has been diluted from its practical focus through the theoretical topics introduced in level 6 and 7 to support progression, with *"the IoT sector in general moving away a little bit from developing these kinds of skill sets in the lab"*. Thomas similarly saw diminution in the skills focus of the technician identity, viewing that, rather than a level 7 graduate being equivalent to the old National Diploma, *"It doesn't give you technician level"*.

Thomas feels that the technician focus of the national diploma has been lost, with the ordinary degrees that replaced them taken by students for the purposes of progressing to level 8 rather than to secure employment. He sees El's influence also, considering that accreditation criteria has led to an overemphasis on maths in level 6 and 7 engineering programmes, in order *"to tick the box for El"*. In his view El accreditation is positive, as is progression, but he questions if we are *"serving the needs of our students by designing courses with such a high level of knowledge content in it, mainly in the maths and they're never going to use it"*. Thomas proposed that we should be emphasising engineering practice at level 7, and that, to teach technicians, we need to be less theoretical and, *"do more skills-based learning"*.

Edith related that, over the course of her career in the IoTs, programmes have become less technician focussed and *"much more academic focussed"*. She regards current level 6 engineering programmes as being *"an academic level 6 as opposed to a technician"*.

Henry feels that the de-emphasising of skills is also related to the lack of funding of the IoTs: blaming the Department of Education "*because it takes funding* [...] *it takes facilities*", an issue also identified by Edith³¹. Alan also regarded the reduction in skills emphasis as being funding related, "to keep people in the system as long as possible" (the IoTs are funded (partially) based upon student numbers (HEA, 2014)). Alan also feels the change in emphasis on practical skills is related to pressure to reduce contact hours. This became apparent to Alan during programme reviews, relating that every time there is "an industrial advisory board or some sort of external panel coming in they are going, 'there is too much contact in here'".

I have earlier discussed the views of some of my research participants who associate the NFQ with the massification of higher education (Hazelkorn et al, 2015). The views expressed above leads to a concern that the reduction in practical focus has come about through a 'credentialism' (Allais, 2014), where an over-emphasis on qualifications for all leads to "*an ever-diminishing relationship with the skills needed for specific jobs*" (p. 9). Whilst Henry, similarly to Alan, felt that where formerly for students being trained to be technicians there was an emphasis on engineering practice, in the new post-NFQ programmes the emphasis is on analytical subjects. However, the engineering focus of such programmes is not being

³¹See p. 145

questioned, rather the balance of knowledge and skills being taught to students. Edith viewed the cause of level 6 and 7 programmes becoming more academic as policy driven. She considers it a *"selling point"* for foreign direct investment (FDI) to have a population that is seen as highly educated, and that this requires more university and IoT level 8 graduates. Edith, however, doesn't *"think it's right that everybody has to go to university [...] or an IoT"*. These perspectives imply a professional engineering focus to level 6 and 7 engineering education, with Henry stating *"they're not technicians"*.

George identified the change in skills emphasis as related to industry requirements. In considering whether he felt current level 7 graduates were equivalent to the technicians that used to be produced with the old national diploma qualification, George discussed the changing roles of technicians in industry. Drawing from his own industrial experience, George gave a detailed view of the type of work technicians used to do:

years ago the technicians when they came out would often be doing things like board repair, they would have a very good understanding of using tools like oscilloscopes and you know, and they would take instructions. They do have a fairly high technical level. Like, when I was working in xxx the technicians were able to debug boards, they would be able to run code, change codes slightly, run tests and debug and be able to figure out what was wrong with different chips or, you know? They were very skilled.

He feels that the technician role may have become less technical, ascribing this change to industry requirements, "most of that work is gone now", rather than the NFQ. He views the technician role of today as ill-defined, further stating that "I think the roles have changed but it is very hard to figure out what their role should be". Henry feels that the renaming of the National Diploma to Ordinary Degree (level 7), has "muddied the waters" with employers, who are now not always clear what it is they are looking for in a graduate employee, and may need to be advised that "they may be looking for a technician"

Rudolf also gave consideration to employers' views on progression. His experience is that employers understand the honours degree concept, but not the concept of progression across levels. He did feel that some employers, particularly HR departments, see a role for technicians, they *"understand the level 6 space because they may be looking to develop programmes where they can up-skill craft-workers or operatives"*. However, Rudolf feels employers do not understand level 7 awards. He does feel that this can be resolved through working with industry, and is part of an industry/academia group that is carrying out *"work* on progression and to create pathways, easily understandable pathways for the industry partners and for potential staff within those industry partners to look at".

The general thrust of my research participants' views suggest that they now regarded level 7 ordinary degrees in engineering as preparatory programmes for level 8 honours degrees, with an increased emphasis on theory at the expense of practical skills.

I explored this sub-theme in the focus group, where we considered education through a vocational lens, discussing whether progression has influenced the balance of skills vs. theory on level 7 and 8 programmes. There was broad consensus in the focus group that IoT engineering education had become less vocational, with one participant referring to *"gentrifying"*. The focus group recognised the conflicting educational requirements of technician and professional engineering education, and the difficulties in trying to satisfy both simultaneously. One participant queried if *"somebody who's in third year […] are they both a fully-fledged technician and half way on the road to being an engineer as well?"*.

The focus group did not view progression as the only cause of change in technician education, and offered further insights. One view was that the learning of skills, which involves, in many cases, repetition until the task can always be completed correctly ("*repetition is the natural instinct of learning*"), does not map easily on to the³² learning outcomes (LOs) approach, where "*if you have tested an outcome once don't test it again*". The constraints of semesterisation were also regarded as having a strong effect "given all these LOs over a semester that you must tick, you must hit all of those, so we can't do the repetition".

To summarise, level 6 and 7 engineering programmes are seen as less-skills based and more academic than the certificates and diplomas they replaced. Indeed, Kees noted that holders of national diplomas were not allowed to replace them with transcripts indicating degrees:

We would have had hundreds of enquiries from people saying 'I have a diploma from IoT from 1975. Can I now get my ordinary degree?' [...] The answer to that was very simple, it was 'No, you have a qualification, it's a great qualification, and congratulations'

The fact that the pre- and post-NFQ awards are not seen as directly equivalent by awarding bodies is an indication that it was expected some differences would arise. However, I will

³² Of course a LO could call for a skill to be successfully repeated a number of times.

argue later that the manner in which what were previously higher certificate and national diploma programmes would change their focus was not anticipated.

I have already contended that progression in the NFQ acts to reconfigure the stratification of engineering education into technician and professional engineering³³. The analysis suggests progression has been part of a transformation of the focus of IoT engineering education towards more academic, less skills focussed, engineering qualifications. Someone who graduates with an ordinary degree can be regarded as not having managed to make the engineer grade, but not necessarily educated to be a technician. Several of the interviewees related concerns that the emphasis on progression was disadvantaging those who wanted to be technicians, which concern the focus group shared.

Although it as a factor, the interviewees did not fully ascribe this change to progression. Other perceived causes were: the influence of EI accreditation; changing industry requirements; decrease in funding of engineering programmes; and the use of LOs in engineering education.

The fieldwork suggests that, influenced by the NFQ concept of progression, alongside other factors, level 6 and 7 programmes now tend to be regarded by academics as preparatory programmes for level 8 degrees, with a consequent increased emphasis on theory over practice in such programmes. The fieldwork also suggests that, consequent to this, progression has contributed to a diminution in the skills focus of the technician identity in engineering education, with level 6 and 7 programmes now more academic and less vocational.

Some of the interviewees discussed whether progression was appropriate for all students. Although it is an aim of the NFQ to provide all students with the opportunity to progress to honours degree, George feels they may not all be capable, "the standard of students that you're getting in in first year are finding it quite difficult". Related to this, Edith regarded progression for those "that want to progress further up the ladder" as disadvantaging those who want to become technicians. Ayah similarly viewed progression as not necessarily appropriate for all: "not everybody wants a degree and not everybody wants to be a professional engineer [...] where would we be if there were no operators and technicians". There is suggestion of a negative socio-economic impact if all engineering students take the

³³ See p. 14

professional engineering route, resulting in a consequent technician shortage, which would have serious implications for industry and attracting FDI (HESG, 2011).

8.5 The Effect on the Higher Certificate

The Higher Certificate in Engineering was the original building block of IoT engineering education (McLaughlin, 1999), and, in considering the views expressed by my research participants with regard to it, deserves specific consideration. My research participants considered that progression was seen as contributing to a decline in enrolments to higher certificate, but that there were other factors contributing to this, particularly industry demand.

Bill viewed that "*if somebody has a certificate in the 1980s, early 1990s, you could probably still expect to get a reasonably decent enough sort of technician level job in some industry*". However, in more recent times he saw a reduced demand for technicians with higher certificates. George also sees less industry demand for the higher certificate, and that now for technician roles you "*need level 7s*".

George also feels that the higher certificate is downgraded through being referred to as a level, stating that *"level 6 sounds like it is nothing"*, and Edwin views that *"higher certs, unfortunately they are fading off"*. He feels that employers no longer value them: *"they don't say I'm looking for a higher certificate. Now they look for a BTech*³⁴".

Thomas similarly believes the higher certificate has become defunct: "they appear to be gone", considering that "it's as if the state doesn't want them anymore". However, educational policy (HESG, 2011) is to ensure that the IoT sector continues to produce level 6 graduates. Kees, in referring to pre-NFQ policy from the Department of Education to keep up the supply of technicians, regarded it as ineffectual as he could already, pre-NFQ, see a trend for students to stay in the system once they had gained a certificate, who "unless they drop out, were going on to the diploma".

Thomas believes the multinationals played a role in the diminution of the higher certificate, where they changed the game in starting to insist upon employing degree level graduates,

³⁴Edwin was referring to level 7 Ordinary Degrees, which are, on occasion, referred to as Bachelor of Technology (BTech). As noted earlier (p. 70), in the DIT level 7 degrees were Bachelor of Technology, a deliberate decision, one interviewee related "so we're not confusing them with Bachelor of Engineering".

instead of seeking to employ higher certificate holders. This would have had a huge influence on students' take-up of level 6 programmes. Thomas still sees value in the higher certificate qualification as "*a great stepping stone*" for someone to become a technician.

Edith similarly saw a reduction in the demand for level 6 technicians from multinationals, who previously up-skilled their workers to be technicians using certificate programmes in her IoT. However, this is no longer a common approach to up-skilling the workforce for multinationals, which requirement she feels has now become "*much more* [...] academic".

Edwin, similarly to Kees, considers that students perceive a lessening in the value of level 6 qualifications. Edwin perceives societal influences, which calls for "higher and higher degrees [...] in everything", which "is not for everybody". However previously someone who had the potential to become a very competent technician, i.e. who "could be a fantastic technician, able [...] to fix everything", would have been quite happy to do so. A technician, highly skilled and capable, is rewarded financially and mentally through their career.

A conclusion from analysing the fieldwork with regard to the higher certificate is that this is regarded (seven of the interviewees) as significantly less in demand than previously. Referring to *Figure 2-2,* it can be seen that, in fact, numbers of full-time level 6 engineering students had reduced in 2017 to 40% of their 2008 figure, with an even higher drop in those pursuing part-time programmes (*Figure 2-3*). The NFQ concept of progression seems to have been a factor, but, the perception is that industry requirements are a significant influence.

A finding is that, for those of the interviewees who expressed views regarding level 6 higher certificates, they regard them as significantly less important than they were pre-NFQ, and that, although progression may be a factor, this decline in their status and in enrolments in education is principally associated with reduced industry demand.

8.6 Progression and Equality of Access

Bill's interview provided evidence that he regarded progression as facilitating a greater equality and acting to counter elitism in engineering education. Similar sentiments were directly apparent in the views of two of the other interviewees. Edith saw progression as improving access and opportunity for students: "I do think it's nice and I think it's fair that there are students that you get that come in with very poor Leaving Cert, but have fallen in love with engineering and they get an opportunity to progress upwards". For Thomas "the NQF to me it was about access and progression", in particular:

people that really gained from that framework at the time appeared to be the apprentice guys, that they were given progression. So where they were, they stopped at their particular level with their apprenticeship and this framework allowed them to progress. So it took away that barrier, so it forced the hand of IoTs really to make progression for those guys, to accommodate them if they wished to progress.

This resonates with my own experience, where I have led the design of programmes with inbuilt routes for advanced entry for tradespeople into level 7 and onwards to level 8 degrees (TU Dublin, 2019a, 2019b, 2019c). The NFQ's level structure, and mapping of National Craft Certificates to level 6 of the NFQ, allowed us to make the argument at engineering degree programme validation for advanced entry of qualified tradespeople to level 7 programmes.

The issue of elitism in education, and improved access to education, arose as a significant topic in the focus group, as has been discussed in the earlier chapter on the language of levels. To re-iterate in summary form, the focus group felt that the NFQ and progression helped provide more equitable access to education, and improved opportunities with regard to how far an individual could take their education.

A finding, based upon the evidence provided from the fieldwork, which has contextualised my own experience of the design of engineering programmes, is that my interviewees consider the concept of progression as a specific feature of the NFQ that acts to improve access to education.

8.7 Conclusion

The fieldwork suggests that progression has contributed to important structural changes in IoT engineering education. The old pre-NFQ certificate and diploma technician qualifications have been replaced with new level 6 and 7 qualifications, but the fieldwork reveals concerns that they are not equivalent. Given they are regarded as principally pre-cursors to honours engineering degrees, the skills focus has been downgraded in favour of a more analytical, theoretical, education. Considering *Table 8-1*, the rows representing *"Progression to Level 8 viewed as the Aspirational Target for Students"* and *"NFQ has influenced Technician Education to become more academic"*, appear to be different sides of the same coin. Viewing them as one, then eleven of the interviewees regarded progression within the NFQ as having changed the nature of technician education to having a more academic focus.

This change in the theoretical/practical weight of level 6 and 7 programmes is also seen to be related to industry influences, funding difficulties and pressures to reduce class contact time. El accreditation of programmes was seen as influential in changing the balance of theory and practice of level 7 programmes by one interviewee, who perceived a tendency for accredited programmes to have increased theoretical content.

	Research Participant														
	Isambard	Ayah	Kees	Rudolf	Vitruvius	Bill	Thomas	Gordon	Edwin	George	Edith	Henry	John	Alan	Grace
Progression to Level 8 viewed as the Aspirational Target for Students															
NFQ has Influenced Technician Education to become more academic															
The decline of the Higher Certificate															

Table 8-1 Influences of Progression on Engineering Education in the IoT Sector

A significant, policy-related issue emerges through the voices of the research participants, which suggest that progression is another factor, alongside the skills focus of government policy, El's influence and industry requirements, contributing to structural changes in the education of engineering students at level 6, 7 and 8. The literature associates a skills focus with the use of OBE (Young, 2008; Young & Muller, 2010; Allais, 2014; Corbell, 2014), and government policy promotes skills in higher education (HESG, 2011; DES, 2016). The missions of the IoTs included a focus on sub-honours degree qualification to help implement this policy focus. However, engineering academic staff on the ground, whilst recognising this skills focus, feel a stronger counter pressure from progression for more academic oriented programmes. A finding of this thesis is that the views of my interviewees regarding progression, amongst other factors, as contributing to the increased academic focus of technician education appears to be in contradiction to government policy promoting a skills focus for higher education. This will be explored more in the analysis in *Chapter 10* of the views of the interviewees regarding the balance of knowledge and skills in curriculum, and also in the discussion.

Chapter 9 The Impact on the Pedagogy of Engineering Academics

The self is not something ready-made, but something in continuous formation through choice of action

John Dewey (1916, chap 26, para. 8).

9.1 Introduction

I earlier identified that an important line of inquiry for my research would be to investigate the effect that outcomes based education (OBE) might be having on the pedagogy of my research participants (Havnes & Prøitz, 2016). The literature review revealed the lack of consideration of pedagogy in the development of the national framework of qualifications (NFQ), and Engineers Ireland (EI) make no stipulations or suggestions regarding pedagogy in their learning outcomes (LOs) based accreditation requirements (EI, 2014). Collins et al (2009) reported little evidence of an impact on pedagogy in the early years of the NFQ, although noting some disquiet in higher level education that it might be curtailing flexibility in that regard.

I had expected, when I conducted the fieldwork some 7 years on from Collins et al, with the NFQ more firmly embedded in higher education, any impact would be more clearly discernible. This was borne out through the findings reported in *Chapter 6*, which included that a majority of my research participants regard the use of LOs as an implicit influence over their pedagogy, largely regarding this in a positive, constructive, fashion. This chapter builds upon this earlier analysis, examining the effect this influence has had, inculcating in my research participants a pedagogy focussed around teaching towards assessment of LOs. Although I have noted (*section 3.10*) that constructive alignment was advanced, during the transition to OBE, as an approach to link LOs, assessment and pedagogy, I saw no evidence in the fieldwork to contradict my earlier conclusion that it had not been influential. The majority of my interviewees were aware of the reorientation of their pedagogy, which could be characterised as moving from assessing what they taught towards teaching what they assess. However, even where they did not regard their pedagogy to have changed, the analysis revealed this focus.

I commence the analysis in this chapter through illustrating my interviewees' emphasis, in discussing their pedagogy, on the teaching of engineering practice. I then present a sample of

my interviewees' views to illustrate their focus, in this teaching, on assessment of LOs. I further analyse this theme through the voice of John, who was very self-aware of the impact on his pedagogy, and had reflected in some depth on the topic. Although very focussed on the assessment of LOs, John uses his identity as an engineering academic to give this context. Ayah's interview presents a managerial view of the change in pedagogic identity of engineering academics, in which she is somewhat critical of the prescriptive nature of the LOs approach. I make use of the focus group discussion on pedagogy to provide further insight.

9.2 An Emphasis on the Practical

My research participants all chose to contextualise discussion of their pedagogy in the teaching of the application of engineering, linking theory, skills and practical work, and relating, as they did so, their own approaches to that of their colleagues. I illustrate this with some representative views.

Bill gave an account of this full of engineering detail, a practical focus within a practical focus:

it's through talking to other colleagues. Finding out what particular chips will be used or if we have used something with micro controllers or something like that, what micro controller are we going to use, and what ones are people using. I suppose a good example of that would be the way we use a lot of Arduinos now for project work.

John described how he linked the teaching of engineering practice to the workplace from an earlier stage in a students' education, and the engagement he perceived from this approach:

I took a first year class and I structured the practicals where they worked in groups and I facilitated the groups and they produced a work product at the end. Really enjoyed it. Students really enjoyed it.

Thomas, who emphasised skills in his interview, highlighted the need to link theory and practice in a skills context:

there should be more criss-cross between we'll say a maths tutorial and a practical way of measuring temperature and integrating it and averaging it and things like that. See the application of a particular skill, of a particular bit of knowledge that you're trying to convey to them.

Edith explained "I try to make the subject as practical as possible". She does this so that:

the problems that they're working on they're real problems and you try and do it in class time. You know and I find that that approach, because mechanics is my subject and you could stand up all day and give them a lecture about how to do the analysis, but unless they sit down and actually try it, they don't have a chance. In the next section I use the views of my research participants to show what this focus on teaching means for them in our OBE engineering education system.

9.3 Teaching to Assessment of LOs

Isambard regarded the NFQ as demanding a particular pedagogical approach, even though not explicitly specified. An extract from his interview used earlier in a related context (p. 141) illustrates this in a powerful manner: *"if you are developing an assessment process for a particular module, you actually have to build in the pedagogy, the requirement that will enable the LOs to be achieved"*.

Vitruvius pointed out that pre-NFQ curriculum documents in his Institute of Technology (IoT) included a section on the teaching methodology to be used for particular modules, which post-NFQ, has been dropped. Despite feeling that including a teaching methodology was useful, he argues that under the NFQ it is not really necessary, as LOs make it clear what is to be taught, and how it is linked to assessment. Vitruvius describes a module as being written in terms of LOs, taught (*"communicated"*) to the students in terms of LOs, and assessed in terms of LOs. Kees provided an insight in discussing the simultaneous introduction in his IoT of semesterisation and the NFQ, where he related that academics *"had to figure out how to assess modules on a semester basis"*. The significance is that he emphasised not that academics had to learn how to *teach* modules using LOs, but how to *assess* them.

Bill felt that "LOs and the programme outcomes try to put a structure into how this fits together as an educational package as much as a technical package for an engineer", and that "LOs reflect the process" of teaching.

Henry was "very positive" about the use of LOs, and the effect on his pedagogy. He describes initially not quite understanding LOs, as he was focussed on "content" (knowledge as defined in the syllabus). He describes LOs as having changed his approach to teaching, becoming more focussed on assessment, particularly continuous assessment, of which he further states "I would have changed labs and I would have changed how I do the assessment of labs". He also feels that LOs have had a strong influence on his colleagues' approach to teaching, and made him and his colleagues much more conscious of the process of education.

He does see a downside, in that whilst some LOs are quite specific to a module, others "are very generic" and could be applied "to any module", raising connotations of a fundamental

concern as to whether generic LOs can be used to effectively describe discipline specific curriculum, (Young, 2008; Simpson & Jackson, 2003).

Rudolf, in discussion of leading and being involved with course design, described assessment of LOs as the main pedagogical consideration during module development, with any further considerations taking a back-seat. Rudolf considers staff actively try to adapt their pedagogy to the new academic structures they face (LOs, semesterisation) and that they are concerned with the effect that over-assessment may be having on student welfare:

we see lecturers struggling with how to manage the delivery and the creation of assessments [...] you mentioned semesterisation, modularisation, you know where people are moving from year-long to semesterised programme or module delivery and I suppose recalibrating themselves in terms of their pedagogy and their assessments and trying to avoid the student suffering from over assessment.

George regards the NFQ as a guideline for writing and delivering modules. As with many of the interviewees, George views that "LOs help to drive the assessment". However, George also makes it clear that, in his case, assessments are not fully driven by the LOs, but are based upon his own knowledge of what it is appropriate to deliver on the programme. Indeed, in writing a module he will "generate the LOs either as I go along or at the very end".

However, not all interviewees viewed themselves as influenced in pedagogical terms by the NFQ. For example, Alan describes his industrial experience as being much more influential on his pedagogy than the NFQ. On the other hand, regardless of his stated view, Alan describes that "*I do all the assessment based on the LOs*", and, for module design states that "I dictate the text of the LOs based on the verb" (from Blooms taxonomy).

When questioned on whether the NFQ is an influence on her pedagogy, Edith responded that "I don't think it has". She went on to describe her pedagogical approach as "I try to make the subject as practical as possible, so that [...] the problems that they're working on they're real problems". However, she also seems to focus on LOs in teaching: "You know you have to follow these. You've set the plan out. You've set the list of LOs and you have to basically cover those in your class, in some shape or form, whether its lab, lecture or tutorials or exams".

Edwin recognises some influence from the NFQ on his pedagogy, in that "*I have to make sure all the LOs are covered*". However, he feels he is largely teaching as he was before, albeit in a

more structured context. On the other hand in discussing the relationship between LOs and assessment Edwin described a teaching approach inextricably linking LOs and assessment.

Kees, in academic management and not actively teaching, considered that the engineering professional bodies are trying to improve the application of LOs based accreditation standards "by trying to modify how the material is learned by the students. So I think they actually do foster a more active and collaborative approach to learning". However, he questions if pedagogy should be prescribed, stating "whether a framework should, in some sense promote or require a particular active pedagogy, as opposed to passive, I don't know".

To summarise, a pedagogical approach of teaching to assessment of LOs emerges strongly from the fieldwork (fourteen interviewees). The next section, through analysis of John's interview, gives a detailed account of what this means for one individual engineering academic.

9.4 The Influence of LOs on John's Pedagogy

John's interview provides a compelling account of actively developing his pedagogy through his engagement with the LOs approach, which provided him with guidance that was previously lacking. Although John's pedagogy is tightly focused towards the assessment of LOs, there is no sense of someone slavishly following a teaching to assessment approach. Instead John has thought deeply since becoming an IoT engineering academic on how to work with a LOs approach to engineering education.

John feels that it imperative on him to use the NFQ and its LOs approach, recognising its regulatory function: *"society has put the NFQ in place right, we are hired at one level to teach both learning and assessment in the context of the NFQ"*. He relates that:

I have accepted the controlled [...] way society is structured', right? So, I am not at that level critiquing that control aspect where you could get into an argument about being in control. But what I am saying is, we have structured our society and I've benefited and I think you've benefited.

However he also sees value in the NFQ approach to teaching and learning: "*if you read the NFQ*, [...] *it is actually well structured, right and it makes some useful points*". As the analysis below reveals, John values aspects of the NFQ such as LOs, its view of knowledge, and its inclusion of teamwork.

In discussing how it has affected his pedagogy (which he terms enactment) he feels:

it gives me a context in which to enact what I want to do, right? But having said that, I like it because I have a range of enactment skills now that I didn't have ten years ago in terms of looking at different learning theories, being more open to summative and formative assessment. I am now prepared to pull reflection in which is key in the NFQ as well, [...] in terms of self-assessment and peer assessment, things I would never have looked at.

John refers to LOs on occasion in terms of the "verbs" used to write them using Bloom's taxonomy. He regards teaching as an approach in which "you map your LO to your assessment", in an effective manner: "if you come up with the right assessment type that puts meat on the verb". However, he perceived that not all of his colleagues integrate LOs and assessment in this manner: "a lot of lecturers here, no formal teaching or learning experience, very good in their discipline, constantly go back to the old assessment methods that they have relied on in the past". John thus perceives that the pedagogy of engineering academics either had changed, or should have changed, under the influence of the NFQ.

John regards knowledge and assessment as intertwined through the NFQ: "we can't teach all knowledge or we can't assess all knowledge for example but the standard is only interested in knowledge that can be assessed under the three strands". Although John recognises his role in imparting knowledge, he feels that:

I am not teaching you knowledge that is permanent, some of it might be, some of the concepts are permanent but particularly with the skills, the tools, the approaches and if you are in a different context, if you are in a medical device company one day and telecommunications company another day, context is different.

Instead, students must learn to learn, or, in fact, learn outcomes that can be assessed: "you have to learn to adapt and part of the learning to adapt is assessing where you are at this point and assessing where you want to go to".

As a new lecturer John "still assessed content", but, as he has reflected on the NFQ, in particular the three strands of knowledge, skills and competence, he has changed his pedagogical approach, restructuring his classes "So I go in now every day and say, 'This is the LO we are going to have today'". John had assumed that his students should understand LOs, but, he found "they knew none of that". He now tells students at the start of each class "that we are looking at this LO, this is how it's going to be assessed, be it a practical or written exam or whatever and this is what you need in order to understand the LO". It is striking that John

does not describe what knowledge, or even skills, that the students will learn, but rather that they learn outcomes in the context of how they are to be assessed.

John, although conscious of the three strands of knowledge, skills and competencies of the NFQ, regards the focus of curriculum as the assessment of LOs, stating that:

LOs have to be viewed in the context of assessment and I think LOs have to move past just summative assessment really particularly for the know-how and skill and the competence strands, you are looking at formative assessment, self-assessment and peer assessment [...] It's how I structure and enact the curriculum, a lot of it is coming back to, and something I am looking at, is my assessments.

However, John's pedagogy is more complex than a focus on assessment driven by use of LOs. It is contextualised through his reflection on how this should be used in engineering education, which, for him is "about becoming a software engineer". He describes, as a new lecturer, teaching software engineering through group work, where he tried to get the students to form teams and relationships similar to those he had seen himself in industry. He uses terms such as "engage", "brought into play", and assertive language such as "right, we are doing group work", implying an active, exciting approach, where he formed educational "relationships" with the students, and got them to "believe in themselves". He explained he did not use any particular educational theory in doing this, instead it was based on his industry experience, and a dissatisfaction with alternative approaches. However, he is redesigning his approach, using the team-work aspects set out by the NFQ, stating "that's what I want to use now. It didn't happen in the past because to be honest I didn't know it was in the NFQ".

Although John is very assessment focused, he does see issues associated with this approach:

we are overloading then with assessments then in the semester. So you have six modules in a semester and in engineering we find that all the assessments hit in week four and week eight, and all modules have to have at least two. So [...] I believe in this idea of cross modular assessment. Modules are related but we don't look at how they are related.

He suggests how this could be addressed: "we could reduce the number of assessments and we could share LOs across modules if we went and designed better and still meet the rules of the game and improve learning".

At the conclusion of the interview, John made it clear that he regarded the NFQ as a positive influence on his pedagogy: "*I am not critiquing that, I am saying 'OK I have accepted that*', and then the question is '*How can I improve what I am doing*?'". Similarly to Bill, a finding is

that John's interview shows he has actively embraced the change in his pedagogic identity brought about through the influence of the NFQ. In his case the assessment of LOs is a particularly strong influencing factor, in line with the skills-oriented policy drivers of the NFQ. He is very conscious of the regulatory nature of the NFQ, and feels that his colleagues should also change their approach to assessment to align with its LOs approach.

9.5 Ayah's Managerial Perspective

Ayah spoke with regard to the effect pedagogy was having on academics, and students, rather than on her personally. She describes the pedagogic identity of engineering academics as a group reorienting towards a LOs and assessment based approach. Although Ayah is positive with regard to some aspects of the NFQ, she adopted a somewhat critical view regarding pedagogy. As discussed earlier, Ayah viewed the prescriptive approach LOs entailed as quite appropriate for levels 6 and 7, but not for levels 8 and 9. She stated that, for engineering lecturers:

because they have to work through this very prescriptive approach. So they're basically working through these ten or twelve LOs that they need to stick to which doesn't really allow the time for [...] demonstrating the knowledge that they've built up.

In the past Ayah would have encountered lecturers that she considered really engaged the students "*it was the knowledge back that they had built up, so whether it was somebody who had a passion for marine, nearly all of their examples throughout mechanics, materials, thermodynamics were all related back to that field*". Now however, Ayah views the prescriptive approach of the NFQ as curtailing the ability of academics to fully bring their knowledge and experience to the classroom. At this point in Ayah's interview I got a sense of teaching under the NFQ as a pedestrian activity, the working through by the lecturer of a number of LOs, rather than demonstrating their knowledge of the topic to the benefit of the students. Ayah's words that "knowledge is lost sometimes now" resonate with *The Neglect of Knowledge* (Allais, 2014) and *Bringing Knowledge Back In* (Young, 2008).

Ayah does feel that an academic's pedagogical approach might depend somewhat on their confidence, with some getting support from *"following a framework and a process that's very clearly outlined*". She felt that academics *"who are more confident and have greater experience out in the real world in industry*" would be better placed to teach outside the prescription of the NFQ, highlighting her view of the importance of staff industry experience.

9.6 The Focus Group Views on Pedagogy

I sought the views of the focus group with regard to the change in pedagogy under the influence of the NFQ and EI's LOs approach. They agreed that, under these dual influences, they were generally teaching to assessment of LOs. One participant keenly felt the lack of a teaching qualification, and the difficulty they considered this created for lecturing: "we have to teach ourselves first how to teach and then how to assess whether the student has understood, and by the time we do that the term is over". Indeed, he agreed that "a default mode of teaching towards performance is what would result which is effectively teaching to assessment within constrained timeframes".

Another focus group participant felt that "*it is appropriate then that the NFQ is quiet on the matter of pedagogy so that we don't feel constrained* [...]. *I think the LOs are useful that way*". However, another participant argued that LOs are constraining, identifying the manner in which they reduced a module to passing LOs which represent a "*bit of knowledge*". Another view put forward in the focus group, reflecting views expressed by Ayah in her interview, was that the level of a module influences the manner in which it is taught: "*a level 6 or a level 7 would be dealing with more closed questions,* [...] *how much, how many, yes or no. But then the level 8* [...] *would be throwing more open questions at them*".

A related focus group conversation developed in response to a comparison I gave on teaching pre- and post- NFQ. Previously, teaching the subject of micro-processors, I would have taught the students what I valued, for them, about the topic, whereas I proposed that post NFQ I would be bound to teach a prescriptive list of LOs about micro-processors. The consensus of the focus group was that, although a lecturer may feel constrained by LOs, in fact they had considerable freedom to change and adapt these as they saw fit, which would be clear to an experienced lecturer.

This view was further expanded upon by one participant, who stated that:

remember that LOs should not constrain you to how you achieve the end. LOs just tells you what the learner should be able to do and should be able to know at the end of it [...] there is a bit of flexibility there that people tend to misinterpret and put rigidity and you can be quite innovative in the way that you convey your message.

Indeed, within the group there was consensus that as lecturers they would teach what they felt needed to be taught, regardless of what might be stated in the LOs, but would seek, later

to correct the curriculum to reflect that. I should note that the lecturers in the focus group were all highly experienced, although not all had experience in the IoT sector pre-NFQ.

9.7 Conclusion

A finding of this chapter is that under the influence of OBE, it appears that nearly all of my research participants (fourteen) have adopted a pedagogy which emphasises teaching towards assessment, with LOs used to decide what those assessments should be based upon. John's interview reveals that, although an engineering academic may adopt such a pedagogical approach, there is still room for them to contextualise this in their engineering experience, and in how they approach the identity formation of their students. Reflecting the insight from John's interview, the focus group also considered that this pedagogical approach still allowed them to bring their engineering experience to the classroom.

Although the NFQ does not "promote or require a particular active pedagogy" (Kees), it has been argued that it causes LOs to "become central to teacher–learner interaction" (Kenny, 2009, p. 24), promoting a skills focussed education. Teaching towards assessment is not described or prescribed within the NFQ or El's OBE accreditation approach, but the analysis suggests that, for my research participants, such a pedagogical approach is contingent.

However, a further finding is that the influence of the parallel student-centred, lifelong learning policy drivers of the NFQ was also evident, although to a lesser extent, as the interviewees described their pedagogy (for fourteen of the interviewees). Bill, for example, stated that he would try to find "new ways of doing things that were more relevant to students". John, though very focussed on assessment, states of the students, in describing a new assessment approach that "it was all about getting them to work and getting them to believe in themselves and ask questions".

Kees expressed sentiments that presented a balanced view of the two influences. Although he felt the importance of "the right blend of theoretical knowledge with an understanding of how we connected with industry" he also feels that we need "what's best for the individual and what's best for society, we need to have that balance".

The findings from this chapter will be further considered in the discussion, where I will draw on literature that considers the pedagogical approaches that may emerge in the context of OBE (Muller, 1998; Deacon & Parker, 1999; Torrance, 2007; Havnes & Prøitz, 2016).

Chapter 10 The Impact on Knowledge, Skills and the Curriculum

Its objectives being an introduction to the world of knowledge. John Heywood (2016, p 296).

10.1 Introduction

In my conceptual framework I have stressed the importance of knowledge in engineering curriculum, and the difficulty that has can arise in ensuring that this is correctly balanced with engineering skills. This chapter analyses my interviewees' perceptions of change in their approach to knowledge, skills, and the curriculum through the influences and prescriptions of the National Framework of Qualifications (NFQ), Engineers Ireland (EI) and other drivers.

I examine whether the introduction and embedding of outcomes based education (OBE) in IoT engineering education has influenced my research participants' approaches to acquiring new engineering knowledge required for programme and module design and delivery. I then explore their views of the appropriate balance of knowledge and skills for engineering curriculum. To conclude the analysis, the views and practice of my research participants are explored to gain insight into the influence of learning outcomes (LOs) on their approach to recontextualisation of knowledge in the curriculum.

10.2 Engineering Knowledge

For engineers, knowledge is viewed as that which can be usefully applied in engineering activities³⁵. My research participants (fourteen) expressed sentiments consistent with this perspective. For example, Bill explained that he was able to assess whether a particular topic was relevant because "you know from industry that it's an important topic". Isambard regarded that the student should "be able to contextualise what you have in theory in respect of what you can achieve" in practical application. In Ayah's view "everything that you've learnt in the confines of the classroom" is only relevant if it can be applied "to work through all of that in a pressurised environment when your team is falling apart, when your manager's telling you how much money you're losing for every hour that we're down". Similarly, Edwin talked of someone becoming an engineer "when you start making money from your knowledge". Thomas questioned whether a particular type of knowledge was relevant for

³⁵See p. 86

students as they were "*never going to use it*". For my interviewees, the utility of engineering knowledge is extant under the influence of OBE, which is not to be unexpected given the competency focus of LOs, which, if anything, is supporting of this worldview (Allais, 2014).

10.3 Knowledge Acquisition

Educators recognise that they must continue to develop their knowledge of the field over their career. For example Edwin considered." I can't say, look I've learned something in my university and that it's enough knowledge for the next thirty years. So we have to keep reading, reading, reading" and Bill contended that "if you were just using the stuff that you had at the start of your career, you'd just be dinosaurs". We discussed how this might be changing in the context that knowledge is now recontextualised and taught to students in modules defined by levels and LOs. The interviewees presented a range of views, in a small number of cases tightly aligning knowledge acquisition with the NFQ levels, with the majority regarding knowledge acquisition as independent of the NFQ, although recognising that it had to be recontextualised into NFQ levels and LOs afterwards. One interviewee questioned whether new knowledge acquisition should be a priority at all, whereas in contrast others focussed on the importance of research.

Consider John, in whose view the type of knowledge that a lecturer should be concerned with is dictated by the NFQ: "we can't teach all knowledge or we can't assess all knowledge for example but the standard is only interested in knowledge that can be assessed". This is suggestive of claims for qualification frameworks, reported (critically) by Allais (2014), that they will fundamentally alter the understanding of what knowledge is, i.e. will change epistemology itself. However, it is not clear what form it is proposed these "new notions of knowledge" (Allais, 2014, p. 14) would take, other than a hint that they are somehow learner centred. Is knowledge, in John's epistemology, represented through LOs that can be assessed? Certainly when he discussed pedagogic practice, knowledge was represented in those terms. Furthermore, John's interview showed strong evidence of a learner centred aspect, speaking of how he "engaged or enacted the curriculum with my students", and in the analysis presented earlier (p. 189) he could be regarded as attempting to inculcate in the students a LOs approach to constructing their personal worldview as engineers.

For Alan the acquisition of new knowledge is now closely associated with NFQ levels, being

very conscious of the level knowledge is to be taught at as he acquires it, "Yeah, that's the approach I take". However, this is so that: "it gives me more confidence to adapt how I am presenting it then or how I am trying to engage the students on it". He expects more capable students will want "more and more", and ensures he is not in a position where he has to say "I actually didn't read that part".

For Bill, the acquisition of new knowledge has been ongoing throughout his career: "*most of what I teach now didn't exist when I left college*". Bill, working in an academic field characterised by rapidly changing technology, experienced the need to learn new technologies, systems and skills on many occasions. Bill described the process, in his early days as a lecturer, pre-NFQ, of acquiring knowledge and structuring it so as to be able to teach it (the process of recontextualisation):

The very first semester, one of the subjects I taught was a subject that existed at the time of all the various characteristics of components, capacitors and stuff like this, how different ways resistors are made and so on. And I of course knew what these things were, but didn't really know in that level of detail, so all of that had be learned. Other stuff that I had done in college just had to be learned, maybe again or just had to be clarified in what you needed to actually get it across to somebody else.

Bill does use the level at which a module to is to be taught as a guide in acquiring knowledge, however, he implies that he has taken a similar approach throughout his academic career.

However, for more of the interviewees, the knowledge acquisition process was independent of the NFQ level, although, once knowledge was acquired, the process of recontextualisation into curriculum was LOs and levels focussed.

Vitruvius explained that "new knowledge itself, that would still be me, the original [...] me, as it were, reading up something, delving into something". However:

as soon as I move from acquiring the knowledge to even beginning to put it in some sort of framework to teach others, I'm already thinking of NFQ. I'm already thinking of who am I delivering this to, by which I mean, where are they, on the NFQ? Are they 6, 7 or 8, or 9?

In recontextualising this knowledge into module syllabi, Vitruvius recognises that some academics feel "there's too much emphasis on the LOs", but he himself feels that it works well "if you write your LOs in sufficient detail [...] to communicate it properly".

George similarly describes the process of acquiring knowledge required to design and teach a new programme as being independent of the LOs, although the knowledge is afterwards used to develop the LOs. George does not consider LOs during the development of a syllabus, finding them "to be quite difficult to actually base a course on". George explains that he would first prepare a draft syllabus "before I would actually go into the LOs".

Gordon, in describing acquiring knowledge for new subject areas, considered employers' requirements and the relationship to other modules. Once he has a general concept of what the curriculum entails he considers how to assess it, and then develops LOs (using Bloom's taxonomy), which become key drivers for establishing what knowledge is to be taught.

One of the interviewees, Thomas, questioned whether new knowledge acquisition should be a priority. Thomas felt quite strongly that the IoTs should focus on level's 6 & 7 of the NFQ, *"where we should be"*. He feels the technology at these levels is well defined, and *"how much research do we need to do to be good at that? I don't think we need to do a whole lot"*. In this world-view, new knowledge acquisition is not a priority as the knowledge to be taught in relation to level 6 and 7 of the NFQ is relatively static and hence research is not necessary.

In contrast, two interviewees focussed on research in discussing new knowledge acquisition. Isambard considered that the NFQ has influenced curriculum such that *"anything that we deliver has to be research informed"*. Henry who had been research active in industry and academia, related his impression that the research process had changed over the years, to focus much more on questions, and could not pinpoint any influence from the NFQ.

I discussed the theme of knowledge acquisition with the focus group. They considered that, rather than the NFQ level at which a module is situated guiding the lecturer in their acquisition of the knowledge required to teach it, the level was more likely to guide how it is taught. They considered that the act of teaching itself may influence knowledge acquisition: "as I have acquired material I have always measured my understanding by being able to explain it to students"; and "you have to look for supporting material depending on how the class reacts".

My interviewees portrayed their acquisition of new knowledge through working with employers, considering what students need to be taught and how to teach it to them, reflecting on their own experience and expertise, and through being research informed. These, from my perspective as an engineering academic, all appear quite reasonable approaches. Although our conversation may have been couched in the context of LOs and NFQ levels, I do not believe the interviews provided evidence that OBE has significantly influenced knowledge acquisition in these cases. Nonetheless, for a small number of my interviewees, OBE has been a significant influence. John appears to regard knowledge largely in the context of assessment of LOs, and Bill and Alan indicates they use the levels of the NFQ as a guide to determining what knowledge they need as academics.

The analysis of the fieldwork leads to a finding that the NFQ has not had a significant impact on the acquisition of new knowledge for the majority of my interviewees, although a small minority use NFQ levels as a guide to what that knowledge should be.

Some of the interviewees considered that their workload, 18-20 hours a week of lecturing, alongside administrative and other duties, was interfering with engaging with acquiring knowledge, including engaging in research. They identified the increased administrative work consequent to the NFQ as a further contribution to this, e.g. Isambard, who felt that workload "does not give you any room whatsoever to delve in depth into enhancing your knowledge".

Ayah also believes that OBE has placed an increased workload on academic staff which completely dominates their work³⁶. Ayah feels that consequently academics are "not as research active" and that LOs and their associated workload are "stifling innovation and it's stifling research".

George also perceived "that there has been a lot of additional workload" in the form of quality assurance (QA) and related procedures and requirements, although not directly attributing this to the NFQ. He made reference to the difficulty of doing 18 hours of teaching per week and trying to maintain, for example, good industry contacts. Henry felt that the workload on Heads of Department is inordinate, making it difficult to properly manage their staff whilst simultaneously leading activities like programmatic review, "they don't have the time". He felt this contributed to this important activity being "a paper exercise" that "doesn't have a positive input [...] as it could have or should have" in refreshing the knowledge in syllabi.

Edith questioned the workload associated with conducting both EI accreditation and programmatic review, which interfered with "*getting the knowledge to teach*". She regarded EI accreditation more positively, although still considering it "*paperwork*".

A clear impression emerged (from five of the interviewees) of the heavy workload on lecturing

³⁶ See p. 131.

staff, from administration and particularly from 18 hours teaching. Although the NFQ is perceived by some of my research participants as playing a role in this increasing workload, it appears to be as part of the overall system, rather than a principal determining factor. A heavy workload was seen as being detrimental to their capacity to engage in other activities such as research and other aspects of knowledge acquisition.

10.4 The Balance of Knowledge and Skills

Engineering curriculum are a type that face "both inwards to disciplines and outwards to professional or workplace practice" (Blackie et al, 2016, p. 763), and with the IoTs so linked to industry and business, it would be expected that there would be a skills emphasis in IoT engineering education. Skills would thus be expected to feature strongly in my research participants' views of curriculum, which OBE, given it is generally associated with an increased focus on skills (Allais, 2014; Young, 2008), would strengthen. However, an earlier finding, in considering progression, was of the academicization of engineering technician education. I sought to tease out these apparently contradictory positions through investigating my interviewees' views of the balance of knowledge and skills in curriculum.

John, in a recent programmatic review, related that he and his colleagues placed less emphasis on skills than on knowledge, they "chose verbs that valued definitely the knowledge strand over know-how and skill and definitely over competence". He regarded the different strands as complementary, allowing skills to be delivered in relationship to the other strands.

Reflecting the importance of skills, and the context provided by the NFQ, Isambard, when "at the initial stage of my academic career", considered that "you need to teach the students the theory first and then support that with practice to enable it to embed". However, as he became more experienced, his views changed:

I then realised that actually in order to embed the practical concepts that are the practical skills, they should actually be done in tandem, it should be together. There are some cases for example like in problem based learning, you find that you actually learn the theory better when you are actually doing it and to be able to contextualise what you have in theory in respect of what you can achieve or what you can implement practically.

He feels that under the NFQ, particularly at honours degree level, it has "distilled [...] refined what area needs research", making it clearer what is needed for and from the student "in terms of skills and knowledge".
Bill described having to reconceptualise the manner in which he approached engineering student projects due to the NFQ. For final year level 8 students he considered "you're certainly expected to have very detailed writing skills for reports, organisation, timetabling, to make sure you get the thing done in time". A level 8 engineering project will typically be trying to solve a real-world problem using theory and practice, applying analysis and design techniques, but the level of emphasis on skills from Bill is pronounced.

Partly sharing this perspective Ayah described how, under the NFQ, lecturers need to consider "knowledge, skills and competencies", and that this maps very well, for example "when you're talking about explaining how a machine works and it's a step-by-step approach and that fits very much into the framework for a [...] 6 & 7". However, for a level 8 graduate Ayah expects that "they need to be able to go out into the world and solve problems and navigate through different systems that they've never seen". Ayah does not feel that "the LOs approach helps that particular position at level 8, particularly with [...] skills".

However, some of the lecturers felt that there was not enough emphasis on skills, e.g. Edith "they're going in to be a technician, so they need the practical skills. Now we do try and incorporate a bit of practical stuff, but you're very limited in what you can do". Thomas also feels that the IoTs should be focussing more on teaching skills, and less on theory: "We should stick to teaching the technical subjects, the skills". Henry felt that "we've lost our applied bit", referring to a reduction in the teaching of practical skills.

In summary, a perception of the relationship between skills and knowledge was evident in the views of twelve of my research participants. In some cases they viewed the NFQ as providing a context within which to combine the three strands of the NFQ, knowledge, knowhow and skills together, and were accepting of the skills emphasis even at level 8 (Bill, Isambard, John, Gordon, Rudolf, Kees). In one case (Ayah) the interviewee felt that the skills emphasis, whilst acceptable at level 6 and 7, was not appropriate at level 8. However, four of the interviewees (Edith, Thomas, Henry, Alan) felt that there was not enough emphasis on skills, attributing this at least partly to the NFQ. One interviewee (Vitruvius) felt that the question was not relevant to what they taught.

A finding is that the interviews provide evidence of conflicting opinions amongst my research participants with regard to the balance of knowledge and skills: some regard there as being

an insufficient emphasis on skills, whereas others felt that the use of the NFQ led to an appropriate balance of skills and knowledge. This stands in comparison to an earlier finding that the views of my interviewees regarding progression as contributing to an increased academic focus for technician education appear to be in contradiction with government policy promoting a skills focus for higher education. As will be discussed in *Chapter 11*, that these contradictory positions became apparent from the analysis are quite significant.

10.5 Recontextualisation of Knowledge in the Curriculum

Curriculum design, the recontextualisation of knowledge into syllabi, is a key aspect of the use of LOs. I sought to understand my research participants' perceptions of what has changed, for this important activity, under the influence of the NFQ and EI. *Table 10-1* provides a representation of how the interviewees' view this in terms of: LOs as a driving force in recontextualisation; knowledge secondary to LOs; and how they bring their personal knowledge to the curriculum.

Table 10-1 Recontextualisation of Knowledge

	Research Participant														
Recontextualisation of Knowledge	Isambard	Ayah	kees	Rudolf	Vitruvius	Bill	Thomas	Gordon	Edwin	George	Edith	Henry	John	Alan	Grace
LOs a driving Force															
Knowledge Secondary															
Bringing Your Personal Knowledge to the Curriculum															

10.5.1 What has changed in the Design of Curriculum?

Under the NFQ, modules are represented using LOs, alongside an indicative syllabus. The QQI engineering standards provide a reference point for module design. Suggested practice is to design the module LOs first (Kennedy et al, 2006), and then develop the indicative syllabus.

Isambard perceived the use of LOs in a positive manner, explaining that pre-NFQ he regarded graduate attributes as not well defined, even if graduates were "very competent engineers", whereas post-NFQ it is easier to stand over the details of a qualification. He perceived a reduction in the depth of programmes since the NFQ, where in his IoT honours engineering degrees were "squeezed" from five to four years. He attributes this change to financial pressure and the extra time involved for students. As discussed in the literature review, this

change was, in fact, consequential to El's accreditation process changing to align with Bologna, albeit it occurred in the same timeframe as the introduction of the NFQ.

Vitruvius, was similarly positive, stating that, pre-NFQ as "you were just relying on your own intuitive grasp of what a particular module" should contain. He considers the NFQ structure makes module design and delivery easier, and as an academic manager:

explaining to a colleague [...] and in particular [...] explaining to a relatively new colleague how to write a module, how we should go about it, what you should look out for, and also how to deliver a module, was made much easier by the framework.

Bill feels that knowledge still has a role post-NFQ, where "you design your subject around a syllabus and a set of labs and everything like that, and then from that you say – well what are the key things that this is achieving? And there's my LO". He feels that "the emphasis is probably more on LOs", whereas pre-NFQ "in the old system it was that syllabus that was the defining thing". This is consistent with the policy shaping the formation and use of the EQF (and the NFQ), where "the emphasis is on defining key competences and LOs to shape the learner's experience, rather than giving primacy to the content of the subjects that make up the curriculum" (CEDEFOP, 2011, p. 9). However, Bill thinks the overall design process is:

Just different more than anything. I think what we always did was still the same, when we were here or any other staff for that matter, or a group of staff in our department wherever they were would do the same thing. You would sit down you'd say well what kind of a degree do we want to do, and what areas are we good at, and what is our expertise – and that's your starting point".

John describes himself, pre-NFQ, as being un-inspired by just thinking about a "content based approach to the knowledge". However, as his understanding of the NFQ has grown, John associates knowledge with LOs, and after that "it's the fact that you map your LO to your assessment" that creates learning. Edith regards syllabi as subservient to LOs, describing the syllabi content as something that "feed in from your LOs".

The use of LOs in curriculum design was not without criticism. John described a programmatic review where they "ended up retrospectively mapping [...] module LOs to programme LOs". Here John revealed a back-to-front approach, where the need for programme LOs was only identified towards the end of the review process, as opposed to early on. Although this may have satisfied programmatic review, it was not how LOs were intended to be written, which requires definition of the programme outcomes first, and design of modules LOs to meet

those. However, such an approach was quite characteristic of the approach during the introduction of the NFQ, referred to by Kees as *"re-badging"*, and is something thing I myself took part in at that time.

As discussed earlier, John uses assessment as a lens to relate to knowledge, which is secondary to assessment (of LOs). Indeed, John implies that knowledge is secondary to LOs themselves, where, when, referring to lecturing students he states that he gives them an *"overview of what the content of the knowledge associated with that LO is"*.

Edwin provided a view of LOs based curriculum from his experience as an external examiner, stating you "have to make sure that the learning outcomes are there", with the indicative syllabus a secondary consideration. This is consistent with the QQI (2015) guidelines for external examiners, which focus on ensuring alignment LOs and student achievement with the appropriate national standards as related to the NFQ. This reflects QQI policy where "the impact of LOs on QA processes is significant. In general terms, the object of QA shifts from the curriculum and the teaching to the QA of the learning that is achieved as a result of the curriculum and the teaching" (Coles, 2016, p. 11).

Thomas implied an at times lip-service approach to the NFQ, which only really assumed importance at programmatic review:

There needs to be a way of ensuring that programme outcomes are being met, that knowledge is being conveyed, you know, how one does it, I don't know, but and maybe that should be the job of [...] the framework here, but they don't seem to do any of that. Every three years, programmatic review, that's about it, come out for a day or two.

However, Thomas is keenly aware of El's influence, regarding them as a key driving force in the use of LOs. He feels this approach has no regard to *"syllabus content really. They just wanted to see LOS. We don't care how you achieve them so long as you achieve those LOs"*. In this worldview, LOs have risen to the fore, with knowledge quite side-lined. Thomas's view does seem a reasonable, if somewhat emotive, interpretation, in that El's accreditation procedure (El, 2010), focuses around providing evidence that programme outcomes match accreditation requirements, and have been met.

Blooms taxonomy is an enabler for the design of module LOs. However, five interviewees expressed sentiments of the taxonomy being the primary driver in LOs design, as opposed to knowledge. For example, Alan states that *"sometimes I actually find it difficult to find an*

appropriate verb from Bloom's Taxonomy [...] to actually reflect what you're trying to do".

To summarise, the majority of the interviewees (fourteen) described that they now saw LOs, under the influence of the NFQ and EI, as a key aspect of curriculum design, with the module content (knowledge) being dependent upon LOs, and, for some (nine), knowledge secondary to LOs. However, some academics were critical in their perception that the LOs approach is not fully followed in module design and review processes, although they agreed the end result of such processes required that modules and programme were presented using LOs.

The focus group agreed that engineering academics regard LOs as the primary driver in LOs in curriculum design. In exploring the implications of this, they took the view that well written LOs in syllabus design actually empower the lecturer with regard to the delivery of knowledge e.g. *"the module leader retains the freedom to cover all or most or some of the points in the syllabus"*. However, a dissenting voice stated *"in general, do lecturers pay much heed to the LO's? Or are they driven by the syllabus content"*.

A finding is that the fieldwork provides strong evidence that engineering academics now perceive LOs (influenced by both the NFQ and EI) as the key driver for the recontextualisation of knowledge in the curriculum, with, for some, knowledge secondary to LOs.

10.5.2 Bringing an Academic's Personal Knowledge to the Curriculum

A further topic of discussion, raised by six of the interviewees, was with regard to how lecturers brought their own personal knowledge to the curriculum.

For Rudolf, the fact that a syllabus mapped to the NFQ is *indicative* allows scope for academics to be innovative in teaching the knowledge they feel is appropriate, i.e. "*if a relevant topic area is now irrelevant or superseded by another subject area*" it can be replaced.

Isambard similarly sees the LOs approach as providing the necessary flexibility:

I would be surprised if given two academics with the same level of qualification working in exactly the same area [...] would have exactly the same sequencing even of the material that they deliver [...]. You draw examples from different spheres and you are free to do that, as long as the LOs which in most cases should be broadly defined, allows you to achieve the same end.

Thomas, expressing similar sentiments to Bill, provides a related viewpoint that regards disciplinary knowledge as being held within a group of academics, rather than individually. He

speaks of course design teams: "a good small team of people that have the skill, they have the knowledge, experience, you could develop a good course from the bottom up". Thomas values not just the knowledge of the group, but also their skills and experience, and sees LOs as helping to bring all of this together: "you've got to have good programme outcomes".

Ayah was concerned that the prescriptive nature of LOs may inhibit academics from fully utilising their knowledge and experience. In her own case she "would have brought an awful lot of experience" from industry and considered "a more holistic approach to learning was needed [...] than what we were using at level 6 and 7", which she found "quite restrictive".

Alan described his experience on joining an IoT from industry: "I worked in telecoms and networks for seven or eight years but it wasn't until I had to come back and design a course for it that I actually started to learn it in a different way again". He had to recontextualise his industry knowledge in order to use it in teaching, using NFQ levels to assist in gauging how to pitch lecture material. However, he finds this does not always fully delineate the knowledge to teach: "a level 7 versus a level 8, really there might not be an awful lot of difference".

The literature on LOs is largely silent on how to approach this recontextualisation of a lecturer's disciplinary expertise in the delivery of a LOs based programme. Consideration of guides on the use of LOs illustrates the deficit (Office of University Evaluation, 2001; Kennedy et al, 2006; COBE, 2007). These were created by HEI's to assist/direct their academic staff in the implementation of OBE. They avoid discussion on academic staff disciplinary expertise, focussing on generic issues related to delivery and assessment using LOs. EI state that academic staff should *"have the ability to design, develop and implement courses"* (EI, 2014, p. 5), but other that this rather general statement, do not comment on this matter. It could be argued that the QQI Engineering Standards provide this disciplinary guidance (QQI, 2014c), but, as already discussed, they are quite generic across different engineering disciplines.

However, the focus group consensus was that it was important to keep LOs generic, as opposed to specific. This to ensure "*we are not constrained by them*", and that "*the module leader retains the freedom*" to cover material they consider relevant.

The analysis reveals what could be described as a doublethink in some of the interviewees' views on how to use LOs. On the one hand they are accepting as a group of this approach to curriculum design which emphasises what a student should know and be able to do on

completion of a module. On the other hand, for some academics, including the focus group, LOs should be written more generically, or "written well" (focus group), not specifying precisely what the LOs should be or how they are assessed. In the capable hands of an experienced academic this provides a framework within which they can deliver the module in the context of their own expertise, and does not tie them down to a prescriptive approach. This leads to a finding that, for some academics, LOs, if written such that are not overly prescriptive, can be enabling, facilitating individual expertise to module delivery.

A further finding from the analysis was that curriculum design and review was regarded, by all of my interviewees, and the focus group, as a social process, whether at department level or involving smaller groups. For example, Rudolf described, in relation to programmatic review, that "we had a really interesting debate about mapping module learning outcomes to programme learning outcomes". John described pre-NFQ curriculum design as "I don't think we engaged in the verb game then".

10.6 Conclusion

The analysis reveals the interviewees approach to new knowledge acquisition for curriculum design: through being research informed; using industry consultation; reflecting on what to teach and how to teach it; and through considering their own expertise and how it might need to be further developed. They do not appear to have been overly influenced by the OBE approach, although a small number did appear quite cognisant of NFQ levels and LOs in acquiring knowledge. Once new knowledge is acquired, it is recontextualised into LOs in curriculum mapped to NFQ levels. The use of LOs in curriculum was generally regarded in a positive fashion, with LO's regarded as a key tool of curriculum design.

In investigating the views of my research participants with regard to what they considered an appropriate balance of skills and knowledge in curriculum, contradictory positions became apparent. Some participants saw too little skills, whereas other participants were happy with the balance of knowledge and skills promoted by the NFQ. The fact that my research participants hold such disparate views on this key issue appears related to an earlier finding that progression is contributing to an increased academic weighting in technician education, in apparent contradiction to government policy promoting a skills focus for higher education.

The discussion will consider the implications of these issues for engineering curriculum.

Chapter 11 Discussion

For what we wish we readily give credit to, and what we think ourselves, we hope is the opinion of other men

Julius Caesar (Commentaries on the Civil War, para 2.27)

11.1 Introduction

My research is situated in a time which sees the alignment of political, societal, cultural and economic structures and values with neo-liberal doctrine (Finnegan, 2016). The re-orientation of higher education, including engineering education, to having a more skills focussed orientation targeted towards the needs of employers is a key enabler for this change. As noted earlier, this employability focus has been supported internationally by the introduction of national qualification frameworks (NQFs) (Allais, 2014), and reinforced in engineering education through the adoption of outcomes based education (OBE) accreditation (Matos et al, 2017). However, as will be apparent from the analysis, and as will be developed further in this discussion, in Institute of Technology (IoT) engineering education the parallel learner-centred facet of the National Framework of Qualifications (NFQ) has also had some influence.

The views and practice of my interviewees in relation to impact and change they perceive from the LOS based engineering education system in IoTs have been revealed through the analysis of the fieldwork. The findings from the fieldwork, supported by earlier findings and themes regarding IoT engineering education, the National Framework of Qualifications (NFQ), the government skills agenda for higher education, and Engineers Ireland (EI) accreditation, suggest that there has been substantial consequential change in IoT engineering education.

Although some of the change I have observed, for example the emphasis on learning outcomes (LOs) in curriculum, were intended to follow from the adoption of OBE, other changes appear contingent, and in some cases their appropriateness is questionable. For example it is clear that it was intended that the nomenclature of the NFQ should become part of the policy discourse of higher education (Collins et al, 2009). However the usage of the related language of levels I observed in the pedagogic discourse of my research participants at times displaces disciplinary terminology, and, in emphasising the importance of progression for my research participants, has further curriculum and structural implications for engineering education. Furthermore, international experience of the implementation of

NQFs reveals that is it common for them to lead to contradictory policy positions (Fernie & Pilcher, 2009; Allais, 2010a; Havnes & Prøitz, 2016). The fieldwork revealed that not only are such contradictory policy positions apparent in the implementation of OBE in IoT engineering education, but, also, that they have had important, and at times concerning, resultant effects on engineering educators and engineering education.

In this discussion I consider the implications of my research for IoT engineering educators and the education we offer. I also consider what my fieldwork reveals about the effectiveness of the NFQ, and EI's OBE accreditation processes. These considerations are part of a broader context where the societal logic relating to education and knowledge is being reconfigured through the use of OBE, and I show how my research contributes to debate on these matters.

I include a reflection on my framing of the analysis of the fieldwork using the pedagogic device, which I developed, for IoT engineering education, through: a synthesis of ideas from social realism; my own experience as an engineering educator; consideration of the sociohistoric context within which our contemporary IoT engineering education system has arisen; and working with my research participants in the fieldwork.

11.2 Engineering Educators and Engineering Education in the IoTs

In seeking to describe and understand the social logic (Morgan et al, 2018) of IoT engineering education through a conceptual framework influenced by social realism and EER critical or questioning of OBE (Woolston, 2008; Riley, 2012, Rajaee et al, 2013; Wolff, 2015; Heywood, 2016; Akera, 2017; Matos et al, 2017; Klassen, 2018; Nudelman, 2018), my findings are inherently mediated by what Glynos and Howarth (2008) term "the *constructed* and *political* character of social objectivity" (p. 7). Consequently, in framing the analysis of the fieldwork using the pedagogic device, my findings already incorporate a considerable aspect of discussion in which I have adopted "ideological and normative arguments" (Edström, 2018, p. 39) which take a critical stance towards the use and impact of OBE in engineering education. In this section I expand the discussion in considering my findings from an engineering education research (EER) perspective.

As I have shown, the field of EER pays considerable attention to research issues associated with the application of LOs in engineering education, where, generally, OBE is considered a benign approach, leading to a better engineering education. EER in relation to engineering educators is largely focussed on how to improve their engagement with the use of LOs in teaching and curriculum. However, as identified by Matos et al (2017), there is a research gap, both theoretically and empirically, with regard to the impact of the use of OBE on engineering faculty, for which, through my findings, I contend that OBE may have, at times, adverse and unintended effects on pedagogy and curriculum structure.

11.2.1 Pedagogy

Pedagogy had not been a significant consideration in my own experience of our transition to OBE in IoT engineering education, although over time I noted the pre-eminence that LOs assumed in collegiate discussions regarding teaching. Consequently, a particular concern I had from the outset of my research, and which the fieldwork provided the opportunity to explore, was whether, despite my perception of a lack of explicit focus on pedagogy, the use of OBE might, nevertheless, have an implicit effect on our pedagogy as IoT engineering academics. My research revealed mine was not an isolated experience, where a key finding was to show that the pedagogical basis of the NFQ received little attention during the development process that led to its implementation (NQAI, 2001a, 2001b, 2002c, 2003a). Similarly EI give little guidance on pedagogy (EI, 2014), and Heywood contends that EC2000 (ABET, 1998a), the genesis of the international transition to OBE for engineering accreditation, was developed without due consideration of student "learning and development" (Heywood, 2016, p. 290), and that this has not been corrected in more recent proposals from ABET.

My research participants, however, were deeply interested in their pedagogy and that of their peers, similar to the experience related by Andrews et al (2019) in their exploration of their colleagues' pedagogy in the WMG engineering department in the University of Warwick. Our discussion on pedagogy generally centred on teaching the practice of engineering, both theory and skills together, *"the IoT way of doing things"* (Vitruvius). The fieldwork confirmed the influence of OBE, where a finding was that the NFQ acts as an influence over the pedagogy of my interviewees, which they largely regard as positive and constructive. For example consider Bill who stated: *"certainly in the last 10 years, I've spent more time on the process than necessarily on the content"*. The fieldwork revealed that, influenced by OBE, my interviewees have adopted pedagogical approaches emphasising teaching towards assessment, with LOs used to decide what those assessments should be.

As Collins et al (2009) have argued, the use of LOs requires "a radical shift in teaching" (p. 39); and Kenny (2009) proposes that pedagogy is directly affected as teaching turns towards the requirement to meet LOs. OBE for engineering can lead to "the generation of long lists of checklists" (Heywood, 2016, p. 290), stifling educational innovation, and may affect engineering academics approach to pedagogy (Matos et al, 2017), as my research has shown. In the case of the NFQ it, like other NQFs, simultaneously promotes a learner centred approach alongside an employability emphasis, which leads to tension and contradictory pressures regarding pedagogy. Although OBE in IoT engineering education does not stipulate, or suggest, a pedagogical approach, these two different drivers for OBE in the Irish educational system appear to require different approaches.

If NQFs truly support a learner centred approach, assessment will examine whether learners meet LOs within a learning environment encompassing "a shift from a transmission-content pedagogic model to an acquisition-competence model" (Muller, 1998, p. 182). This can be considered a competency pedagogic approach which "emphasises learner opportunities to shape their own engagement in learning, create learning conditions and enhance their own learning" (Havnes & Prøitz, 2016, p. 216). However, it has been argued that the reality is that the learner centred approach of OBE is offset by engineering students not necessarily having the self-awareness and capability to engage with it, alongside the traditional role of engineering academics as presenters of knowledge rather than guides (Rajaee et al, 2013).

On the other hand, a principal purpose of OBE is to provide an employment focus for qualifications (Young 2003; NQAI, 2003a; CEDEFOP, 2009), including engineering (Prados, 2004; Matos et al, 2017). These must meet the skills needs of employers, within a time-frame determined by the market in order to be of value, leaving little scope for learner-centeredness. A concern I share is that "extensive use of LOs and assessment criteria might risk resetting the horizon of learning or situate learning in the context of passing tests or achieving specific LOs, instead of perceiving these as milestones in an expanded qualification process" (Havnes & Prøitz, 2016, p. 218). This process oriented influence from OBE in engineering education (Woolston, 2008) results in the "valuation of product over process in traditional engineering education" (Riley, 2012, p. 5). The potential for this focus on assessment was identified by Tyler (1949), who cautioned that "the diverse objects appraised by evaluation" (p. 124) might come to dominate. This suggests teaching may become focussed

on the assessment of LOs, a performance pedagogic approach, which can be referred to as *"assessment as learning"* (Torrance, 2007, p. 282).

In order to gain a deeper understanding I drew on literature that examined which of performance or competence pedagogy might come to dominate under the influence of OBE (Muller, 1998; Deacon & Parker, 1999; Torrance, 2007; Havnes & Prøitz, 2016). Consider:

"in response to weakened institutional boundaries, increased calls for public accountability in the new reflexiveness of risk society and [...] market calls for relevance, effectiveness and efficiency, the pendulum in all national systems conscious of their competitive position in the global economy swings towards performance models (Muller, 1998, p. 189).

I explored this with the focus group, who highlighted that IoT engineering academics are not expected to be trained in education. They regarded this as an impediment to effective teaching, which, combined with the workload of a LOs based, semesterised system, entailed that achieving anything other than teaching to assessment was quite difficult. Indeed none of my interviewees discussed receiving any significant training in the pedagogical aspect of LOs.

Given the economic imperatives on the IoTs, that the NFQ and EI are relatively silent on pedagogy, and that IoT academics do not require teaching qualifications and are most unlikely to have been trained in competency oriented pedagogy, then conditions are favourable for the emergence of performance pedagogy. Furthermore, for accreditation a key focus is on "assessment of what students have learned" (Felder, 2012, p. 9), and which can lead to "accompanying rigidities" (Heywood, 2016, p. 290). From a Bernsteinian perspective my interviewees have adopted what he called tightly 'framed' pedagogical approaches (Bernstein, 2000), where they make it clear to the students, and themselves, what will be taught and assessed, set in the context of LOs (Nudelman, 2018) and the time constraints of a semesterised system. This provides evidence, supported by related studies (Muller, 2004a; Rami, 2012), that a consequence of the use of LOs, regardless of policy aims of promoting a learner centred approach (Rajaee et al, 2013), may be to influence academic's pedagogic identity towards the adoption of performance pedagogy.

In contrast to my findings, Woolston (2008), despite regarding the OBE focus on process as inappropriate and ineffectual, and identifying the emphasis on assessment rather than on how learning takes place, believed that it had little effect on engineering pedagogy. Similarly to the deepening of the impact of the NFQ I have observed since Collins et al (2009), I suspect

that, as OBE accreditation has become more established, so has its impact become more pronounced for engineering educators. As Kees speculated regarding the NFQ, it appears that *"over time [...] people are normalised into a particular way of looking at [...] learning"*.

My critical, practitioner oriented, analysis of the consequence for our pedagogy of the move to OBE for IoT engineering academics is in contrast to El's position (Owens, 2016a), who consider that their accreditation assessment process provides space for academics to reflect on their pedagogical approaches. My findings, however, concur with those who question the contention that OBE is associated with an improvement in the quality of teaching and learning in engineering education (McCullough, 2007; Woolston, 2008). While I accept that there is still some room for the "various pedagogical processes" (EI, 2014, p. 10) El suggest may be applied, they will, inevitably, given their positioning, be focussed towards assessment of LOs.

Although I have been somewhat critical of the pedagogical consequences of OBE for my research participants, the fieldwork also revealed that their pedagogic approaches are not fully defined through their orientation towards performance pedagogy. For example, the focus group made it clear that, although they recognised constraining aspects to the OBE approach, for some, LOs, if written to avoid being overly prescriptive, can be enabling, facilitating academics bringing their individual expertise to module delivery. This finding is reminiscent of Coffield et al's (2007) description of teachers in UK further education, for whom policy constraints were "not all powerful and pervasive" (p. 738), with many still feeling "they have the space to exercise their most cherished professional values" (p. 739). Similarly, the fieldwork provided considerable evidence of mixed-mode pedagogies (Muller, 1998; Bernstein 2000), in particular aspects of the competence based pedagogy that can be associated with the learner centred facet of an NQF. For example Isambard, who felt we should "meet the needs of individual or [...] any of the students that you have in the class".

My interviewees emphasised the importance of the application of teaching and learning approaches in teaching engineering practice, similarly to other engineering academics (Brent & Felder, 2003; Knowles et al, 2019). This points towards my interviewees having embraced OBE precisely because they perceived it as a framework within which they could develop their pedagogy for the betterment of the practical aspects of students' education. This is exemplified by John's interview, who, although he has strongly aligned his pedagogy with a LOs approach, was very conscious of how this was impacting on the student experience.

Discussion

This discussion causes me to reflect on my own pedagogical position as an engineering educator. I see value in outcomes or competencies, as it is important for engineers to be able to achieve tangible, useful, effect. However, the emphasis on LOs creates, for me, a significant difficulty, reflecting a pragmatist influence. In valuing the educational experience students receive on their journey to becoming engineers (Woolston, 2008; Somerville-Midgette, 2014), I cannot divorce the outcome from the journey, where new knowledge becomes part of the experience, and the experience in acquiring it is part of the knowledge. As Frezza and Nordquest (2015) write "in some sense the purpose of design education is not to create an artefact but rather to grow as an engineer" (section *Implications for Engineering Pedagogy*). This journey to becoming an engineer should provide adequate opportunity for reflection (Socha et al, 2003; Bowe & Duffy, 2010; Adams & Forin, 2013) as a key part of the process of engineering student identity formation (Haase, 2014). I thus have a concern that the overemphasis on assessment I have observed, on achieving outcomes rather than learning from experience, interferes with the inculcation of this reflective capacity.

Clearly assessment is important for engineering educators, but, in my experience, in engineering education it is formative assessment (Heywood, 2016), the use of which is regarded by Froyd et al (2013), alongside LOs and other aspects derived from educational research, as part of a major shift in engineering education, that is critical. This gives students feedback on what they have done incorrectly as well as correctly, which is most important in giving them the opportunity to reflect not only on what they know, but also on what they don't know, and how they have arrived at their current disposition. Summative assessment, where outcomes are measured, is also important, but principally in terms of ascertaining that the other aspects of a students' education that I value have taken place.

In the absence of a pedagogical model as part of our OBE engineer education system a default, performance oriented pedagogy has emerged, which I consider does not fully serve our educational goals, and which has, arguably, inculcated in our pedagogic identities an alignment with a neo-liberal oriented employability focus. I suspect that, previously, my pedagogy was quite aligned with the performance approach I have observed in my interviewees. However, I do not consider my pedagogy *now t*o be LO focussed, which I believe has followed from the awakening that this research has engendered. This leads me to believe that IoT engineering faculty enthusiasm for becoming better educators, and their

consideration of the importance of their role in educating future generations of engineers (Haase, 2014), can be harnessed through encouraging the development of a clear view of their own educational philosophy (Heywood, 2016). This could be inculcated through appropriate, engineering oriented, pedagogical training, as has been recognised as important for engineering educators (Heywood, 2016; Kersten, 2018). For example the IGIP (International Society for Engineering Pedagogy) offer certification as an 'International Engineering Educator' (Morell et al, 2018).

In the next section I consider the impact on curriculum, where the tension between the skillsoriented employability facet of OBE and its learner centred aspect is also a significant factor.

11.2.2 Curriculum

My research revealed counter pressures on curriculum in IoT engineering education. The first is the employability emphasis associated with OBE, a tool of government strategy to achieve a more skills focussed higher education (HESG, 2011; DES, 2016). I have identified the importance of considering how the skills emphasis of OBE may have impacted on IoT engineering academics regard for curriculum, and in the fieldwork all of my research participants acknowledged the influence of LO's on curriculum, as exemplified by Gordon's view that "for me it's been very, very positive".

However apparently acting in contention to this skill emphasis is progression and the NFQ's level structure, associated with the NFQ's parallel promotion of learner-centeredness (Allais, 2014). As I noted earlier, progression arose as a prominent theme in the fieldwork, providing evidence that, alongside other factors including industry requirements, EI accreditation and funding, progression has led to level 6 and 7 programmes being regarded by IoT engineering academics as preparatory programmes for level 8 degrees, leading to an increased emphasis on theory over practical skills. IoT engineering academic focus on engineering practice, so my interviewees were already very skills focussed in any case, meaning this has not been a complete re-orientation, rather a tempering of our approach to engineering education. The views of my interviewees with regard to the effect of progression can be summarised through the focus group consensus that "we now see a level 7 as a person on a journey to a level 8".

Progression has implications for engineering education beyond the opportunity it provides to students. Whist recognising the role of progression in countering elitism, Llorens et al (2014)

Discussion

consider that the level of expectation of students to progress "almost removes the legitimacy of the Level 7 programme in its own right; the vast majority of students see it as a back-door to [...] Level 8" (p. 5). Allais (2010a) reports a similar effect in Scotland, where "Higher National Diplomas became more accepted as a route to a degree" (p. 11), rather than employment. My fieldwork supports concern (Llorens et al, 2014) that level 7 engineering programmes are changing their focus from a path to employment to further study, reflecting change at national level. This is resulting in all IoT engineering students being educated from the outset to be honours degree graduates, rather than technicians. Those who leave early receive a qualification that "doesn't give you technician level. No. I guess the ordinary degree now is someone who couldn't cut it and make it to the honours" (Thomas).

Given my earlier argument³⁷ that technician and professional engineering education require different emphases, the implication for curriculum is significant. The diminution in the skills focus of technician programmes can be considered through the contention that NQFs can lead to a weakening of the boundaries between vocational and academic education (Young and Muller, 2010). However, "boundaries can serve educational purposes – they underpin skill and knowledge specialisation" (Young, 2011, p. 98), providing opportunities for learners "in establishing their specialist occupational identities" and serving "as guides for those designing curricula". This provides argument for retaining boundaries between technician and professional engineering education, to preserve their occupational identities (IEI, 2001d).

These issues are illuminated by the South African experience (Wolff, 2015), where higher education is driven by a socio-economic agenda, with progression through levels of a qualification framework regarded as a key enabler for a "social justice agenda" (p. 12). As part of the reconfiguration of South African engineering education towards these aims, a new, more theoretical, Bachelor of Engineering Technology (BTech) qualification has been established, to run in parallel with more traditional skills oriented technician diplomas. However, it is not clear that its theoretical emphasis meets industry requirements, and the new curriculum, which it has been argued does not clearly position the type of graduate identity it seeks to develop, has created difficulties for educators (Doorsamy & Padayachee, 2019). This is not an argument against progression, but certainly sounds a note of caution as

³⁷ See p. 23.

to how it is facilitated by the NFQ, which concern is reflected in my findings.

In considering how engineering academics construe student identity formation in contemporary Irish engineering technician education, I am not going so far as to say that level 6 and 7 engineering graduates are not technicians, but, certainly there has been a change in the graduate identity, orienting it towards preparation for further study, as opposed to an immediate role as a technician. This is captured succinctly in *"it is much more academic"* (Edith), or as articulated by Thomas, *"if we're teaching to technician level […] we need to do more skills-based learning"*. It may not have been intended that progression would reduce the skills focus of technician education, but it appears that it has contributed to change, alongside other factors, in this important aspect of IoT engineering education.

However, this change can also be regarded positively, where the more theoretical focus engendered by progression in my interviewees reinforces those aspects of engineering knowledge which encompass a hierarchical, disciplinary, discourse in curriculum (Wolff, 2015; Wolmarans, 2017). Indeed, as noted earlier, this has been suggested may be a feature of the level structure of NQFs (Baynham, 2012; CEDEFOP, 2017). Progression also appears to have an integrating effect, for my research participants, across NFQ levels, on the knowledge of engineering practice, as has been suggested is a characteristic that is essential for modern engineering education (Duderstadt, 2008). For example Gordon, Bill, and Thomas stress the relationship between projects at different NFQ levels: level 6 projects prepare students for the more involved projects at level 7, which in turn prepare students to engage with level 8 projects; a vertical discourse, "hierarchically organised" (Bernstein, 1999a, p. 159).

When the impact of progression is coupled with that of the skills emphasis of LOs, a complex picture emerges. Contradictory policies, again associated with the parallel employability and learner centred facets of Irish OBE, simultaneously promote more skills, whilst orienting programmes to be more theoretical, acting in contention with the academic discourse on technician education (IEI, 2001c, Shay, 2012; Wolff, 2015). Although my interviewees identified the influence they perceived from progression to increase the academic content of technician programmes, they were, also, and apparently in contradiction to this, generally accepting of the skills focussed LOs approach to engineering education. Furthermore, through the analysis of the fieldwork I identified that they at times held conflicting opinions on the appropriate balance of knowledge and skills in curriculum: some seeing an insufficient

emphasis on skills, whereas others felt that the NFQ supported an appropriate balance. I suggest their contradictory positions on this critical aspect of engineering education have arisen through being influenced to different degrees by the opposing pressures suggested above. This confusion is not too dissimilar to the difficulties reported for curriculum developers in decoupling the new, more theoretical, South African BTech from the more established technician diploma programmes (Doorsamy & Padayachee, 2019), although at least in that case the difficulty had been identified.

As noted earlier, it is not without precedent that contradictory, unintended, positions could arise through the use of NQFs, although the problem I had identified appeared quite novel. In considering the challenge of how to explain it more fully, I explored it through Bernstein's (2000) contention that tension can arise between official knowledge and academic disciplines. Bernstein (1999b, 2000) makes reference, without great elaboration (Singh, 2018), to the term "pedagogic schizoid position" (Bernstein, 2000, p. 71), to describe situations where policy considerations and the market drive pedagogy, as opposed to "the intrinsic value of the discourse". This can cause academics to adopt pedagogy that reflects "external contingencies" (Pausigere & Graven, 2013, p. 14) "rather than focusing upon learners' understanding of disciplinary knowledge".

This raises the question as to whether the cognitive discourse in IoT engineering education has been stable historically, or whether external contingencies have been an ongoing feature? The Regional Technical College (RTC) sector was conceived to implement government policy, and has prided itself on being reactive to industry and business needs (McLaughlin, 1999), accepting of the role external contingencies have had in its direction. In considering the influence of external contingencies associated with OBE on the pedagogic discourse, the fieldwork suggests that conflicting policy aims and effects have led to manifestation of the pedagogic schizoid position. Official knowledge is, it seems, not always coherent, and, at times, Irish educational policy is at odds both with itself and with the academic discourse of engineering technician education. I have noted earlier that engineering education struggles at times to find the appropriate balance of knowledge and skills (Prados, 1992; Johnston et al, 1996; Seeley, 1999; Dooge, 2006; Heywood, 2016; Dempsey, 2017; Edström, 2018) with the more recent debate leading to the adoption of OBE. The market has always had a role in these debates, hence Bernstein's characterisation of engineering as a region. However, in this

case the difficulty arises principally from the external, rather than, cognitive, interests.

These curriculum issues raise questions regarding the appropriateness of IoT technician engineering education. This issue is linked to the further impact I have observed from OBE on the structure of IoT engineering education, and I explore them together in the next section.

11.2.3 Structure of IoT Engineering Education

In describing the creation of the RTCs, and their subsequent evolution into the IoTs, I identified the importance with which technician education was, and continues to be, regarded from a national policy perspective (Hillery, 1963; OECD, 1964; Gol, 1965, 1967; Rafferty, 1968; McLaughlin, 1999; Barry, 2005; HESG, 2011; DES, 2016). The more recent refocussing of IoT engineering education towards honours engineering degrees, rather than on level 6 and 7 programmes, has contributed to concern that the IoTs are prone to 'mission drift' (OECD, 2004; HESG: 2011; Hannon, 2017) towards a replication "of the role played by the traditional research intensive Universities" (CIT, 2017, p. 2).

Mission drift, which has been a characteristic of non-university engineering education, both in Europe and the USA (Christensen, 2012; Christensen & Newberry, 2015), is associated with a variety of causes. These include: both students and academics seeking more prestigious credentials; a tendency towards the "gentrification" that my focus group identified; and the consequences of government funding for research. It may also be a manifestation of the pendulum effect that has been noted in engineering education which periodically emphasises and de-emphasises engineering practice (Christensen, 2012; Edström, 2018), and which was a concern of my research participants. Certainly IoT engineering education has changed quite substantially since the introduction of the NFQ, to where the majority of full-time enrolments are now on level 8 honours degrees, which, at the least, is facilitated by progression. Not all of my research participants agreed with this refocussing, in particular Thomas considered the IoTs should focus on the technician education he considered the strength of the sector, that we should "concentrate on level 7. It's down the [...] NFQ ladder we should go, not up".

It is important to consider the form IoT level 8 engineering education progression opportunities take. As noted earlier, consequential to El's decision, through their interpretation of Bologna (European Ministers of Education, 1999)³⁸, to increase the

³⁸ See p. 78 and p. 137.

educational requirements for Chartered Engineer, many five year IoT honours engineering degrees were replaced by four year honours degrees: as Edwin remarked "*everybody moved to a one year add-on*". Thus, when an IoT engineering student progresses to level 8, although they *may* be moving from technician to professional engineering education, it often entails enrolling on a programme I consider better classed as a BTech degree, suitable for accreditation as associate engineer (EI, 2014). Rather than being an example of OBE influenced policy driven reform of engineering education (Hanson & Norell, 2004; Schuster & Hees, 2010; Doorsamy & Padayachee, 2019) as EI (IEI, 2001d) sought to achieve, it was instead an 'on the ground' response by the IoTs to events.

A number of my interviewees expressed concern regarding this change. For example, Henry considered that "before I was always happy to say we're producing a student at the end of the five years who has a (professional) degree [...] different to the university student, has different abilities and we take longer to get there". Now however, with the new four year degrees that replaced then, "we're producing in the same amount of time" as the universities, "but can we really put our hand on our heart and say the student is equal but different".

In comparison to professional engineering degrees, these new degrees, which carry the same title, Bachelor of Engineering Honours, have less emphasis on theory, and more on the application of technology, offering a route for technicians to further their education whilst maintaining their applied focus. My research participants clearly regarded graduates of such programmes as engineers rather than technicians, whether their qualifications were suitable for accreditation towards professional status or not. It is not even evident that the difference between the two types of level 8 honours engineering degrees is particularly apparent to prospective students, or employers, a point raised by Alan. This is consistent with the USA experience, where, despite the plentiful job opportunities for graduates of BTech type programmes, it is in "many ways a 'stealth' profession, existing under the radar of many prospective students, other postsecondary educators, and employers" (NAE, 2017, p. 156). However, it has been argued that industry sees little distinction between the two types of graduate in overall capability, whilst respecting their different strengths (Kelnhofer et al, 2010; Land, 2012; NAE, 2017). Indeed, in the USA graduates of such qualifications are often engaged in equivalent work to professional engineering graduates, with Land (2012) proposing that "engineering technologists are engineers" (p. 1).

Discussion

Despite my concerns with regard to how the changes in curriculum and the structure of our engineering programmes have arisen, employment prospects for all engineering graduates are bright (HEA, 2017b). The knowledge economy in which Ireland is so highly invested (HESG, 2011) requires suitably qualified graduates (Skilbeck, 2003). The associated "higher value added activities that can form a sustainable basis for competitiveness" (McIver Consulting, 2003, p. 48) lead to a requirement, for both "engineers and engineering technicians", to engage with "more knowledge intensive manufacturing and logistics processes". This can lower the demand for the skills and knowledge associated with what were previously key technician activities. EI (2019d) identify that "that demand for manual skills is falling while the need for analytical thinking and innovation continues to grow" (p. 1), specifically identifying a decline in the need for "technology installation and maintenance" (p. 2).

My interviewees identified these new industry requirements, advancing it as one of the additional determining factors, alongside progression, for the changing skills/knowledge balance in technician programmes. Young (2008) argues that higher valued job opportunities need to be in place to make progression meaningful. Consideration of honours engineering graduate employment (HEA, 2017b) shows that such employment opportunities are, indeed, available, which appears to reflect an alignment of graduate capability with industry needs.

A further finding (*Figure 2-3*) is that the pattern of part-time enrolments confirms industry demand for technicians, and level 7 programmes in particular, remains strong, as predicted by government (DES, 2016), and industry (IBEC, 2015). Furthermore, HEA (2017b) indicates that, although the majority of full-time level 6 and 7 graduates go on to further study or training, for those who choose to seek employment job opportunities are readily available. Thus employers, and employees, continue to see value in contemporary level 7 technician programmes, despite my concerns regarding their reduced skills focus.

My research shows that the curriculum, and the OBE engineering education system within which it is taught, continue to serve industry requirements. However, I have argued that this continuing relevance has, to an extent, followed from unintended policy related consequences which happen to align in the direction of changing industry requirements. As I explore next, my research suggests a further enabling factor has been the resilience of IoT engineering academics approach to what constitutes engineering knowledge (de Vries, 2006; Sheppard et al, 2006; Wolff, 2013; Montfort et al, 2014; Korte, 2015; Wolmarans, 2017).

11.2.4 Knowledge

I have earlier articulated a view that knowledge is key to curriculum design (Young, 2008; Allais, 2014), particularly for engineering (Wolff, 2015; Heywood, 2016), where skills are also important. As evidenced in the fieldwork, my interviewees have maintained the engineering ontological and epistemological position of the utility of knowledge (de Vries, 2006; Montfort et al, 2014). This includes consideration of its industry and professional relevance, an important aspect of engineering educators' perspective on the purpose of education (Duderstadt, 2008; O'Byrne, 2009; Andrews et al, 2019). Although my interviewees perceive LOs as a key driver for curriculum design, significant aspects of traditional approaches remain. In particular, pre-NFQ practices for knowledge selection and acquisition are retained. The approaches the majority described seemed appropriate from my perspective as an engineering educator, allowing the identification of "appropriate forms of knowledge and ways in which they interact" (Wolff, 2015, p. 36). Additionally, as discussed earlier, the NFQ's level structure and progression seem to have inculcated a reinforcing of the hierarchical nature of much engineering knowledge, and an integrating aspect across NFQ levels for engineering practice.

Furthermore, in recognising the social nature of curriculum design, my interviewees ensure that curriculum design remains a department level activity, building upon the expertise of engineering faculty as a group. This is epitomised by Bill's comment, when discussing programme design, that, although he may not be an expert on all topics, he is assured of the overall integrity of a programme because "the knowledge is in the department", suggestive of the screening approach to curriculum design (Furst, 1958; Heywood, 2008, 2016).

As Young (2008) states, "the content and forms that the curriculum takes are not and should not be static; new curriculum forms and content will always emerge" (p. 89). The approach to knowledge and curriculum described by my interviewees seems to reflect the practicalities of engaging with the new system for "labour-market orientated 21st century engineering education" (Wolff, 2015, p. 82), whilst retaining the existing practices for knowledge acquisition they recognised as effective. Allais (2014) puts forward the view that it is centres of academic strength and excellence that are required to provide meaningful education, rather than programmes designed using LOs. I believe this strong approach towards engineering knowledge that I have observed in the fieldwork has been a key factor in enabling the relevance of our engineering education and graduates to be maintained.

11.2.5 Contributing to the Debate Regarding Engineering Curriculum

I have earlier questioned whether the stratification of engineering into professional engineering and technician career paths is necessary, and it is clear that is has at least partially arisen, and to an extent is maintained, through societal influences (Dempsey, 2017, 2018). However, the professional engineering/engineering technician dichotomy has been effective (McLaughlin, 1999), and, as I have shown, in the case of the RTC engineering education system, arose through a policy development and implementation process that involved detailed consideration of the implications for proposed actions.

I consider that the changes I have observed in IoT engineering education consequent to OBE have not had the benefit of a similar level of forethought and follow-through. Educational policy regards a supply of level 6 and 7 technician graduates as important for our economy (DES, 2016), as supported by industry requirements (IBEC, 2015). Clearly there has been a recognition that what technicians do has changed and that reforms in higher education need to be harnessed to ensure that enough technicians are educated and that they are educated appropriately (DES, 2019; EGFSN, 2013; EI, 2019e; McIver Consulting, 2003). However, not only has much of the change I have observed, as a consequence of policies which were designed to achieve this, been contingent, rather than fully planned, more worryingly it is not necessarily either fully in the direction intended, or coherent from a disciplinary perspective. As Henry said, in discussion of the structural effect of OBE on IoT engineering education, "*we haven't sat back and said 'what kind of engineer do we want to produce*".

The challenges faced by engineering educators tend to persist across the decades (Edström, 2018). Although the practical skills, the theoretical knowledge, and the application of technology are different, in some respects the pressures identified in the *Training of Technicians in Ireland* report (OECD, 1964) are still with us. However, the debate and investigation evident in the 1960s to decide what that education should constitute appear somewhat less well developed in these contemporary times, and, as I have argued, are overly focussed on the perceived benefits of OBE. However, the earlier debate did not just begin with policy makers, but, my research shows, was associated with concern amongst educators,

business interests, and civic authorities, supported by the technical education sector of the time (Warren, 1961, Latchford, 1962). IoT engineering education as a sector needs to contribute to contemporary debate, to help define what our engineering education should be (Edström, 2018) in these neoliberal times (Pawley, 2019), and I would hope, this research forms part of the effort that is required.

11.3 The Effectiveness of the NFQ

My research also speaks to the overall effectiveness of the NFQ and other similarly configured qualification frameworks (Raffe, 2009). QQI are quite positive about the benefits that stakeholders perceive the NFQ has brought to Irish education (QQI, 2017). However, my practitioner-centred findings are somewhat in contrast to their policy level perspective, and, to begin with, I make further use of the tension between the learner centred and skills focussed facets of the NFQ to provide a focus for my concerns.

I have noted earlier that transparency, a fundamental value of the NFQ, in common with other NQFs (Allais, 2014), and which is enabled through the use of LOs, can be regarded as contributing to learner-centeredness (Attard et al, 2010). However, a recent QQI Green Paper paints a complex picture of LOs, stating that "writing LOs is an art. The statements must be comprehensible at an appropriate level to learners, to teachers, to members of the discipline's wider communities of practice and to other key stakeholders" (QQI, 2018, p. 50). It is very difficult to see how LOs can be constructed with the multi-layered meaning required in order make them accessible to the different groups listed above. I would have to agree that if it were possible it would, indeed, be an 'art'. Take, for example, a LO from a mechatronics programme "analyse, design, communicate, reflect and critically evaluate technology and resource choices". This would carry significant meaning for a mechatronics academic, but the proposition that any student in the early years of study could read it and get real meaning from it is quite questionable (Cort, 2010; Young, 2011).

This concern was apparent in the fieldwork, where John described the process of trying to teach students on the basis of LOs, asking them "*do you know what a programme LO is? What is a module LO?*", but found a complete lack of awareness. John still persevered with using LOs to engage with the students as part of his pedagogy, but realised that, rather than LOs being transparent to them, he had to explain each in detail.

To investigate transparency for prospective or current students further, I used readability analysis (Reck & Reck, 2007) to examine programme outcomes of a number of engineering Higher Certificates mapped to the NFQ, and also an EQF programme, (see *Appendix C*). The resultant readability indices represent how understandable the programme outcomes are in terms of the number of years of educational attainment required to easily comprehend them. The results suggests a high degree of educational attainment, college graduate and beyond, would be required, consistent with LOs being constructed by experts, using a complex and domain-oriented terminology. However, it is not consistent with LOs assisting in qualifications being transparent to prospective or current students, particularly in the earlier years of study. This brings into question the claim that transparency "has made it easier to evaluate qualifications for [...] study" (QQI, 2017, p. 16).

A further concern I raised earlier was whether LOs even have the same meaning for all engineering academics (Heywood, 2016). However, the fieldwork shows that transparency can be regarded by engineering academics as a valuable tool to help them compare and benchmark awards. The focus group considered that, when used in combination with Bloom's taxonomy, anyone "who has written a programme or course knows what you are talking about", with the use of LOs making programmes transparent across different HEI's.

Transparency is a fundamental value of the NFQ, but it appears overly ambitious to believe it fully useful to students and prospective students, even if it has value for experts in the field.

A further area of digression from QQI's positive view of the impact of the NFQ is with regard to the significance of the use of the nomenclature of the NFQ. Collins et al (2009) identified that the NFQ "would introduce a new language and set of concepts including the levelling of qualifications, learning outcomes and award-type descriptors" (p. 9) and that "*the language of learning outcomes and Levels*³⁹ creates common reference points for identifying and assessing non-formal and informal learning" (p. 28). More recently QQI (2017) noted that a number of stakeholders agreed that "the NFQ is now part of the infrastructure and language of Irish education" (p. 13), similar to the South African experience (Tuck et al, 2004). The fieldwork revealed more than just the use of the nomenclature associated with the NFQ and EI's accreditation processes, where I observed the phenomena of the language of levels in use

³⁹ My emphasis

by my interviewees, and myself, as a language of engagement with the regulative discourse, and also in the instructional discourse of our academic practice.

I have earlier contended that this language of levels acts to emphasise progression for my research participants, but it is pervasive in the pedagogic discourse on academic matters, through which it has become part of our disciplinary terminology as engineering educators. The significance can be considered through the concept of learnification, which raises concern of the side-lining of knowledge in the modern curriculum by languages of learning (Biesta, 2005, 2014, 2015) rather than of education, such as that associated with LOs (Biesta, 2014).

A key question is whether such a language of learning rather than of education, where the curriculum emphasises skills and competences as opposed to, and at the expense of, knowledge of the field, is sufficient to represent educational concepts of "content, purpose and relationships" (Biesta, 2015, p. 76). Given that the ontology of engineering disciplines is reflected in their terminology (Moses, 2008; Heywood, 2011), it may not be compatible with generic LOs terminology (Young, 2010), such as is represented by the language of levels. For example, the focus group recognised that the language of levels was not useful in an international engineering education context. Furthermore, Heywood (2016), as discussed earlier, reported on differences in interpretation of LOs between lecturers, and between lecturers and students, suggesting both ontological (between academics) and epistemological (between academics and students) discord.

In considering the impact of OBE on the pedagogy of IoT engineering academics, my research highlights issues of more general impact across higher education. QQI (2018) quite correctly regard assessment and LOs as fundamentally interlinked. In their 2017 survey (QQI, 2017), involving government, learners/employees, higher education institutes, further education, employers, schools, and educators, they report positive benefits the NFQ has brought to assessment. 64% of respondents agreed that "the LOs Approach of the NFQ has Improved Teaching and Learning Practice" (QQI, 2017, p. 24), and nearly 70% of respondents agreed that "the LOs Approach of the NFQ has lmproved Teaching and Learning Practice" (QQI can claim success on this matter. The NFQ was bound to have an impact on teaching and learning, but my research shows that the National Qualifications Authority of Ireland (NQAI), which later subsumed by QQI, at the least did not set out what it was to be, and, perhaps, had not considered it at all.

Discussion

QQI's measure of success seems to be a circular argument that OBE promotes assessment, to which I respond that my research suggests that where previously we assessed what we taught, now we teach what we assess. QQI (2018) do raise the criticism that "teaching to the test" (p. 13) may arise, where academic integrity is compromised through academics overly focus on teaching what is to be assessed. However, surely this is a consequence of LOs, which emphasise the role of assessment over teaching, where QQI consider that when "assessment in all its dimensions is healthy it is likely that teaching and learning are also healthy" (p. 9).

I have already shown that NFQ's skills focus has not been fully effective in its influence on my research participants. However, QQI (2017) reports strong evidence of the NFQ being effective in matching employer skills requirements to qualifications, based upon an employer survey related to higher and further education and training (HEA, 2015b). Closer analysis reveals a significant proportion of employers (31%) were unaware of the NFQ, and only 36% referred to it during recruitment. Allais (2017), in a contemporary study, similarly reported on limited employer awareness of the NFQ, and this was highlighted by Rudolf in the fieldwork. This suggests, as HEA (2015b) acknowledges, that respondents to the employer survey may have been self-selecting. Given the skills emphasis, it is hardly surprising that, for those employers aware of the NFQ, the matching of their requirements to the skills focus of qualifications on the framework is apparent. However, the level of employer awareness casts further doubt on the effectiveness of the NFQ's skills focus.

11.4 OBE and the Relationship between IoT Engineering Education and Society

In considering the pivotal relationship (HEA, 2012) between engineering and society (Heywood, 2016: Pawley, 2019), a concern I share is that the neo-liberal skills emphasis of OBE could have a detrimental effect not just on engineering education, but on the society it serves (Matos et al, 2017). However, for my interviewees, our education system, whilst maintaining its industry relevance, has not become more skilled focussed under the influence of OBE. IoT engineering education has been an effective instrument of government educational policy, and its ongoing socio-economic focus, from the outset (Christensen & Newberry, 2015), and it is possible that alignment with the more recent neoliberal emphasis of government educational policy is not a major re-orientation. Reflecting the utilitarian turn in engineering education with which Heywood (2016) is concerned, it may be that we "unwittingly indoctrinate students into neoliberalism as the only possible mode of economic

development" (Pawley, 2019, p. 6).

A broader issue relates to the strengthening of participation in our democracy associated with the IoTs (Department of Education and Science, 2001; Fleming & Harfords, 2014). However, a challenge to this success is the influence of OBE, which is targeted towards producing a workforce who will be compliant with the needs of industry rather than educating a citizenry with the capacity to engage in democratic debate (Rata, 2014; Pawley, 2019). The distribution of knowledge in engineering education, and the power and control issues associated with it (Akera, 2017; Edström, 2018), reflect wider societal hierarchies (Bernstein, 2000). When I consider the performance pedagogy approach I have observed, and the impact this might have on the disposition of our graduates, I consider it does not inculcate the development of the reflective capacity which is an aspect of the ability to engage not just in effective engineering practice, but also in the democratic process (Bernstein, 2000; Nudelman, 2018).

The IoT's are particularly associated with access to education, and a key societal benefit that it is proposed will accrue from an NQF is further improvement in this regard (Carney & Schweisfurth, 2018; O'Reilly, 2018), with HEA (2015a) linking equity of access to what is good for society *and* the economy. Improved access and diversity is necessary for engineering (JEE, 2006), with El's registrar urging "students to consider all routes into an engineering sector full of job opportunities" (Owens, 2016b). However, engineering has much to do to improve diversity and access (Dempsey, 2017; El, 2017c; Pawley, 2019), and I consider El's role in the restructuring of Irish engineering education through Bologna has acted in opposition to this.

The international and Irish experience is that there is little evidence to support claims that any major positive change regarding access and diversity has resulted from OBE (Young, 2003, 2005). Lynch (2018) argues that "equality of opportunity in education" (p. 21) cannot overcome fundamental socio-economic inequalities, and Clancy (2015) writes it terms of "access to what" (p. 303) in considering equity of access to more prestigious higher education courses. Furthermore, there are concerns that the NFQ might have led to a devaluing of vocational qualifications (Allais, 2017). However, the fieldwork provided evidence that access is regarded as important by my interviewees, who consider the NFQ's level structure, and progression, act to counter elitism, which is emphasised for them through the pervasiveness of the language of levels in the pedagogic discourse. Furthermore, my research does point towards improved access to degree level engineering education for tradespersons. Fleming et al (2017), in considering the overall success of the access agenda in Irish higher education, remark that "progress has been patchier and slower than policymakers hoped" (p. 3), and the progress revealed through my research is clearly not the major success that it was anticipated would result from the NFQ. Nevertheless, progression has raised the awareness of this important matter for my interviewees and provided them with new structures within which to advance the access agenda. Consequently, some students have benefited through educational opportunities that would not otherwise have been available.

The engineering professional bodies have a larger role, where, as communities of trust (Beck & Young, 2005; Young, 2008), they mediate between engineering education and society with regard to engineering qualifications (EI, 2018b; Engineering Council, 2020). EI promote their relevance to society and the economy, and consider OBE based accreditation a core activity (EI, 2017c) in maintaining this. In recognising the importance of EI accreditation, my interviewees regarded it as a sectoral imperative, a 'must have' (Klassen, 2019, p. 21) to keep up with competitors, although they did not all consider that accreditation is fully relevant for all graduate engineers. However, accreditation has a prominence with Irish engineering educators (Homan, 2020), including my research participants, who recognise it as an important benchmark for our programmes. Rather than being unaccepting of EI's societal role, the views of a number of my research participants suggest that, instead, they were concerned that it was not sufficiently effective and relevant for their engineering fields.

11.5 Reflection on the Use of the Pedagogic Device

In the preceding discussion the complexity of what I have observed and conceptualised concerning the impact of OBE on IoT engineering education is apparent. Bernstein (1977) considers that "Curriculum, Pedagogy and Evaluation form a whole and should be treated as a whole" (p. 73). To this I would add the dispositions of the academics who construct, teach, and assess the curriculum, the academic structures within which they are situated, the policy drivers that shape them, and the wider societal context. In providing a theoretical model which seeks to "explain the entire environment of pedagogic discourse" (Halliday, 2014, p. 32), the pedagogic device offered a heuristic through which I could regard IoT engineering education, and which I consider may be of wider use to my peers. Utilising the pedagogic device in my conceptual framework required a synthesis of my personal positioning as an engineering educator within the field of inquiry, alongside socio-historic accounts of the

development and evolution of IoT engineering education since the 1960s, and the more recent transition to an OBE engineering education system. This served as a perspective to build upon the fieldwork to develop the understanding I sought.

A key issue I addressed in applying the theoretical framework was establishing the distributive rules of the pedagogic device, which Bernstein (2000) refers to, as I have noted earlier, as "who may transmit what to whom and under what circumstance" (p. 31). In my reading of the literature to gain understanding of the application of the pedagogic device I noted that authors were often quite clear in their discussion of the recontextualising rules of the educational context they were considering. However, whilst the distributive rules were generally acknowledged, this was often without any significant elaboration (Smithwick, 2001; Wolff, 2015; Klassen, 2019), although in suggesting a role in determining access to education and knowledge they seemed to me quite critical

Wheelahan (2005) gave the distributive rules more consideration, where in discussing their role in the context of vocational education and training, she related them to policy level issues related to access to education and knowledge. Further research suggested an interpretation of the distributive rules as a means to represent overarching aspects of educational policy (Kwok, 2017; Lim, 2014). This offered a heuristic through which to advance my research, where I chose to interpret the distributive rules as associated with the policy debate around the formation and development of RTC/IoT engineering education, and the later consequences of OBE for the sector. This was critical to my research, allowing me to situate the socio-historic context as it related to policy formation within the distributive rules of the device, which in turn influenced my establishment of the recontextualising rules. I consider I used this to useful effect, and I suggest it as a way forward for others.

A further important consideration was how to represent El's role. The engineering professional bodies are part of the community of trust for engineering education (Young, 2008), and could be regarded as contributing to the instructional discourse of the pedagogic recontextualising field (PRF). However, through their decisions and positioning they often have an influence more consistent with being part of the regulative discourse (Homan, 2020). My research suggested El's role pertaining to OBE aligned with the extension of official knowledge to encompass the engineering professional bodies (Klassen, 2018; Moodley, 2014). However, this may not reflect every situation, and it may be appropriate to revisit this

in the application of the pedagogic device in other engineering education contexts, allowing a role for the professional bodies in both the regulative and the instructional discourse.

The pedagogic device of IoT engineering education, as revealed through consideration of the fieldwork, is presented in *Figure 11-1*. A key influence on the distributive rules is educational policy (Kwok 2017; Lim 2014) promoting a skills agenda in higher education, in which the NFQ plays a key role. EI, as the engineering professional body also contribute to the distributive rules through their accreditation requirements, including their interpretation of Bologna.

The discourse of the PRF is seen through the analysis of the fieldwork, for the selection of engineering academics I interviewed, to be shaped by the following.

- The use of LOs, which has been established as a core part of the pedagogic discourse;
- Progression has become an important consideration in pedagogy and course structure and design, and influences the balance of skills and knowledge taught to students.
- The language of levels, which has emerged from the NFQ, is used by engineering academics in their engagement with the ORF, i.e. with the policy discourse on higher education. It has become part of the terminology of their everyday academic practice, part of the pedagogic discourse that shapes recontextualisation.

The evaluative rules determine the pedagogic practice in the field of reproduction. The analysis has shown that, for my interviewees, pedagogy follows a performance approach (Bernstein, 2000), focussed on assessment. Bernstein (2000) used the pedagogic device in applied research into particular educational contexts, adapting it where appropriate to particular situations. The pedagogic device incorporates, through the concept of the PRF, the disposition of those engaged in teaching. In considering the implication of our focus as engineering academics on assessment in the theory of instruction, I propose that the field of reproduction has become oriented more towards being a field of evaluation.

The pedagogic device can be used in conjunction with fieldwork to assist in the understanding and explanation of particular educational systems, as in my research and others (Bernstein, 2000; Wolff, 2015). Kelly (2013) proposed the pedagogic device can be used to inform policy development through comparing pedagogic approaches in different educational contexts. The pedagogic device also offers the potential for the effect of change to be anticipated, and I suggest it may also serve as a tool to allow more informed decisions to be made with regard to the development of educational policies and their implementation.

Discussion

Field of Production

Sites for creation of new knowledge

Distributive rules: skills and

competencies as what is to be taught

to new generations of students can

be seen as the consequence of the

dominant influence over the

distributive rules by government

policy promoting a skills agenda in

The NFQ promotes the teaching of

skills in higher education, in line with

El's accreditation requirements

increase vertical stratification in

higher education

government policy.

engineering education.

Field of Recontextualisation

The IoT Engineering Education Sector as a site where knowledge from the field of production are selected, rearranged and transformed to become pedagogic discourses. This pedagogic discourse includes programme and module design, and also the selection of pedagogy to be used in teaching in the field of reproduction

Recontextualising rules

The Discourse of the Official Pedagogic Field

The **ORF** is formed from the influence and prescription of the NFQ, and also El's accreditation requirement, i.e. by Official Knowledge

Aspects of this discourse include:

Government Policy, in particular the skills driven knowledge economy and related economic imperatives

Transparency, a key under-pinning of the NFQ, manifests as use of learning outcomes, leading to new course design requirements

The nomenclature of the framework, which manifests as a *Language of levels*, is part of the policy *discourse*, helping to communicate policy and shape implementation.

Progression is foregrounded

Engineers Ireland require learning outcomes for accreditation, and also influence structure of engineering education, increasing stratification

The Discourse of the Pedagogic Recontextualising Field

The **PRF** incorporates the cognitive interests of academics

Learning outcomes have become an established part of the PRF

Concept of progression has become a consideration in pedagogy and course structure

Pedagogic identities of academics are being re-shaped by the **ORF**

Language of levels has become part of the pedagogic discourse

- Part of the discourse regarding knowledge
- part of the pedagogic principles that shape recontextualisation

Field of Reproduction (Evaluation)

The IoT Engineering sector as sites of teaching and learning

Evaluative rules

Teaching is focussed on assessment. The field becomes oriented towards being a *field* of evaluation rather than a *field of reproduction*

Identity of new graduates is being formed, but technician identity is being weakened in comparison to pre-NFQ technician education

Figure 11-1 The Pedagogic Device of Engineering Education in the IoT sector as Revealed through the Analysis, adapted from Bernstein (2000)

11.6 Conclusion

In this discussion I have considered the complexity of causes for, and effects of, the impacts I have observed on IoT engineering education since the transition to OBE. The views of my research participants suggest there are conflicting educational policies at play, particularly associated with tension between the employability and learner centred facets of the NFQ. These manifest in our pedagogy; in how we balance the impact of progression vs. the skills oriented education agenda; and in the problematic nature of transparency.

I discussed my findings regarding the pedagogy of my interviewees, the curriculum they offer, and overall structure of IoT engineering education through an EER lens, which allowed me to contextualise my research in the contemporary debate on what should constitute engineering education. I was quite questioning of the impact of OBE as revealed through the fieldwork. However, my research highlighted my interviewees' enthusiasm for being better engineering educators, their student focus, and their strong approach to what constitutes engineering knowledge, which I believe have all been factors in ensuring the continuing relevance of our engineering education. I articulated concern that the changes I observed have been largely unplanned and contingent, and that IoT engineering educators need to take a more prominent role in debate as to what constitutes Irish engineering education.

Of more general relevance to the application of OBE, I highlighted areas where my practitioner oriented research perspective digresses from QQI's policy level outlook. This includes consideration of the effectiveness of the skills focus of LOs and the validity of the concept of transparency. I also discussed implications of the usage of the language of levels, which goes beyond the adoption of the nomenclature of the framework that was intended.

In discussing how my findings illuminated the relationship between engineering and society I acknowledged the role of the neoliberal drivers for OBE and highlighted: the impact on the democratic project; the relationship to the commitment to diversity and access in engineering education; and El's role as a mediator between engineering education and society.

In my reflection on the use of the pedagogic device as part of my conceptual framework to assist in my understanding of the complex field I have described, I suggested that it may of use to my peers as a means to consider related engineering educational contexts, and I highlighted particular challenges I faced, and how I resolved them.

Chapter 12 Conclusions

This is not to say that the challenges of curriculum in the contemporary age cannot be adequately *met, but that we should become understanding of the nature of the challenges before us.* Ronald Barnett and Kelly Coate (2005, p. 9).

12.1 Introduction

When I began this research into the impact of outcomes based education (OBE) on the Institute of Technology (IoT) engineering education system that I, and my colleagues, have made a career, the changes in our approach to curriculum, consequent to OBE, appeared to me so substantial that I felt sure it was a research area that required attention. From the outset, I had a feeling that our approach to pedagogy would have been impacted, and my early readings into the field suggested there might be change to how we regarded knowledge. These considerations coalesced into my research question of considering what effect the introduction of outcomes based education (OBE) has had, and is having, on engineering educators and engineering education in the Institutes of Technology (IoT) sector in Ireland, and for which, through my research, I have provided the answers presented in this thesis.

In order to achieve this, I adopted a qualitative approach for my research, opening up avenues of inquiry to address that I consider I would just not have visited otherwise. For example, the highlighting of progression, initially by Bill, in the fieldwork, led me, through careful consideration of the literature and the fieldwork, to the realisation that profound related change has occurred in our education system. My conceptual framework may have only fully developed as I moved towards the later stages of my research, but important issues that it highlights including: the stratification of knowledge; whose knowledge it is that is selected in the curriculum (Young & Muller, 2014); and the factors exerting control over our engineering education system, became apparent, if not fully conceptualised, quite early in the fieldwork.

My conceptual framework called for the generation of detailed socio-historic accounts to help me understand the context in which my research was situated. Initially these comprised material on the genesis and early development of engineering education in Ireland; and the later development of the RTC/IoT engineering education system. As I moved to consider the development and implementation of the national framework of qualifications (NFQ), and the adoption of OBE for engineering accreditation, I began to describe events in which I myself had participated: where the changes I described had a direct effect on me, and, I, in helping implement them, on others. I used my insider status to these events to useful effect in identifying and teasing out issues, and it provided me with, at times, access to important, but not readily accessible, literature, and opinion from my peers.

A limitation of my study is that I have chosen to focus the changes in IoT engineering through the lens of OBE. As Allais (2017) remarks "analyzing qualifications frameworks is complicated; impact analysis of most education and training policies are contested and complex" (p. 2). I have considered other influences, but only in so far as they directly relate to the introduction of the NFQ and Engineers Ireland's (EI's) learning outcomes (LOs) based accreditation process. This does, however, mean that my research goes beyond describing the impact of the NFQ and EI's accreditation as perceived by my research participants, in that their views were situated in the totality of the political, economic and social pressures on engineering education, and the subsequent impact that they have experienced.

The application of knowledge and skills in practice lies at the core of engineering. OBE provides a means for engineering educators to teach them together, where the LOs focus should lead to an emphasis, for students, on the importance for engineers of achieving tangible, useful, effects. However, as my research reveals, OBE oriented educational policies and associated factors, as they seek to achieve their intended impact, interact with the cognitive interests of engineering educators to produce other, and at times contradictory, contingent changes.

The next section presents the contributions of my thesis to the literature with regard to the impacts I have observed, analysed and conceptualised in my research.

12.2 Contributions of my Research to the Literature

This thesis contributes to engineering education research (EER) into engineering faculty, examining the consequences, for engineering educators, and the education they offer, of the use of OBE. My literature review revealed this to be an under-researched, but important, aspect of the field. It is the first research study of the experience of IoT engineering academics of the use of OBE, and of the effect on the education they offer. Using a qualitative methodology, and drawing from a theoretical perspective on social realist, EER literature, and other critique of OBE, it allows the voices of my research participants to speak with regard to

Conclusions

how they themselves, and the engineering education they offer, have been affected by OBE under the dual influence of the NFQ and Engineers Ireland (EI), and related policies. In doing so I offer a practitioner perspective, rather than a policy led perspective, on the impact of OBE on IoT engineering education. My contributions to the literature relate to the effect of OBE on engineering education and educators in the IoTs, the effectiveness of OBE as a policy instrument, and the representation of IoT engineering education as a pedagogic device.

An overarching contribution of my research to the literature is to the national policy discourse on engineering education, where I have set out a critical view of the effects of OBE on the technical engineering education sector in Ireland. This is in contrast to QQI's (2017) review of the impact of the NFQ, which was largely positive with regard to the benefits they perceive it has brought to education in Ireland, and EI's similarly positive view of LOs (EI, 2014).

I have shown that, for my interviewees, the 'official knowledge' of the NFQ alongside Engineers Ireland's (EI's) accreditation requirements has exerted a dominance over the pedagogic discourse of IoT engineering education. This includes: in the recontextualisation of knowledge and the balance of knowledge and skills in the curriculum; on our pedagogy; in the impact of progression; and on the overall structure of engineering education. My research provides evidence, through the fieldwork, of the influence of EI, as a community of trust (Tuck, 2007; Young, 2008) in the engineering region (Bernstein, 2000), in establishing the credibility of LOs with IoT engineering academics (Brent & Felder, 2003; Heywood, 2005, 2016).

A key contribution to the literature is to demonstrate the manner in which my research participants have given a positive, if at times qualified, acceptance to OBE. They regard it as effective: impacting positively on their pedagogy; serving as a useful framework for curriculum design; facilitating communication; and improving access to education. However, I set out a critique of the pedagogy they appear to have adopted as a consequence, question the appropriateness of the language of levels that has emerged in the pedagogic discourse, and raise concerns about the impact on the structure of engineering education.

My research reveals, through analysis of the available records, that the pedagogical basis of the NFQ received little attention during its development (NQAI, 2001a, 2001b, 2002c, 2003a). Through the fieldwork, I provide qualitative evidence, supported by related studies (Rami, 2012, Muller, 2004a), for the contention in the literature that a potential consequence of the
use of LOs is a move to performance pedagogy (Deacon & Parker, 1999; Havnes & Prøitz, 2016; Muller, 1998; Torrance, 2007).

Kenny (2006), in exploring the impact of policy on contemporary higher education in Ireland, reported that "LOs, as a new discourse, are embedded in the academic language and programme content of HE" (p. 201). In providing qualitative evidence to support this, a further contribution to the literature is to suggest, through the voice of the interviewees, the emergence of a *language of levels* consequent to the introduction of the NFQ, as is proposed will result from the use of qualification frameworks (Allais, 2010b; Raffe, 2011), and the NFQ in particular (Kenny, 2006, Collins et al, 2009). My research describes the manner in which this is used by engineering academics, in their engagement with policy discourse and also in their everyday academic lives, consistent with Biesta's (2015) concept of learnification. I question whether the use of this language of levels, in displacing the terminology of the discipline, is fully appropriate, whilst recognising the utility of this language that my research participants identify, and the necessity they feel for engaging with it and through it.

My research shows the strong influence OBE has exerted over the instructional discourse of my research participants through the use of LOs in curriculum design. Although this influence was particularly apparent in the recontextualisation of knowledge in the curriculum, my research participants showed they had maintained effective practices regarding engineering knowledge. Furthermore, my research provides qualitative evidence that the NFQ has facilitated the presence of hierarchically organised discourses (Bernstein, 1999) of engineering knowledge, as has been proposed for national qualification frameworks (NQFs) (Baynham, 2012; CEDEFOP, 2017). This allays a concern raised in the literature (Allais, 2014), and by one of my research participants (Ayah) that LO's reduce curriculum to the transmission of information rather than the teaching of knowledge. A further contribution to the literature (McLaughlin, 2001; ABET, 2006; Heywood, 2016, 2005; Owens, 2016a) is to provide qualitative evidence of the overarching influence of El's OBE accreditation requirements on IoT engineering curriculum design.

A finding revealed the NFQ as a policy tool in the re-orientation of Irish higher education towards a skills focussed emphasis. The NFQ is regarded as a key enabler for the knowledge economy through its promotion of knowledge and skills *together*, as part of life-long learning (HESG, 2011). However, the literature suggests that NQFs downplay knowledge and

emphasise a skills based education (Heywood, 2000; Corbell, 2014; Allais, 2014). Through the use of Corbell's (2014) methodology to analyse NQAI (2003a), my research confirms this skills emphasis in the case of the NFQ, which is seen as supporting the national requirement for a continuing supply of level 6 and 7 technicians (DES, 2016).

However, my research revealed the dramatic change in full-time enrolments in IoT engineering over the period 2008-2017. Whereas previously higher certificates and ordinary degrees (level 6 and 7) were the most popular, by 2017 the majority of students were enrolled in honours degrees (level 8). A finding from the fieldwork is that, influenced by the principle and consequences of the NFQ concept of progression (NQAI, 2003a), alongside other factors, my interviewees now regard level 6 and 7 programmes as preparatory programmes for level 8 degrees, leading them to emphasise theory over skills. There are two implications, the first of which is in providing qualitative evidence, in the specific case of IoT engineering, of the use of OBE leading to the weakening of boundaries between vocational and academic education (Young & Muller, 2010; Allais 2010a; Shay, 2012). The second is in providing qualitative evidence to support earlier research (Llorens et al, 2014) that suggests level 7 engineering programmes are becoming precursors to level 8 degrees rather than paths to employment.

Despite the skills emphasis of the NFQ, my interviewees held somewhat conflicting views of the correct balance of knowledge and skills in curriculum. I argue this is related to dual influences from the NFQ as a policy instrument, simultaneously promoting a more skills-based education, (HESG, 2011; DES, 2016), whilst, my research has shown, progression encourages the opposite, leading to a diminishing in the skills focus of technician programmes. I suggest that, consequent to these opposing pressures, Bernstein's (2000) pedagogic schizoid position is manifest in the differing attitudes of IoT engineering academics to the level of skills that should be taught.

A contribution to the literature, from the vantage of 2020, is to confirm McLaughlin's (2001) prediction that EI's interpretation of Bologna would have an impact on the structure of IoT engineering education, with many of the five year IoT engineering degrees, suitable in principle for accreditation towards Chartered status, being discontinued. These were replaced with four year programmes suitable for accreditation as associate engineer. I have argued that the changes I have observed in IoT engineering education consequent to OBE have, to an extent, been contingent, rather than planned, and it is not clear the educational

implications were always fully considered, with Edith relating "we did it half-heartedly because we didn't think it was right". However, there appears to be an alignment of graduate capability with employer requirements, and I propose that this continuing relevance is maintained through the strong engineering knowledge base of IoT engineering academics, and their understanding of industry requirements.

QQI (2017) makes strong claims with regard to the effectiveness of the NFQ value of transparency (NQAI, 2001b), but my research indicates the broadness of QQI's claims to be problematic. In particular my research shows that readability analysis (Reck & Reck, 2007) applied to programme outcomes of curriculum mapped to the NFQ and the EQF supports the contention that transparency cannot be fully achieved through LOs (Cort, 2010; Young, 2011). However, the fieldwork indicates that the principle of transparency is of value if regarded in a more limited fashion, in particular for communication between academics.

Consistent with the international experience of NQFs (Fernie & Pilcher, 2009; Allais, 2010a; Havnes & Prøitz, 2016), a contribution the literature is to suggest that the NFQ, and associated policies, leads to contradictory policy positions. In the case of the NFQ these are particularly associated with the tension between the learner centred facet of the NFQ and the parallel employability emphasis. The NFQ promotes more skills, but progression, alongside other factors, inculcates a more theoretical approach; pedagogy is not specified, but a performance pedagogy appears contingent, at odds with the competency model that would be associated with student-centeredness; the NFQ value of transparency, from which it follows that LOs should be accessible to all stakeholders, is shown to be problematic.

From a methodology perspective I show the applicability of the pedagogic device (Bernstein, 2000) to assist in the consideration of the impact, from the perspective of a selection of engineering academics, of OBE on IoT engineering education. I further propose that, in considering the resultant pedagogic device, under the assessment emphasis of OBE the field of reproduction of the device has become oriented towards being a field of evaluation. A contribution to the field of EER is in suggesting the pedagogic device has the potential for a wider application in research into engineering education systems.

12.3 Future Directions

A limitation of my research is that, as opposed to using a quantitative approach, it does not

provide for definitive conclusions to be taken from the findings (Donmoyer, 2008). In considering future research direction, the research could be expanded to allow for more definitive findings regarding the impact of the use OBE in IoT engineering education. For example, the impact on pedagogy could be further examined to confirm the extent and modalities (Bernstein, 2000) of the performance pedagogy approaches. My qualitative research could form the basis for a mixed methods research methodology (Creswell, 2011), informing the design of survey instruments to be used to gather quantitative data for a statistically representative sample of IoT engineering academics.

A further future research direction could be to expand the research to examine the impact in other academic areas. In particular, academic disciplines could be chosen that did not have professional bodies with the same influence as EI, or, indeed, might have associated professional bodies that had not moved so wholeheartedly to embrace LOs. The research question, in this case, would examine the role of professional bodies in the take up of OBE.

My research shows there to have been structural change, under the influence of OBE and other factors, to IoT engineering education. The views of my interviewees suggest that technician programmes have become more theoretical, in preparation for progression, for most students, to honours degrees. Other causes identified by my interviewees included the reduction in government funding for third level education, and changing industry requirements, as also identified in the literature (EI, 2019; McIver Consulting, 2007). I believe a debate is necessary, which the technical education sector should lead, in order to decide what is meant by contemporary engineering technician education in Ireland, to ensure it is aligned with our country's requirements. The stratification of engineering into technician and professional engineering should form part of that debate (Dempsey, 2017, 2018), including revisiting an earlier proposal for a parallel Chartered Engineering Technologist path (IEI, 2005a).

This leads me to future developments related to the NFQ. I was encouraged, in reading QQI (2013), by the importance which it ascribed to the influence of communities of practice in helping determine the shape of educational programmes. There was an acknowledgement that LOs are not in themselves complete descriptors, leaving space for the disciplinary expert. The ascendancy of LOs within the NFQ was diminished in that it was described as LO referenced as opposed to based, and that "not everything that matters can be expressed

using LOs." (QQI, 2013, p. 17). There was a recognition that different types of educational institution will produce different types of graduates, albeit their programmes may be mapped to the same NFQ level, which acknowledges the importance of established centres of disciplinary excellence, and the differentiation between them. In order to consider whether a programme meets the standard to which it aspires, it is not just the compliance of the programme documentation with the QQI standards that is important, but also the institutional disciplinary capacity, i.e. "textual analysis is not a sound basis for comparing standards" (QQI, 2013, p. 17). However, more recent publications make clear that LOs will continue to be in the ascendency (QQI, 2017, 2018), and, indeed, QQI are bound by statute to implement the NFQ (Irish Statute Book, 2012). However, I believe my research acts as a counterpoint to some of their claims for the NFQ, and will inform policy makers in this regard.

I believe reducing the journey to becoming an engineer to achieving a series of LOs misses the substantial, critical, point, which I raised earlier in *Chapter 2*, that it is what we learn on the way to achieving those outcomes that is important, allowing us to develop the engineering mind-set. If OBE is to be the effective educational approach its protagonists claim, we need to use it within an appropriate educational model, where we integrate what Heywood (2016) refers to as "content knowledge and understanding, theories of learning" and "a defensible theory and philosophy of assessment" (p. xiii). The literature points a way forward: Bowe and Duffy (2010), Clark and Andrews (2014), Heywood (2016) regarding educational models for engineering education; Heywood (2016) regarding assessment; Shay (2012), Wolff (2015) regarding technician vs. professional engineering education; Barnett and Coate (2005), Heywood (2008), Wolff (2015) regarding curriculum design; Muller (1998), Heywood (2016) and Deacon and Parker (1999) regarding pedagogy; and Allais (2014, 2017), Maton and Moore (2009), Young and Muller (2010) on broader issues associated with NQFs.

I believe this perspective can be inculcated in my fellow engineering academics through taking advantage of the interest identified by my research participants in training as educators. If I consider this from the vantage of a Head of Department of engineering in TU Dublin, I would be in a position to address this through the facilitation of a series of workshops where, working with my colleagues, we considered what OBE means to us as engineering educators, what type of engineering graduate we intend to educate, and whether we feel we are achieving this in our current programme structures. Further workshops would focus on particular pedagogical approaches, where we explore, for example, the applicability of constructive alignment, reflective practices in engineering education, and how the problem based learning activity within my department contributes to our students' identity formation.

I believe it will also be of benefit to my fellow engineering education researchers to publicise my work, and, through communities of practice such as the UK-Ireland Engineering education Research Network (EERN, 2021), TU Dublin's CREATE (2019) group, and SEFI (2021), I will seek to disseminate my research and engage in debate with regard to it.

12.4 Final Reflection

As I move to conclude this thesis, a final finding is my realisation of just how influenced I am, as an engineering academic manager and leader, by government policy, of yesteryear and today, and by El's positioning as a professional body. In leading the implementation of the OBE approach for engineering programmes within my IoT I helped to bring about other, related changes, the consequences of which I question, and with regard, in particular, to pedagogy, I am quite critical. In my role as head of department (HoD) I had to engage with the move to OBE, but I consider I became overly focussed on operational issues. In my defence, the move to the NFQ was initially presented as a re-packaging exercise for our existing programmes, and, indeed, that was what it was to begin with. Over the years, under the dual influences of the NFQ and EI, we have, however, experienced quite radical change in terms of the engineering education we offer.

In retrospect I should have been conscious of the need to question the claims that were made for OBE, and query the effects its introduction might have, in order to guide and support my colleagues appropriately. My research provides me with the perspective, and the opportunity, to provide more appropriate support and leadership to my colleagues going forward, to encourage them to develop and use their engineering expertise to deliver an engineering education relevant to our students and society.

The fieldwork was carried out in 2016, 4 years before the thesis was completed. The moves within our sector to consolidate IoTs into technological universities has led to creation of Technological University Dublin, where I am now employed. This changing landscape for technical education in Ireland will no doubt have an impact on the education we offer and on us as educators. At time of writing, although TU Dublin has been formed for over 2 years, the

impact on academics, the programmes we offer, and our students, has as yet been quite minimal. However, the adoption of a TU Dublin strategic plan, an impending radical change in our organisational structure, and the development of an educational model for the University will bring significant change across all levels of the engineering education we offer. My research, informed by the views of my research participants, places me in an advantageous position to contribute to the impending changes for the positive benefit of my colleagues, our students, and the industry and society in which they will find careers.

Government policy and the engineering professional bodies may dictate an OBE approach to this education, but within that we should strive to develop our students' identities as future engineers and technicians (Nudelman, 2018) rather than educate them to be level '6, 7, 8 and 9' graduates. As engineering educators, rather than emphasising assessment, we need to reinvigorate our focus on the identity formation process of educating the next generation of engineers, asking ourselves "what would you like to see in an engineer" (Thomas). Our academic expertise, our pedagogic identities, the curriculum we offer and the structure of our programmes must be aligned to allow us to inculcate in our students the engineering view of knowledge as that which is useful, along with the associated body of expertise, know-how and skills they need to usefully apply themselves in engineering practice. We must be aware of the pressures on our engineering education system as we undertake this important task. Change should be driven by, and accepted for, the betterment of engineering education and the society it helps shape, rather than come about through the unintended contingency that may result from educational reforms that have not been fully thought through.

In adopting a social realist influenced conceptual framework for my research, it is necessary to acknowledge that the description and understanding of the impact of OBE on IoT engineering education I have developed in this thesis is only one way to analyse the consequences of these educational reforms (Sayer, 2000). I offer my findings not as a definitive position but as a reasoned account (Smit, 2016) of the impact I have observed. I hope the reader agrees that this perspective, which I arrived at through the assistance of my research participants, offers insight, and where appropriate, explanations, into the impact that OBE has had on engineering educators and education in the IoT sector.

Appendices

We have finished the job, what shall we do with the tools?

Haile Selassie, as quoted in Ambrosia and Small Beer, Marsh E and Hassall C, (1964, p. 160).

Appendix A Level Indicators for the NFQ for Levels 6 to 10

	Level 6	Level 7	Level 8	Level 9
Knowledge Breadth	Specialised knowledge of a broad area	across a variety of areas and methods pertaining to a field (or fields) d flearning t		A systematic understanding of knowledge, at, or informed by, the forefront of a field of learning
Knowledge Kind	Some theoretical concepts and abstract thinking, with significant underpinning theory	Recognition of limitations of current knowledge and familiarity with sources of new knowledge; integration of concepts across a variety of areas	Detailed knowledge and understanding in one or more specialised areas, some of it at the current boundaries of the field(s)	A critical awareness of current problems and/or new insights, generally informed by the forefront of a field of learning
Know-How & Skill Range	Demonstrate comprehensive range of specialised skills and tools	Demonstrate specialised technical, creative or conceptual skills and tools across an area of study	Demonstrate mastery of a complex and specialised area of skills and tools; use and modify advanced skills and tools to conduct closely guided research, professional or advanced technical activity	Demonstrate a range of standard and specialised research or equivalent tools and techniques of enquiry
Know-How & Skill Selectivity	Formulate responses to well-defined abstract problems	Exercise appropriate judgement in planning, design, technical and/or supervisory functions related to products, services, operations or processes	Exercise appropriate judgement in a number of complex planning, design, technical and/or management functions related to products, services, operations or processes, including resourcing	Select from complex and advanced skills across a field of learning; develop new skills to a high level, including novel and emerging techniques
Competence Context	Act in a range of varied and specific contexts involving creative and non-routine activities; transfer and apply theoretical concepts and/or technical or creative skills to a range of contexts	Utilise diagnostic and creative skills in a range of functions in a wide variety of contexts	Use advanced skills to conduct research, or advanced technical or professional activity, accepting accountability for all related decision making; transfer and apply diagnostic and creative skills in a range of contexts	Act in a wide and often unpredictable variety of professional levels and ill-defined contexts
Competence Role	Exercise substantial personal autonomy and often take responsibility for the work of others and/or for the allocation of resources; form, and function within, multiple, complex and heterogeneous groups	Accept accountability for determining and achieving personal and/or group outcomes; take significant or supervisory responsibility for the work of others in defined areas of work	Act effectively under guidance in a peer relationship with qualified practitioners; lead multiple, complex and heterogeneous groups	Take significant responsibility for the work of individuals and groups; lead and initiate activity
Competence Learning to Learn	Learn to evaluate own learning and identify needs within a structured learning environment; assist others in identifying learning needs	Take initiative to identify and address learning needs and interact effectively in a learning group	Learn to act in variable and unfamiliar learning contexts; learn to manage learning tasks independently, professionally and ethically	Learn to self-evaluate and take responsibility for continuing academic/professional development
Competence Insight	Express an internalised, personal world view, reflecting engagement with others	Express an internalised, personal world view, manifesting solidarity with others	Express a comprehensive, internalised, personal world view manifesting solidarity with others	Scrutinise and reflect on social norms and relationships and act to change them

Table A-1 Level Indicators for NQF Levels 6-9 (NQAI, 2011)

Appendix B Sample Focus Group Exercise

Progression

The table below shows where the interviewees had views on progression.

	Dr	ograccion			
	Progression				
	Progression to Level 8 viewed as the Aspirational Target for Students	Influence on Technician Education	The Higher Certificate		
Interviewee					
Isambard					
Ayah					
Kees					
Rudolf					
Vitruvius					
Bill					
Thomas					
Gordon					
Edwin					
George					
Edith					
Henry					
John					
Alan					
Grace					

The old Certificate and Diploma technician qualifications have been replaced with new level 6 and 7 qualifications:

- Are they equivalent?
- Or has the skills focus has been downgraded in favour of a more analytical, theoretical, education
- Or do you teach in a manner more consistent with the claims for the knowledge economy, i.e. **knowledge and skills** together?

Are the new level 6 and 7 qualifications becoming merely pre-cursors to professional engineering qualifications (level 8)?

What has happened to the Higher Certificate? Is this due to progression, or other factors?

Appendix C Readability Analysis of Programme Outcomes

One of the claims associated with qualification frameworks is that they promote transparency with regard to qualifications (Cort, 2010). This can be taken to mean that a prospective student should be able to read the programme LOs, and understand them sufficiently well to make an informed decision with regard to whether the programme in question is one they would pursue (Lindholm, 2009). However, my research suggested that this claim is aspirational, and, in fact, the language of levels and LOs provided by qualification frameworks is largely opaque to such prospective students (and others) (Cort, 2010).

In order to investigate this I applied readability analysis (Reck & Reck, 2007), using an online tool (Online-Utility.org, 2017), to the programme outcomes of a number of NFQ level 6 awards, and to an EQF award (level 6). The utility takes text as input, and applies a number of algorithms to assess the text's readability at an initial reading. As can be seen from the results below, in each case the algorithms calculate that many more years of formal education are necessary in order to be able to easily understand the programme outcomes. Assuming someone commences college at age 18, having started primary school at 5, they would have, at that point 13 years of formal education, whereas the average number of years of formal education required for easy readability of the programme LOs, in, for example, the case of the ITB Mechatronics Higher Certificate, is 19 years. This is equivalent to someone having spent enough years in education to not be pre-college, but to already have received a bachelor's degree and master's (and even possibly a PhD). In one case (the EQF mapped Air Traffic Control Engineer) the readability analysis implies that the text is largely incomprehensible (at first reading).

This strongly suggests that programme outcomes, as written, do not provide prospective students with the transparency related to the qualification that is expected from the use of qualification frameworks.

This is supported by the following fieldwork interview extract, discussing the use of LOs with students "I'll try to convert them into [...] everyday language" (Rudolf). The interviewee recognises the difficulty of expecting students to understand learning outcomes, and instead explains them to the students in terms of more accessible terminology.

ITB Mechatronics Higher Certificate Readability Analysis, level 6 NFQ

Programme Outcomes

Display a knowledge and understanding of the mechatronic engineering discipline and the fundamental science, mathematics and technologies underpinning this discipline.

Demonstrate knowledge of the integration of the various mechatronics disciplines and relevant codes of practice.

Demonstrate proficiency in the use of workshop, laboratory and computer equipment used in the construction and testing of mechatronic products

Be able to select and apply mathematical formulae and scientific techniques to solve mechatronic engineering problems and communicate the results in an appropriate format.

Apply the knowledge and skills learned to plan and produce mechatronic engineering projects, reports and assignments, within a specified time-frame.

The ability to work both independently and as part of a team to plan, manage and organise mechatronic engineering assignments and projects.

Reflect on and evaluate the quality of his/her own learning. Knowledge of the importance of continuing professional development.

Show awareness of the significance of the wider ethical, environmental and business contexts within which engineering operates and to participate in engineering activities in a responsible manner

Readability Analysis.

Indication of the number of years of formal education that a person requires in order to easily understand the text on the first reading	
Gunning Fog index :	21.52
Approximate representation of the U.S. grade level needed to comprehend the text :	
Coleman Liau index :	19.09
Flesch Kincaid Grade level :	17.28
ARI (Automated Readability Index) :	18.06
SMOG :	17.75

CIT Electrical Higher Certificate, level 6 NFQ

Programme Outcomes

A knowledge of areas of electrical circuitry and practice, installation and equipment. A knowledge of mathematics, ICT, design, business and engineering practice relevant to the electrical engineering technician.

The ability to apply knowledge of electrical science, practice and design, to the solution of well-defined electrical engineering technology problems

The ability to use basic electrical engineering techniques, skills and modern computer-based engineering tools and packages necessary for engineering practice.

The ability to assist in the design of an electrical facility, component or process to meet specified needs and to contribute to the assessment of the technical performance of the system.

The ability to contribute to the solution of common engineering technology problems in electrical engineering.

The ability to work autonomously and safely and as a member of a multidisciplinary team in well-defined work settings.

The ability to identify and address learning needs within a structured learning environment and an awareness of the need for continued professional development.

An appreciation of the wider social, political, business and economic contest within which engineering operates and the need for high ethical standards in the practice of engineering, including the responsibilities of the engineering profession towards people and the environment

Readability Analysis

Indication of the number of years of formal education that a person requires in order to easily understand the text on the first reading	
Gunning Fog index :	22.22
Approximate representation of the U.S. grade level needed to comprehend the text :	
Coleman Liau index :	17.67
Flesch Kincaid Grade level :	19.03
ARI (Automated Readability Index) :	18.68
SMOG :	19.09

Air Traffic Control Engineer, Level 6 EQF

http://ec.europa.eu/ploteus/en/content/air-traffic-control-engineer-rai

Programme learning Outcomes

To obtain knowledge about air traffic services and organisation in accordance with the national, EU, ICAO and EUROCONTROL regulations as well as to obtain knowledge about meteorology, aircraft and navigation;

To acquire skills in management and control of air traffic, communication with the air crews about meteorology, equipment operation and flight planning, to use the data of the IT management systems for the choice of safe, efficient and rational flight routes;

To acquire skills necessary to use radio electronic and automated satellite systems for air traffic control and to manage the work of emergency, search and rescue services;

To develop a competence in air traffic control by studying the binding regulations, assuring flight security and efficient work of air traffic control service.

Readability Analysis

Indication of the number of years of formal education that a person requires in order to easily understand the text or the first reading	
Gunning Fog index :	57.65
Approximate representation of the U.S. grade level needed to comprehend the text :	
Coleman Liau index :	16.16
Flesch Kincaid Grade level :	54.43
ARI (Automated Readability Index) :	65.32
SMOG :	33.00

Appendix D, HEA Data on Engineering Enrolments

Table D-1 Enrolments in Full-Time Engineering Programmes in the IoT Sector, Based on Analysis of Data from HEA (2021).

Year	No of Level 6&7 Enrolments	No of Level 7 Enrolments	No of Level 6 Enrolments	No of Level 8 enrolments in the IoT sector	Level 6/7 as a percentage of total	Total enrolments in engineering in the IoT sector
2008-09	8418	7399	1019	3581	70	11999
2009-10	8021	6927	1094	4907	62	12928
2010-11	7582	6487	1095	4831	61	12413
2011-12	6737	5942	795	4759	59	11496
2012-13	6381	5688	693	4325	60	10706
2013-14	6148	5457	691	4278	59	10426
2014-15	5772	5176	596	4478	56	10250
2015-16	5569	5005	564	4942	53	10511
2016-17	5047	4539	508	5223	49	10270
2017-18	4705	4,283	422	5703	45	10408

Table D-2 Enrolments in Part-Time Engineering Programmes in the IoT Sector, Based on Analysis of Data from HEA (2021)

Year	No of Level 6&7 Enrolments	No of Level 7 Enrolments	No of Level 6 Enrolments	No of Level 8 enrolments in the IoT sector	Level 6/7 as a percentage of total	Total part- time enrolments in engineering in the IoT sector
2008-09	2110	785	1325	272	89	2382
2009-10	1738	956	782	340	84	2078
2010-11	1194	1013	181	313	79	1507
2011-12	1503	1057	446	390	79	1893
2012-13	1667	1098	569	508	77	2175
2013-14	1704	907	797	388	81	2092
2014-15	1848	950	898	434	81	2282
2015-16	2027	951	1076	481	81	2508
2016-17	1308	799	509	388	77	1696
2017-18	1360	904	456	413	77	1773

Appendix E The Bologna Process

The Bologna process began with the Sorbonne Declaration (Ministers of Education France, Germany, Italy and UK, 1998), which, linked to moves to reform the education systems of the initiating countries (Huisman et al, 2012), formally established moves to create an EHEA, and invited other countries to participate. The Bologna declaration (European Ministers of Education, 1999) marked the expansion of this to 29 countries, including Ireland. It sought to establish a system where degree's awarded in the EHEA could be easily compared, in order to promote employability and international competitiveness. It advocated a credit transfer system, such as the ECTS, to aid in mobility and in lifelong learning, and promoted semesterisation. It had as an objective the promotion of mobility of students, teachers, researchers and administrative staff. It promoted the concept of a common European view of higher education, alongside associated quality assurance processes.

It initially established a two-cycle degree process (an undergraduate and a graduate cycle), with a third (doctoral education) added in the Berlin Communiqué (Ministers Responsible for Higher Education, 2003) as part of continuing bi-annual ministerial meetings (Huisman et al, 2012). The Berlin Communiqué also committed the signatories to the use of OBE at a national level, in the context of an overarching European qualification framework.

Ireland and the NFQ played a role in this adoption of a qualification framework approach, e.g. through its influence on the earlier Copenhagen seminar (Bergan, 2003), which set out that "an overarching qualifications framework for the European Higher Education Area is a *conditio sine qua non* to the setting up of a European Higher Education Area". A series of meetings following the Berlin Communiqué (Joint Quality Initiative Informal Group, 2004b), culminated in the agreement of the Dublin descriptors (Joint Quality Initiative informal group, 2004a). The report setting out the Dublin descriptors makes explicit reference to the influence of the NFQ, and notes the participation of representatives of the NQAI, HETAC and the IoTs. The Dublin descriptors differentiate between the different degree cycles through outcomes specified in terms of knowledge and understanding, applying knowledge and understanding, making judgements, communication, and learning skills. This served as a defining moment in the development of the Bologna framework, in that the Dublin Descriptors enshrined the use of LOs in the Bologna process (Feeny & Horan, 2016).

The Bologna Framework (ECA, 2020), formally designated the Framework for Qualifications of the European Higher Education Area is, essentially, the three cycles agreed under Bologna, implemented using the Dublin descriptors, and was officially announced in the Bergen Communiqué (European Ministers Responsible for Higher Education, 2005). This communiqué also agreed the desirability of compatibility with the developing EQF, as later established by the Bologna Follow-Up group (BFUG, 2005). Indeed the EQF was required to build upon the achievements of the Bologna process (CoEU, 2002).

The compatibility of the NFQ with the Bologna framework was determined in 2006, and compatibility with the EQF was certified in 2009 (NQAI, 2009). This was hardly surprising given the degree of influence of the NFQ in the development of the European frameworks. See below for a comparative table of the levels of the Bologna framework, the EQF and the NFQ.

EQF Levels	Bologna Framework	lrish NFQ Levels	Irish Major Award-Types
1		1	Level 1 Certificate
1		2	Level 2 Certificate
2		3	Level 3 Certificate, Junior Certificate
3		4	Level 4 Certificate, Leaving Certificate
4		5	Level 5 Certificate, Leaving Certificate
5	Short Cycle within First Cycle	6	Advanced Certificate*, Higher Certificate
		7	Ordinary Bachelor Degree
6	First Cycle	8	Honours Bachelor Degree, Higher Diploma
7	Second Cycle	9	Masters Degree, Postgraduate Diploma
8	Third Cycle	10	Doctoral Degree, Higher Doctorate**

Table E-1 Comparison of the Bologna Framework, the EQF and the NFQ (FIN, 2017)

Bibliography

- ABET. (1995). *The Vision for Change: A Summary Report Of the ABET/NSF/Industry Workshops.* Baltimore: Accreditation Board for Engineering and Technology, Inc.
- ABET. (1998a). *Engineering Criteria 2000, 1998-2000.* Baltimore: Accreditation Board for Engineering and Technology, Inc.
- ABET. (1998b, January 31). Proposed Criteria for Accrediting Engineering Technology Programs. Retrieved December 2, 2018, from The Wayback Machine Internet Archive: https://web.archive.org/web/19980131060930/http://www.abet.org:80/TAC/propcrit.htm
- ABET. (2006). *Engineering Change: A Study of the Impact of EC2000.* Baltimore: Accreditation Board for Engineering and Technology, Inc.
- ABET. (2018a). *ABET Website*. Retrieved December 2, 2018, from Website of ABET: https://www.abet.org/
- ABET. (2018b). Criteria for Accrediting Engineering Technology Programs, 2019 2020. Retrieved December 2, 2018, from ABET: https://www.abet.org/accreditation/accreditationcriteria/criteria-for-accrediting-engineering-technology-programs-2019-2020/
- ABET. (2018c). *Criteria for Accrediting Engineering Programs, 2019 2020*. Retrieved December 2, 2018, from ABET: https://www.abet.org/accreditation/accreditation-criteria/criteria-for-accrediting-engineering-programs-2019-2020/
- Adam, S. (2004). Using Learning Outcomes: A Consideration of the Nature, Role, Application and Implications for European Education of Employing 'Learning Outcomes' at the Local, National and International Levels. *Report Presented at the United Kingdom Bologna Seminar*, (pp. 1-30). Edunburgh.
- Adams, A. (2011). *Drawing the Dragon*. Amazon, Kindle Edition.
- Adams, E. (2010). The Joys and Challenges of Semi-Structured Interviewing. *Community Practitioner*(83), 18-21.
- Adams, R., & Forin, T. (2013). Working Together Across Disciplines. In B. Williams, J. Figueiredo, & J. Trevalyn (Eds.), *Engineering Practice in a Global Context* (pp. 101-128). Boca Raton: CRC Press.
- Adelman, C. (2009). *The Bologna Process for U.S. Eyes:Re-learning Higher Education in the Age of Convergence.* Washington, DC: Institute for Higher Education Policy.
- Adelman, J. (2006). Communities of science: the Queen's Colleges and scientific culture in provincial Ireland, 1845-1875: PhD Thesis . Galway: NUIG.
- Akera, A. (2017). Setting the Standards for Engineering Education: A History. *Proceedings of the IEEE, 105*(9), 1835-1843.
- Allais, S. (2010a). National Qualifications Frameworks: Solving the Education/Labour Market 'Mismatch'? *Education and Employers Taskforce Inaugural Research Conference* (pp. 1-21). University of Warwick.

- Allais, S. (2010b). *The Implementation and Impact of National Qualification Frameworks: Peport of a Study in 16 Countries.* Geneva: International Labour Office; Skills and Employability Dept. doi: 9789221241195
- Allais, S. (2014). Selling out Education: National Qualification Frameworks and the Neglect of Knowledge. Rotterdam: Sense Publishers. doi:978-94-6209-576-2
- Allais, S. (2017). *Labour Market Impact of National Qualification Frameworks in Six Countries.* Geneva: International Labour Organization.
- Ando, H., Cousins, R., & Young, C. (2014). Achieving Saturation in Thematic Analysis: development and refinement of a codebook. *Comprehensive Psychology*, *3*(4), 1-7.
- Andrews, J., Clark, R., & Knowles, G. (2019). Life on the Chalk Front: Learning & Teaching in Engineering Education - A Reflection on Colleagues' Observations. UK & Ireland Engineering Education Research Network Annual Conference 2019. Warwick: UK & Ireland Engineering Education Research Networ.
- Andrews, J., Clarke, R., & Evans, M. (2011). Reviewing and Evaluating CDIO: An Empirical Approach to Engineering Education Curriculum Development. *Coventry University Problem Based Learning Annual*, (pp. 1-13). Coventry.
- APEL. (2001). Submission to the NQAI by the National Apel Network. Retrieved November 26, 2017, from The Wayback Machine Internet Archive: https://web.archive.org/web/20020830211116/http://www.nqai.ie:80/apel.htm
- Archer, M., Bhaskar, R., Collier, A., Lawson, T., & Norriet, A. (Eds.). (1998). *Critical Realism, Essential Readings.* London: Routledge.
- Archer, M; Decoteau, C; Gorski, P; Little, D; Porpora, D; Tutzou, T; Smith, C; Steinmetz, , G;
 Vandenberghe, F. (2016, Dec 23). What is Critical Realism. *Perspectives, A Newsletter of the ASA Theory Section*. Retrieved from http://www.asatheory.org/current-newsletter online/what-is-critical-realism
- Armstrong, J., & Lumsden, D. (1999). Impact of Universities' Promotional Materials on College Choice. *Journal of Marketing for Higher Education*, *9*(2), 83-91.
- ASEE. (2019). *Journal of Engineering Education*. Retrieved July 7, 2019, from American Society for Engineering Education: https://www.asee.org/papers-and-publications/publications/jee
- ASEE. (2020, August 28). K-12 STEM Guidelines for All Americans . Retrieved from ASEE, American Society for Engineering Education: https://www.asee.org/member-resources/councils-andchapters/corporate-member-council/special-interest-group/cmc-k12-stem-guidelines-forall-americans.pdf
- Attard, A., Di Lorio, E., Geven, K., & Santa, R. (2010). *Student Centered Learning: An Insight into Theory and Practice.* Bucharest: Education and Culture DG.
- Augusti, G. (2009).) EUR-ACE: The European Accreditation System of Engineering Education and Its Global Context. In A. Patil, & P. Gray (Eds.), *Engineering Education Quality Assurance*. (pp. 41-49). Boston: Springer.

- Augusti, G., J. Birch, J., & Payzin, E. (2011). EUR-ACE: A System of Accreditation of Engineering Programmes Allowing National Variants. *INQAAHE*. Madrid.
- Ayres, L. (2001). Semi-Structured Interviews. In L. Given (Ed.), *The SAGE Encyclopedia of Qualitative Research Methods* (pp. 810-811). London: SAGE Publishing.
- Barbour, R. (2007). Doing Focus Groups. London: SAGE Publications.
- Barnett, R., & Coate, K. (2005). *Engaging the Curriculum in Higher Education*. Maidenhead: Open University Press.
- Barret, B., & Rata, E. (2014b). Introduction. In B. Barrett, & E. Rata (Eds.), *Knowledge and the Future of the Curriculum: International Studies in Social Realism* (pp. 1-23). Basingstoke: Palgrave-MacMillan.
- Barrett, B., & Rata, E. (Eds.). (2014a). *Knowledge and the Future of the Curriculum*. Basingstoke: Palgrave-MacMillan.
- Barry, F. (2005). Third-level education, Foreign Direct Investment and Economic Boom in Ireland. *Int. J. Technology Management, 38*(3), 198-219.
- Barry, F. (2007). Foreign Direct Investment and Institutional Co-Evolution in Ireland. *Scandinavian Economic History Review*, 55(3), 262-288.
- Barry, F. (2014). Outward-Oriented Economic Development and the Irish Education System. *Irish Educational Studies*, *33*(2), 213-223.
- Baynham, M. (2012). Cultural Geography and the Retheorisation of Sociolinguistic Space. In S.Gardner, & M. Martin-Jones (Eds.), *Multiligualism, Discourse and Ethnography* (pp. 114-130).London and New York: Routledge.
- Beagon, U. (2018). The Academic Perspective: A Study of Academic Conceptions of the Importance of Professional Skills in Engineering Programmes in Ireland. *The UK & IE EERN Spring Colloquium in Newcastle.*
- Beck, J., & Young, M. (2005). The Assault on the Professions and the Restructuring of Academic and Professional Identities: A Bernsteinian Analysis. *British Journal of Sociology of Education*, 26(2), 183-197.
- Bell, L., & Stevenson, H. (2006). Education Policy Process, Themes and Impact. London: Routledge.
- Benaquisto, L. (2008). Codes and Coding. In L. Given (Ed.), *SAGE Encyclopedia of Qualitative Research Methods* (pp. 85-88). London: SAGE Publishing.
- Bergan, S. (2003). Qualification Structures in European Higher Education; Danish Bologna Seminar, Copenhagen. Retrieved November 28, 2017, from EHEA: http://media.ehea.info/file/Qualification_structures_Copenhagen_2003/10/5/030327-28Report_General_Rapporteur_576105.pdf
- Bernhard, J. (2018). Engineering Education Research in Europe Coming of Age. *European Journal of Engineering Education, 43*(2), 167-170.
- Bernstein, B. (1971). On the Classification and Framing of Educational Knowledge. In M. Young (Ed.), *Knowledge and Control* (pp. 47-69). London: Collier MacMillan.

- Bernstein, B. (1977). *Class, Codes and Control, Volume III, Towards a Theory of Educational Transmission.* New Youk: Routledge.
- Bernstein, B. (1999a). Vertical and Horizontal Discourse, an Essay. *British Journal of Sociology of Education*, 20, 157-173.
- Bernstein, B. (1999b). Official knowledge and Pedagogic Identities. In F. Christie (Ed.), *Pedagogy and the Shaping of Consciousness: Linguistic and Social Processes* (pp. 246-161). London and New York: Cassel.
- Bernstein, B. (2000). *Pedagogy, Symbolic Control and Identity*. Lanham: Rowman and Littlefield.
- Bernstein, B. (2001). Das pedagogias aos conhecimentos. Educaçao, Sociedade e Culturas, 15, 9–17.
- BFUG. (2005). Bologna Working Group on Qualifications Frameworks: Overarching Frameworks for Qualifications. Copenhagen: Bologna Follow-Up Group. Retrieved from http://media.ehea.info/file/20051012-13_Manchester/05/0/BFUG7_8a_WG-QualificationsFramework-LifelongLearning_584050.pdf
- Bhat, S., Bhat, S., Raju, R., D"Souzac, R., & Binu, K. (2019). Collaborative Learning for Outcome Based Learning. *9th World Engineering Educactioin Forum 2019* (pp. 927-936). Chennai: Elsevier.
- Biesta, G. (2005). Against Learning. Reclaiming a Language for Education in an Age of Learning. *Nordisk Pedagogik, 25,* 54–66.
- Biesta, G. (2009). Good Education in an Age of Measurement: On the Need to Reconnect with the Question of Purpose in Education. *Educational Assessment Evaluation and Accountability*. 21., 21(1), 1-14. Retrieved September 10th, 2019, from https://www.researchgate.net/publication/41529875_Good_Education_in_an_Age_of_Measurement_On_the_Need_to_Reconnect_with_the_Question_of_Purpose_in_Education
- Biesta, G. (2014). Pragmatising the Curriculum: Bringing Knowledge Back into the Curriculum Conversation, but via Pragmatism. *The Curriculum Journal, 25*(1), 29-49.
- Biesta, G. (2015). What is Education For? On Good Education, Teacher Judgement, and Educational Professionalism. *European Journal of Education*, *50*(1), 75-87.
- Biggs, J. (2014). Constructive Alignment in University Teaching. *HERDSA Review of Higher Education*, 5-22.
- Biggs, J., & Tang, C. (1999). *Teaching for Quality Learning at University*. Maidenhead: McGraw-Hill.
- Bjornavold, J., & Coles, M. (2010). Added Value of National Qualifications Frameworks in Implementing the EQF. Luxemburg: Publications Office of the European Union. doi:978-92-79-15214-6
- Blackie, M., le Roux, K., & McKenna, S. (2016). Possible Futures for Science and Engineering eEucation. *Higher Education*, 755-766.
- Bloom, B. (1956). *Taxonomy of Educational Objectives: Book 1 Congitive Domain*. London: Longman Group.
- Bobbit, F. (1918). Scientific Method in Curriculum-Making. In D. Flinder, & S. Thornton (Eds.), *The Curriculum Studies Reader* (2nd (2004) ed., pp. 9-16). New York: Routledge.

- Boddy, C. (2016). Sample size for Qualitative Research. *Qualitative Market Research: An International Journal, 19,* 426-432.
- Borderieux, J., & Jolly, A. (2017). Use of Agile Methods in the Education of Engineering Students. *Proceedings of the 46th SEFI Annual Conference* (pp. 629-235). Azores, Portugal: SEFI.
- Borrego, M., & Bernhard, J. (2011). The Emergence of Engineering Education Research as an Internationally Connected Field of Enquiry. *Journal of Engineering Education, 100*(1), 14-47.
- Borrego, M., & Streveler, R. (2014). Preparing Educators for Engineering . In A. Johri, & B. Olds (Eds.), *Cambridge Handbook of Engineering Education Research* (pp. 467-478). New York: Cambridge University Press.
- Bourdieu, P. (1971). Systems of Education and Systems of Thought. In *Knowledge and Control* (pp. 189-207). London: Collier MacMillan.
- Bourdieu, P. (1990). The Logic of Practice. (R. Nice, Trans.) Stanford: Stanford University.
- Bowe, B., & Duffy, G. (2010). A Strategy for the Development of Lifelong Learning and Personal Skills throughout an Undergraduate Engineering Programme. *IEEE conference Transforming Engineering.* Retrieved from https://arrow.tudublin.ie/cgi/viewcontent.cgi?article=1124&context=engscheleart
- Brent, R., & Felder, R. (2003). A Model for Engineering Faculty Development. *International Journal of Engineering Education*, 19(2), 234-240.
- Brey, P. (2014). The Physical and Social Reality of Virtual Worlds. In M. Grimshaw (Ed.), *The Oxford Handbook of Virtuality* (pp. 42-54). Oxford University Press.
- Brinkman, S., & Kvale, S. (2011). *Interviews: Learning the Craft of Qualatitive Research Interviewing*. London: Sage.
- Bruce, C. J. (1997). Engineering in Ireland. *Engineering Science and Education Journal, 6*(4), 183-184.
- Bucciarelli, L., Coyle, E., & McGrath, D. (2009). Engineering Education in the US and the EU. In S.Christensen, B. Delahouse, & M. Megank (Eds.), *Engineering in context* (pp. 105-128,).Aarhus: Academica.
- Buckley, J., Niall Seery, N., Gumaelius, I., Canty, D., Doyle, A., & Pears, A. (2020). Framing the Constructive Alignment of Design within Technology Subjects in General Education. *International Journal of Technology and Design Education*. doi:https://doi.org/10.1007/s10798-020-09585-y
- Bulman, U., Bornhoft, S., Vosgerau, K., Ellinger, D., & Knutzen, S. (2017). *Combining Teaching and Research in Engineering*. Azores, Portugal: Proceedings of the 46th SEFI Annual Conference.
- Caesar, J. (n.d.). Commentaries on the Civil War. Retrieved from http://mcadams.posc.mu.edu/txt/ah/Caesar/CaesarCiv02.html
- Campus France. (2016). The Titre d'ingénieur in France (National Master's degree in Engineering). Retrieved from Campus.france.org: https://ressources.campusfrance.org/catalogues_recherche/diplomes/en/ingenieur_en.pdf
- CAO. (2017). Central Applications Office. Retrieved May 20, 2017, from http://www.cao.ie

Carney, S., & Schweisfurth, M. (Eds.). (2018). *Equity In and through Education*. Leiden: BrillSense.

- Carroll, D., & Gallery, R. (2006). Rapid-Prototyping Emulation System using a SystemC Control System Environment and Reconfigurable Multimedia Hardware Development Platform. *WSEAS Transations on Circuits and Sytems, 5*(8), 1198.
- Case, J. (2015). A Social Realist Perspective on Student Learning in Higher Education: the Morphogenesis of Agency. *Higher Education Research & Development, 34*(5), 841-852.
- Cavendish, L. (2011). *Stories of International Teachers: A Narrative Inquiry about Culturally Responsive Teaching, PhD Thesis.* University of Iowa: Iowa Research Online.
- CDIO. (2020, January 21). UK and Ireland European Annual Meeting of the CDIO initiative. Retrieved from CDIO: http://www.cdio.org/meetings-events/uk-and-ireland-european-annual-meeting-cdio-initiative
- CDIO. (2021, May 18). CDIO Home Page. Retrieved from Conceive Design Implement and Operate: http://www.cdio.org/
- CDIO. (2021, May 1). Conceiving Designing Implementing Operating (CDIO). Retrieved from www.cdio.org: http://www.cdio.org/
- CEC. (1995). White Paper on Education and Training: Towards the Learning Sociery. Luxemburg: Commission of the European Communities, Publications Office of the European Union.
- CEDEFOP. (2009). *The Shift to Learning Outcomes: Policies and Practices in Europe*. Luxembourg: Office for Official Publications of the European Communities.
- CEDEFOP. (2011). Using Learning Outcomes European Qualifications Framework Series: Note 4. Luxemburg: Publications Office of the European Union.
- CEDEFOP. (2015a). Global Inventory of Regional and National Qualifications Frameworks Volume I: National and Regional Cases. Hamburg: UNESCO Institute for Lifelong Learning.
- CEDEFOP. (2015b). Global Inventory of Regional and National Qualifications Frameworks Volume II: National and Regional Cases. Hamburg: UNESCO Institute for Lifelong Learning.
- CEDEFOP. (2017). *Defining, Writing, and Applying, Learning Outcomes*. Luxemburg: Publications Office of the European Union.
- Charmaz, K. (2006). *Constructing Grounded Theory: A Practical Guide Through Qualitative Analysis.* London: SAGE Publications Ltd.
- Cheville, A. (2016). A Philosophical Perspective on ABET's Proposed Changes to Criterion 3. 123rd
 Annual Conference, American Society of Engineering Education. New Orleans:
 https://peer.asee.org/a-philosophical-perspective-on-abet-s-proposed-changes-to-criterion-3.
- Christensen, S. (2012). Academic Drift in European Professional Engineering Education: The End of Alternatives to the University. In S. Christensen, C. Mitcham, B. Li, & Y. An (Eds.), Engineering, Development and Philosophy: American, Chinese and European Perspectives (pp. 145-147).
- Christensen, S., & Newberry, B. (2015). The Role of Research in Academic Drift Processes in European and American Professional Engineering Education Outside the Universities. In S.

Christensen, C. Didier, A. Jamison, M. Meganck, C. Mitcham, & B. Newberry (Eds.), International Perspectives on Engineering Education: Engineering Education and Practice in Context, Volume 1 (pp. 33-69).

- Christensen, S., Meganck, M., & Delahousse, B. (2007a). Definitions of Key Concepts. In S.Christensen, M. Meganck, & B. Delahousse (Eds.), *Philosophy in Engineering* (pp. 409-417).Aarhus: Authors and Academia.
- Christensen, S., Meganck, M., & Delahousse, B. (Eds.). (2007b). *Philosophy in Engineering.* Aarhus: Academia.
- CIT. (2017). TU Criteria Consulation Response. Retrieved February 2017, from Cork Institute of Technology: http://www.cit.ie/contentfiles/TU%20Criteria%20-%20CIT%20response%20to%20HEA%20FINAL%20031011.pdf
- CIT. (2021, MAy 2021). SCHOOL OF MECHANICAL, ELECTRICAL & PROCESS ENGINEERING 2013 -2015. Retrieved from Munster Technological University: https://www.cit.ie/aboutcit/reports_plansandpolicies/programmatic-reviews/schoolengineering
- Clancy, P. (1993). Goal Enlargement and Differentiation: The Evolution of the Binary Higher Education System in Ireland. In C. Gellert (Ed.). London and Philadelphia: Jessica Kingsley.
- Clancy, P. (2008). The Non-University Sector in Irish Higher Education. In J. Taylor, J. Brites, M. Machado, & R. Santiago (Eds.), *Non-University Higher Education in Europe* (pp. 123-145). Dordecht: Springer.
- Clancy, P. (2015). *Irish Higher Education: A Comparitive Perspective*. Dublin: Institute of Public Administration.
- Clandinin, D. J. (2006). Narrative Inquiry: A Methodology for Studying Lived Experience. *Research Studies in Music Education*, 27(1), 44-54.
- Clark, H., & Wasow, T. (1998). Repeating Words in Spontaneous Speech. *Cognitive Psychology, 37*, 201–242.
- Clark, R., & Andrews, J. (2014). Relationships, Variety & Synergy: the vital ingredients for scholarship in engineering education? A case study. *European Journal of Engineering Education*, 39(6), 585-600.
- Clarke, A. (2006). Anonymising Research Data. ESRC National Centre for Research Methods.
- Clarke, L., & Winch, C. (2015). Have Anglo-Saxon Concepts Really Influenced the Development of European Qualifications Policy? *Research in Comparative and International Education*, 10(4), 593–606.
- COBE. (2007). Using Learning Outcomes. The Open University.
- CoEU. (2002). COUNCIL RESOLUTION of 27 June 2002 on Lifelong Learning. Council of the European Union.
- Coffield, F., Edward, I. F., A, H., Spours, K., Steer, R., & Gregson, M. (2007). How Policy Impacts on Practice and How Practice Does Not Impact on Policy. *British Educational Research Journal*, 33(5), 723-741.

- CoHE. (1967a). *Commission on Higher Education, I, Presentation and Summary of Report.* Dublin: The Stationary Office.
- CoHE. (1967b). Commission on Higher Education, II. Report. Dublin: The Stationary Office.
- CoHE IoT. (2001). Submission to the NQAI by Council of Heads of Engineering of the Institutes of Technology Regarding the Proposed National Framework of Qualifications. Retrieved November 25, 2017, from The Wayback Machine Internet Archive: https://web.archive.org/web/20020901115849/http://www.nqai.ie:80/chseit.htm
- Coles, M. (2016). National Qualifications Frameworks Reflections and Trajectories. QQI.
- Collins, J. W., & O'Brien, N. P. (2011). *Greenwood Dictionary of Education*. Santa Barbara: Greenwood.
- Collins, T., Kelly, f., Murdoch, H., Raffe, D., & Murphy, A. (2009). *Framework Implementation and Impact Study*. Dublin: QQI.
- Conlon, E. (2008). Integrating Engineering Ethics and Research Skills in a First Year Programme. International Symposium for Engineering Education 2008, (pp. 145-56). Dublin. Retrieved from https://arrow.tudublin.ie/cgi/viewcontent.cgi?article=1011&context=schmuldistcon
- Conlon, E. (2015). Engineering Ethics: Ontology and Politics. *Research in Engineering Education Symposium*, (pp. 1-9). Dublin. Retrieved from https://arrow.tudublin.ie/cgi/viewcontent.cgi?article=1009&context=schmuldistcon
- Connolly, B. (2001). *Submission to the NQAI*. Retrieved November 26, 2017, from The Wayback Machine Internet Archive: https://web.archive.org/web/20020831221706/http://www.nqai.ie:80/bridconnolly.htm
- Cooke, J. (1999). The Dublin Mechanics' Institute, 1824-1919. Dublin Historical Record, 52(1), 15-31.
- Coolahan, J. (1981). Irish Education: History and Structure. Dublin: Institute of Public Administration.
- Coolahan, J. (1990). The Commission on Higher Education, 1967, and Third Level Policy in Contemporay Ireland. *Irish Eudcational Studies*, *9*(1), 1-12.
- Coolahan, J. (1994). National Education Convention (1993 : Dublin, Ireland). Dublin: The Secretariat.
- Corbell, C. (2014). The Missing Voice of "Knowledge" in Knowledge and Skills. In B. Barrett, & E. Rata (Eds.), *Knowledge and the Future of the Curriculum* (pp. 104-122). Basingstoke: Palgrave-MacMillan.
- Corden, A., & Sainsbury, R. (2006). *Using Verbatim Quotations in Reporting Qualitative Research.* University of York: Report to the ESRC.
- Cort, P. (2010). Stating the Obvious: the European Qualifications Framework is not a neutral evidence-based policy tool. *European Educational Research Journal*, *9*(3), 304-316.
- Cort, P. (2014). Trailing the Unpredictable Pathways of European Union Lifelong Learning Policy: Methodological Challenges. In M. Milana, & J. Holford (Eds.), Adult Education Policy and the European Union (pp. 127-140). Rotterdam: Sense Publishers.
- Cox, R. (2019). *Called To Serve Two, Presidents of the Institution of Engineers of Ireland 1969-2018.* Dublin: Engineers Ireland.

- Cox, R., & Callanan, F. (2006). The Irish Chartered Engineering Institution. In R. Cox (Ed.), *Engineering Ireland* (pp. 55-88). Cork: The Collins Press.
- Cox, R., & Dwyer, D. (2014). Called to Serve. Dublin: Engineers Ireland.
- Coyle, E., & Rebow, M. (2009). Sustainable Design: a Case Study in Energy Systems. Chapter 16 in Engineering in Context. In S. Christensen, B. Delahouse, & M. Meganck (Eds.), *Engineering in Context* (pp. 309-326). Coyle, E; Rebow, M.
- Coyle, E., Murphy, M., & Grimson, W. (2007). Engineering Science as Opposed to Natural Science and Applied Science. In S. Christensen, M. Meganck, & B. Delahousse (Eds.), *Philosophy in Engineering* (pp. 139-160). Aarhus: Authors and Academica.
- CREATE. (2019). *CREATE Contributions to Research in Engineering & Applied Technology Education*. Retrieved October 3, 2019, from TU Dublin: http://www.dit.ie/create/
- Creswell, J. (2011). Controversies in Mixed Methods Research. In N. Denzin, & Y. Lincoln (Eds.), *The SAGE Handbook of Qualitative Reserch Methods* (pp. 269-284). London: SAGE publishing.
- Crogan, P. (2010). The Nintendo Wii, Virtualisation and Gestural Analogics. *Culture Machine*, 11(1), 82-101.
- Cross, J., Ekawati, E., Fukahori, S., Obi, S., Saito, Y., & Tan, P. (2017). Development of a Mechanical Engineering Test Item Bank to Promote Learning Outcomes-Based Education in Japanese and Indonesian Higher Education Institutions. *Tuning Journal*, *5*(1), 41-73.
- CSO. (1970). Census 1966-Volume 7, Education. Dublin: The Stationary Office.
- Cullen, P. (1996, August 30). *NCEA Threatens Legal Action over Report*. Retrieved November 11, 2017, from Irish Times: https://www.irishtimes.com/news/ncea-threatens-legal-action-over-report-1.81298
- Cummins, B. (2013, April). *Institutes of Technology are Still Victims of Academic Elitism*. Retrieved February 2017, from Irish Times: http://www.irishtimes.com/news/education/institutes-oftechnology-are-still-victims-of-academic-elitism-1.1352957
- Dáil Éireann Debate. (12 Feb 1998). Written Answers. National Qualifications Framework.
- Dáil Éireann Debate. (1966). December 1st, Vol. 225 No. 13. 2007-2074.
- Dáil Éireann Debate. (26 March 1998). Written Answers. National Qualifications Framework.
- Daniels, J., Wood, S., & Kemnitzer, S. (2011). The Role of NSF's Department Level Reform Program in Engineering Education Practice and Research. Advances in Engineering Education, 2(4). Retrieved from https://advances.asee.org/publication/the-role-of-nsfs-department-levelreform-program-in-engineering-education-practice-and-research/
- Davies, J. (2008). Part-time Undergraduate Study in Civil Engineering Students from the Workplace. *Engineering Education, 3*(1), 21-29,.
- DCU. (2019). Prof Dermot Brabazon. Retrieved Oct 2, 2019, from School of Mechanical & Manufacturing Engineering: https://www.dcu.ie/mechanical_engineering/people/dermotbrabazon.shtml

de Graaff, E. (2017). Ten Years in Engineering Education Research: looking back ahead. *European Journal of Engineering Education*, 42(6), 587-590.

de Vries, M. (2006). Teaching and Learning the Nature of Technology. Proceedings of the 14th Annual SAARMSTE Conference. http://www.saarmste.org/images/Conference_Proceedings/SAARMSTE2006-University_of_Pretoria/deVriesMJ.pdf.

- Deacon, R., & Parker, B. (1999). Positively Mystical. An interpretaton of South Africa's Outcomes Based National Qualifactions Framework. In C. Balchin (Ed.), *Changing Curriculum, Studies on Outcomes-based Education in South Africa* (pp. 59-76). Kenwny: Juta and Co.
- Dempsey, R. (2017). Gateway or Gatekeeper: Engineering Technology as a Pathway for African Americans into the Field of Engineering. *The Bridge, National Academy of Engineering, 47*(2), 19-26.
- Dempsey, R. (2018). *The Role of Engineering Technology as a Pathway for African Americans into the Field of Engineering, PhD Thesis.* Atlanta: Georgia Institute of Technology.
- Denscombe, M. (2003). *The Good Research Guide for Small Scale Social Research Projects*. Maidenhead: McGraw-Hill.
- Denzin, N., & Lincoln, Y. (2011). Introduction: Discipling the Practice of Qualitative Research. In N. Denzin, & Y. Lincoln, *The SAGE Handbook of Qualitative Research* (pp. 1-20). London: SAGE publications.
- Department of Education and Science. (2001). *Report of the Action Group on Access to Third Level Education.* Dublin: Government Stationary Office.
- DES. (1998). *Green Paper:Adult Education in an Era of Lifelong Learning*. Department of Education and Science. Dublin: The Stationary Office.
- DES. (2000). *Learning for Life: White Paper on Adult Education*. Department of Education and Science. Dublin: The Stationary Office.
- DES. (2005). *Circular Letter: Arrangements for Progression from Assistant Lecturer to Lecturer.* Ireland: Department of Education and Science.
- DES. (2016). Ireland's National Skills Strategy 2025. Department of Education and Skills.
- DES. (2019). *Technology Skills 2022: Ireland Third ICT Skills Action Plan.* Dublin: Department of Education and Skills.
- Dewey, J. (1907). The School and Society. Chicago: University of Chicago Press. (1907).
- Dewey, J. (1916). Democracy and Education (reprint ed.). Harvard University: McMillian.
- DHET. (2014). *Policy Document on Differentiation in the South Afcian Post-School System*. Pretoria, Republic of South Africa: Departmeny of Higher Education and Training.
- Dick, P. (1966). Now Wait for Last Year. London: Orion Publishing Group.
- DIT. (2009). Enhancing Student Learning through Assessment and Feedback. Dublin Institute of Technology.

DIT. (2017, March). *DIT Postgraduate Diploma Third Level Learning and Teaching DT906*. Retrieved from

http://www.dit.ie/catalogue/Programmes/Details/DT906?tab=Programme%20Structure

- DKIT. (2021, May 22). Course at DKIT. Retrieved from courses.DKIT.ie: https://courses.dkit.ie/
- Donmoyer, R. (2008). Quantitative Research. In L. Given (Ed.), *The SAGE Encyclopedia of Qualitative Research Methods* (pp. 713-18).
- Donnelly, R. (2008). Lecturers' Self-Perception of Change in their Teaching Approaches: Reflections on a Qualitative Study. *Educational Research*, *50*(3), 207-222.
- Dooge, J. (2006). Engineering Training and Education. In R. Cox (Ed.), *Engineering Ireland* (pp. 36-54). Cork: Collins Publishing.
- Doorsamy, W., & Padayachee, K. (2019). Conceptualising the Knower for a New Engineering Technology Curriculum. *Journal of Engineering Design and Technology, 17*(4). Retrieved from https://www.researchgate.net/publication/333070752_Conceptualising_the_knower_for_a _new_engineering_technology_curriculum/link/5dc557a792851c8180378034/download
- Duderstadt, J. (2008). *Engineering for a Changing World: A Roadmap for the Future of Engineering Pratice, Research and Education.* Michigan: University of Michigan, Millenium Project.
- Duff, T. (2011). Reflections on Ireland's Education/Training Policy Making-Process Leading to the National Framework of Qualifications: National and International Influences. *European Journal of Qualifications*(2).
- Duff, T., Hegarty, J., & Hussey, M. (2000). *The Story of the Dublin Institute of Technology*. Dublin: DIT Arrow.
- Duffy, G. (2017). Spatial Thinking in the Engineering Curriculum: an Investigation of the Relationship Between Problem Solving and Spatial Skills Among Engineering Students, PhD Thesis. Dublin: DIT.
- Duffy, G., & Bowe, B. (2014). An Analysis of Consistency in, and between, Ontology, Epistemology and Philosophical Perspective as Contained Studies of the First Year Experience. *Frontiers in Education Conference (FIE)*. IEEE.
- Dukas, H., & Hoffmann, B. (Eds.). (2013). *Albert Einstein, The Human Side: Glimpses from His Archives.* Princeton: Princeton University Press.
- Dwyer, S., & Buckle, J. (2009). The Space Between: On Being an Insider-Outsider in Qualitative Research. *International Journal of Qualitative Methods 2009, 8*(1), 54-63.
- EC. (2008a). *Explaining the European Qualifications Framework for Lifelong Leaning*. European Commission. Luxemburg: Office for official publications of the European Communities.
- EC. (2008b). The European Qualifications Framework for Lifelong Learning(EQF). European Commission. Luxemburg: Office for Official Publications of the European Communities. doi:978-92-79-08474-4
- EC. (2010). New Skills for New Jobs: Action Now. A Report by the Expert Group on New Skills for New Jobs. European Commission. European Union.

- ECA. (2020, July 29). Framework for Qualifications of the European Higher Education Area. Retrieved from European Consortium for Accreditation: http://ecahe.eu/w/index.php/Framework_for_Qualifications_of_the_European_Higher_Edu cation_Area
- Edström, K. (2017). Exploring the Dual Nature of Engineering Education: Opportunities and Challenges in Integrating the Academic and Professional Aspects in the Curriculum, Doctoral Thesis in Technology and Learning. Stockholm: KTH Royal Institute of Technology.
- Edström, K. (2018). Academic and Professional Values in Engineering Education: Engaging with History to Explore a Persistent Tension. *Journal of Engineering Studies, 10*(1), 38-65.
- Edström, K. (June 12-16, 2016). Aims of Engineering Education Research The Role of the CDIO Initiative. *Proceedings of the 12th International CDIO Conference,* (pp. 974-985). Turku, Finland: Turku Univesity.
- Edström, K., & Malmqvist, J. (2010). *The CDIO Approach to Engineering Education: Designing and Intergrated Curriculum*. Retrieved from CDIO: http://www.cdio.org/files/document/file/TW2A_Presentation.pdf
- EERN. (2021, May 3rd). *UK-Ireland Engineering Education Research Network*. Retrieved from Royal Academic of Engineering: https://hefocus.raeng.org.uk/eern-home/
- EGFSN. (2013). *Future Skills Requirements of the Manufacturing Sector to 2020*. Expert Group on Future Skills Needs and ForFás.
- EI. (2010). *Procedure for Accreditation of Engineering Education Programmes.* Engineers Ireland. Engineers Ireland.
- EI. (2012). *Path to Chartered Engineer post 2012*. Retrieved December 4, 2017, from Engineers Ireland:

https://www.engineersireland.ie/EngineersIreland/media/SiteMedia/services/accreditation/ GuidanceNotesPost2012.pdf

- EI. (2014). Accreditation Criteria for Professional Programmes. Engineers Ireland. Engineers Ireland.
- EI. (2017a). Chartered Engineer Accredited Programmes. Retrieved December 4, 2017, from Engineers Ireland: http://www.engineersireland.ie/Services/Accredited-Courses/Chartered-Engineer.aspx
- EI. (2017b). *About Engineers Ireland*. Retrieved December 4, 2017, from Engineers Ireland: http://www.engineersireland.ie/About.aspx
- EI. (2017c). Engineers Ireland Strategy 2017-2020. Dublin: Engineers Ireland.
- EI. (2018, December 10). Accredited Engineering Programmes. Retrieved from Engineers Ireland: http://www.engineersireland.ie/Services/Accredited-Courses.aspx
- El. (2018b). The State of Ireland 2018: A review of infrastructure in Ireland. Dublin: Engineers Ireland.
- EI. (2019a, November 15). CPD Accredited Organisations. Retrieved from Engineers Ireland: http://engineersireland.ie/cpd-training/cpd-accredited-employer-scheme/accreditedcompanies.aspx

- EI. (2019b). *Chartered Engineer*. Retrieved August 28, 2019, from Engineers Ireland: http://www.engineersireland.ie/Membership/Registered-Titles/Chartered.aspx
- EI. (2019c). Engineering Your Future. Retrieved November 2019, 2019, from Steps to Engineering: https://issuu.com/thinkmedia/docs/engineersireland_engineeringyourfut?e=1242192/9508 135
- EI. (2019d). Engineering 2019: A Barometer of the Engineering Profession. Dublin: Engineers Ireland.
- EI. (2019e, October). Engineers Ireland conference on future of engineering education. Retrieved from Engineers Ireland: https://www.engineersireland.ie/Brexit/conference-generatesinsights-on-future-of-engineering-education
- EI. (2020a, April` 6). Associate Engineer. Retrieved from Engineers Ireland: https://www.engineersireland.ie/Professionals/Membership/Registered-professionaltitles/Associate-Engineer
- EI. (2020b, April 6). *Engineering Technician*. Retrieved from Engineers Ireland: https://www.engineersireland.ie/Professionals/Membership/Registered-professionaltitles/Engineering-Technician
- ENAEE . (2015). EUR-ACE[®] Framework Standards and Guidelines. ENAEE (European Network for Engineering Accreditation). Retrieved from http://www.enaee.eu/wp-assetsenaee/uploads/2015/04/EUR-ACE-Framework-Standards-and-Guidelines-Mar-2015.pd
- ENAEE. (2017). *The ENAEE 10th Anniversary E-Book.* Retrieved from ENAEE: https://www.enaee.eu/wp-assets-enaee/uploads/2012/01/ENAEE-10th-Anniversary-E-Book-single-page.pdf
- Engels, F. (1845). *The Condition of the Working Class in England*. Retrieved from https://www.marxists.org/archive/marx/works/download/pdf/condition-working-classengland.pdf
- Engineering Council. (2020, Nov 22). *Home Page*. Retrieved from Engineering Council: https://www.engc.org.uk/
- ESG. (2015). *Standards and Guidelines for Quality Assurance in the European Higher Education Area.* Brussels: Eurashe.
- ESIB. (2002). European Student Handbook on Quality Assurance in Higher Education. National Union of Students of Europe. Retrieved from http://www.aic.lv/bolona/Bologna/contrib/ESIB/QAhandbook.pdf
- European Communities. (1992). Maastricht Treaty. *Treaty on European Union*. Luxembourg: Office for Official Publications of the European Communities.
- European Ministers of Education. (1999, November 28). *The Bologna Declaration*. Retrieved 2017, from Eurashe: https://www.eurashe.eu/library/modernising-phe/Bologna_1999_Bologna-Declaration.pdf
- European Ministers Responsible for Higher Education. (2005, May 19-20). The European Higher Education Area - Achieving the Goals: Communiqué of the Conference of European Ministers Responsible for Higher Education, Bergen. Retrieved from EHEA:

http://media.ehea.info/file/2005_Bergen/52/0/2005_Bergen_Communique_english_58052 0.pdf

- Expert Group in Future Funding for Higher Education. (2016). *Investing in National Ambition:A Strategy for Funding Higher Education*. Dublin: Department of Education and Skills.
- Feeny, S., & Horan, C. (2016). The Bologna Process & the European Qualifications Framework (EQF):
 A Routines Approach to Understanding the Emergence of Educational Policy Harmonisation From Abstract Ideas to Policy Implementation. In J. Hogan, & M. Howlett (Eds.), *Policy Paradigms in Theory and Practice. Studies in the Political Economy of Public Policy* (pp. 198-216). London: Palgrave-Macmillan.
- Felder, R. (2012). Engineering Education: A Tale of Two Paragims. In B. McCabe, M. Pantazidou, & P. D (Eds.), Shaking the Foundations of Geo-Engineering Education (pp. 9-14). Leiden: CRC Press.
- Felder, R., Brent, R., & Prince, M. (2014). Engineering Instructional Development: Programs, Best Practices and Recommendations. In *The Cambridge Handbook of Engineering Education Research* (pp. 409-436). New York: Cambridge University Press.
- Fensham, P. (2004). *Defining an Identity: The Evolution of Science Education as a Field of Research.* Boston: Kluwer Academic.
- Fernie, S., & Pilcher, N. (2009). National Qualification Frameworks: Developing Research Perspectives. *Quality in Higher Education*, 15(3), 221-232.
- Fetterman, D. (2001a). Fieldwork. In L. Given (Ed.), *The SAGE Encyclopedia of Qualitative Research* (pp. 347-348). London: SAGE Publishing.
- Fetterman, D. (2001b). Ethnography. In L. Given (Ed.), *The SAGE Encyclopedia of Qualitative Research* (pp. 288-292). London: SAGE Publishing.
- Figueiredo, A. (2008). Toward an Epistemology of Engineering. *Workshop on Philosophy of Engineering* (pp. 94-95). London: The Royal Academy of Engineering.
- FIN. (2009). University awards and the National Framework of Qualifications: Issues around the Design of Programmes and the Use and Assessment of Learning Outcomes. Dublin: Irish Universities Association.
- FIN. (2011). *Recognition of Prior Learning (RPL) in the University Sector.* Framework Implementation Network.
- FIN. (2017). University Sector Framework Implementation Network. Retrieved March 2017, from http://www.nfqnetwork.ie
- Fingleton, A., Lougnane, M., McGuiness, L, & McKenna, K. (2014). *Towards Gender Balance in Engineering*. Dublin: UCD Engineering Graduates Association.
- Fink, L. D., Ambrose, S., & Wheeler, D. (2005). Becoming a Professional Engineering Educator: A New Role for a New Era. *Journal of Engineering Education, 94*(1), 185-194.
- Finnegan, F. (2016). The Future is Unwritten: Democratic adult education against and beyond neoliberalism. *The Adult Learner, 10*(4), 46-58.

- Firmin, M. (2001a). Unstructured Interview. In L. Given (Ed.), *SAGE Encyclopedia of Qualitatiive Research Methods* (p. 907). London: SAGE Publishing.
- Firmin, M. (2001b). Structured Interview. In L. Given (Ed.), *The SAGE Encyclopedia of Qualitative Research Methods* (pp. 837-838). London: SAGE Publishing.
- Fitzmaurice, M. (2013). Constructing Professional Identity as a New Academic: A Moral Endeavour. *Studies in Higher Education, 38*(4), 613-622.
- Fitzsimons, C. (2015). *Re-Kindling Community Education in Neoliberal times, PhD Thesis.* Maynooth University.
- Fitzsimons, C. (2017a). *Community Education and Neoliberalism : Philosophies, Policies and Practices in Ireland.* London: Palgrave Macmillan.
- Fitzsimons, C. (2017b). Rhetoric and reality: The Irish experience of Quality Assurance. *Irish Journal* of Adult and Community Education, 15-32.
- Fleming, T., Loxley, A., & Finnegan, F. (2017). *Access and participation in Irish Higher Education*. Palgrave Macmillan.
- Forero Rueda, M., & Gilchrist, M. (2011). Innovations in Undergraduate Engineering Mechanics Education: Use of Team-Based Research-Led Project Methods for Large Student Cohorts. International Journal of Engineering Education, 27(4), 821–830.
- Forester, E. (1909). The Machine Stops. *The Ofxord and Cambridge Review*. Retrieved from http://www.visbox.com/prajlich/forster.html
- Fox, A., & Patterson, D. (2013). Is the New Software Engineering Curriculum Agile. *IEEE Software*, pp. 85-88.
- Frace, K., Latanision, R., & Pearson, G. (Eds.). (2016). *Engineering Technology Education in the United States.* The National Academies Press.
- Frezza, S., & Nordquest, D. (2015). Engineering Insight. The Philosophy of Bernard Lonergan Applied to Engineering. In R. Korte, M. Mina, I. Omidvar, S. Frezza, D. Nordquest, & A. Cheville, "Philosophical and Educational Perspectives on Engineering and Technological Literacy, II. http://lib.dr.iastate.edu/ece_books/2.
- Friedman, A., Hogg, K., Nadarajah, K., & Pitts, R. (2017). *Professional Body Accreditation in Higher Education Institutations in Ireland*. Quality and Qualifications Ireland.
- Froyd, J., & Lohmann, J. (2014). Chronological and Ontological Developments of Engineering Education as a Field of Scientific Study. In A. Johri, & B. Olds (Eds.), *Cambridge Handbook of Engineering Education Research* (pp. 3-26). New York: Cambridge University Press.
- Froyd, J., Wankat, P., & Smith, K. (2012). Five Major Shifts in 100 Years of Engineering Education. *Proceedings of the IEEE, 100,* 1344-1360.
- Fugard, A., & Potts, H. (2016). Supporting thinking on sample sizes for thematic analyses: a quantitative tool. *International Journal of Social Research Methodology*, *18*(6), 669-684.
- Furst, E. (1958). Constructing Evaluation Instruments. New York: David McKay.

- Gallery, R, & Trew, T. (1997). US Patent US6173077B1:Image Segmentation. United States Patent and Trademark Office.
- Gallery, R. (1989). *Development of a Practical Model of the Cochlea using Digital Filters; M. Eng. Sc Thesis.* Galway: University College Galway.
- Gallery, R. (1996). Virtual Playspace. SIGGRAPH '96 ACM SIGGRAPH 96 Visual Proceedings: The art and interdisciplinary programs of SIGGRAPH '96 (p. 90). New York: ACM.
- Gallery, R. (1999a). US Patent US5900849: Display Headset. United States Patent and Trademark Office.
- Gallery, R. (1999b). US Patent US5973689:Cursor Control With User Feedback Mechanism. United States Patent and Trademark Office.
- Gallery, R. (2013). Mechatronics Education at Institute of Technology Blanchardstown. *unpublished article*.
- Gallery, R. (2015). My Intellectual Position, Assignment for Ed.D coursework, unpublished.
- Gallery, R., & Ballesty, D. (1999). Slow Motion Audio using VIOLA. *Irish Signals and Systems Conference.*
- Gallery, R., & Bliss, N. (2000). US Patent US6034690: Post-processing generation of focus/defocus effects for computer graphics images. United States Patent and Trademark Office.
- Gallery, R., & De Bruyn, A. (2002). US Patent US6493293: Disc controlling apparatus for displaying the status of playback information. United States Patent and Trademark Office.
- Gallery, R., & Gibson, B. (1993). Low-Cost Virtual Reality. In K. Warkwick, J. Gray, & D. Roberts (Eds.), Virtual Reality in Engineering (pp. 137-150). Institution of Electrical Engineers.
- Gallery, R., & Kuijpers, H. (2003). US Patent US6643495: Communication system and device for use with both a broadband distribution network and a telephone network. United States Patent and Trademark Office.
- Gallery, R., & Shakya, D. (2004). Hardware/Software Partitioning and Simulation with System C. WSEAS Transations on Computers, 3(1), 111-114.
- Gallery, R., & Trew, T. (1992). An Architecture for Face Classification. *IEE Colloquium on Machine Storage and Recognition of Faces* (pp. 2/1-2/6). London: IEE.
- Gallery, R., & Trew, T. (1994). US Patent US5561718A: Classifying Faces. United States Patent and Trademark Office.
- Gallery, R., McSweeney, D., & Smith, M. (2009, September 16). SECASE. Retrieved August 30, 2018, from The Wayback Machine Internet Archive: https://web.archive.org/web/20090916161752/http://www.secase.eu/
- Gallery, R., Shah, J., & Heron, D. (1999). US Patent US5982353: Virtual body modeling apparatus having dual-mode motion processing. United States Patent and Trademark Office.
- Gallery, R., Trew, T., Thanassas, D., & Badiqué, E. (1992). Automatic Face Location to Enhance Videophone Picture Quality. *British Machine Vision Conference* (pp. 488-497). British Machine Vision Association.

- Garvin, T. (2004). *Preventing the Future: Why was Ireland so poor for so long?* Dublin: Gill and Macmillan.
- Gleeson, M. (1956). *Development of Technical Education in Dublin*. Dublin: Association of Technical Institutions.
- Glynos, J., & Howarth, D. (2008). Critical Explanation in Social Science: a Logics Approach. Schweizerische Zeitschrift für Soziologie. Revue suisse de sociologie., 34, 5-35.
- GMIT. (2019). Programmatic Review School of Engineering. Engineering. Galway-Mayo Institute of Technology. Retrieved from https://www.gmit.ie/sites/default/files/public/general/docs/programme-review-schoolengineering.pdf
- Gol. (1958). *First Programme for Economic Expansion. Dublin:*. Government of Ireland. Dublin: Stationery Office.
- Gol. (1964). *Second Programme for Economic Expansion. Dublin: Stationery Office.* Government of Ireland. Dublin: Stationary Office.
- Gol. (1965). *Investment in Education-Report of Survey Team Appointed by the Minister for Education, in Cooperation with the OECD.* Government of Ireland. Dublin: The Stationary Office.
- Gol. (1967). Steering Committee on Technical Education Report to the Minister for Education on Regional Technical Colleges. Irish Department of Education and Skills publications. Dublin: The Government Stationary Office.
- Gol. (1992). Education for a Changing World. Government of Ireland. Dublin: The Stationary Office.
- Gol. (1995). *Charting our Educational Future, White Paper on Education.* Government of Ireland. Department of Education and Science.
- Goold, E. (2012). *The Role of Mathematics in Engineering Practice and in the Formation of Engineers, PhD Thesis.* Maynooth: NUIM.
- Granville, G. (2003). "Stop Making Sense": Chaos and Coherence in the Formulation of the Irish Qualifications Framework. *Journal of Education and Work, 16*(3), 257-270.
- Grimson, W. (2007). Engineering An Inherently Philosophical Enterprise. In S. H. Christensen, M.
 Meganck, & B. Delahousse (Eds.), *Philosophy in Education* (pp. 89-102). Aarhus: Authors and Academia.
- Guest, G., Brunce, A., & Johnson, L. (2006). How Many Interviews are Enough? *Field Methods, 18*, 59-82.
- Guy, K. (2010). Forweword. In T. R. Engineering, *Philosophy of Engineering Volume 1 of the* proceedings of a series of seminars held at The Royal Academy of Engineering (p. 3). London: The Royal Academy of Engineering.
- Haase, S. (2014). Professional Identity and Role of the Engineer in a Challenged Society Empirical Investigation of Engineering Student Conceptions, PhD Thesis. Aarhus: Aarhus Univerity.
- Halliday, L. (2014). An Analysis of How Knowledge is Differentiated in a Vocationally Based Curriculum for a New Profession, PhD Thesis. Cape Town: University of Cape Town.

- Hannon, M. (2017). *Examining Shifts in Institutional Positioning in the Evolving Irish Higher Education System, PhD Thesis.* Bath: University of Bath.
- Hanson, M., & Norell, M. (2004). Restructuring of Mechanical Engineering Education. *Design Education* (pp. 27-34). Delft: The Design Society.
- Havnes, A., & Prøitz, T. (2016). Why Use Learning Outcomes in Higher Education? Exploring the Grounds for Academic Resistance and Reclaiming the Value of Unexpected Learning. Educational Assessment Evaluation and Accountbility, 28, 205–223.
- Hazelkorn, E., & Moynihan, A. (2010). Ireand: The Challenges of Building Research in a Binary Higher
 Education Culture. In S. Kyvik, & B. Lepori (Eds.), *The Research Mission of Higher Education Institutions outside the University Sector* (pp. 175-198). London: Springer.
- Hazelkorn, E., Gibson, A., & Harkin, S. (2015). From Massification to Globalisation: Reflections on the Transformation of Irish Higher Education. In K. Rafter, & M. O'Brien (Eds.), *The State in Transition* (pp. 235-260). Dublin: New Island Books.
- HEA. (2012). *Code of Governance of Irish Institutes of Technology*. Dublin: Higher Education Authority.
- HEA. (2014). *HEA Recurrent Grant Funding Model: Summary.* Retrieved January 2017, from Higher Education Authority: http://www.hea.ie/sites/default/files/rgam_summary_2014_final.pdf
- HEA. (2015a). *National Plan for Equity of Access to Higher Education 2015-2019*. Higher Education Authority.
- HEA. (2015b). National Employer Survey. Higher Education Authority, QQI, Solas.
- HEA. (2016). Key Facts and Figures: Higher Education 2015/16. Higher Education Authority.
- HEA. (2017a). Review of the Funding Model for Higher Level Institutions: Working Paper 6 Cost Drivers and the Costing System Underpinning Higher Education. Retrieved Septmeber 6, 2019, from Higher Education Authority: https://hea.ie/assets/uploads/2017/06/HEA-RFAM-Working-Paper-6-Costs-of-Higher-Education-Provision-06217.pdf
- HEA. (2017b). Graduate Outcomes Survey. Dublin: Higher Education Authority.
- HEA. (2021, May 18). *Statistics*. Retrieved from Higher Education Authority: http://hea.ie/statisticsarchive/
- Hector, D., Christensen, C., & Petrie, J. (2018). Toward a Practical Philosophy of Engineering: Dealing with complex problems from the sustainability discourse. In C. Mitcham, B. N. Li, & B. ZHANG (Eds.), *Philisophy of Engineering, East and West.* Springer.
- Heeney, A., & Foster, J. (2010). Integrating Sustainable Development Into The Undergraduate Engineering Curriculum Through A Mandatory First Year Engineering Design Course At The University Of Toronto . *Knowledge Collaboration & Learning for Sustainable Innovation* (pp. 1-27). Delft: ERSCP-EMSU conference.
- HESG. (2011). *National Strategy for Higher Education to 2030*. Higher Education Strategy Group. Dublin: Department of Education and Skills.

- HETAC. (2001). *Submission to the NQAI by HETAC.* Retrieved November 26, 2017, from The Wayback Machine Internet Achive: https://web.archive.org/web/20020906124328/http://www.nqai.ie:80/hetac.htm
- Hewlett, L. (2013). *Recontextualising Knowledge in the Curriculum in Public Management, PhD Thesis.* Johanessburg: University of the Witwatersrand.
- Heywood, J. (2000). Assessment in Higher Education, Student Learning, Teaching, Programmes and Institutions. London: Jessica Kingsley Publishers.
- Heywood, J. (2005). *Engineering Education: Research and Development in Curriculum and Instruction.* Hoboken: John Wiley & Sons.
- Heywood, J. (2008). Screening Curriculum Aims and Objectives Using the Philosophy of Education. *38th ASEE/IEEE Frontiers in Education Conference* (pp. S4H1-6). Saratoga Springs: IEEE.
- Heywood, J. (2011). A Historical Overview of Recent Developments in the Search for a Philosophy of Engineering Education. *41st ASEE/IEEE Frontiers in Education Conference*. Rapid City: IEEE.
- Heywood, J. (2016). *The Assessment of Learning in Engineering Education: Practice and Policy*. Wiley-IEEE press.
- Heywood, J. (2017). *The Human Side of Engineerinng*. San Rafael: Morgan & Claypool.
- Hillery, P. (1963, May 20). Post-Primary Education. Hillery Papers, UCD Archives.
- Hoadley, U. (2015). Michael Young and the Curriculum Field in South Africa. *Journal of Curriculum Studies*, *47*(6), 733-749.
- Hodge, S. (2010). Student-Centred Learning in Higher Education and Adult Education, Occasional Papers at Unisa. Retrieved November 6, 2019, from Semantic Scholar: https://pdfs.semanticscholar.org/b12a/22e383f26d690b209e002778082f58cc1b24.pdf
- Hofmeyer, A., & Scott, C. (2007). Moral Geography of Focus Groups with Participants Who Have Preexisting Relationships in the Workplace. *Qualitative Research Methods*, *6*(2), 69-79.
- Homan, E. (2020). *The Engineer in Society: an exploration of the treatment of ethics in engineering education and practice: Doctoral Thesis.* Maynooth: Maynooth University.
- Hugo, W., Bertram, C., Whitty Green, W., & Naidoo, D. (2008). Bernstein, Bloom and the Analysis of Pedagogy in South African Schools. *Journal of Education, 43*, 31-56.
- Huisman, J., Adelman, C., Hsieh, C., Shams, F., & Wilkins, S. (2012). Europe's Bologna Process and its Impact on Global Higher Education. In D. K. Deardorff, H. de Wit, J. D. Heyl, & T. Adams (Eds.), *The SAGE Handbook of International Higher Education,* (pp. 81-100). Thousand Oaks: Sage Publications.
- Hyland, A. (2014). The Investment in Education Eeport 1965 Recollections and Reminiscences. *Irish Educational Studies, 3*(2), 123-139.
- Hyland, T., Buckley, J., Seery, N., Gordon, S., & Canty, D. (2018). Assessing design activity in engineering education: A proposed synthesis of adaptive comparative judgement and the CDIO framework. *Proceedings of ASEE Engineering Design Graphics Division 72nd Mid-Year Conference.* Montego Bay: ASEE.

IBEC. (2015). National Skills Strategy 2015-2025. IBEC.

- IEA. (2015a). 25 Years: The Washington Accord. International Engineering Alliance.
- IEA. (2015b). A History of the International Engineering Alliance and its Constituent Agreements: Toward Global Engineering Education and Professional Competence Standards. International Engineering Alliance.
- IEA. (2016). Educational Accords. Retrieved December 4, 2017, from International Engineering Alliance: http://www.ieagreements.org/assets/Uploads/Documents/Policy/IEA%20Rules%20and%20 Procedures%20(3%20June%20206).pdf
- IEI. (2001a). Submission to the NQA by Institution of Engineers of Ireland Regarding the Proposed National Framework of Qualifications. Retrieved November 25, 2017, from The Wayback Machine Interet Archive:

https://web.archive.org/web/20020906125111/http://www.nqai.ie:80/ieii.htm

- IEI. (2001b, January 6th). Institute of Engineers Ireland Accreditation. Retrieved February 26, 2019, from The Wayback Machine Internet Archive: https://web.archive.org/web/20010106004400/http://www.iei.ie/accred/accreditation.html
- IEI. (2001c, November). The Bologna Declaration and Engineering Education in Ireland. *Seminar Proceedings.* Dublin: Institute of Engineers of Ireland.
- IEI. (2001d, July). The Bologna Declaration and Engineering Education in Ireland: Submisson to Government. Retrieved December 4, 2017, from Institute of Engineers Ireland: The Wayback Machine Internet Archive:

https://web.archive.org/web/20030419143846/http://www.iei.ie:80/govsubmit/bologna.pd f

- IEI. (2003, June 8th). Institute of Engineers of Ireland Accreditation Criteria. Retrieved February 26, 2019, from The Wayback Machine Internet Archive: https://web.archive.org/web/20030608094644/http://www.iei.ie/Accred/accofeng.pdf
- IEI. (2004). Annual Report 2003-2004 . Dublin: The Institution of Engineers of Ireland.
- IEI. (2005a). Chartered engineering Technologist, a Proposal. Personal Copy.
- IEI. (2005b). Annual Report 2004-2005. Dublin: Insitute of Engineers of Ireland.
- IEI Western Region. (2000, October 16). *History of Engineering at NUI, Galway*. Retrieved from Realized Vision: http://www.realizedvision.com/nuig.php
- IGIP. (2021). International Society for Engineering Pedagogy. Retrieved from http://www.igip.org/
- Ingram, N. (2016). Time past: Impacts of ICT on the Pedagogic Discourse in the Interactive Project. *Technology, Pedagogy and Education, 25*(1), 1-18.
- Irish Statute Book. (1926). University Education (Agriculture and Dairy Science) Act. Office of the Attorney General.
- Irish Statute Book. (1969). The Institution of Civil Engineers of Ireland (Charter Amendment) Act. Office of the Attorney General.
- Irish Statute Book. (1979). *National Council for Educational Awards Act*. Office of the Attorney General.
- Irish Statute Book. (1992). Regional Technical Colleges Act. Office of the Attorney General.
- Irish Statute Book. (1999a). Regional Technical Colleges Act. Office of the Attorney General.
- Irish Statute Book. (1999b). *The Qualifications (Education and Training) Act.* Office of the Attorney General.
- Irish Statute Book. (2006). Institute of Technology Act. Office of the Attorney General.
- Irish Statute Book. (2012). *Qualifications and Quality Assurance (Education and Training) Act.* Attorney General of Ireland.
- Irish Statute Book. (2018). Technological University Act 2018. Dublin: Dáil Éireann.
- IT Sligo. (2019). Programmatic Review. Sligo: Faculty of Engineering and Design, IT Sligo.
- ITB. (2006). *Delegation of Authority Submission*. Blanchardstown, Dublin: Institute of Technology Blanchardstown.
- ITB. (2006). School of Informatics and Engineering Programmatic Review. Institute of Technology Blanchardstown. Retrieved from https://www.itb.ie/aboutitb/QualityAssuranceReports/Archived%20Reports/Programmatic %20Review%20School%20of%20Informatics%20and%20Engineering%202%20June%202006. pdf
- ITB. (2015). Programmatic Review School of Informatics and Engineering. Institute of Technology Blanchardstown. Retrieved from https://www.itb.ie/aboutitb/QualityAssuranceReports/Programmatic%20Review%202015% 20-%20School%200f%20Informatics%20and%20Engineering.pdf
- ITB. (2016, April 8). *Approved Programmes at Institute of Technology Blanchardstown*. Retrieved April 8, 2016, from courses.itb.ie: courses.itb.ie
- ITB. (2017). Institute Review Policy. Retrieved December 9th , 2018, from ITB Quality Assurance Policies: http://www.itb.ie/AboutITB/QualityAssurancePolicies/2MP31%20Institute%20review%20po licy%20-%2026th%20May%202017.pdf
- ITB. (2018, November 26). *Engineering*. Retrieved from Institute of Technology Blanchardstown: http://www.itb.ie/StudyatITB/engineering.html
- ITT. (2001). Submission to the NQAI by Institute of Technology Tallaght. Retrieved November 26, 2017, from The Wayback Machine Interet Archive: https://web.archive.org/web/20020907070903/http://www.nqai.ie:80/ittallaght.htm
- JEE. (2006). Editorial: The Research Agenda for the New Discipline of Engineering Education. *Journal* of Engineering Education, 95(4), 259-261.
- Jesiek, B., Borrego, M., Beddos, K., & Hurtado, M. (2009). Internationalising Engineering Education Research: Mapping Countries and Keywords to Identify New Collaborative Horizons. *Proceedigs of the American Society for Engineering Education Annual Conference* (pp. 14.792.1-14.792.16). Austin Texas: ASEE.

- Jessup, G. (1991). *Outcomes: NVQs and the Emerging Model of Education and Training*. London: Falmer Press.
- Johnston, S., Lee, A., & McGregor, H. (1996). Engineering as Captive Discourse. *Techné: Research in Philosophy and Technology*, 1(3/4), 128-136.
- Johri, A., & Olds, B. (Eds.). (2014). *The Cambridge Handbook of Engineering Education Research*. New York: Cambridge University Press.
- Joint Quality Initiative informal group. (2004a, October 18). Shared 'Dublin' descriptors for Short Cycle, First Cycle, Second Cycle and Third Cycle Awards: A report from a Joint Quality Initiative informal group. Retrieved November 28, 2017, from NVAO website: https://www.nvao.net/system/files/pdf/Dublin%20Descriptors.pdf
- Joint Quality Initiative Informal Group. (2004b). *Agenda*. Retrieved from Joint Quality Informal Group: The Wayback Machine Intenet Archive: https://web.archive.org/web/20040825112038/http://www.jointquality.org:80/
- Kamberlis, G., & Dimitriadis, G. (2011). Focus Groups: Contingent Articulations of Pedagogy, Politics and Inquiry. In N. Denzin, & Y. Lincoln, *The SAGE Handbook of Qualitative Research* (pp. 545-562). London: SAGE Publishing.
- Kamberlis, G., & Dimitriadis, G. (2014). Focus Group Research: Retrospect and Prospect. In P. Leavy (Ed.), *The Oxford Handbook of Qualitative Research* (pp. 315-340). Oxford: Oxford University Press.
- Karwat, D., Eagle, W., Woodridge, M., & Princen, T. (2014). Activist Engineering: Changing Engineering Practice By Deploying Praxis. *Sci Eng Ethics*.
- Keenahan, J., & McCrum, D. (2021). Collaborative teaching of engineering and architecture students. Retrieved from Engineers Ireland: https://www.engineersireland.ie/Covid-19-informationbase/collaborative-teaching-of-engineering-and-architecture-students
- Kelham, B. (1967). The Royal College of Science for Ireland (1867-1926). *Studies: An Irish Quarterly Review, 56*(223), 297-309.
- Kelly, J. (2001, November 20). The Benefits of a 4 Year engineering Degree. *The Bologna Declaration and Engineering Education in Ireland, The Institution of Engineers of Ireland*, pp. 25-45.
- Kelly, K. (2009). Response Of Academics To Demands For Change In Engineering Education. *The IGIP Conference*. Graz, Austria: IGIP.
- Kelly, K. (2010). A Doctoral Thesis Examining Change in a HEI in Ireland: Changing Universities and the response of Academics to Change in the Dublin Institute of Technology . Dublin: Dublin Institute of Technology.
- Kelly, K., & Bowe, B. (2011). Qualitative Research in Engineering Education. *118th. ASEE Annual Conference and Exposition*. Vancouver: American Society for Engineering Education.
- Kelly, M. (2005). The Effects of Increasing Numbers of Mature Students on the Pedagogical Practices of Lecturers in the Institutes of Technology. *Irish Educational Studies, 24*(2-3), 207-221.
- Kelly, P. (2013). Comparative Pedagogy: making sense of cultural complexity. *Research in Comparative and International Education, 8*(4), 415-427.

- Kelnhofer, R., Strangeway, R., Chandler, E., & Petersen, O. (2010). Future of Engineering Technology, a Proposal. ASEE Annual Conference & Exposition (pp. Page 15.608.1-15.608.12). Louisville: American Society for Engineering Education.
- Kennedy, D. (2007). *Writing and Using Learning Outcomes: A Practical Guide.* Cork: University College Cork.
- Kennedy, D., Hyland, A., & Ryan, N. (2006). *Writing and Using Learning Outcomes: A Practical Guide.* Cork: University College Cork.
- Kenny, A. (2006). The emergence of Quality Assurance in Irish Higher Education. *level3*(4). Retrieved February 24, 2016, from http://level3.dit.ie/html/issue4/aidan_kenny_paper_2/kenny2.pdf
- Kenny, A. (2009). The Dynamics of Human Capital and the World of Work: Towards a Common Market in Contemporary Tertiary Education. *Level 3, 9*.
- Kenny, A. (2010). *Exploring Contemporary Higher Education and Academic Work in Ireland: The Impact of Policy on Practice: PhD Thesis.* Dublin: TCD.
- Kenny, A., Clarke, M., & Loxley, A. (2015). Creating a Supportive Working Environment for Academics in Higher Education:Country Report Ireland. (1 ed.). Dublin: The Teachers Union of Ireland and the Irish Federation of University Teachers.
- Kersten, S. (2018). Approaches of Engineering Pedagogy to Improve the Quality of Teaching in Engineering Education. In J. Drummer, G. Hakimoc, M. Joldoshov, T. Köhler, & S. Udartseva (Eds.), *Vocational Teacher Education in Central Asia* (pp. 129-139). Cham: Springer Open.
- Klassen, M. (2018). *The Politics of Accreditation: A Comparison of the Engineering Profession in Five AngloSphere Countries, Master of Arts in Higher Education.* Toronto: Ontario Institute for Studies in Education, University of Toronto.
- Klassen, M. (2019). Extending Bernstein's Pedagogic device to Higher education and the Professions. *Cambridge Symposium on Knowledge in Education* (pp. 1-29). Cambridge: The Cambridge Symposiums. Retrieved December 17, 2019, from https://www.researchgate.net/publication/334611924_Extending_Bernstein%27s_pedagogi c_device_to_higher_education_and_the_professions
- Knowles, G., Clark, R., & Andrews, J. (2019). Herding Cats? Reflections on Colleagues' Perceptions of Learning & Teaching in Engineering Education. SEFI (pp. 639-650). Budapest: European Society for Engineering Education.
- Korte, R. (2015). Pragmatism, Practice, and Engineering. In R. Korte, M. Mina, I. Omidvar, S. Frezza,
 D. Nordquest, & A. Cheville, "Philosophical and Educational Perspectives on Engineering and Technological Literacy, II. http://lib.dr.iastate.edu/ece_books/2.
- Kvale, S. (2006). Dominance Through Interviews and Dialogues. *Qualitative Inquiry*, 12(3), 480-500.
- Kwok, H. (2017). Bernstein's Sociological Theory of Pedagogic Device and its Relevance for Researching Curriculum Politics in Hong Kong. *The 90th Annual Meeting of the Japan Sociological Society* (pp. 1-20). Tokyo: https://www.researchgate.net/publication/317661034_Bernstein%27s_sociological_theory_

of_pedagogic_device_and_its_relevance_for_researching_curriculum_politics_in_Hong_Kon g.

- Kyne, M. (2019). Incorporation of the Programmatic Review and Accreditation Process in Engineering Education. UK & IE EER Spring Colloquium: Building Capacity and Growing Collaborations. UK and Ireland Engineering Education Research Network.
- Laguador, J., & Dotong, C. (2014). Knowledge versus Practice on the Outcomes-Based Education Implementation of the Engineering Faculty Members in LPU. *International Journal of Academic Research in Progressive Education and Development;, 3*(1), 63-74.
- Lahiff, A. (2006). *Curricula Debates in Higher Edcuation an overiew*. Trinity College.
- Land, R. (2012). Engineering Technologists Are Engineers. *Journal of Engineering Technology*, 32-29.
- Latchford, G. (1962). *Technicians For Irish Industry*. Dublin: City of Dublin Vocational Educational Committee. Retrieved September 19, 2020, from https://arrow.tudublin.ie/cgi/viewcontent.cgi?article=1017&context=ditbk
- Lauritsen, A. B. (2012). Bridging the Gap between Theory and Praxis in Engineering Education. *Proceedings of the 8th International CDIO Conference.* Brisbane: Queensland University of Technology.
- Lawlor, R. (Ed.). (2016). Engineering in Society. Beyond the Technical: what every engineering student should know. Royal Academy of Engineering.
- Leckie, G. (2001). Researcher Roles. In L. Given (Ed.), *The SAGE Encylopedia of Qualitative Research Methods* (pp. 771-776). London: SAGE Publishing.
- LeCompte, M., Preissle, J., & Tesch, R. (1993). *Ethnography and Qualitative Design in Educational Research*. New York: Academic Press.
- Lennox, T., & O'Brien, J. (2013). The Civil Engineering Technologist and the Civil Engineer According to the Authorities What's the Difference? *120th ASEE Annual Conference and Exposition* (pp. 1-15). Atlanta: ASEE.
- Liamputtong, P. (2011). *Focus Group Methodology: Principles and Practice.* London: SAGE publications.
- Lichtman, M. (2014). Qualiatative Reseach in Education: A user's guide. Sage publications.
- Lim, L. (2014). *Knowledge, Control and Critical Thinking in Singapore: State Ideology and the Politics of Pedagogic Recontextualisation.* Abingdon: Routledge.
- Lindholm, J. (2009). *Guidelines for Developing and Assessing Student Learning Outcomes for Undergraduate Majors.* Los Angeles: UCLA.
- Lipton, P. (2010). Engineering and Truth. In *Philosophy of Engineering Volume 1 of the proceedings of a series of seminars held at The Royal Academy of Engineering* (pp. 7-14). London: The Royal Academy of Engineering.
- LIT. (2017). Programmatic Review of the Department of Electrical and Electronic Engineering, Panel Report. Limerick Institute of Technology, Electrical and Electronic Engineering. Retrieved from https://lit.ie/getmedia/c43ff66f-34ef-40f7-91d9-bdcca7c66684/E_EE.pdf

- LIT. (2018). *LIT Teaching and Learning Strategy, 2018-2023*. Limerick Institute of Technology.
- Llorens, M., O'Shaughnessy, S., Carr, M., Sheridan, D., Sorby, S., & Bowe, B. (2014). Articulation and Progression: An Investigation into the Transition of Engineering Students from Applied to Theoretical Programmes. *SEFI 42nd annual conference*. Birmingham UK.
- Lohmann, J. (2005). Building a Community of Scholars: The Role of the Journal of Engineering Education as a Research Journal. *Journal of Engineering Education, 94*(1), 1-3 ;.
- Lynch, K. (2006). Neo-liberalism and Marketisation: the implications for higher education . *European Educational Research Journal*, 5(1), 1-17.
- Lynch, K. (2012). On the Market: Neoliberalism and New Managerialism in Irish Education. *Social Justice Series*, 88-102.
- Lynch, K. (2018). Written submission to the Joint Oireachtas Committee on Education and Skills -Barriers to education facing vulnerable groups. Dublin: University College Dublin Library.
- Lynch, K., Grummell, B., & Devine, D. (2012). *New Managerialism in Education: Commercialization, Carelessnes and Gender.* Basingstoke: Palgrave-Macmillan.
- Lynch, R., Seery, N., & Gordon, S. (2007). An evaluation of the CDIO approach to engineering education. *Proceeding of the International Symposium for Engineering Education*. Gemini International LTD; Dublin City University.
- Mahon, R. (2010). After Neo-Liberalism? Global Social Policy(10), 172-192.
- Malley, E. (1981). The Decline of Irish Industry in the Nineteenth Century. *The Economic and Social Review,, 13*(1), 21-42.
- Marchetta, M. (2009). Finnikin of the Rock,. Melbourne: Penguin Group.
- Marsh, E., & C, H. (1964). Ambrosia and Small Beer,. London: Longmans Books.
- Marusik, M., Azmat, ,. E., Ahmad, F., Bilala Harrison, B., Colombo, R., & Walter, A. (2019). The T-CHAT Teaching Approach: An Introduction. UK & Ireland Engineering Education Research Network Annual Conference 2019. Warwick: UK & Ireland Engineering Education Research Networ.
- Marzano, R., & Kendall, J. (Eds.). (2006). *The New Taxonomy of Educational Objectives.* Thousand Oakes: Corwin Press.
- Maton, K. (2014). Knowledge and Knowers. Abingdon: Routledge.
- Maton, K., & Moore, R. (2009). Coalitions of the Mind. In K. Maton, & R. Moore (Eds.), *Social Realism: Knowledge and the Sociology of Education: Coalitions of the Mind* (pp. 1-13). London: Continuum.
- Matos, S., Riley, D., & Akera, A. (2017). WannABET: Historical and Organizational Perspectives on Governance in Engineering Education. *ASEE Annual Conference & Exposition.* Columbus.
- Maynooth University. (2015, January 20). *Maynooth University Research Integrity Policy*. Retrieved March 16, 2016, from Maynooth University website: https://www.maynoothuniversity.ie/sites/default/files/assets/document/Research%20Integ rity%20Policy%20Jan%2015.pdf

- McCabe, H., Manley, R., & Duffy, J. (2009). Teaching Engineering Students about Electronics using PBL Methods: A Case Study from Ireland. *Ninth Finish Probem Based Learning Conference*. University of Tampere: Ninth Finnish Problem Based Learning conference.
- McCarthy, T. (1977). Apprenticeships In Ireland. Social Policy Series.
- McCullough, C. (2007). ABET EC2000: How Has It Changed? Has It Accomplished What Was Intended? . ASEE Southeast Section Conference.
- McGoldrick, J. (1992). *Third Level Qualifications by Alternative Modes of Study, Masters Degree Thesis.* RTC Sligo, NCEA.
- McGrann, R. (2008). Philosophy of Technology in Engineering Education. *38th ASEE/IEEE Frontiers in Education Conference* (pp. S4H29-S4H34). Saratoga Springs: IEEE.
- McIver Consulting. (2003). *The Demand and Supply of Engineers and Engineering Technicians*. Dublin: Expert Group on Future Skills Needs.
- McIver Consulting. (2007). Transfer And Progression Into Undergraduate Programmes Leading To University Awards A Study Undertaken On Behalf Of National Qualifications Authority Of Ireland Irish Universities Association Higher Education Authority. Dublin: National Qualifications Authority of Ireland.
- McKenna, A., & Light, G. (2009). The Role of Collaborative Reflection on Shaping Engineering Faculty Teaching Approaches. *Journal of Engineering Education*, 17-26.
- McLaughlin, P. (1999). Engineering Education in Ireland's Institute of Technology Sector. International Journal of Engineering Education, 15(2), 99-106.
- McLaughlin, P. (2001, November 20). Implications of Bologna Declaration for Engineering Programmes in Institues of Technology. *The Bologna Declaration and Engineering Education in Ireland, The Institution of Engineers of Ireland*, pp. 46-74.
- McManus, A. (2016). The Transformation of Irish Education: The Ministerial Legacy, 1919–1999. In B. Walsh (Ed.), *Essays in the History of Irish Education* (pp. 267-296). Palgrave-Macmillan UK.
- McNutt, L. (2010). *Tension, Frustration and Compromise in the Field: An Exploratory Study of the Habitus of Educational Technologists.* NUIM: Education Doctorate.
- McPhail, G. (2017). Rethinking What it means to be a Progressive Teacher: Key Ideas from Social Realism. *Pacific-Asian Education*, 75-90.
- McPhail, J. (2008). *The OECD, Neoliberalism, and the Learning City:Promoting Human Capital in the Guise of Lifelong Learning, M. A Thesis.* Halifax: Mount St Vincent University.
- Memon, J., Demirdogen, R., & Chowdhry, B. (2009)). Achievements, Outcomes and Proposal for Global Accreditation of Engineering Education in Developing Countries. *Procedia Social and Behavioral Sciences World Conference on Educational Sciences 2009*, 2557–2561.
- Mercille, J., & Murphy, E. (2015). The Neoliberalization of Irish Higher Education under Austerity. *Critical Sociology*, 1-17.
- Milligan, M. (2014). ABET, Global Competencies in STEM Education, and Future Challenges. *NAFSA*. San Diego. Retrieved February 25, 2019, from ABET, Global Competencies in STEM Education, and Future Challenges

- Minister for Education and Skills. (2012, November 6th). *Minister Quinn Merges Educational Agencies to Establish QQI.* Retrieved from Department of Education and Skills: https://www.education.ie/en/Press-Events/Press-Releases/2012-Press-Releases/PR2012-11-06.html
- Ministers of Education France, Germany, Italy and UK. (1998). *The Sorbonne Declaration: Joint Declaration on Harmonisation of the Architecture of the European Higher Education System.* Retrieved November 28, 2017, from EHEA website: https://media.ehea.info/file/1998_Sorbonne/61/2/1998_Sorbonne_Declaration_English_55 2612.pdf
- Ministers Responsible for Higher Education. (2003). *Realising the European Higher Education Area: Communiqué of the Conference of Ministers Responsible for Higher Education, Berlin.* Retrieved November 28, 2017, from EHEA: https://media.ehea.info/file/2003_Berlin/28/4/2003_Berlin_Communique_English_577284. pdf
- Mitchell, J., Nyamapfene, A., Roach, K., & Tilley, R. (2019). Philosophies and pedagogies that shape an integrated engineering programme. *Higher Education Pedagogies, 4*(1), 180-196.
- Montfort, D., Brown, S., & Shinew, D. (2014). The Personal Epistemologies of Civil Engineering Faculty. *Journal of Engineering Education, Social Science Premium Collection, 103*(2), 38-416.
- Moodley, M. (2014). *Mathematics for Future Engineers: A Study of Teaching and Learning Mathematics in an Engineering Curriculum, PhD Thesis.* Durban: University of KwaZulu-Natal.
- Moore, R. (2014). Social Realism and the Problem of Knowledge in the Sociology of Education. In B. Barrett, & E. Rata (Eds.), *Knowledge and the Future of the Curriculum* (pp. 23-40). Basingstoke: Palgrave-MacMillan.
- Moore, R., & Muller, J. (1999). The Discourse of 'Voice' and the Problem of Knowledge and Identity in the Sociology of Education. *British Journal of Sociology of Education, 20*, 189-206.
- Morell, L., Buxeda, R., Cukierman, U., Rivera, W., Vendrell, E., & Sjursen, H. (2018). A Developmental Framework for Teaching Expertise for Engineering and Related Disciplines. *World Engineering Education Forum* (pp. 1-6). Albuquerque: Global Engineering Deans Council (WEEF-GEDC).
- Morgan, D. (1998). *The Focus Book Guidebook (Focus Group Kit, Book 1).* Thousand Oaks: SAGE Publishing.
- Morgan, D. (2008). Focus Groups. In L. Given (Ed.), *The SAGE Encyclopedia of Qualitative Research Methods* (pp. 352-354). London: SAGE publishing.
- Morgan, J., Hoadley, U., & Barrett, B. (2018). Chapter 1-Introduction: Social realist perspectives on knowledge, curriculum and equity. In B. Barrett, U. Hoadley, & J. Morgan (Eds.), *Knowledge, Curriculum, Equity: social realist perspectives* (pp. 1-16). Abingdon: Routledge.
- Morie, J. (2008). Ontological Implications of Being in Immersive Virtual Reality. In I. McDowall, & M. Dolinsky (Eds.), *Proc. SPIE 6804, The Engineering Reality of Virtual Realit.*

- Moses, J. (2008). Towards an onotology for systems related terms in engineering and computer studies. *Workshop on Philosophy and Engineering* (pp. 74-75). London: Royal Academy of Engineering.
- Moynihan, A. (2015). Normative Isomorphism: Is Irish Academic WorkLife the Same in Different Institutional Types in the Universal Phase of Higher Education? PhD Thesis. Dublin: Dublin Institute of Technology.
- MTU. (2021, May 22). *Courses at CIT*. Retrieved from Munster Technological University: https://courses.cit.ie/
- Muller, J. (1998). The Well-Tempered Learner: Self-Regulation, Pedagogical Models and Teacher Education Policy. *Comparative Education*, *34*(2), 177-193.
- Muller, J. (2004a). Assessment, Qualifications and the National Qualifications Framework in South African Schooling. In L. Chisholm (Ed.), *Changing Class:Education and Social Change in Post-Apartheid South Africa* (pp. 223-244). Cape Town: HSRC Press.
- Muller, J. (2004b). Introduction: The Possibilities of Basil Bernstein. In J. Muller, B. Davies, & A. Morais (Eds.), *Reading Bernstein, Researching Bernstein* (pp. 1-12). Routledge-Falmer.
- Munzenmaier, C., & Rubin, N. (2013). *Perspectives, Bloom's Taxonomy: What's Old is New Again.* Santa Rosa: The Elearning Guild.
- Murphy, P., O'Donnell, P., & Jameson, J. (2019). Business in Engineering Education: Issues, Identities,
 Hybrids, and Limits. In C. S, B. Delahousse, C. Didier, M. Megank, & M. Murphy (Eds.), *The Engineering-Business Nexus* (pp. 365-396). Cham: Springer.
- Murray, J. (2013). Quality Assurance in Qualifications Frameworks. *Dublin Conference*. Quality and Qualifications Ireland (QQI), on behalf of the Irish Presidency of the European Council, with the support of the European Commission .
- Myers, M. (2016). *Implementation Of Agile Project Management In the Classroom: MSc Thesis.* Chapel Hill: University of North Carolina.
- NAE. (2001). *Educating the Engineer of 2020; Adapting Engineering Education to the New Century.* Washington DC: National Academy of Engineering, National Academies Press.
- NAE. (2017). *Engineering Technology Education in the United States.* Washington DC: National Academy of Engineering, National Academies Press.
- NCEA. (2001). Examinations Marks and Standards. National Council for Educational Awards.
- NESF. (1993). National Development Plan 1994-1999. National Economic and Social Forum.
- Newman, J. (1851). Duties of Catholics Towards the Protestant View. *Lectures on the Present Position of Catholics in England*. Retrieved from http://www.newmanreader.org/works/england/index.html#contents
- Niemi, T., Kalliomaki, H., & Pajarre, E. (2017). A Teacher? A Mentor? A Friend? Teaching Mentoring Experience at Tampere University of Technology. 45th SEFI Annual Conference (pp. 1352-1361). Azores, Portugal: SEFI.
- Nightingale, S., Carew, A., & Fung, J. (2007). Application of Constructive Alignment Principles to Engineering Eduation: Have We Really Changed? *Proceedings of the 2007 AaeE Conference*.

Melbourne. Retrieved from https://www.semanticscholar.org/paper/Application-of-Constructive-Alignment-Principles-to-Nightingale-Carew/ef845de42cbe3a25653ad149d8afbf975765a93f

Niiniluoto, I. (2011). Virtual Worlds, Fiction, and Reality. *discus.filos* [online]. 2011, 12(19), 13-28.

- NQAI. (2001a). *Submissions Made to the Authority*. Retrieved November 26, 2017, from National Qualifications Authority of Ireland: The Wayback Machine Internet Archive: https://web.archive.org/web/20020821165626/http:/www.nqai.ie:80/submissions.htm
- NQAI. (2001b). *Towards a National Framework of Qualifications: A Discussion Document.* Dublin: National Qualifications Authority of Ireland.
- NQAI. (2002a). *Frameworks of Qualifications: A Review of Developments Outside of the State.* National Qualifications Authority of Ireland.
- NQAI. (2002b). *Towards a National Framework of Qualifications: Establishment of Policies and Criteria*. Dublin: National Qualifications Authority of Ireland.
- NQAI. (2002c). Discussion Forum Towards a National Framework of Qualifications Summary of Trends in Submissions. Retrieved November 25, 2017, from The Wayback Machine, Internet Archive: National Qualifications Authority of Ireland: https://web.archive.org/web/20040513074049/http://www.ngai.ie:80/seanforum.pdf
- NQAI. (2002d). *Towards a National Framework of Qualifications A Discussion Forum, Analysis of Issues Groups*. Retrieved November 27, 2017, from The Wayback Machine Internet Archive: https://web.archive.org/web/20020609134434/http://www.nqai.ie:80/forumagenda.htm
- NQAI. (2003a). *Policy and Criteria for the Establishment of the National Framework of Qualifications.* Dublin: National Qualifications Authority of Ireland.
- NQAI. (2003b). Outline National Framework of Qualifications Determinations made by the National Qualifications Authority of Ireland. Dublin: National Qualifications Authority of Ireland.
- NQAI. (2009). *Referencing of the Irish National Framework of Qualifications (NFQ) to the European Qualifications Framework for Lifelong Learning (EQF)*. Dublin: National Qualifications Authority of Ireland.
- NQAI. (2011). *National Framework of Qualifications, Grid of Level Indicators*. National Qualifications Authority of Ireland.
- Nudelman, G. (2018). A Social Realist Study of Employability Development in Engineering Education, PhD thesis. Grahamstown: Rhodes University.
- NUIG. (2019). *Ms. Mary Dempsey*. Retrieved October 3, 2019, from NUI Galway, Engineering and Informatics: https://www.nuigalway.ie/our-research/people/engineering-andinformatics/marydempsey/
- Nyamapfene, A. (2017). Otherness and Belonging: Integration of Practitioner Academics into an Engineering School at a Research Intensive University. *45th SEFI Annual Conference* (pp. 1198-1205). Azores, Portugal: SEFI.
- O'Byrne, C. (2011). RESEARCH AND THEORY: Against the Odds: Researcher Development in Teaching-Focused HEIs. *International Journal for Researcher Development, 2*(1), 8-25.

- O'Connor, M. (2014). Investment in Edification: Reflections on Irish Education Policy Since Independence. *Irish Educational Studies*, *33*(2), 193–212.
- O'Byrne, C. (2009). Plus ça change, plus ça devient la même chose The Convergence of Academic Professional Identities in the Institute of Technology Sector and the Implications for Teaching and Learning. NUI Maynooth: All Ireland Society for Higher Education Conference.
- O'Byrne, C. (2014). Structure and Agency in an Irish Institute of Technology. In G. L, C. Cook, L. Daunton, J. Salisbury, & B. Thomas (Eds.), *Academic Working Lives* (pp. 15-25). London: Bloomsbury.
- OECD. (1964). Training of Technicians in Ireland. Paris: OECD.
- OECD. (1996). The Knowledge Based Economy. Paris: OECD.
- OECD. (2003). The Role of National Qualifications Systems in Promoting Lifelong Learning: Background Report for Ireland. Paris: OECD.
- OECD. (2004). *Review of National Policies for Education: Review of Higher Education in Ireland.* Paris: OECD.
- OECD. (2005). The Role Of National Qualifications Systems In Promoting Lifelong Learning ,First Draft Of The International Synthesis Report. Paris: OECD.
- OECD. (2006). Higher Education in Ireland. Paris: OECD.
- Office of University Evaluation. (2001). *Guidelines For The Assessment Of Learning Outcomes*. Temple: Arizona State University.
- Oireachtas Library & Research Service. (2014). Higher Education in Ireland: For Economy and Society. *Spotlight, 5,* pp. 1-16.
- O'Malley, M. (2012). University and the State in Ireland: From a Negotiated Exchange Relationship Based on Trust to Prescriptive Requirements in University Governance.: EdD Thesis. Maynooth: Maynooth Universty.
- O'Neill, G., & McMahon, T. (2005). Student Centred Learning: What Does it Mean for Students and Lecturers? In G. O'Neil, S. Moore, & B. McMullin (Eds.), *Emerging issues in the practice of University Learning and Teaching* (pp. 27-36). AISHE.
- Online-Utility.org. (2017). *Readability Calculator*. Retrieved January 23, 2017, from Online-utility.org: https://www.online-utility.org/english/readability_test_and_improve.jsp
- O'Reilly, M., & Kiyinba, N. (2015). Advanced Qualatitive Research: A Guide to Using Theory. London: Sage.
- O'Reilly, N. (2018). *Lifelong Learning & Mobility*. Retrieved from Aontus: https://www.aontas.com/knowledge%2Fblog%2Flifelong-learning-and-mobility-niamho'reilly-ceo-of-aontas
- Orivel, F. (2005). The Bologna process in France. Origin, Objectives and Implementation. Palomba (Donatella). Comparative Study of the Implementation of the Declaration of Bologna in Six European Countries. Roma: Tor Vergata University. Retrieved from https://halshs.archivesouvertes.fr/halshs-00086796/document

- Osborne, R. (1996). *Higher Education in Ireland: North and South.* London: Jessica Kingsley Publishers.
- O'Sullivan, N. (2014). An Exploration of Effective Leadership at Head of Department Level in the Institute of Technology (IoT) Sector Ireland, PhD Thesis. Waterford Institute of Technology.
- Owens, D. (2016a). Outcome-based Education and Engineering Education Accreditation, Presented to China Association of Science and Technology Beijing. Retrieved from Engineers Canada: https://engineerscanada.ca/sites/default/files/20160411_damien_owens_-_cast_obe_paper.pdf
- Owens, D. (2016b). Engineers Ireland urges students to look at all routes into engineering. Retrieved April 20, 2020, from Irish Building: https://www.irishbuildingmagazine.ie/2016/08/17/engineers-ireland-urges-students-tolook-at-all-routes-into-engineering/
- Oxford University Press. (1992, 2012). *Oxford Advanced Learners Dictionay.* Oxford: Oxford University Press.
- Park, C. (2007). Redefining the Doctorate. Higher Education Academy.
- Pausigere, P., & Graven, M. (2013). Using Bernstein to Analyse Primary Maths Teachers' Positions and Identities in the Context of National Standardised Assessment: The Case of the ANAs. *Journal of Education*, 58, 1-22.
- Pawley, A. (2019). OPINION: "Asking questions, we walk": How should engineering education address equity, the climate crisis, and its own moral infrastructure? *JEE: The Research Journal for Engineering Education., 108*(4), 447-452.
- Peräkylä, A., & Ruusovouri, J. (2011). Analysing Talk and Text. In N. Denzin, & Y. Lincoln (Eds.), *The* SAGE Handbook of Qualitative Research (pp. 529-543). London: SAGE Publishing.
- Plummer, K. (2001). Documents of Life 2. London: Sage Publications.
- Poser, H. (1998). On Structural Differences between Science and Engineering. *Philosophy and Technology*, 4(2), 81093.
- Prados, J. (1992). 1991-92 ABET Presidential Address. 60th Annual Report.
- Prados, J. (2004). Can ABET Really Make a Difference. *International Journal of Engineering Education*, 315-17.
- Prados, J., Peterson, G., & Lattuca, L. (2005). Quality Assurance of Engineering Education through Accreditation: The Impact of Engineering Criteria 2000 and Its Global Influence. *Journal of Engineering Educatio*, 165-184.
- Priestly, M., & Higham, J. (1999). *New Zealand's Curriculum and Assessment Revolution*. Leeds: Occasional Publication No, 8, Leeds University School of Education. Retrieved from http://www.leeds.ac.uk/educol/documents/00002221.doc
- QQI. (2013). *QQI Comprehensive Policy Development Programme 1 Qualifications systems and related concepts a QQI background paper.* Quality and Qualifications Ireland.
- QQI. (2014a, August 20). Awards Standards-Generic Higher Education and Training. Retrieved 2015, from Quality and Qualifications Ireland:

http://www.qqi.ie/Publications/Generic%20Major%20Awards%20-%20QQI%20Awards%20Standards.pdf

- QQI. (2014b). National Qualifications Framework. Retrieved from QQI: www.nfq.ie
- QQI. (2014c). Awards Standards- Engineering. Retrieved May 2016, from Quality and Qualifications Ireland: www.qqi.ie
- QQI. (2015). Effective Practice Guidelines for External Examining. Quality and Qualifications Ireland.
- QQI. (2016a). Understanding the National Framework of Qualifications (NFQ). Quality and Qualifications Ireland.
- QQI. (2016b). *Core Statutory Quality Assurance (QA) Guidelines Statutory QA Guidelines developed by QQI for use by all Providers.* Quality and Qualifications Ireland.
- QQI. (2016c). Progression Opportunities into Higher Education. Quality and Qualifications Ireland.
- QQI. (2017). Policy Impact Assessment of the Irish National Framework of Qualifications. Quality and Qualifications Ireland. Retrieved from https://www.qqi.ie/Downloads/Policy%20Impact%20Assessment%20of%20NFQ_Indecon%2 0Report%20with%20Cover_FINAL.pdf
- QQI. (2018). Green Paper on Assessment of Learners and Learning. Qualifications and Quality Ireland.
- QQI. (2019). Accreditation/Approval of Higher Education Programmes by Professional Bodies. Dublin: Quality and Qualifications Ireland.
- RAE. (2008). Workshop on Philosophy in Engineering. London: Royal Academy of Engineering.
- RAE. (2010a). Philosophy of Engineering Volume 1. London: Royal Academy of Engineering.
- RAE. (2010b). Philosophy of Engineering Volume 2. London: Royal Academy of Engineering.
- RAE. (2014). *Thinking like an Engineer: Implications for the Education System.* London: Royal Academy of Engineering.
- Raffe, D. (2009). National Qualifications Frameworks in Ireland and Scotland: A Comparative Analysis. Vienna: European Conference on Educational Research.
- Raffe, D. (2011). National Qualifications Frameworks:What Can Be Learnt From the International Experience? *Journal Of Contemporary Educational Studies*(4), 66-80.
- Raffe, D. (2013). What is the evidence for the impact of National Qualifications Frameworks? *Comparative Education, 49*(2), 143-162.
- Rafferty, P. (1968). Presidential Address. *Transactions of the Institute of Civil Engineers of Ireland* (1967-68), i-vi.
- Rajaee, N., Junaidi, E., Taib, S., Salleh, S., & Munot, M. (2013). Issues and Challenges in Implementing Outcome Based Education in Engineering Education. *International Journal for Innovation Education and Research*, 1(4), 1-9.
- Rata, E. (2014). Knowledge and Democracy: The Strife of the Dialectic. In B. Barrett, & E. Rata (Eds.), *Knowledge and the Future of the Curriculum* (pp. 79-91). Basingstoke: Palgrave-Macmillan.

- Reck, R. P., & Reck, R. (2007). Generating and Rendering Readability Scores for Project Gutenberg Texts. *Corpus Linguistics Conference*, (p. Article #120).
- Rigney, T. J. (2009). The Birth of Change in Irish Education with Particular Reference to Business Studies. *Irish Life and Lore Cork Institute Of Technology Its Origin And Development*, pp. 33-55.
- Riley, D. (2012). Aiding and ABETing: The Bankruptcy of Outcomes-based Education as a Change. *ASEE Annual Conference & Exposition* (pp. 1-13). American Society for Engineering Education.
- Ropers-Huilman, B., Carwile, L., & Lima, M. (2007). Service-learning in engineering: a valuable pedagogy for meeting learning objectives. *European Journal of Engineering Education*, 155-165.
- Royle, E. (1971). Mechanics' Institutes and the Working Classes, 1840-1860. *The Historical Journal,* 14(2), 305-321.
- Ruprecht. (2000). Curriculum Development: The Whole and its Parts. *European Journal of Engineering Education*, 25(4), 359-367.
- Rüütmann, T. (2017). IGIP Philosophy Effective Basis for Contemporary STEM Teaching and Learnin. 2017 IEEE EDUCON. Athens. Retrieved from http://www.igip.org/documents/Paper_R%C3%BC%C3%BCtmann.pdf
- Ryan, A. (2006). Post-Positivist Approaches to Research. In *Researching and Writing your thesis: a guide for postgraduate students.* (pp. 12-26). Maynooth: MACE: Maynooth Adult and Community Education.
- Ryan, A. B., & Walshe, T. (2014). *Doctorate in Higher and Adult Education Student Handbook*. Maynooth: Department of Adult and Community Education, Maynooth University.
- Saldaña, J. (2009). The Coding Manual for Qualitative Researchers. London: Sage Publications ltd.
- SAQA. (2017). South African Qualifications Authority. Retrieved from http://www.saqa.org.za
- Sargeant, J. (2013). Qualitative Research Part II: Participants, Analysis, and Quality Assurance. Journal of Graduate Medical Education, 1-3.
- Saunders, B., Kitzinger, J., & Kitzinger, C. (2015). Anonymising Interview Data: challenges and compromise in practice. *Qualitative Research*, *15*(5), 616-632.
- Saunders, M. (2012). Choosing Research Participants. In G. Symons, & C. Cassell (Eds.), *The Practice of Qualitative Organizational Research:Core Methods and Current Challenges* (pp. 37-55). London: Sage publishing.
- Savage, G. (2017). Neoliberalism, Eeducation and Curriculum. In B. Gobby, & R. Walker (Eds.), Sociological Perspectives on Education (pp. 143-165). Oxford: Oxford University Press.
- Sayer, A. (1992). *Method in Social Science: A Realist Approach* (2nd ed.). London: Routledge.

Sayer, A. (2000). Realism and Social Science. London: Sage.

- Schuster, K., & Hees, F. J. (2010). Dipl-Ing Rest in Peace? The Implementation of the Bologna Process in Germany's Engineering Education. *Proceedings of the 2010 AaeE Conference*, (pp. 38-45). Sydney.
- Seanad Éireann Debate. (1979). National Council for Educational Awards Bill, Second Stage, Wednesday 14 November. *Houses of the Oireachtas, 93*(2), 1-8.
- Sedelmaier, Y., & Landes, D. (2014). A Multi-Level Didactical Approach to Build up Competencies in Requirements Engineering. 8th International Workshop on Requirements Engineering Education & Training. Banff: IEEE.
- Seeley, B. (1999). The Other Re-Engineering of Engineering Education, 1900-1965. *Journal of Engineering Education*, 88(3), 285-294.
- SEFI. (2002). The Implementation of the Bologna Declaration in Higher Engineering Education. (T. Hedberg, Ed.) Retrieved from Société Européenne pour la Formation des Ingénieurs: http://sefibenvwh.cluster023.hosting.ovh.net/wp-content/uploads/2017/08/Collection-ofopinions-on-Bologna-Declaration.pdf
- SEFI. (2017). *Proceedings of the 45th SEFI Annual Conference*. Azores, Portugal: Proceedings of the 46th SEFI Annual Conference.
- SEFI. (2018). *Proceedings of the 46th SEFI Annual Conference*. Copenhagen: SEFI and University of Copenhagen.
- SEFI. (2019). *European Society for Engineering Education*. Retrieved July 7, 2019, from Société Européenne pour la Formation des Ingénieurs: https://www.sefi.be/
- SEFI. (2021, May 3rd). *European Society for Engineering Education*. Retrieved from Société européenne pour la formation des ingénieurs: https://www.sefi.be/about/
- Shannon, D. (2019). A Take of a Discursive Shift, Analysing EU Policy Discourse in Irish Adult Education Policy: From the 'White Paper' to the 'Further Education and Training Strategy'. Irish Journal of Adult and Community Education, 98-118.
- Shapin, S., & Barnes, B. (1977). Science, Nature and Control: Interpreting Mechanics' Institutes. *Social Studies Of Science*, 7(1), 31-74.
- Shay, S. (2012). Conceptualising Curriculum Differentiation in Higher Education: a sociology of knowledge point of view. *British Journal of Sociology of Education*, 1-20. doi:0142-5692
- Sheppard, S., Colby, A., Macatangay, K., & Sulliva, W. (2006). What is Engineering Practice? International Journal of Engineering Education, 429-438.
- Simpson, D., & Jackson, M. (2003). John Dewey's View of the Curriculum in The Child and the Curriculum. *Education and Culture, XX*(2), 21-27.
- Singh, M. (2017). Organizing RVA at National, Regional and Local Levels. In R. Duvekor, D. Couglan, &
 K. Aagaard (Eds.), *The Learner at the Centre* (pp. 11-38). Houten/Aarhus: European Centre
 Valuation of Prior Learning/ VIA University College.
- Singh, P. (2018). Performativity, Affectivity and Pedagogic Identities. European Educational Research Journal., 17(4), 498-506. Retrieved from https://journals.sagepub.com/doi/pdf/10.1177/1474904117726181

- Sitwell, E. (n.d.). *Dame Edith Sitwell, Answers*. Retrieved from InternetPoem.com: https://internetpoem.com/dame-edith-sitwell/answers-poem/
- Skilbeck, M. (2003). Towards an Integrated System of Tertiary Education. unpublished.
- Skills Ireland. (2017, January). http://www.skillsireland.ie/About-Us/EGFSN-Mandate-2007.pdf. Retrieved from www.skillsireland.ie.
- Slough-Kuss, Y. (2015). The Possibility of a 'New Imaginary': Discursive 'De/Humanising' 'Framing' of Teacher Education for Teaching Diverse Student Populations in OECD and CoE Texts, Working Paper Series, International and Global Issues for Research. Bath: Department of Education, University of Bath.
- Smit, R. (2016). *The Nature of Engineering and Science Knowledge in Curriculum: A case study in thermodynamics: PhD Thesis.* Cape Town: University of Cape Town.
- Smithwick, N. (2001). *Code Theory and Changing Pedagogic Discourse, Master's Thesis.* Totonto: University of Toronto.
- Socha, D., Razmov, V., & Davis, E. (2003). Teaching Reflective Skills in an Engineering Course. *ASEE* 2003 Annual Conference (pp. 8.1080.1-8.180.20). Nashville: ASEE.
- SOLAS. (2014). Further Education and Training Strategy 2014-19. Dublin: SOLAS.
- Somerville-Midgette, K. (2014). An Engineering Journey: A Transcendental Phenomenological Study of African-American Female Engineers' Persistence: PhD Thesis. Lynchburg: Liberty University.
- Sorby, S., Williams, B., Oiveira, J., Duffy, G., Brabazon, D., & VanHattum, N. (2014). The Evolution of Engineering Education Research in Portugal and Ireland. ASEE 2014. Indianapolis: American Society for Engineering Education. Retrieved from https://www.researchgate.net/publication/264343930_The_evolution_of_Engineering_Edu cation_Research_in_Portugal_and_Ireland/
- SQA. (2017). Scottish Credit and Qualification Framework. Retrieved from SCQF: http://www.sqa.org.uk
- Squires, A., & Cloutier, R. (2011). Comparing Perceptions of Competency Knowledge Development in Systems Engineering Curriculum:. 2011 ASEE Annual Conference & Exposition (pp. 22.347.1-21). Vancounver: American Society for Engineering Education.
- TCD. (2017a). *School of Engineering, Trinity College Dublin*. Retrieved December 3, 2017, from Brief History of the School: https://www.tcd.ie/Engineering/history.php
- TCD. (2017b). Trinity Professor Elected 2016 IEEE Fellow. Retrieved February 2017, from Trinity College Dublin: https://www.tcd.ie/news_events/articles/trinity-professor-elected-2016ieee-fellow/6425
- TERG. (2019). *Dr Niall Seery*. Retrieved Oct 3rd, 2019, from Technology Education Research Group: http://terg.ie/index.php/members/dr-niall-seery/
- Teustsch, M. (2010, Dec 9th). An opportunity for validation of non-formal learning. Retrieved from Nordic Network of Adult Learning: https://nvl.org/content/an-opportunity-for-validation-ofnon-formal-learning

- Thomas, R. (1979). The Mechanics' Institutes of the Home Counties, c. 1825–70 Part One. *The Vocational Aspect of Education*, *31*(79), , 67-72.
- Thorn, R. (2018). No Artificial Limits. Dublin: Institute of Public Administration.
- Tilley, N. (2016). EMMIE and Engineering: What Works as Evidence to Improve Decisions. *Evaluation*, 22(3), 304-22.
- Tilley, S. (2019). The Role of Critical Qualitative Research in Educational Contexts: A Canadian Perspective. *Educar em Revista, 35*(75), 155-180.
- Torrance, H. (2007). Assessment as learning? How the Use of Explicit Learning Objectives, Assessment Criteria and Feedback in Post-Secondary Education and Training can come to Dominate Learning. *Assessment in Education: Principles, Policy & Practice, 14*(3), 281–294.
- Trant, M.-L. (2002). The Quest for an Inclusive Curriculum and Assessment Culture: The National Council For Vocational Awards 1991-2001. *Irish Educational Studies, 21*(1), 19-33.
- Trevelyan, J. (2019). Transitioning to Engineering Practice. *European Journal of Engineering Education, 44*(6), 821-837.
- Triventi, M. (2011). The Role of Higher Education Stratification in the Reproduction of Social Inequality in the Labour Market. A Comparative Study of Recent European Graduates. Retrieved September 15, 2019, from MPRA: https://mpra.ub.unimuenchen.de/35996/1/MPRA_paper_35996.pdf
- Triventi, M. (2013). Comparative Study of 11 European Countries. *European Sociological Review,* 29(3), 489-502.
- TU Dublin. (2019a). *Bachelor of Engineering in Mechatronic Engineering*. Retrieved November 30, 2019, from Technological University of Dublin: https://www.itb.ie/StudyatITB/bn306.html
- Tu Dublin. (2019b). *Bacholor of Science in Process Instrumentation and Automation*. Retrieved November 30, 2019, from Technological University Dublin: https://www.itb.ie/StudyatITB/BN315.html
- TU Dublin. (2019c). https://www.itb.ie/StudyatITB/BN309.html. Retrieved November 30, 2019, from Bachelor of Science in Data Centre Operations and Management: https://www.itb.ie/StudyatITB/BN309.html
- Tuck, R. (2007). An Introductory Guide to National Qualifications Frameworks: Conceptual and Practical Issues for Policy Makers. Geneva: International Labour Office Publications.
- Tuck, R., Hart, J., & Keevy, J. (2004). The Relevance Of The National Qualifications Framework Impact Study to Qualification Framework Development In The Southern African Development Community. SAQA Bulletin, 6(2), 5-29.
- TUI. (2016a, February 19th). *Lecturer Contract*. Retrieved from Teachers Union of Ireland: http://www.tui.ie/_fileupload/Image/PWT%20LECTURER%20IOT%20CONTRACT.doc
- TUI. (2016b). *Third-Level Agreed Contracts*. Retrieved February 19th, 2016, from Teachers Union of Ireland: http://www.tui.ie/welcome-to-our-website/third-level-agreed-contracts-.2196.html
- Turner, C. (1980). *Politics in the Mechanics Institutes, 1820-1850: A Study in Conflict, PhD Thesis.* Leicester: Leicester University.

- Turunenm, I., & Byers, B. (2012). *Implementing Outcomes-Based Education in Chemistry and Chemical Engineering*. European Chemistry and Chemical Engineering.
- Tyler, R. (1949). Basic Principles of Instruction. Chicago and London: The University of Chicago Press.
- Tyler, W. (2004). Silent, Invisible, Total: Pedagogic Discourse and the Age of Information. In J. Muller,B. Davies, & A. Morais (Eds.), *Reading Bernstein, Researching Bernstein* (pp. 15-29). London:RoutledgeFalmer.
- UCD. (2011). *The Building of the State Science and Engineering with Government on Merrion Street.* (C. Cullen, & Feely, Eds.) Dublin: University College Dublin.
- UCD. (2019). *Dr Tom Curran*. Retrieved Oct 3, 2019, from Universitty College Dublin: https://people.ucd.ie/tom.curran
- UCD. (2021, May 1). Using Biggs' Model of Constructive Alignment in Curriculum Design. Retrieved from Open Educational Resources of UCD Teaching and Learning, University College Dublin: http://www.ucdoer.ie/index.php/Using_Biggs'_Model_of_Constructive_Alignment_in_Curri culum_Design/Introduction
- UK & IEE EER Network. (2019, May 3rd). *Building Capacity, Spring Colloquium*. Retrieved from UK and Ireland Engineering Education Network: https://drive.google.com/file/d/1SvU7pasXpiLnHCXm5hbv4PI4QJTppxH_/view
- Van, N., Chism, N., Douglas, E., & Hilson W, J. (2008). Qualitative Research Basics: A Guide for Engineering Educators. Retrieved from Centre for Research on Teaching and Learing in Education: https://crlte.engin.umich.edu/wp-content/uploads/sites/7/2013/06/Chism-Douglas-Hilson-Qualitative-Research-Basics-A-Guide-for-Engineering-Educators.pdf
- Viennet, R., & Pont, B. (2017). Education Policy Implementation: A Literature Review and Proposed Framework: OECD Education Working Paper No. 162. Paris: OECD.
- Virkii-Hatakka, T. (2018). Playing Safe in Teaching? *Proceedings of the 46th SEFI Annual Conference* (pp. 1394-1401). SEFI and University of Copenhagen: Copenhagen.
- Walling, J. (2005). *THE Mullard/Philips Research Laboratories, Redhill. A Short HISTORY 1946 2002,.* Philips Research Laboratories.
- Walsh, E. (2008). University for Waterford. Retrieved from http://www.edwalsh.ie/.
- Walsh, J. (1999). A New Partnership in Education. Dublin: Institute of Public Administration.
- Walsh, J. (2018). *Higher Education in Ireland, 1922–2016: Politics, Policy and Power—A History of Higher Education in the Irish State.* London: Palgrave Macmillan.
- Walsh, J., McCovy, S., Seery, A., & Conway, P. (2014). Investment in Education and the Intractability of Inequality. *Irish Educational Studies*, 119-122.
- Warren, H. (1961). *The Nature and Scope of Technical Education*. Dublin: City of Dublin Vocational Education Committee. Retrieved September 19, 2020, from https://arrow.tudublin.ie/cgi/viewcontent.cgi?article=1020&context=ditbk
- WhatIsEngineering. (2016). *Engineering Disciplines*. Retrieved November 3, 2019, from What is Engineering: http://whatisengineering.com/engineering-disciplines/

- Wheelahan, L. (2005). The Pedagogic Device: The relevance of Bernstein's Pedagog Device for VET.
 Vocational Learning : Transitions, Interrelationships, Partnerships and Sustainable Futures. Gold Coast: Australian Academic Press. Retrieved from https://research repository.griffith.edu.au/handle/10072/2515
- White, T. (2001). *Investing in People: Higher Education in Ireland from 1960 to 2000* (1 ed.). Dublin: Future Print.
- Wilby, P. (2018, October 9). *The Counterculture Class Warrior who turned to Gove*. Retrieved from The Guardian: https://www.theguardian.com/education/2018/oct/09/counterculture-classwarrior-turned-to-gove
- Winberg, C., Engel-Hills, P., Jacobs, C., Garraway, J., & Winberg, S. (2012). 'I Take Engineering With Me':Theory and Practice in an Engineering Curriculum. *Knowledge and Curriculum in Higher Education Symposium, CHED.* UCT.
- Wolff, K. (2013). 'Reservoirs' and 'Repertoires': Epistemological and Discursive Complexities in Multidisciplinary Engineering Practice. *Journal of Academic Writing*, *3*(1), 84-94.
- Wolff, K. (2015). *Negotiating Disciplinary Booundaries in Engineering Problem Solving Practice, PhD Thesis.* Cape-Town: University of Cape-Town.
- Wolmarans, N. (2017). *The Nature of Professional Reasoning: An Analysis of Design in the Engineering Curriculum: PhD Thesis.* Cape Town: University of Cape Town.
- Woolston, D. (2008). Outcomes-Based Assessment in Engineering Education: A Critique of Its Foundations and Practice . *38th ASEE/IEEE Frontiers in Education Conference* (pp. S4G1-5). Saratoga Springs: IEEE.
- Young, M. (Ed.). (1971). *Knowledge and Control: New Directions for the Sociology of Education*. London: Collier MacMillan.
- Young, M. (2002). The Curriculum of the Future: From the 'New Sociology of Education' to a Critical Theory of Learning. Routeledge.
- Young, M. (2003). National Qualifications Frameworks as a Global Phenomenon: A comparative perspective. *Journal of Education and Work, 13*(3), 223-237.
- Young, M. (2005). National Qualifications Frameworks: Their Feasibility for Effective Implementation in Developing Countries. Geneva: International Labour Office.
- Young, M. (2008). Bringing Knowledge Back In. London: Routledge.
- Young, M. (2010). Foreword. In L. Wheelehan, *Why Knowledge Matters in Curriculum: a social realist argument* (pp. xi-xii). London: Routledge.
- Young, M. (2011). Qualification Frameworks: Some Cautionary Thoughts. *Journal Of Contemporary Educational Studies, 4*, 92–100.
- Young, M. (2014). Curriculum Theory and Why it is Important. *Cadernos de Pesquisa, 44*(151). Retrieved November 4, 2019, from http://dx.doi.org/10.1590/198053142851
- Young, M., & Muller, J. (2010). Three Educational Scenarios for the Future:Lessons from the Sociology of Knowledge. *European Journal of Education, 45*(1), 12-27.

- Young, M., & Muller, J. (2014). On the Powers of Powerful Knowledge. In B. Barrett, & E. Rata (Eds.), Knowledge and the Future of the Curriculum (pp. 41-64). Basingstoke: Palgrave-MacMillan.
- Zelic, B., & Stahl, B. (2005). The Influence of Realist Ontology on Technological Projects: The Case of Irish Electronic Voting Professional Knowledge Management . *Third Biennial conference, WM* 2005, (pp. 599-602). Kaiserslautern, Germany.